

LUND UNIVERSITY

General purpose technology diffusion and labour market dynamics

A spatio-temporal perspective

Martynovich, Mikhail

2016

Document Version: Publisher's PDF, also known as Version of record

Link to publication

Citation for published version (APA): Martynovich, M. (2016). General purpose technology diffusion and labour market dynamics: A spatio-temporal perspective. [Doctoral Thesis (compilation), Department of Human Geography]. Lund University.

Total number of authors:

General rights

Unless other specific re-use rights are stated the following general rights apply:

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights. • Users may download and print one copy of any publication from the public portal for the purpose of private study or recorder.

or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00 General purpose technology diffusion and labour market dynamics A spatio-temporal perspective

MIKHAIL MARTYNOVICH | FACULTY OF SOCIAL SCIENCES | LUND UNIVERSITY



General purpose technology diffusion and labour market dynamics

A spatio-temporal perspective

Mikhail Martynovich



DOCTORAL DISSERTATION by due permission of the Faculty of Social Sciences, Lund University, Sweden. To be defended at Världen (111), Geocentrum I, Sölvegatan 10, Lund on December 16, 2016 at 10.00

> Faculty opponent Prof. Richard G. Shearmur School of Urban Planning McGill University, Montreal, Canada

	1	
	Document name	
Department of Human Coography	Doctoral Dissertation	
Department of Human Geography	November 2016	
Author(a)	Sponsoring organization	
Author(S) Mikhail Martypovich	Sponsoring organization	
General purpose technology diffusion	and labour market dynamics:	A spatio-temporal perspective
Abstract	•	
This dissertation aims at advancing the the technology-induced economic tran micro-, meso-, and macro-levels of the dissertation investigates the co-evoluti across and within regional labour mark general purpose technology (GPT) in s The analytical framework employed in long-term economic development with the labour market in the process of tec research are integrated into the evolut level processes of response to GPT-in and outcomes of these processes.	e knowledge about the role o sformation, taking into accou e economy and at different ge onary dynamics of industrial ets induced by the diffusion Sweden in the period betwee the dissertation combines th the insights from the labour i chnology-induced structural c ionary model of economic ch duced transformation pressu	i the labour market in the process of nt the variety of factors involved at ographical scales. Empirically, the restructuring and worker reallocation of the information technology as a n 1985 and 2010. The theoretical perspectives on the market studies on the dual role of hange. These two strands of ange that links explicitly the micro- res to the macro-level determinants
The findings of this dissertation indicate that the dynamics of technological change as well as its implications for functioning of regional labour markets should be approached through the prism of technology diffusion process which <i>unfolds in time and space</i> . To be able to understand the technology diffusion process one should address four intertwined questions: <i>what kind of technology</i> is diffused, <i>how</i> it is diffused, <i>when</i> it is diffused as well as <i>where</i> it is diffused at different points in time. Also, the dissertation suggests that the role of the labour market in the technology-induced transformation is far from the smoothly operating mechanism of adjustment to economy-wide economic shocks, as it is often assumed in mainstream approaches to economic theorising. In that respect, the dissertation provides a reference point for studies of labour market evolution beyond the standard equilibrium approach.		
Key words: general purpose technology, technological change, information technology, IT, diffusion, labour market dynamics, labour mobility, worker reallocation, Sweden		
Classification system and/or index terr	ns (if any)	
Supplementary bibliographical information		Language
		English
ISSN and key title		ISBN
		978-91-7753-069-5
Recipient's notes	Number of pages 315	Price
	Security classification	1

I, the undersigned, being the copyright owner of the abstract of the above-mentioned dissertation, hereby grant to all reference sources permission to publish and disseminate the abstract of the above-mentioned dissertation.

Signature

fi finstyworth

Date

2016-11-07

General purpose technology diffusion and labour market dynamics

A spatio-temporal perspective

Mikhail Martynovich



Copyright © Mikhail Martynovich

Faculty of Social Sciences | Department of Human Geography

ISBN 978-91-7753-069-5 (print) ISBN 978-91-7753-070-1 (pdf)

Printed in Sweden by Media-Tryck, Lund University Lund 2016

Cover image: Rawpixel



If everything occurred at the same time, there would be no development. If everything existed in the same place there could be no particularity. Only space makes possible the particular, which then unfolds in time.

(Lösch 1954, p. 508; emphasis in the original)

Table of contents

Table of contents	i
Acknowledgements	iii
List of articles	vi
List of figures	vii
List of tables	viii
List of acronyms	ix
1. Introduction	1
1.1. Beginnings	1
1.2. Research background	2
1.3. Aim, research questions, and contributions	4
1.4. Research setting	6
1.5. Core concepts	7
1.6. Outline of the dissertation	8
2. Perspectives on the long-term economic development	11
2.1. Long waves of economic growth - 'the pulsation of modern	
capitalism'	11
2.2. Endogenising technology into Schumpeter's framework	14
2.2.1. General purpose technologies – the 'engines of growth'2.2.2. Techno-economic paradigms	' 15 18
2.3. Decomposing the technology diffusion process	23
2.3.1. Unfolding technological transformation in time	24
2.3.2 and space	26
2.4. A comprehensive summary	29
3. Technological change and labour market dynamics	31
3.1. General remarks	31

3.2. Labour market response to structural change	33
3.3. Labour market as the mediator of knowledge transfer	36
3.4. The GPT diffusion and the dual role of the labour market	39
4. GPT diffusion and labour market dynamics in an evolutionary model	of
economic development	41
4.1. Evolutionary theorising of economic change	41
4.1.1. Foundations of evolutionary thinking	41
4.1.2. Evolutionary research in economic geography 4.2. Analytical model	40
5 Methodological considerations	55
5.1 Data	55
5.1.1. Data sources	55
5.1.2. Regional divisions	56
5.1.3. Industry divisions	58
5.2. Ontological and epistemological perspective	59
5.2. Methodological perspective	61
5.4. Methods	63
6. Current wave of technological change: Information technology as a	
GPT	67
6.1. Growth and productivity impact of the IT	69
6.2. Labour market impact of the IT	72
7. Research setting: Sweden	75
7.1. Swedish economic growth and labour market trends, 1985-	
2010	75
7.2. The IT in Sweden	78
7.3. The 'Swedish model' of the labour market	81
8. Conclusion	87
8.1. Summary of the articles	87
8.2. Synthesis of results	97
References	103

Acknowledgements

'So this is it, here it is' Bon Jovi sings in my headphones. It is almost 2 a.m., and in the morning this piece of text is going to print. A bit more than four years ago – and it still feels like it was yesterday – I entered Geocentrum for the first time as a new member of the Department of Human Geography and started my journey as a Ph.D. student. Now, sitting in my office and furiously re-reading the dissertation, I still can't believe that this journey is about to come to an end.

My time as a Ph.D. student has taught me to be ambitious but realistic, to be ready to deviate from the original path, when it is necessary, but also insist on my ideas when no one else believes in them. More than anything else, however, it has taught me that any journey is much easier when you have great companions. So many people have contributed, in one way or another, to this dissertation that it fascinates me that there is only my name on the cover. I would like to express my sincere and endless gratitude to everyone who joined me on this journey. Without you, this dissertation would never have been written.

First and foremost, I would like to thank my supervisors – a.k.a., my academic dads – Karl-Johan Lundquist and Martin Henning for their guidance, encouragement, and patience. You have made an invaluable contribution to my development, both academic and personal. Kalle, thanks for keeping your door open whenever your assistance was needed. And, of course, for the hat trick in 'Hattrick's'. This is yet to be beaten. Martin, thanks for being supportive, when it was needed, and critical, when I deserved it. I will never forget our (late) evening conversations that generated so many great ideas. Do you still think we should dig into the Soviet Union archives to build the relatedness matrix for Soviet industries? Three of us have been in this together for more than four years. This dissertation is the final dot in a huge chapter of our collaboration. But I truly hope that the whole book is yet to be written.

The next group of people I would like to thank is my great Ph.D. colleagues I have met over these years. I will not mention all, even though all of you undoubtedly deserve it. Noura Alkhalili and Salvatore Paolo del Rosa, we have started together and have been through this together no matter where our work brought us at different stages of our Ph.D. studies. That was great! And, of course, thanks for being great office mates for most of this time. I would like to thank Hjalti Nielsen for our numerous coffee breaks in Sky High, which gave rise to so many research ideas that the lifetime will not be enough to investigate them all. Siri Kjellberg and Mads Barbesgaard, thanks for making us realise that Ph.D. students are not just a group of employees but the community. Kadri Kuusk and Katherine Burlingame, thanks for making the last year of this journey so much fun.

But above all, I owe many thanks to Srilata Sircar. From the salary night in Oslo to the submission of our dissertations, we shared all the ups and downs together, often lubricated with plenty of beers. If I mention every time that you have helped me, academically and personally, this acknowledgement section will likely be longer than the whole dissertation. Thus, I will just say that I owe you big time! And no matter where life brings us, I hope that we both will remember why and when Sheldon plays bongos.

I would like to thank all colleagues at the department, present and past. I can't express how much our teaching collaborations, group meetings, and just small talks in the corridors of Geocentrum helped me through the Ph.D. journey. Thank you all for being such a great collective.

This dissertation would also not be possible without support of my friends. Thank you all for listening to my crazy ideas all these years and giving me great times away from my computer. It is impossible to remember everyone but few names deserve a special mention.

Robin Carlsson, well, I don't even know where to start. We met long before I decided to become a Ph.D. student. Parentesen, bränboll, Kiev, Prague, Thailand – these are just some glimpses of the experiences that we shared together. Thanks for those. Our friendship was one of the most important non-academic factors, which made this dissertation happen. And, surely, thanks for being the reality check for my ideas about the IT.

Keyvan Alvandi, our nights out in 360 (and other places-that-must-not-benamed) always helped me to let off academic steam and meet new challenges with flying colours. Thanks for the endless number of pool games. I hope that one day I will tame the 'curvature' well enough to win at least half of those.

Svetlana Efimkina, you were the first visitor from Russia when I moved to Sweden and remained the most frequent one since then. Your unbounded optimism and enthusiasm always stimulated me to strive for goals which I considered unachievable.

Elle and Johan, thanks for amazing dinners and board games sessions which often lasted from sundown till sunrise.

I would like to thank my family. Without them I would never be where I am today. Мама и папа, на протяжении последних 12 лет, вы поддерживали меня в каждом моём решении, которое уводило меня всё дальше от дома. Мы радовались и смеялись, огорчались и плакали вместе. Никакие слова никогда не смогут передать всю глубину моей благодарности за вашу безграничную поддержку. Ксюша, когда казалось, что даже родители могут не понять, ты была рядом. Ты всегда знала, когда нужно надо мной посмеяться, а когда поддержать. Ты – лучшая!

In the end, I would like to say thanks to everyone who I met along my Ph.D. journey and whose names I have not mentioned.

Thank you so much! Tack så mycket! Спасибо огромное!

Mikhail November 7, 2016 Geocentrum, Lund, Sweden

List of articles

Article I

Martynovich, M. and Lundquist, K.-J. (2016). 'Technological Change and Geographical Reallocation of Labour: On the Role of Leading Industries'. *Regional Studies*, Vol. 50(10), pp. 1633-1647.

Article II

Martynovich, M. and Henning, M. (*under review*). 'Labour sourcing in a rapidly expanding sector'. *Re-submitted to a peer-reviewed journal*

Article III

Martynovich, M. (2016). 'Specialised pervasiveness: Diffusion of IT competences in the Swedish economy'. *An unpublished manuscript*

Article IV

Martynovich, M. (*under review*). 'Local or not? Spatial biographies of Swedish IT entrepreneurs'. *Re-submitted to a peer-reviewed journal*

List of figures

Figure 1. The dual role of the labour market

Figure 2. Evolutionary theory of economic change à la Nelson and Winter (1982)

Figure 3. Co-evolutionary dynamics of industrial restructuring and worker (re)allocation: A schematic summary

Figure 4. Economic growth in Sweden, 1985-2010 (annual GDP growth rate, fixed 2005 prices)

Figure 5. Total employment in Sweden, 1985-2010 (individuals aged 16-64)

Figure 6. Employment growth across Swedish regions, 1985-2010 (1985=1)

Figure 7. IT employment across Swedish regions, 1985-2010

List of tables

Table 1. Versions of SNI

 Table 2. Summary of research methods

Table 3. IT employment in Sweden

Table 4. Overview of the articles

List of acronyms

ABC 80	Advanced BASIC Computer 80
BARK	Binär Aritmetisk Relä-Kalkylator (Binary Arithmetic Relay Calculator)
BESK	Binär Elektronisk SekvensKalkylator (Binary Electronic Sequence Calculator)
DEVIL	Databases of Evolutionary Economic Geography in Lund
EEG	Evolutionary economic geography
GDP	Gross domestic product
GMM	Generalised method of moments
GPT	General purpose technology
IT	Information technology
LISA	Longitudinell integrationsdatabas för sjukförsäkrings- och arbetsmarknadsstudier (Longitudinal Integration Database for Health Insurance and Labour Market Studies)
MMN	Matematikmaskinnämnden (Swedish Board for Computing Machinery)
NACE	Nomenclature statistique des activités économiques dans la Communauté européenne (Statistical classification of economic activities in the European Community)

OLS	Ordinary least squares
РС	Personal computer
R&D	Research and development
RDC	Receiver and development competences
SNI	Standarden för svensk näringsgrensindelning (Swedish Standard Industrial Classification)
ТЕР	Techno-economic paradigm

1. Introduction

1.1. Beginnings

The economy is in a constant flux. The ability to incessantly transform themselves from within is the central feature of capitalist economic systems (Schumpeter 1942; Nelson 1996). The restlessness of capitalism builds upon its unlimited capacity to generate new knowledge and put it to the productive use, creating new and reviving existing economic activities. Economic agents – individuals, firms, and institutions – do not simply adjust to the current economic conditions, but rather creatively reinterpret the environment, in which they operate and which itself is the product of human design (Metcalfe et al. 2006). The creativity of economic agents is essentially linked to processes of investment and innovation that lie at the core of economic change.

The importance of innovation has for a long time been recognised by scholars of economic development. In the first half of the twentieth century, Joseph Schumpeter (1939, 1942, 1961) suggested that radical innovations are the major driving force behind the long-term upheavals in the rhythm of economic growth. The diffusion of radical innovations moves the technological frontier outward and redefines the set of opportunities and constraints for activities of economic agents, their investment behaviour and business strategies, leading to deep structural changes in the economy.

Some radical innovations – often regarded as general purpose technologies (Bresnahan and Trajtenberg 1995; Bresnahan 2010) – have an impact on economic activities in a wide variety of industries, far beyond the sector they originate in. Their diffusion leads to the emergence of new industries, revitalisation of growth in existing ones, as well as obsolescence and eventual crowding out of others. In that respect, the restlessness of capitalism manifests itself in the whirlwind frenzy of ongoing economic change that is propelled by the everlasting gale of creative destruction (Schumpeter 1942).

It would be highly unrealistic to expect that processes of technology-induced transformation and resulting economic outcomes are evenly distributed in geographic space. As there are regional differences in economic structure, business dynamics, skill endowments, investments and so on, there can never be a perfect convergence in wages, per capita incomes, productivity, etc. across different regions. In particular, creative destruction is unlikely to be spatially neutral in its dynamics and its outcomes since barriers to movements of labour, capital, and knowledge prevent the automatic self-correcting adjustment mechanisms praised by mainstream economists (Martin 2015).

Schumpeter-inspired research on the spatiality of economic change laid the basis for the research paradigm known as evolutionary economic geography (Boschma and Frenken 2006; Boschma and Martin 2007; Martin and Sunley 2015). Its main proposition is that experiences and competences acquired over time by economic agents – characterised by bounded rationality and operating in the situation of information asymmetries – determine, to a great extent, present configurations of spatial economic landscapes as well as future trajectories of regional development (Kogler 2015). The consequence is that regional economies are in constant unrest, resulting in uneven geographies of knowledge (and, thus, innovation) generation and diffusion. The technology-induced economic transformation is, therefore, a process that is inherently geographical in character (Lundquist and Olander 2009; Henning et al. 2016).

1.2. Research background

In his keynote address to the 10th Conference of the International Joseph A. Schumpeter Society Conference in 2004, Richard Nelson claimed that

Schumpeter's theories had important implications for labor economics, but [...] this was a much neglected topic, because of the historical accident that most of those with an interest in Schumpeter had backgrounds in industrial organization, and related areas. (cited in Diamond 2010, p. 44)

Indeed, Schumpeter-inspired evolutionary research in economics and economic geography has relatively overlooked the role of the labour market dynamics during the technology-induced economic transformation. This neglect of the topic is rather surprising, though. On the one hand, in the course of creative destruction flexible labour markets smooth the processes of industrial restructuring and resulting structural change as well as lower the costs of adjustment to new industrial systems by ensuring that redundant labour laid off in declining parts of the economy is reabsorbed by those that are emerging or revitalised (Aghion et al. 2006). In the same vein, the labour market is often considered as an important adjustment mechanism that mitigates the imbalances across regional labour markets (Lindley et al. 2002; Caroleo and Pastore 2010; Pastore 2013; Hauser 2014) by ensuring the optimal reallocation of human resources and more efficient distribution of employment in space in response to the process of technology-induced transformation of regional economies.

On the other hand, '[t]he heart of the whole process of industrialization and economic development is intellectual: it consists in the acquisition and application of a corpus of knowledge' (Landes 1980, p. 111). As individuals are the principal carriers of knowledge (Bienkowska 2007; Breschi and Lissoni 2009; Maliranta et al. 2009), endowments of regions with skilled labour as well as reallocation of workers across regions may have a profound impact on the regional ability to undergo economic transformation. In other words, the labour market – being the mediator for knowledge transfer – may both enable and restrict opportunities for technology-induced economic change in regions.

In that respect, investigating the mutual dynamics of technology-induced industrial restructuring and labour market transformation may provide an important insight into where, why, and how economic change happens in geographical space.

Another, even though less obvious, drawback of the Schumpeter-inspired research, that can be inferred from Richard Nelson's quote above, is the excessive focus on industrial organisation in evolutionary research. More recently, discussing the development of the evolutionary economic geography tradition, Martin and Sunley (2015) asserted that

evolutionary economic geography has been mainly concerned with the construction of a sort of 'evolutionary industrial geography' rather than an explicit evolutionary theory of uneven regional development. There have been studies of how particular industries have evolved across space; studies of the evolution of one or more industries in a given place [...]. What has been much less in evidence is a concern with the synergy of different economic processes and structures [...] and with the systemic tendency towards uneven regional

development. Hence evolutionary economic geography has struggled to connect micro-scale processes with large-scale processes, patterns and regularities. (p. 720)

In other words, evolutionary theorising has been mainly concerned with understanding processes taking place at the micro-level of the economy. At the same time, the role of the surrounding environment and macro-level processes, both as outcomes of and conditioning mechanisms for micro-scale processes, has not been addressed in a proper way to date.

This dissertation is an attempt to connect, through the prism of labour market dynamics, the macro-level outcomes of technology-induced economic change to the micro-level behaviour of economic agents. This is done by identifying the rules operating at the meso-level of the economy which, on the one hand, condition the micro-level processes and, on the other hand, define how the latter are translated into the outcomes at the macro-level of the economy.

1.3. Aim, research questions, and contributions

On a theoretical level, this dissertation aims at *advancing the knowledge about the role of the labour market in the process of technology-induced economic transformation, taking into account the variety of factors involved at micro-, meso-, and macro-levels of the economy and at different geographical scales.* A particular attention is devoted to the temporal dimension of labour market dynamics.

The dissertation sets itself around the process of diffusion of a general purpose technology (GPT) that has a potential to generate restructuring in all parts of the economy. In this process, the labour market may be expected to play a dual role (Figure 1). On the one hand, the GPT-induced industrial restructuring manifests itself in the changing patterns of demand for labour, which generate the response of the labour market in the form of worker reallocation across jobs, industries, and regions. In that respect, the labour market may be considered a response mechanism to the technology-induced economic change. On the other hand, the very opportunities for industrial restructuring may be expected to depend on the availability of skilled labour in the new and old parts of the economy as well as across different regions.

Thus, labour market may also facilitate and/or restrict the dynamics of technology-induced transformation.



Labour market as enabling/restricting mechanism

Figure 1. The dual role of the labour market

The overarching research question to be addressed in this dissertation is then formulated in the following way:

In what ways and to what extent is the mutual co-evolution of patterns of industrial restructuring and labour market dynamics across and within regional labour markets conditioned by the introduction and subsequent diffusion of a general purpose technology?

Specifying the overarching research question further, the following subquestions are posited:

RQ1: how does the diffusion of a GPT affect patterns of worker reallocation across and within regional labour markets?

RQ2: how are *GPT*-related competences accumulated in the *GPT*-producing sectors?

RQ3: what patterns do characterise the diffusion of GPT-related competences to the GPT-using sectors?

RQ4: in what way do properties of regional economy, and particularly the position of a region in the regional hierarchy, condition the opportunities for regional GPT-induced transformation?

RQ5: how does the role of the labour market change at different stages of the GPT diffusion in the economy?

While the first sub-question addresses the role of the labour market as response mechanism to the diffusion of a GPT, the sub-questions 2-4 focus primarily on the potential role of the labour market as an enabler/restrictor of technology-induced industrial restructuring. Finally, the fifth sub-question aims at investigating whether these roles of the labour market change over time, as the GPT makes its way further into the economy.

By answering these research questions, the dissertation seeks to contribute to filling the research gaps in the field of economic geography. In particular:

- 1. The dissertation develops the evolutionary model that conceptualises the relationship between the GPT-induced industrial restructuring and labour market dynamics;
- 2. The dissertation provides an integration of the literature in the fields of economics, economic history, and economic geography to characterise this relationship as unfolding at *different geographical levels* as well as *over time*;
- 3. The dissertation links the micro-level processes of technologyinduced transformation with the macro-level determinants and outcomes of these processes by identifying the meso-rules that ensure the translation of the former into the latter.

1.4. Research setting

Empirically, the dissertation is set around the latest wave of technological change induced by introduction and diffusion of the information technology (IT). It is commonly agreed that the IT is the GPT of nowadays (David and Wright 1999; Bresnahan 2010). It has been demonstrated to be adopted by a wide range of firms in different industries and to be a source of productivity growth at the firm, industry, and national levels (for reviews of the literature, see Draca et al. 2007; Stiroh 2010; Kretschmer 2012; Cardona et al. 2013; Stanley et al. 2015).

In particular, the dissertation investigates the co-evolutionary patterns of industrial restructuring and labour market dynamics induced by the diffusion of the IT as a GPT in Sweden in the period between 1985 and 2010. Over this time period, Sweden became one of the global leaders as a user of IT products

and services (Giertz et al. 2015). Numerous rankings place Sweden among the top digital economies. To name a few, Sweden scored 2nd in the 2013 Digital Evolution Index (Chakravorti et al. 2014), 3rd in the 2016 EU Digital Economy and Society Index (EU 2016) and the 2016 World Economic Forum's Networked Readiness Index (Baller et al. 2016), and 5th in the 2015 Global ICT Development Index (ITU 2016).

At the same time, Sweden is also one of the major IT-producing economies in the world. In 2011, Sweden satisfied 1.59 per cent of the world's foreign demand – that is, the demand posed by economic agents outside the country – in IT products and services (Ali-Yrkkö et al. 2015). That is a huge success for a country that in the same year produced around 0.09 per cent of the world's GDP (The World Bank 2016).

Sweden's success story is, therefore, of a particular interest for studying the processes of technology-induced economic transformation. Analysing how Sweden ended up where it is today, it is possible to disentangle the success factors and barriers which arose in the process of diffusion of the IT as a GPT.

Also, Sweden represents an interesting case for analyses of long-term regional economic change for two features of its regional system. On the one hand, despite being a rather small economy, Sweden features the regional system characterised by the combination of regions that are very diverse in their economic structures and those which are extremely specialised (Enflo et al. 2014). On the other hand, Swedish regional system is characterised by highly uneven patterns of population distribution. The vast majority of Swedish population lives in Stockholm or to the south of it, which leaves Northern Sweden with vast and sparsely populated areas and large distances between regional centres (Henning et al. 2011). These features makes it interesting to investigate the mechanisms of regional diffusion of technology-induced transformation pressures and the resulting patterns of regional economic change.

1.5. Core concepts

There are two important groups of concepts that are used in the dissertation:

- 1. technological change, structural change, industrial restructuring;
- 2. worker reallocation and labour mobility.

To establish a common understanding with the reader as well as to ensure the consistency in the usage of terms these concepts are clarified below.

Technological change is understood generically as the process of introduction and diffusion of new technologies on the market. The market dimension ensures that these are not mere technical inventions, that are considered as leading to technological change, but rather commercialised technologies being put to the productive use. Technological change may be radical (e.g., introduction of a new GPT) leading to the revolutionary upgrade of the technological frontier within which the economy operates. It may also be incremental (e.g., refining a new technology and putting it to a broader scope of productive uses) allowing the economy to utilise the whole potential of technologies within given technological frontier.

Technological change entails changes in a composition of the economic system – that is, emergence of new sectors, products, and services, extinction of the old ones as well as redefined weights and overall importance of sectors, products, and services present on the market. This phenomenon is captured by the notion of *structural change*. It should be distinguished from *industrial restructuring* which is an economic transformation observed at lower levels of aggregation (e.g., individual industries). It describes changes in how industries produce their output and in competition conditions, under which the production process takes place.

The notions of *worker reallocation* and *labour mobility* are used interchangeably in this dissertation. The basic category is a *job mobility* which takes place whenever an individual switches a firm of employment. Depending on whether the job mobility involves certain characteristics, one can further distinguish between *occupational mobility* (when a job switch involves a change in the occupation that an individual is employed at), *inter-industry reallocation* (if industry of employment is changed), as well as *inter-regional* or *geographical labour mobility* (when a job mobility involves switching a region of employment). One may also mention the *mobility across labour market states* which takes place when an individual becomes employed from unemployment/non-participation or vice versa.

1.6. Outline of the dissertation

The core of this document contains four articles that are introduced with an opening part – the *kappa* (Swedish for 'coat'). The latter provides the discussion of the overall theoretical and methodological approaches employed in this dissertation. It aims also at contextualising the research performed in individual articles and synthesising their major findings. The following chapters are included in the *kappa*.

Chapter 1 is an introduction. It provides the background and motivation for the research project underlying the dissertation. In this chapter, aim and research questions to be answered are formulated, expected contributions are presented, research setting is summarised, and core concepts are introduced.

Chapter 2 introduces the reader to various perspectives on the long-term (regional) economic development. It provides a review of several theoretical frameworks that were developed to explain the long-term fluctuations in the rhythm of economic growth, the so-called Kondratiev waves. The discussion in this chapter suggests that the long-term economic development is an immensely complex process, explaining which requires accounting for a variety of mechanisms at different levels of the economy.

Chapter 3 provides a discussion of labour market dynamics during the technology-induced economic transformation. It suggests that the labour market plays a dual role in this process. On the one hand, it acts as an adjustment mechanism smoothing the transition to newly emerging industrial systems. On the other hand, being the mediator of knowledge transfer, the labour market may itself facilitate or restrict the opportunities for economic transformation.

Chapter 4 integrates the insights from previous two chapters into the evolutionary model of economic change that lies at the core of the explanatory framework behind this dissertation.

Chapter 5 introduces the ontological and epistemological perspectives informing the research. It also provides a discussion of major data and methodological considerations that emerged during the research process.

Chapter 6 is a contextual chapter that discusses the current wave of technological change induced by the introduction and diffusion of the

information technology as a general purpose technology. Particular attention is paid to the labour market impact of the information technology.

Chapter 7 places the research in Sweden by focusing on three major topics. First, the major patterns of the development of the Swedish economy in the period between 1985 and 2010 are overviewed. Second, an overview of the history and current state of Swedish IT sector is performed. Finally, institutional aspects of the Swedish labour market are discussed.

Finally, Chapter 8 summarises findings from individual articles, synthesises them through the prism of the current debates in the literature as well as discusses the overall contribution of the dissertation.

2. Perspectives on the long-term economic development

2.1. Long waves of economic growth – 'the pulsation of modern capitalism'¹

The seven years of plenty that occurred in the land of Egypt came to an end, and the seven years of famine began to come, as Joseph had said

(Genesis 41: 53, 54)

That economic growth is not linear over time but exhibits cyclical patterns – i.e., periodic upswings and downswings in the growth rate – is not a new story in economic research. The Kitchin inventory cycles (2-4 years), Juglar fixed investment cycles (7-11 years), and Kuznets infrastructural investment cycles (15-25 years) have solid theoretical and empirical foundations, and are considered a stylised fact by most economic growth scholars nowadays (van Duijn 1983; de Groot 2006).

At the same time, the evolution of the economy is characterised by longer dynamic phases of booming economic development that are followed by more recessed and unstable economic conditions (O'Hara 2001). As put by Hansen (1964),

the past economic experience [...] discloses prolonged periods of relatively buoyant times, extending far beyond the boundaries of the major business cycle, and similarly prolonged periods of more or less chronic hard times, within which, however, the swings of the business cycle occur. (p. 56)

¹ Louçã (2007, p. 766)

These longer cycles transcend short-term fluctuations in economic activity and make a more lasting impact on economic development not only in the sense that they last longer, but also because their repercussions embrace the entire socio-economic system (Göransson and Söderberg 2005). They are known as *long waves* of economic growth, or *Kondratiev waves*, after the Russian economist Nikolai Kondratiev who was among the first scholars to bring an empirical observation of the long-term fluctuations in economic activity to international attention (Kondratiev 1925).

Not being uncontested², the phenomenon of long waves has attracted a substantial amount of research. Some contributions were predominantly methodological and aimed at the empirical identification of long waves in economic data (Haustein and Neuwirth 1982; van Ewijk 1982; Metz 1992; Dietbolt 2005). Other scholars focused on investigating triggering factors behind long waves.

The latter strand of research has been based on the premise that durable structures of the economy evolve over time in complex patterns of prosperity and decline (O'Hara 2001). These structures emerge around different forms of capital (fixed capital, knowledge capital, institutional capital, etc.) that have been emphasised by different scholars as triggers of Kondratiev waves. In particular, researchers have accentuated long-term fluctuations in profit rates (Mandel 1975; Gordon 1980; Mandel 1980), (re)investments in fixed capital (Kondratiev 1925; Graham and Senge 1980; Forrester 1981; Sterman 1985), and prices of primary commodities (Marchetti 1980; Rostow 1985; Volland 1987). The role of wars has also been repeatedly stressed (Rose 1941; Thompson and Zuk 1982; Goldstein 1985).

All these scholars reserved a prominent role for technological innovation. However, it was not until Joseph Schumpeter (1939, 1942, 1961) that radical technological breakthroughs were identified as the major force propelling economic dynamics and put at the core of the explanatory mechanism behind long waves.

The starting point of a long wave à la Schumpeter is an introduction of one or several radical innovations that open up new investment opportunities and have an innate potential to boost economic growth rates in the long run. These opportunities unleash a great commercial potential and attract a swarm of

² The phenomenon of long waves has been criticised, in particular, for the lack of convincing evidence of their strict periodicity and, therefore, cyclical nature.

imitators, resulting in the emergence of new industries and rapid expansion of new demand. In the words of Schumpeter, radical innovations 'tend to cluster, to come about in bunches, simply because first some, and then most, firms follow in the wake of successful innovation' (Schumpeter 1939, p. 75).³ These bunches are gradually disseminated to adjacent sectors, attracted by cost advantages of new technologies. As imitators add incremental improvements to the radical innovation and expand their activities, new technologies are pushed into the economy, taking it into the upswing of a long wave.

In this process, competition between firms on the basis of quality of products and services takes the place of price competition as the market coordination mechanism. Prices are still significant, but whereas they are central to the adjustment to limiting conditions, innovation is key for overcoming previous limiting conditions and setting the new ones (Hanusch and Pyka 2007).

As new technologies, products, and processes are introduced, existing ones become obsolete and are driven out of the market. Schumpeter used the notion of *creative destruction* to characterise this process. As new technologies diffuse in the economy, productive capital specific to old technologies – i.e., installed machinery and equipment, skills and experience embodied in workers, infrastructures used for transportation and energy distribution, etc. – can no longer be used. Firms face the choice between either adopting the new technologies or leaving the market. In that respect, creative destruction is the dual process of expansion and competition between technologies that takes place during the upswing phase of a long wave (Verspagen 2004).

However, there are inherent limitations to the developmental potential of new technologies. As the bandwagon of imitators grows, the scope for incremental innovation narrows down gradually and technological opportunities approach exhaustion. At this stage, the technology reaches maturity, its diffusion gets saturated, and the competition between firms increasingly takes the form of price competition. The ongoing entry of new firms increases competition even further, driving down profits as well as the aggregate economic growth rate. Eventually, when the saturation process is complete and the intensity of price competition is at its peak, the recession turns into a depression. For the economy to be taken into a new upswing of economic growth, a new set of

³ However, for an innovation to trigger a long wave Schumpeter nowhere insists that it must be part of a cluster of innovations; only that it is radical enough itself (Silverberg 2007).

radical innovations is needed. Schumpeter himself, however, did not offer any plausible explanations for this lower turning point.

Two important qualifications with regards to the Schumpeterian framework should be mentioned. First, for Schumpeter the notion of innovation is all encompassing.

Technological change in the production of commodities already in use, the opening up of new markets, or of new sources of supply, Taylorization of work, improved handling of material, the setting up of new business organizations such as department stores - in short, any "doing things differently" in the realm of economic life - all these are instances of what we shall refer to by the term Innovation. (Schumpeter 1939, p. 84)

Second, Schumpeter distinguished between an innovation – a commercialised new product – and an invention, which he treated as something belonging purely to the realm of science and technology.

In many important cases, invention and innovation are the result of conscious efforts to cope with a problem independently presented by an economic situation or certain features of it [...]. Some *innovation* is so conditioned, whereas the corresponding *invention* occurred independently of any practical need. This is necessarily so whenever innovation makes use of an invention or a discovery due to a happy accident, but also in other cases. (Schumpeter 1939, p. 85; emphasis in the original)

In other words, what is technologically possible is broader than what is economically feasible and meaningful. Schumpeter, therefore, claimed that 'economic logic prevails over technological' and that technologies are exogenous to the economy and should stay 'outside the domain of economic theory' (Schumpeter 1961, pp. 11, 13-14).

2.2. Endogenising technology into Schumpeter's framework

While some prominent attempts to analyse the development of technologies *per se* and their impact on the economy date back to the 1960s (e.g., Landes 1969), it was not until the 1980s that researchers truly endeavoured to investigate technological change and innovation as such, bringing the issue

of technologies' evolution to the centre of the economic growth debate. Two strands of research emerged in this literature.

The first one introduced the concept of *general purpose technologies* (GPTs) as engines of economic growth (Bresnahan and Trajtenberg 1995; Helpman and Trajtenberg 1998; Jovanovic and Rousseau 2005). Originating from mainstream economics, this approach focused primarily on market-centred mechanisms of diffusion of radical technologies.

The other strand of research emphasised a broader range of aspects – social, institutional, cultural, etc. – and suggested that the introduction and subsequent diffusion of radical technologies brought about the succession of *techno-economic paradigms* (TEPs) that provided long-term frameworks for the socio-economic development (Freeman and Perez 1988; Freeman and Louçã 2001; Perez 2010).

2.2.1. General purpose technologies – the 'engines of growth'⁴

In their seminal contribution, Bresnahan and Trajtenberg (1995) claimed that

at any point of time, there are a handful of 'general purpose technologies' (GPT's) characterized by the potential for pervasive use in a wide range of sectors and by their technological dynamism. As a GPT evolves and advances it spreads throughout the economy, bringing about and fostering generalized productivity gains. (p. 84)

Simply put, GPTs are path-breaking technologies that push the technological frontier outward and, through their diffusion in the economy, boost the rates of economic growth (Strohmaier and Rainer 2013; Ljungberg 2016).⁵ Any GPT possesses three input-side characteristics, namely: pervasiveness, technological dynamism, and innovation spawning (Jovanovic and Rousseau 2005; Lipsey et al. 2005; Bresnahan 2010).

Pervasiveness implies that a technology 'performs some generic function that is vital to the functioning of a large number of using products or production systems' (Rosenberg and Trajtenberg 2004, p. 65). It allows the GPT to

⁴ Bresnahan and Trajtenberg (1995, p. 83)

⁵ Other related terms may be found in the literature, including macro invention (Mokyr 2002), basic innovation (Freeman et al. 1982), strategic invention (Usher 1954), and epochal innovation (Rosenberg and Trajtenberg 2004).

generate new industries and affect the productivity growth across a wide range of uses and sectors (Martin 1995; Hung and Chu 2006).

Technological dynamism means that 'continuous innovational efforts increase over time the efficiency with which the generic function is performed, benefiting existing users, and prompting further sectors to adopt the improved GPT' (Rosenberg and Trajtenberg 2004, p. 65). That is, the GPT provides a wide scope for potential improvements that are an important source of value creation in the process of technological change (Trajtenberg 1990).

Finally, *innovation spawning* implies that 'technical advances in the GPT make it more profitable for its users to innovate and improve their own technologies' (Rosenberg and Trajtenberg 2004, p. 65). Thus, the GPT should exhibit strong complementarities with existing or potential new technologies, fostering complementary investments and technical change in user sectors.

While the research in economic history has identified a wide range of technologies that possess (some of) GPT characteristics (for reviews, see David and Wright 1999; Field 2008), most scholars agree that three (clusters of) technologies may undoubtedly be classified as GPTs, namely: the steam engine, electricity, and information and communication technologies.

Bresnahan (2010) referred to the combination of GPT's technological dynamism and innovation spawning as *innovation complementarities* that magnified effects of incremental innovation in the GPT itself and enabled innovation in other sectors. For instance, the productivity of R&D activities in user sectors increases as a result of innovation in the GPT. At the same time, the application in a wide range of sectors raises the returns to advances in the GPT itself, thus, generating even broader opportunities for new applications. That is, GPT's utility increases in combination with other activities (Dahmén 1988; Ljungberg 2016). These cumulative feedbacks reinforce the process of technological change and lead to sustained productivity gains and higher aggregate growth rates (Petsas 2003).

Contribution of GPTs to economic growth has been investigated in two generations of growth models, which differ in their assumptions of how and when a technology is identified as a GPT (Cantner and Vannuccini 2012). In the first generation of models (Bresnahan and Trajtenberg 1995; models collected and reprinted in Helpman 1998), a new technology is recognized *ex-ante* as a GPT. Once the technology arrives, economic agents face a

decision whether to allocate resources to GPT-related R&D activities, based on their expectations of future returns. In the second generation of models (van Zon et al. 2003; Carlaw and Lipsey 2006; Harada 2010; Carlaw and Lipsey 2011), the assumption is made that the designation of a new technology as a GPT is an *ex-post* mental construct derived from an evidence that the technology provides a wide range of productive functions in the economy. 'During the innovation process, the actual pervasiveness of an innovation, when and if it arrives can only be guessed at' (van Zon et al. 2003, pp. 8-9). Therefore, the mechanism of the GPT arrival is modelled endogenously through the accumulation of knowledge, allocation of resources to R&D activities, or routinised decision-making.

The arrival of a new GPT implies that it needs to compete with technologies that are already established on the market. In this context, an impediment to the adoption of the GPT is a large number of complementary components that have to be developed by application sectors in order to bring changes to their established production processes (Helpman and Trajtenberg 1998). Also, pre-existing technologies are not static, and their incremental improvement may further delay the diffusion of the GPT. In that respect, the larger are investments in old technologies, the higher is the efficiency threshold the GPT faces, and the longer is its adoption time.

At once, if innovation complementarities, that pre-existing technologies have developed in potential application sectors of a GPT, are not (prohibitively) specific to the old technologies, they may become complements to the GPT (Bresnahan 2010). In other words, dynamic complementarities emerge between the GPT and the technologies it replaces. Costs and delays, associated with developing complementary applications of the GPT, are then dependent on the extent to which these applications have already been developed for the pre-existing technologies.

Scholars of technology diffusion have known for long that the diffusion process is very gradual (Griliches 1957; Hägerstrand 1967; Mansfield 1968). Even in presence of complementarities between the old and new technologies, the initial stage of the GPT diffusion is a period of intermittent growth, when early adopters need to make required investments, whereas laggards keep producing with the old technology. Some constraints appear on the supply-side as the value of the GPT is realised after costs of its adoption has fallen below a certain threshold or the technology itself has acquired certain features. There are also demand-side constraints: economic agents are
heterogeneous (e.g., lower value users adopt later); adjustment costs (e.g., from learning-by-adopting) may appear; also, anticipated improvements in the GPT may lead to a rational delay. Besides, the pervasive use of the GPT and emergence of innovation complementarities require aligning incentives of economic agents located far from each other along the time and technology dimensions (Bresnahan and Trajtenberg 1995). This alignment is not unproblematic as the process of knowledge generation and diffusion is characterised by uncertainty and asymmetric information (Arrow 1962). Therefore, the aggregate outcomes of the GPT diffusion take a long time to materialise (David 1990).

From the perspective outlined above, it appears that the research on GPTs is a contemporary endeavour to revitalise neoclassical endogenous growth theory by adding a Schumpeterian flavour to it (Cantner and Vannuccini 2012). Its strength lies in providing elegant analytical tools to understand the long-term fluctuations in economic growth, while focusing rather narrowly on the market mechanisms of the technology diffusion and its effect on the productivity dynamics. It demonstrates how, being introduced in particular niches of the market or in specific sectors, a GPT is developed until it becomes the core technology that shapes new production configurations through industrial interaction, demand pressures and technological competition. Introduced in mainstream economics, the concept of GPT is now widely used in various research fields, such as innovation studies, economic history, economic geography, organisational studies, etc.

At the same time, focusing on the introduction and diffusion of technologies from the market perspective, the GPT literature disregards other important systemic dimensions of technological and economic change, such as social, institutional, and organisational. These became crucial components of research efforts parallel to GPT theorising, namely the research on technoeconomic paradigms.

2.2.2. Techno-economic paradigms

The 1970s-1980s marked a revived interest in Schumpeter's ideas about innovation and its disequilibrium character, creative destruction as the driving force of economic transformation, and aggregate fluctuations as inseparable features of the capitalist process of development. These ideas gave rise to the debate, that would later be characterised as the 'Schumpeterian renaissance' (Freeman 2007), and became the basic tenets of the neo-Schumpeterian economic research (Silverberg 2007).

Taking a historical approach to innovation and its impact on the economy, contributions in this stream of the literature argued against the relevance of clustering of radical innovations in time and focused on the process of their diffusion (Freeman et al. 1982; Freeman and Soete 1997; Freeman and Louçã 2001). As an outcome, several working hypotheses were formulated with regards to the impact of radical innovations on the economy.

First, it is not the introduction of new technologies *per se* that brings the economy into the upswing of a long wave, but rather their diffusion. Their maximum impact is observed when the diffusion has gone around the half way of a logistic process (Silverberg 2007). Therefore, whereas the technology diffusion is a gradual process, it has a tremendous long-term impact (Verspagen 2004).

Second, while radical innovations expand the technological frontier and define directions for new investments, the incremental innovation is instrumental for the diffusion process (Enos 1962). Product and process improvements have an impact not only on productivity growth and market expansion, but also on the new technology itself. In that respect, the radical innovations are not spread in isolation as the opportunities, offered by new and pre-existing technologies, are recombined in new ways. Thus, '[w]hat is being diffused at the end of a diffusion process may be rather different than that which started the diffusion process' (Freeman 1982, p. 6).

Third, if it is radical enough, a new technology leaves a deep structural impact on the economy. Its diffusion redefines a set of opportunities and constraints for innovative activities of economic agents. To capture this, Dosi (1982) introduced the concept of *technological paradigm*.

In broad analogy with the Kuhnian definition of a "scientific paradigm"^[6], we shall define a "technological paradigm" as "model" and a "pattern" of solution of selected technological problems, based on selected principles derived from natural sciences and on selected material technologies. (p. 152)

In other words, technological paradigm is a collectively-shared logic regarding the potential, relative costs, market acceptance, functional

⁶ Paradigm is an all-embracing conception of the world, on which all knowledge and interpretations are built, until a new paradigm replaces the old (Kuhn 1962).

coherence and other aspects of a new technology (Perez 2010).

A new technological paradigm emerges out of the crisis in the previous one. When profit rates in mature industries decline, economic agents search for new investment opportunities in technologies that have a great potential for generating profits (Perez 2002). As the problem of economic transformation is a problem of adaptation, the patterns of resource allocation and demand composition change in response to newly opening technological opportunities (Metcalfe et al. 2006). Perez (2004, 2010) referred to the switch between two successive technological paradigms as a *technological revolution*.

When a new technological paradigm is being established, 'industries differ significantly in the extent to which they can exploit the prevailing general natural trajectories, and these differences influence the rise and fall of different industries and technologies' (Nelson and Winter 1977, pp. 59-60). New, or renewing, industries – whose technological base is (closely) related to the new technologies – are exposed to broader opportunities and experience more dynamic development. Industries, that are not connected to the emerging technological paradigm to the same extent, develop along less dynamic paths (Nelson and Winter 1982; Castellacci 2008). In that respect, changes in the composition of the economy, induced by growth and decline in various industries, redefine the economy-wide relations between productivity, employment, and output as well as contribution of each industry to the aggregate growth of the economy (Metcalfe et al. 2006).

According to their role in the transformation process, the key sectors of each technological revolution may be divided into (Perez 1983, 2010):

- *motive branches*, producing key factors with pervasive applicability and other inputs directly associated with them; they define conditions for development within a new technological paradigm;
- *carrier branches*, the most visible and active users of new inputs that are best adapted to the new organisation of production; they generate a great variety of investment opportunities and have a great impact on the rhythm of economic growth;
- *induced branches* that have existed before but are modernised and assume new roles in the economic system. Their development is both a consequence of and a complement to the expansion of motive and carrier branches and is indispensable for diffusion of new technologies.

A vast reorganisation of the economy, induced by the establishment of a new technological paradigm, implies that an inherent (and recurrent) feature of any technological revolution is a crisis of structural adjustment that brings about large scale structural unemployment and social unrest (Freeman and Louçã 2001; Louçã 2007). Therefore, technological development, economic growth, and crises are inseparably connected in history (Kleinknecht 1987).

In that respect, the impact of technological revolutions goes beyond transforming the economy and requires redefining rules and regulations, such as specialised training, norms and other institutional facilitators (Perez 2010). In other words, the diffusion of radical technologies can only proceed with the major institutional change (Freeman and Perez 1988). The economic system should then be seen as comprised of two related sub-systems - technoeconomic and socio-institutional – whose co-evolution determines the mode of development. When a new set of radical innovations emerges, there is an impulse for the techno-economic sub-system to adopt new technologies that offer new profitable opportunities. However, while the techno-economic subsystem is more flexible in adopting the new, it takes longer for the more inertial socio-institutional sub-system to respond to challenges posed by the newly emerging technological paradigm (Castellacci 2007). The temporal mismatch between the adjustment in each of these two sub-systems may, therefore, delay the take-off of the new paradigm as social, organisational, and institutional changes are required for its transformational potential to be fully realised. As time goes by and complementarities between two subsystems are built, the new mode of development eventually sets in. In that respect, the notion of technological revolution is all encompassing so that it covers not only the technological innovation, but also its institutional, organisational, social and political dimensions (Freeman and Louçã 2001; Hanusch and Pyka 2007).

The impact of technological change on the economy should, therefore, be analysed through the prism of transformation of complete systems – *technoeconomic paradigms* (TEPs – that intertwine technology, culture, economy, and organisation (Freeman and Perez 1988). The succession of TEPs involves a gradual replacement of technical, organisational and managerial modes of production. A technological revolution should then be seen more generally as

a major upheaval of the wealth-creating potential of the economy, opening a vast innovation opportunity space and providing a new set of associated generic technologies, infrastructures and organisational principles that can

significantly increase the efficiency and effectiveness of all industries and activities (Perez 2010, p. 190)

From this perspective, the focus shifts from the long-term fluctuations in economic growth to transformative effects of technological revolutions on all aspects of the socio-economic life.

It is, therefore, no longer possible to ignore particularities of the specific technological revolution and the nature of the particular set of technologies being diffused. Each major technological breakthrough develops in its historical time and in particular context, whereas various systemic factors, interacting in complex ways, shape its development path (Verspagen 2004).

In that respect, the TEP approach does not provide a (completely) deterministic perspective on the long-term development of the economy. It resembles the GPT research with respect to its focus on the diffusion of new technologies as well as the necessity of those to be pervasive and build complementarities with pre-existing technologies. At the same time, by incorporating the environment, in which the diffusion process takes place, into the core of its explanatory framework, the TEP approach emphasises a broader spectrum of potential implications of technological change for the development of the economy.

The TEP approach may, however, be criticised for taking the predominantly macro-level perspective and not paying enough attention to processes at the micro-level of the economy, e.g., the behaviour of economic agents involved in the development and diffusion of new technologies. To become a more powerful tool for analysing processes at the micro- and meso-levels of the economy, the approach would benefit from integrating the insights from innovation studies, evolutionary economics, and evolutionary economic geography, which emphasise the importance of knowledge, learning processes, social networks, etc.

2.3. Decomposing the technology diffusion process

Universal truths without reference to time and space are unlikely to characterise economic affairs.

(Penrose 1989, p. 11)

The technology diffusion process is unpredictable, as potential applications of the technology are continuously explored, tested, expanded, modified, or rejected (Louçã 2007). Even though the technology-induced economic transformation is pervasive, the forces, shaping evolution of demand and technological base in various industries, operate unevenly and cumulatively (Metcalfe et al. 2006). The adaptive development in new and old parts of the economy is, therefore, the primary process, while the aggregate economic growth is a secondary outcome. In that respect, macroeconomic explanations of economic transformation should be complemented by analyses of micro-and meso-level determinants of technology adoption and diffusion as well as their implications for functioning of the economy as a whole (Nelson and Winter 1974; Metcalfe et al. 2006).

Besides, the impact of technological revolutions is not equally distributed in space. The contemporary view of regional economic dynamics builds upon the principle of increasing returns (Krugman 1991; Fujita et al. 1999) that are connected to the capacity of regions to generate innovation and to receive and exploit ideas, technologies and market changes from the outside world (Jaffe et al. 1993; Boschma and Lambooy 1999). This capacity is dynamic and defines the ability of regions to translate technology-induced transformation pressures into the competitive regional performance. This ability as well as the roles of and relations between different regions change over time, implying that technological revolutions are essentially geographical in character (Lundquist and Olander 2009, 2010).

In the words of Mokyr (2005):

Technological innovations [...] do not affect the aggregate level of economic activity abruptly: they need to diffuse from region to region, from activity to activity, cross boundaries and seas, be evaluated, adapted, and refined. Their

promoters have to dislodge the entrenched, persuade the sceptic, and reassure the fearful. (p. 285)

Therefore, understanding the technology diffusion requires acknowledging that it is the process that unfolds over time and proceeds unevenly across different industries and regions.

2.3.1. Unfolding technological transformation in time...

Based on years of empirical investigations of the Swedish economic growth patterns, Schön (1998, 2010, 2012) identified the recurring phases of technology-induced economic transformation that shared common features with regards to investment patterns, technology diffusion processes, and systematic lead-lag relationships between various industries and institutional structures. Each of the phases lasts about 50 years and has the common structure: transformation period – rationalisation period – structural crisis.

The *transformation period* (first investment period) is characterised by the introduction and early diffusion of a new GPT and brings about deep changes in institutions, markets and industrial organisation. This is a period of 'changes in industrial structures, where resources are reallocated between industries, and diffusion of basic innovations within industry [...] provides new bases for such reallocation' (Schön 1998, p. 399).

New activities, emerging around a new GPT, take the lead in this period. Investments are directed to increasing the capacity in new areas of production and are predominantly of a long-term character (Schön 2010). The GPT diffusion across industries leads to the emergence of complementarities between the old and new parts of the economy. First-mover industries⁷, implementing the new technology into their production processes, take the lead in the transformation and contribute greatly to the aggregate growth of the economy. At once, the increasing demand for intermediate products generates growth even in less technology-intensive industries acting as subcontractors. Nevertheless, the productivity and growth rates across various industries diverge during this period. The aggregate productivity growth is driven by the resource reallocation between low- and highproductivity sectors. Factor markets change in response to the transformation

⁷ Motive and carrier branches discussed in Section 2.2.2.

of relative price structures. Institutions adapt to new developments to create stable conditions for continuous growth.

The *rationalisation period* (second investment period) manifests in the maturation and ubiquitous spread of new technologies through the technical standardisation and dissemination of competences. This is a period of the "concentration of resources to the most productive units within the branches and measures to increase efficiency in the different lines of production" (Schön 1998, p. 399).

Technological development fades into the background during this period. The applications of the GPT are increasingly standardised and spread efficiently to older parts of the economy. Investments are directed to reduce costs in existing capacities; they are short-term and resource-saving in character and have a more direct impact on the productivity, growth and real income. Enhanced qualifications of the workforce around the increasingly standardised technology facilitate further productivity gains. Overall productivity growth stems from the intra-industry optimisation as economies of scale increase in importance and trigger an increased competition. The period is characterised by a relative stability in relative price structures and institutions.

With time, falling profits change investment patterns and lead to the structural crisis that weakens old economic interests and creates conditions for a new GPT to set in. It is a dramatic moment of Schumpeterian creative destruction, when new opportunities are created, and investments are re-directed towards technologies demonstrating the GPT potential (Schön 2012).

Of course, the transformation and rationalisation take place simultaneously in the economy. However, capacity-increasing and cost-reducing investments are unevenly distributed over the ordinary business cycle: there are observed long-term shifts of such emphasis (Schön 1998, 2010). In that respect, the switches between the transformation and rationalisation periods occur regularly between successive technological revolutions. Crises form particular sequences in these shifts.

While Schön's model is an empirical heuristic uncovering different stages of the GPT diffusion (Henning 2009), it has important implications for analyses of the long-term economic development. First, the model provides an insight into the behaviour of economic agents at the micro-level with respect to investment patterns, resource allocation, and the nature of competition. In that

respect, it fills in the gap in the TEP approach. Second, it demonstrates that the behaviour of economic agents changes as the technology is spread in the economy, which has an impact on the overall rhythm of economic growth. In that respect, it emphasises the importance of the time dimension for analyses of technology-induced transformation of the economy.

2.3.2. ... and space

The forces of economic transformation are differentiated in space (Henning 2009; Martin 2015). The national trajectory of economic transformation and its aggregate outcomes are composed of a wide range of regional trajectories (Henning et al. 2016). While much research has been devoted to endogenous determinants of regional development paths, such focus draws attention away from other critically important systemic factors that shape regional development, e.g., the shifts in macro-economic conditions (Shearmur 2011; Pike et al. 2016). Particular characteristics of the national economy may impose barriers to movements of capital and labour that inhibit the automatic self-correcting adjustment in the regional system, a mechanism so beloved by mainstream economists (Martin 2015). Imbalances in the regional system are both the outcomes and important determinants of the process of economic transformation. Such imbalances are particularly pronounced in the outcomes of transformation process between the most dynamic regions at the top of regional hierarchies and the most peripheral regions (O'Leary and Webber 2015).

Such systemic perspective, in which roles of and relations between regions change in the process of the GPT diffusion over time, and its implications for Schön's framework laid the basis for the geographical reference cycle model (Lundquist et al. 2008a, b, c; Lundquist and Olander 2009; Henning et al. 2016). It premises on the inherent geographical character of technology-induced economic transformation in that starts in certain industries and regions and, with specific time patterns, spreads through the whole regional and industrial landscape. The national growth cycle is, thus, an aggregate result of regional development, determined by unique characteristics of the regional system.

The theoretical framework of the geographical reference cycle suggests a sequential order of economic transformation in two interlinked dimensions: the GPT diffusion across industries and across regions that possess various

endogenous and systemic characteristics. It provides the insight into when and where the transformation process starts and how it is diffused through the regional economic landscapes.

In compliance with the strands of research discussed above, at the early stages of the GPT diffusion the economic transformation and aggregate growth are driven by the first-mover industries that embed the GPT into their production processes. These industries generate the new demand in the economy. As the GPT is standardised over time, a broader spectrum of industries benefit from building complementarities with the GPT, expanding the new demand in the economy and creating an additional boost to the aggregate economic growth.

These developments are reflected in the spatial pattern of the GPT diffusion over time. Regions are differently equipped with the ability to develop, absorb, implement, and commercially translate the GPT-induced transformation pressures into the productive performance. That is, they possess different regional receiver and development competences (RDC) (Henning et al. 2016). The variation in regional RDC in different parts of the regional system gives rise to systemic lead-lag relationships between regions – that is, diverse transformation patterns – as the economy moves from the transformation stage into the rationalisation stage.

At the early stages of the GPT diffusion the first-mover industries, making use of new technological opportunities, mainly locate in regions at the top of the regional hierarchy (particularly, metropolitan areas). The latter are characterised by the most advanced regional RDC as they provide a superior access to infrastructure and innovating organisations (Feldman and Audretsch 1999; Simmie 2001) and are in a better position to generate knowledge spillovers necessary to facilitate diffusion of new technologies (Audretsch and Feldman 1996; Baptista 2003). The industrial diversity, facilitating opportunities for inter-industry knowledge spillovers (Jacobs' externalities), is of the crucial importance to the new and renewing industries (Neffke et al. 2011). Besides, large regions are magnets for talented individuals possessing high levels of human capital (Tödtling and Trippl 2005; Bosma and Sternberg 2014; Gordon 2015); thus, their advanced RDC are sustained through the constant inflow of knowledge embodied in individuals (Anokhin et al. 2016).

Beginning gently and experimentally in these sites, the new production takes time to become visible. While mature businesses initially dominate the scene, the realisation of the new technology potential leads to the redirection of investments towards, and resulting expansion of, new activities. This allows large regions to defend their absolute employment by sustaining job creation rates high enough to replace jobs that are phased out in declining sectors. Regions at lower levels of the regional hierarchy, lacking sufficient regional RDC, demonstrate slower growth in the new and renewing industries, resulting in an aggregate growth below national average as well as declining employment (Lundquist et al. 2008a). All in all, at the early stages of the GPT diffusion (the transformation period) the forces of technology-induced economic transformation and aggregate growth patterns are pronounced in favour of regions at the top of the regional hierarchy.

As the economy moves into the rationalisation, transaction costs gradually fall, large regional markets become strained on the resource side, new production starts to disseminate regionally and affects growth in value added, productivity and employment in a wider set of regions. The importance of factor costs rises, which benefits smaller regions that are able to provide low relative costs structures. Congestion effects in bigger regions force a regional diffusion and relocation of industries, particularly the ones drawing from the new technological opportunities (Henning et al. 2016). While the largest regions still dominate the scene, the aggregate growth at the national level is no longer constrained by regions at lower levels of the regional hierarchy (Lundquist et al. 2008a).

Particular spatial and sectoral outcomes of the GPT diffusion process depend on a range of interlinked mechanisms and channels, such as internal organisational structures of firms and sectors, imitation by new and existing competitors, emerging complementarities between the new and old parts of the economy, possibilities for decomposition and fragmentation of value chains as well as opportunities offered by standardisation, price competition and falling transaction costs.

All in all, the geographical reference cycle model provides the stylised depiction of the lead-lag relationships that characterise the regional system in the process of technology-induced economic transformation. Its key contribution is in demonstrating that the outcomes of this process are inherently determined by changing roles of different regions at different stages of the GPT diffusion process. One particular implication is that in order to understand the regional mechanisms and outcomes of technology-induced economic transformation one should account for the position of a region in the regional hierarchy.

2.4. A comprehensive summary

Ever since Kondratiev (1925) empirically identified the long waves of economic growth, substantial research efforts from scholars in various disciplines – including economics, economic history and economic geography – aimed at understanding the mechanisms, driving forces and regularities behind these waves and, more generally, the long-term economic development. One of the (early) significant contributions to this debate was in recognising the central role that technological innovation plays in this process: major technological breakthroughs are the initial impetus for generating upswings in the long-term economic growth rates (Schumpeter 1939).

Subsequent research focused on the market mechanisms that facilitate the diffusion of radical innovations (GPT research) as well as on how the socioeconomic environment, in which the diffusion process takes place, is both its determinant and outcome (TEP research). Recent empirical contributions provided additional qualifications to the GPT diffusion process as unfolding in time (Schön's framework) and space (the geographical reference cycle).

Summarising the perspectives on the long-term economic development reviewed in this chapter, one should acknowledge that the technologyinduced economic transformation is an immensely complex process. While its outcomes are clearly observed at the aggregate level, there is a great variety of interlinked driving forces and mechanisms at the sectoral and regional levels that have an impact on the behaviour of economic agents and, eventually, result in the economic transformation. Failing to recognise those, one cannot come up with any comprehensive explanations of the long-term economic development.

3. Technological change and labour market dynamics

3.1. General remarks

The economy is constantly hit by idiosyncratic disturbances that are driven by technological development, changes in economic agents' behaviour, terms of trade, institutional environment, etc. While there is an agreement that these disturbances are behind the (long-term) shifts in regional labour market dynamics (Bachmann and Burda 2010), their sources are debated.

Studying the relationship between employment growth in various industries and regional unemployment rates, Lilien (1982) put forward the hypothesis that the shifts in regional labour market dynamics were the outcome of permanent reductions in the demand for labour in particular industries. These reductions manifest themselves in displacements of workers, increasing regional unemployment rates and creating the need for reemployment of displaced workers in other industries, present in the regional economy.

Lilien's assumption that the labour demand shifts at the industry level take place as an independent source of regional labour demand reductions was criticised by Abraham and Katz (1986). In their view, regional labour market dynamics are propelled by aggregate shocks that have a different impact on different industries. In other words, regional variations in the labour market performance are explained by asymmetric responses of various industries, present in the regional economy, to the same aggregate shock.

In an attempt to discriminate between these two explanations, Robson (2009) demonstrated that the impact of the labour demand shifts on the regional labour market dynamics was transmitted through several channels. First, in the short run, regions, characterised by the high concentration of employment in cyclically-sensitive industries, are more vulnerable to the labour demand

shocks at the industry level. Second, the long-term shifts in the demand for labour across sectors, induced by aggregate shocks, put stronger pressures on regions with higher concentration of employment in declining industries. In that respect, aggregate shocks generate asymmetric responses of regional labour markets through their non-neutral impact on industries (Neumann and Topel 1991). Third, the regional labour market performance is conditioned by the degree of specialisation/diversification in the regional employment mix. More diversified regions tend to have a better capacity to reabsorb displaced workers through more efficient matching process (Moretti 2011). In that respect, analyses of regional labour market dynamics should acknowledge the potential impact of both industry-level demand shocks (following Lilien 1982) and the differential response of various industries to aggregate shocks (in the vein of Abraham and Katz 1986).

Technological change is an overwhelming aggregate shock that has an impact on both old and new parts of (regional) labour markets. Old jobs are destroyed when new technologies arrive (as existing job matches become ineffective and unprofitable), but are replaced by either more productive employment opportunities in existing sectors or completely new jobs in emerging ones (Mortensen and Pissarides 1998).

The discussion of the overall impact of technological change on the labour market divided researchers into technology pessimists, who pointed at the labour-saving and rationalising effects of new technologies that led to higher levels of unemployment as a result of the technology diffusion, and technology optimists, who considered technological change as a necessary condition for attaining full employment as its job creation effect was larger than the job destruction one (Matzner et al. 1990).

One should, however, look beyond the direct effects of job creation in expanding industries and job destruction in declining industries, and consider entire value added chains to take account of indirect effects (Freeman et al. 1982).

It is [...] necessary when evaluating the actual effects on employment of new technology, to differentiate between the direct (primary) employment effects resulting from the introduction of new technology in an enterprise and the indirect (secondary) effects which are transmitted by other mechanisms. (Matzner et al. 1990, p. 688).

Technological change is associated with changing patterns of demand for labour that lead to potentially severe problems of structural unemployment (Hanusch and Pyka 2007). In that respect, since the development of new and old industries in the process of technology-induced economic transformation is uneven over time and space (as it was demonstrated in the previous chapter), analyses of the impact of the new technology diffusion on job creation, job destruction, and other aspects of labour market dynamics should address its spatial and temporal dimensions.

3.2. Labour market response to structural change

The speed with which economies must adjust to change has increased over last decades due to forces of technological development and globalisation (Dahlman 2007). Ever since Mundell (1961), it has been acknowledged that the factor mobility is an important response mechanism of regional and national economies, which balances out the asymmetric effects of aggregate economic shocks. In that respect, the reallocation of workers is an inseparable component of technology-induced structural change (Aghion and Howitt 1992; Kremer and Maskin 1996; Mortensen and Pissarides 1998; Bauer and Bender 2004). Flexible labour markets facilitate an adjustment process by ensuring that changing labour demand patterns meet a response in the form of worker reallocation (Greenaway et al. 2002; Aghion et al. 2006; Bachmann and Burda 2010; Pastore 2013).

As the transformation pressures of the technology diffusion are unevenly distributed across the new and old parts of the economy, technological change leads to the relative or absolute long-run decline of some industries as well as the emergence and expansion of others. This implies changes in profitable opportunities that manifest themselves, among other things, in differential rates of job creation and job destruction between expanding and declining industries (Greenaway et al. 2000). The outcome of this process is the reallocation of labour, as the growth of new and already existing firms is necessarily related to the inflow of workers, while the contraction and closure of firms have the opposite effect (Davis et al. 1996).

This process, however, cannot be reduced to the simple picture of workers being displaced in declining industries and finding jobs in expanding ones. Individuals can still move from expanding to declining industries, while worker reallocations within industries are also common (Greenaway et al. 2000). The latter are particularly pronounced in technologically dynamic and rapidly expanding industries, playing key roles in the technology diffusion process, as they are characterised by high turbulence in terms of firm entries and exits (Antelius and Lundberg 2003; Mamede 2009).

Besides, direct job switches are not exclusive in the adjustment process (Sorm and Terrell 2000; Elliott and Lindley 2006; Bachmann and Burda 2010). There is a time lag between changes in the labour demand and response from the supply-side of the labour market. It takes time for displaced workers to find new jobs due to retraining, limitations in vacancies information accessibility, etc. Therefore, the shifts in the demand for labour lead to temporary increases in structural unemployment (Matzner et al. 1990; Neumann and Topel 1991). Flows to and from unemployment and nonparticipation play an important role in the adjustment process. Declining industries tend to release workers into unemployment, while the employment growth in expanding sectors is often fuelled by workers coming from outside the labour force, for instance, recent graduates from universities (Bachmann and Burda 2010; Arpaia et al. 2015).

Changes in the labour demand patterns, and the subsequent response in the form of inter-industry worker reallocation, have implications for regional labour market dynamics, as it has been for a long time claimed in the literature on dualism and structural change (Lewis 1954; Uzawa 1961, 1963; Chenery et al. 1986; Capasso et al. 2012). Key industries, associated with technological change, tend to exhibit different location patterns at different stages of the technology diffusion process. Thus, the distribution of job creation and job destruction is uneven across regions. Therefore, the shifts in the demand for labour manifest themselves spatially. At different stages of technology-induced economic transformation, labour markets of different regions are characterised by a varying capacity to reabsorb workers displaced in declining sectors, resulting in spatial labour market imbalances (Caroleo and Pastore 2010; Pastore 2013). Thus, the inter-industry reallocation of workers is often accompanied by the additional response of the labour market in the form of geographical labour mobility.

However, such response of the labour market may only be considered an effective adjustment mechanism if it leads to the mitigation of imbalances (that is, convergence) across regional labour markets. Whether this is the case is still debated in the literature, despite the huge number of studies on the issue (for a review, see Magrini 2004).

On the one hand, in mainstream economic models, the worker reallocation is directed to places that are more endowed with physical capital, leading to convergence in the capital-to-labour ratio and, therefore, per capita income and unemployment levels (Boldrin et al. 2001; Boldrin and Canova 2003; Barro and Sala-i-Martin 2004). In that respect, the labour mobility is an equilibrating force reducing disparities between regional labour markets.

On the other hand, the research in economic geography suggests that the worker reallocation widens, rather than reduces, disparities across regional labour markets by reinforcing agglomeration externalities (Krugman 1991; Baldwin 1999; Overman et al. 2002; Puga 2002; Epifani and Gancia 2005; Francis 2009). Two interlinked mechanisms contribute to that (Fratesi 2014). On the one hand, workers are attracted to regions characterised by higher rates of job creation and, thus, offering more job opportunities. These are predominantly large regions with already dynamic labour markets. On the other hand, the outcome of such reallocation is an increased labour supply in receiving regions, providing a supply-side stimulus to the job creation process. In that respect, the worker reallocation and job creation are involved in a virtuous circle of mutual reinforcement that results in divergent trends across elements of the regional system.

Another important aspect is that higher returns to production factors tend to be paid in regions where they already concentrate (Reichlin and Rustichini 1998; Funck and Pizzati 2002, 2003; Moretti 2004). In that respect, there are limits to reallocation of relatively unskilled labour because of the smaller differences in expected returns to relocation across regions. Skilled workers, however, potentially benefit more from relocation (Caroleo and Pastore 2010). Higher level of human capital allows them to adapt more easily to different jobs – particularly, new jobs generated as an outcome of technologyinduced economic transformation – and to benefit more from opportunities offered by various regions (Giannetti 2003; Coulombe and Tremblay 2009). Human capital gains of receiving regions are, in that respect, increasingly a force of the regional divergence (Kanbur and Rapoport 2005; Fratesi and Riggi 2007; Fratesi and Percoco 2014).

All in all, the technology-induced structural change results in complex patterns of worker reallocation across sectors, labour market states, as well as regions. However, it does not necessarily lead to the convergence between regional labour markets as there are cumulative mechanisms that favour the most dynamic labour markets of large regions. Therefore, while the worker reallocation is certainly a response mechanism to aggregate disturbances, it is not necessarily an effective mechanism of adjustment that allows to fully absorb their effects.

3.3. Labour market as the mediator of knowledge transfer

The role of individuals as carriers and distributers of knowledge has been extensively discussed in various fields, including sociology, organisational studies, economics, economic geography, etc. (e.g., Szulanski 1996; Lawson 1999; Lam 2000; Song et al. 2003; Bienkowska 2007; Breschi and Lissoni 2009; Maliranta et al. 2009). Human capital, embodied in individuals, is increasingly seen as the key input to firms' production processes. According to the knowledge-based view of a firm (Grant 1996; Nickerson and Olin 2004), firms' workforce is a key element of constructing and sustaining their competitive advantage (Hatch and Dyer 2004; Lund Vinding 2006; Coff and Kryscynski 2011).

Extensive research on the knowledge distribution across various levels of the economy has shown that the heterogeneity in knowledge endowments is a characteristic of firms (Srholec and Verspagen 2012; Grillitsch et al. 2016), industries (Pavitt 1984; Malerba 2002; Consoli and Rentocchini 2015), clusters (Jensen et al. 2007; Arikan and Knoben 2014) as well as regional (Cooke et al. 1998; Asheim and Coenen 2005) and national innovation systems (Freeman 1995; Carlsson et al. 2002).

Knowledge endowments of firms are not only important at single points in time, but also define dynamic properties of their success. Patterns of firm evolution and turnover of labour at the firm level are intrinsically related, particularly in knowledge-intensive industries, where firms' success is conditional upon their capacity to recruit skilled labour (Mamede 2009), such as hi-tech sectors (Baron 2004) and business services (Mamede 2002). The labour market may be seen as a coordination device that ensures coherence between the labour demand posed by firms and the pool of capabilities that are available on the supply-side. The 'recruitment-based competition'

(Sørensen 2004), and the resulting heterogeneous access of firms to human capital, impacts patterns of sectoral evolution over time. Evolving skill structures of industries are, in that respect, both the cause and the effect of shifting industrial regimes based on the generation, adaptation and diffusion of knowledge (Consoli and Rentocchini 2015).

As knowledge is embedded in individuals (Nelson and Winter 1982), it cannot be transferred easily without individuals' relocation. Ever since Alfred Marshall's work on industrial districts, it has been argued that the worker reallocation across firms is a mechanism of knowledge transfer (Song et al. 2003; Bathelt et al. 2004; Power and Lundmark 2004; Maliranta et al. 2009). Hiring workers is instrumental for firms to gain (relatively cheap) access to the superior knowledge developed elsewhere (Stoyanov and Zubanov 2014). Also, the worker reallocation facilitates the cross-fertilisation of previously unconnected knowledge (Saxenian 1994; Rosenkopf and Almeida 2003). Skilled workers take their knowledge with them and share it in a new workplace, while also acquiring new knowledge, establishing new links and social networks, and promoting new combinations of ideas (Laudel 2003; Trippl and Maier 2010). Moreover, mobile workers are more productive than their non-mobile counterparts (Hoisl 2007, 2009; Lenzi 2009; Singh and Agrawal 2011) as labour mobility improves the employer-employee match quality, which in turn implies efficiency gains and enhanced productivity (Jovanovic 1979; Liu 1986; Topel and Ward 1992).

Knowledge transfer takes place not only across firms, but also across regions. The research has extensively documented the influence of extra-local knowledge sources on firms' innovative performance and knowledge acquisition (Owen-Smith and Powell 2004; Gittelman 2007; Simonen and McCann 2008; Boschma et al. 2009). In that respect, the labour mobility is a principal mechanism of knowledge dissemination which helps to establish cross-regional linkages and spread knowledge and information across space (Coe and Bunnell 2003; Fratesi and Senn 2009; Breschi et al. 2010).

Technological change adds another dimension to the importance of human capital availability and knowledge diffusion. Learning capacity, possessed by highly skilled individuals, facilitates the innovative activity, especially at the early stages of the technology diffusion when the knowledge, related to the new technology, is tacit, sticky and, importantly, outside of the established knowledge domain (Gertler 2003; Baumol 2004). Therefore, the supply of adequate skills lags behind the demand for them, increasing structural skill

mismatches. At this stage, the knowledge transfer is conditional upon the mobility of a few skilled individuals.

As time goes by, the adoption of new technologies by firms and repeated learning-by-doing improve workers' performance, while the gradual standardisation of skills leads to their diffusion beyond the early technology adopters. In that respect, the uneven distribution of knowledge among economic agents provides incentives to explore the new technological paradigm, while coordinated efforts to systematise new knowledge facilitate the diffusion of skills throughout the economic system (Vona and Consoli 2015). The process of catch-up to the technological frontier is then conditional upon increasing the availability of human capital associated with the technology that is diffused (Benhabib and Spiegel 2005).

The worker reallocation does not, however, imply that the knowledge transfer happens automatically. Each potential job switch is characterised by an ease, with which it may be performed, and is influenced by three factors: (costs of) acquiring information about available job opportunities, desirability of a job, and direct costs of job switching (Lindley et al. 2002; Elliott and Lindley 2006; Neffke and Henning 2013). The latter are divided into 'hard' costs (e.g., housing-related costs, credit constraints, economic status of a partner) and 'soft' costs, primarily redundancies in the human capital associated with a job switch. It is costlier for a worker if her skills are not transferrable from an old job to the new one. In that respect, it is possible to distinguish between the general human capital – knowledge and skills used at an old job that remain useful in alternative jobs – and specialised human capital – knowledge and skills that are hardly (if at all) transferrable across jobs (Gathmann and Schönberg 2010).

Nowadays, workers, particularly skilled workers, possess highly specialised human capital as they invest heavily in education and training to acquire specific skills that allow them to carry out certain tasks (Neffke et al. 2016). The specificity of human capital develops in several domains. Earlier contributions underlined that human capital, accumulated by a worker at a particular job, was specific to a firm where she is employed (e.g., Becker 1964). More recent work suggests that the human capital is industry- (Neal 1995; Parent 2000; Sullivan 2010) or occupation-specific (Neal 1999; Poletaev and Robinson 2008; Kambourov and Manovskii 2009; Gathmann and Schönberg 2010; Neffke and Nedelkoska 2010). Besides, there are claims that regional variation in production technologies may impact the

accumulation of human capital (Rigby and Essletzbichler 1997). These domains of human capital specificity are complementary to each other. For instance, industry tenure benefits the worker, but the extent of those benefits depends on the occupation she is employed at (Sullivan 2010).

Specificity of human capital is an important factor affecting job switch decisions (Elliott and Lindley 2006). Workers tend to switch to jobs where they incur less shortages (skills to be acquired) and smaller redundancies (skills not needed anymore) in their human capital (Neffke and Nedelkoska 2010; Neffke et al. 2016). Simply put, workers switch to jobs that provide opportunities to reuse their skills. In that respect, an overlap in human capital requirements across different industries facilitates the inter-industry worker reallocation, while the lack of such overlap may constrain it. That is, workers tend to become employed in industries where skill requirements are similar to those in the industry of current employment – or, skill-related industries (Neffke and Henning 2013).

In the process of technological change, new industries emerge, others die out, whereas the very nature of production process changes. This results in the destruction of established linkages between previously skill-related industries. At the same time, however, the process of the new technology diffusion leads to the emergence of new complementarities between previously unconnected industries. In that respect, the whole network of skill-related industries is reorganised, which manifests itself in restructuring of the worker reallocation opportunities.

3.4. The GPT diffusion and the dual role of the labour market

The diffusion of a new GPT induces structural change, that is expansion of new industries, revitalisation of growth in renewing industries, as well as a decline in obsolescent industries. These processes lead to changing patterns of the demand for labour between the new and old parts of the economy, which, in their turn, generate response of the labour market in the form of worker reallocation across expanding and declining sectors. The problem, however, is that the skill requirements of new industries are continuously redefined as the new GPT undergoes incremental innovation, builds complementarities with existing technologies and is diffused to (more) application sectors. In the situation of technological turbulence and rapidly changing skill requirements, there is only a limited support coming from the educational system, due to its inertial development, in terms of providing suitable skills for the newly emerging technological paradigm. Therefore, particularly at the early stages of the GPT diffusion, the new qualifications in the workforce may not be sufficiently available, while obsolete qualifications abound.

In that respect, the adjustment of the labour market to the technology-induced structural change does not necessarily operate smoothly because of redundancies in human capital that arise in the process of worker reallocation. If such redundancies are too large, the direct reallocation of workers between declining and expanding sectors becomes problematic. Indeed, while the process of rapid employment growth in new sectors requires new skills, employees of mature industries cannot easily obtain them. Therefore, established skill structures on the labour market may restrict the opportunities for worker reallocation, and, more generally, labour market response to the technology-induced structural change.

These processes have important spatial implications. On the one hand, the GPT diffusion is uneven across regions at different levels of the regional hierarchy. The expansion of labour demand in the new and renewing industries may be expected to take place in the largest regions, at least at the early stages of the diffusion process. At the same time, displacement of workers from obsolescent sectors affects regions at all levels of the regional hierarchy, giving rise to spatial mismatches between the labour supply and labour demand and facilitating inter-regional reallocation of workers.

On the other hand, the GPT diffusion across industries and regions is conditional upon the availability and relocation of individuals who possess these skills. As the labour market is the mediator of the knowledge diffusion, it may both enable and restrict opportunities for technology-induced transformation. In that respect, the availability of human capital and its dissemination across regions is instrumental in the process of establishment of the new technological paradigm in regional economies.

4. GPT diffusion and labour market dynamics in an evolutionary model of economic development

4.1. Evolutionary theorising of economic change

The Mecca of the economist lies in economic biology

(Marshall 1948, p. xiv)

4.1.1. Foundations of evolutionary thinking

The point of departure in evolutionary thinking is the idea that the economy is in constant flux, subject to evolution through the ongoing process of change that 'revolutionizes the economic structure *from within*, incessantly destroying the old one, incessantly creating a new one. This process [...] is the essential fact about capitalism' (Schumpeter 1942, p. 82; emphasis in the original). According to Witt (2003, 2008), any evolutionary theory of economic change should be based on the following premises.

First, it should be dynamic. Economic change is the engine of growth (Metcalfe et al. 2006) that is based on the reciprocal dynamics of structural change and technological change (Quatraro 2012). Arising endogenously from within the economy, technological change stimulates industrial restructuring through transformative and adaptive mechanisms (Kogler 2015), which, in their turn, generate the creative response of economic agents and facilitate their innovation efforts leading to further technological change. Industrial restructuring is, in that respect, both the cause and the effect of an

endogenously determined process of technological change. This cumulative process ensures that the economic landscape is in constant unrest, in which previous events affect the probability of future events to occur (Boschma and Frenken 2006). In short, history matters (David 1985). Thus, analyses of economic development should explicitly take the time component into the consideration.

Second, evolutionary thinking deals with irreversible processes. The coexistence of random and systematic determinants of economic evolution (Nelson 1995; Verspagen 2005), together with its inertial and dynamic forces, makes it impossible for the economy to approach any kind of steady state, praised by scholars within mainstream economics.

The system is by itself not moving towards any sort of balance between forces, but is constantly on the move away from such a situation. In the normal case a change does not call forth countervailing changes but, instead, supporting changes, which move the system in the same direction as the first change but much further. Because of such circular causation as a social process tends to become cumulative and often gather speed at an accelerating rate. (Myrdal 1957, p. 13)

Simply put, economic development is 'an autocatalytic process in which change begets change' (Metcalfe et al. 2006, p. 9).

Finally, evolutionary theorising builds upon the generation and impact of novelty as the ultimate source of self-transformation. The restlessness of capitalism is based on its capacity to generate new knowledge; and the endogenous variation of knowledge makes economic systems so dynamic and versatile. Knowledge is not just a regular factor of production, in the manner that it can be detached from the production process; rather, it grows in the day-to-day conduct of economic activity (Metcalfe et al. 2006). It is the endogenous generation of knowledge that makes the process of economic evolution both adaptive and transformative in character (Fine 2000). The uneven distribution of knowledge across economic agents opens up various innovation and investment opportunities. Knowledge-based systems are, in that respect, necessarily evolutionary in their nature.

One of the most influential branches in evolutionary thinking traces back to Nelson and Winter (1982). In a nutshell, their theory explains economic evolution in terms of the dynamic interaction between heterogeneity (variety) of economic agents, competition and selection between them, as well as innovation process that reinforces heterogeneity and perpetuates growth.

Heterogeneity of economic agents is fundamental for evolutionary economic systems. In that respect, evolutionary accounts disregard the concept of representative agent, be it an individual or a firm. In a world, characterised by uncertainty and bounded rationality, firms act through routines, which constitute the organisational memory, through which firms develop their productive and commercial activities, and allow them to act without deliberation (Robert and Yoguel 2016). Within firms, activities are guided by routines at different levels. On the one hand, the routine repertoire includes knowledge held by firms' employees. On the other hand, there are organisational routines, which cannot be reduced to the sum of individual workers' knowledge. The routines determine organisational behaviour in terms of selection of products and production processes, investment behaviour, as well as search for new routines and solutions (Castellacci 2007).

Routinisation of firms' activities generates inertial forces and explains an inherent persistency of firm behaviour, resulting in a relatively stable pattern of economic activities over time. However, this stability is counteracted by dynamic forces pushing the economic system evolution, change and transformation (Castellacci 2007).

Just as animal species compete for their survival in the natural environment, heterogeneous firms compete on the basis of their routines. This competition is predominantly Schumpeterian, based on new products and technologies that require new routines. Competition is a *selection mechanism* that ensures the survival and diffusion of smart, fit routines (that are better adapted to the prevailing environment) and the demise of stupid, unfit routines (Verspagen 2005; Boschma and Frenken 2006).

While competition tends to reduce the initial variety over time, the dynamics of economic systems is not confined to the selection process but 'involves restless minds, constantly seeking to invent, to innovate and to connect with other minds in productive organisations in the quest for profits' (Foster 2011, p. 24). The innovation process enriches the variety so that the economic evolution is a never-ending process.

In reality, the variety generation and competitive selection take place simultaneously and mutually affect each other. In that respect, the economy is a complex adaptive system with positive and negative feedbacks.



Figure 2. Evolutionary theory of economic change à la Nelson and Winter (1982)⁸

One problem with this approach is that its explanatory framework is primarily firm-centred and, to a great extent, abstracts from the environment in which the selection and variety generation take place. However, these micro-level processes are affected by the constraints imposed by meso- and macro-level determinants of economic change (Cimoli et al. 2016).

⁸ Inspired by Castellacci (2007).

The economy may be considered as a complex structure of economic mesorules⁹ that determine the selection process among the routines that were developed to solve a specific problem (Dopfer et al. 2004). Economic rules are embedded in a broader environment of rules, such as behavioural, social, legal, political, etc.

The meso-rule system is hierarchical and ranges from long-lived rules that are applied to the whole economy to relatively short-lived contingent rules (Foster 2011). For instance, meso-rules applied in one particular sector are complementary to the economy-wide meso-rules upon which they rely. At the same time, these rules are not completely deterministic: it is impossible to predict which routines will be developed, given a meso-rule. Neither is it possible to foretell whether any micro-rule will be adopted as a meso-rule in the future.

Evolution of the economy in general, and the processes of selection and variety generation in particular, are, in that respect, reflected in the origination, adoption, adaptation and diffusion of meso-rules. Briefly put, economic evolution is a process of shifts in meso-rule sets (Dopfer et al. 2004) that is systemic rather than purely endogenous or exogenous (Foster 2011).

This perspective allows avoiding the reductionism from the micro to the macro (Robert and Yoguel 2016). The macro should not be seen as an aggregation of the micro. It is composed of a deep level – the level of mesorules – and a surface level – the domain of the aggregate outcomes (Dopfer 2007). In that respect, the macro is a systemic perspective on the meso-rule system as a whole. Neither is the micro just a reduction to an individual economic agent: it is a bottom-up perspective on the meso viewed in terms of its component parts (Dopfer et al. 2004). The meso, therefore, gives the micro its distinct position and represents the building block in the construction of the macro. While evolution of the meso-rules cannot be observed directly, the micro and the macro are two perspectives that *reveal* the structural aspects of changes in the meso.

In that respect, while the outcomes of economic evolution are observable at the macro-level, they can neither be understood, nor be explained at this level (Carlsson and Eliasson 2003). Macro-level phenomena are constructed

⁹ Related terms of 'meta-routines' (Aldrich 1999) or 'higher-order principles' (Kogut and Zander 1996) are also used in the literature.

statistics, reflecting changes in the underlying economic structure (Metcalfe et al. 2006). The sources of these phenomena should be sought in the industry dynamics that is, in its turn, driven by generation of novelty and competitive selection at the micro-level (Hanusch and Pyka 2007).

Heterogeneous economic agents, industries at different stages of maturity, and institutional frameworks varying across sectors and regions are all elements of the complexity of the economic systems characterised by non-linearities, such as unbalanced growth processes, catching-up, lag-behind and forging-ahead dynamics, should be considered (Hanusch and Pyka 2007). In that respect, macro-level phenomena are the emergent properties, i.e. 'the collective and largely unintentional outcome of *far-from-equilibrium micro interactions*' (Dosi and Winter 2000, p. 5; emphasis in the original).

Studies of economic change should, therefore, incorporate the full set of factors conditioning the evolutionary dynamics. This implies combining analyses of micro-level processes (i.e., behaviour of individual economic agents), the macro-level environment that constrains and/or facilitates a particular evolutionary path of the economy, as well as the system's and its components' connections with and dependencies on other systems elsewhere (Martin and Sunley 2015).

4.1.2. Evolutionary research in economic geography

Broadly speaking, economic geography deals with the uneven distribution of economic activity – organisation of production, distribution, and consumption – across space. Evolutionary economic geography (EEG) is concerned with how this spatial distribution is transformed over time. It bases on the premise that (emerging) spatial structures are not just the outcome of the economic evolution, but also condition and constrain micro-economic behaviour of economic agents that shapes this evolution (Boschma and Martin 2007).

More specifically, EEG focuses on *spatialities of economic novelty* (innovation, new firms, new industries), on how the *economic landscape emerges from the behaviour of economic agents* (individuals, firms, institutions) at the micro-level of the economic system, on how emerging spatial structures exhibit *self-organisation features*; and on how the *processes of path creation and path dependence shape geographies of economic*

evolution, and whether such processes are *place dependent* (Martin and Sunley 2006).

EEG aims at explaining the spatial distribution of routines over time, that is, the emergence and diffusion of new routines as well as mechanisms through which 'fitter' routines are diffused. As the dissemination of routines is primarily driven by spin-off processes and labour mobility, their spatial evolution develops along a localised structure (Essletzbichler and Rigby 2007; Boschma and Frenken 2011a), with successful routines having a higher chance to survive and be transmitted to other local firms. This results in regional branching, that is, the process in which new routines at a particular location develop out of technologically related routines already present in that location (Frenken and Boschma 2007; Boschma and Frenken 2011b). In that respect, conditions for generation and diffusion of new routines differ across various spatial scales, depending on local institutions, structure of networks of economic agents, patterns of labour mobility, etc.

Spatial structures emerge out of the myriad individual actions and interactions of economic agents that generate economic outcomes – such as daily behaviours, investment decisions, labour sourcing, profits, etc. – that reproduce these spatial structures. In contrast to the natural biological world, where self-organisation is often considered spontaneous, in the economic realm the development of spatial structures is the complex result of intentional behaviours, not (necessarily) intended outcomes, and learning of economic agents that pursue their own objectives (Martin and Sunley 2015).

For instance, the stable prosperous development of regions over time often depends on the location choice of a few firms. At the same time, the motivation behind that location choice may be traced back to the past to the choices made by other firms in the same industry and region, and in other industries and regions. In that respect, history and space matter for the evolution of firms, industries, and networks of economic agents (Capasso et al. 2012). In other words, uneven spatial distribution of economic activity is an outcome of connected, path-dependent historical processes (Dosi and Nelson 2010). However, the success of regions is only partly determined by their historical experience. In times of radical technological change, the development of knowledge (and, thus, spatial evolution of routines) is characterised by a high degree of uncertainty, which creates the windows of opportunities for places and regions to elevate their knowledge production

and innovativeness under particular circumstances (Storper and Walker 1989; Boschma and Lambooy 1999).

The spatial evolution of the economy as a whole is addressed in a framework of structural change (Boschma and Frenken 2006). The evolving pattern of uneven spatial development is a process of spatially-biased cumulative causation, which takes place at all spatial scales – within firms, within clusters, cities, regions, and globally (Boschma and Frenken 2011a; Martin and Sunley 2015). In that respect, economic systems are discontinuous in space, characterised by fuzzy boundaries, and cannot be separated from their environment, which makes them neither completely closed, nor completely open (Martin and Sunley 2015).

Despite being a contextual approach, EEG aims at understanding whether geography matters for evolution of economic activities rather than theoretically assuming *a priori* that it matters all the time (Boschma and Frenken 2006). Institutional differences across regions, indeed, constrain and/or enable the spatial evolutionary mechanisms in the economy. However, regional institutions do not condition the behaviour and performance of firms in a deterministic manner, as routines of firms may vary a lot even under the same institutions (Gertler 2010). Moreover, these same institutions are subject to evolutionary mechanisms as the institutional environment co-evolves with technological development and industrial dynamics (Nelson 1995; Schamp 2010). In that respect, institutions are both the context and the outcome of economic evolution (Martin and Sunley 2015).

Therefore, from the EEG perspective, places emerge from actions of economic agents, rather than fully determine their activities. Real places emerge from the neutral space as new knowledge is generated, new firms and industries emerge, evolve, become spatially concentrated in some regions, and launch the transformation of the institutional base of these regions. Regional development is, in that respect, more about path dependence than place dependence (Boschma and Frenken 2006).

4.2. Analytical model

In this section, various perspectives on the long-term economic development (Chapter 2) and on the dual role of the labour in the process of technology-

induced structural change (Chapter 3) are synthesised into the evolutionary analytical model that lies at the core of the explanatory framework employed in this dissertation. In essence, it deals with the co-evolutionary dynamics of industrial restructuring and worker reallocation across industries and regions in the process of the GPT diffusion.

Two important reservations should be made before moving on to the discussion of the model. First, while evolutionary theorising is appropriate for investigations of the ongoing transformation, based on incremental innovation, it lacks the solid theoretical foundation explaining the generation of Schumpeterian radical innovation that is behind any technological revolution (Castellacci 2007). Therefore, the very arrival of the GPT is considered exogenous in the model. Rather, the model focuses on the process of the GPT diffusion and its implications once the GPT has arrived.

Second, the model focuses primarily on the market mechanisms and implications of the GPT diffusion process. In that respect, the institutional environment, in which the diffusion takes place, is not taken into the consideration explicitly. However, as the institutional environment is subject to the co-evolution with the technology-induced transformation dynamics, it makes its way into the model as an implicit factor that facilitates and/or constraints the latter. This is reflected in both the overall explanatory framework as well as in the analyses performed in individual articles.

Arrival of a new GPT and the process of its diffusion generate different transformation pressures on existing economic activities. Firms' response manifests in the development of new routines that are necessary to accommodate these pressures. The variety of routines increases leading to the boost in the selection process that benefits those organisations that are more able and willing to adapt their routines to the transformation impulses, generated by the GPT diffusion, and disadvantage those that are lagging behind in transforming themselves. This ability is conditional upon the openness of organisations to change and their capacity to respond to it (known as absorptive capacity (Cohen and Levinthal 1990); for a review see Zahra and George 2002). Besides, (emerging) complementarities between routines, that already exist at the organisational level, and ones, that are required, play an important role in this process. The generation of variety through innovation and the subsequent selection process lead to the restructuring process in existing industries and the emergence of new industries.

Processes of variety generation and selection within industries lay the evolutionary foundation for processes of variety generation and selection between industries. Uneven patterns of generation and dissemination of new knowledge lead to the uneven distribution of growth across sectors, producing structural change in the economy. Consequent changes in the relative importance of various sectors continuously define and redefine the dynamic complementarities between them (Metcalfe et al. 2006).

Therefore, the combination of transformation pressures and heterogeneous abilities of firms and industries to accommodate these pressures generates both qualitative change within industries (that is, how the output is produced and under which circumstances of competition) and structural change in the economy as a whole (that is, changes in the shares of industries in the total output, employment, and productivity growth).

Both qualitative change within and structural change across industries have an impact on the labour market by generating new patterns of demand for labour. In particular, there is an explosive employment growth in new industries, reallocation of workers between occupations in renewing industries, as well as displacement of labour in declining sectors. These changes generate the response of the labour market in the form of the interindustry reallocation of labour.

At the same time, there are spatial implications of the labour demand shifts. Regions at different levels of the regional hierarchy are characterised by different receiver and development capacities. In other words, they provide different environments for accommodating the transformation pressures of the GPT diffusion process. In particular, the most dynamic regions at the top of the regional hierarchy are the most successful ones in attracting new industries. Besides, being characterised by the diversification (Jacobs') externalities, they provide the most fertile ground for establishing complementarities between the old and new parts the economy. These regions may, thus, be expected to dominate the process of technology-induced economic transformation, at least at the early stages of the GPT diffusion. For regions at lower levels of the regional hierarchy it takes much longer time to benefit from this process. Moreover, even the regions at the same level of the regional hierarchy might differ in their ability to reabsorb redundant labour from declining sectors. This depends on the diversification of the regional employment mix, and particularly the structure of regional networks of skillrelated industries. Therefore, patterns of job creation in new/renewing

industries and job destruction in declining industries manifest themselves spatially.

All in all, transformation pressures generated by the GPT diffusion lead to qualitative changes within industries as well as the change in importance of various sectors for the economy as a whole. As regions have different capacities to accommodate these pressures, patterns of industrial restructuring and structural change are unevenly distributed across regions, leading to the emergence of spatial imbalances across regional labour markets and generating the response in the form of worker reallocation across regions.

Therefore, the industrial restructuring and resulting structural change have an impact on patterns of worker reallocation. This motivates the role of the labour market as a response mechanism to the technology-induced economic change. This kind of relationship is addressed in Article I that investigates the impact of the uneven distribution of key industries, associated with the GPT, on the patterns of inter-regional reallocation of labour. The particular focus in the article is made on the dynamic roles of different groups of key sectors on the ability of regions to attract and retain labour at different stages of the GPT diffusion.

There is, however, another side of the story. The generation and diffusion of new routines requires application of new skills and knowledge. Therefore, the availability of relevant knowledge conditions the capacity of firms and industries to respond to the transformation pressures, generated by the GPT diffusion. In that respect, the labour market, being the mediator of knowledge transfer, may facilitate and/or restrict the opportunities for industrial restructuring and structural change across industries and regions. Two interlinked dimensions contribute to that.

First, knowledge, related to the GPT, is predominantly outside of the established knowledge domain. Therefore, the supply of new skills lags behind the demand for them, particularly at the early stages of the diffusion process. In that respect, firms, willing to accommodate the transformation pressures of the GPT diffusion, face the deficit in relevant knowledge and skills, embodied in workers who are available on the labour market. Therefore, mismatches arise between the skills demanded and those supplied. As the potential for upgrading organisational routines is dependent on, among other things, the availability of adequate skills, will be more successful in

responding to the transformation pressures. Therefore, the availability of labour is a restricting factor for the industrial restructuring.

At the same time, if there are complementarities between the new skill requirements and skill structures, already available in the economy, the reallocation of workers, who have developed specialised human capital that is complementary to the GPT, may be the enabling factor for the industrial restructuring process.



Figure 3. Co-evolutionary dynamics of industrial restructuring and worker (re)allocation: A schematic summary

Second, it matters not only whether the complementary knowledge and skills are available, but also where they are available and how they are diffused through the regional system over time. Different regions have different endowments with skilled labour, which depends on a range of both demandand supply-side factors. On the one hand, they are realisations of industrial structures of regions: regions characterised by more diverse industrial landscapes accommodate workers with a wider variety of skills. Also, regions, characterised by denser networks of skill-related industries, might find it easier to accommodate the transformation pressures of the GPT if skills employed in these industries are complementary to the GPT, and more difficult otherwise. On the other hand, workers themselves have different preferences with regards to living conditions in different regions (e.g., creative class and the cities, see Florida 2014). The resulting pattern of uneven distribution of skilled workers is an enabler for industrial restructuring for regions where the availability of such labour is high, and a restrictor - where it is low. Spatial patterns of skilled workers availability and reallocation, therefore, condition the opportunities for the accommodation of the technological pressures in different regions.

More generally, processes of industrial restructuring are shaped by the dynamics of knowledge creation and diffusion in the geographic and the technological spaces. Articles II-IV address these issues from various perspectives. Article II investigates the labour sourcing strategies in the sector, that is paradigmatic to the GPT itself, focusing, in particular, on the temporal dimension of the labour sourcing process. Article III analyses the diffusion of skills, related to the GPT, throughout the whole spectrum of industries and regions. Finally, Article IV considers the most dynamic regions at the top of the regional hierarchy as learning regions for individuals, willing to obtain GPT-related skills, and analyses the role of these regions.

There are several novelties that this model offers for investigations of the determinants and outcomes of the long-term economic development. First, it enriches the economic and economic historian research on general purpose technologies and techno-economic paradigms with a spatial component, which remained largely disregarded in these streams of the literature. Second, the model focuses on long-term aspects of spatial processes. In that respect, it puts forward the temporal dimension of the economic transformation that, to a great extent, is largely ignored in the economic geography research. Finally, the model provides a framework that links explicitly the micro-level
processes of response to GPT-induced transformation pressures to the macrolevel determinants and outcomes of these processes. This allows identifying the mechanisms which translate the micro-processes to the macro-outcomes. In other words, it allows specifying meso-rules that determine the evolution of (regional) labour markets during the technology-induced economic transformation. In that respect, it helps coming closer to the true explanatory mechanisms of economic evolution within the micro-meso-macro ontology.

5. Methodological considerations

5.1. Data

5.1.1. Data sources

Data employed in this dissertation come primarily from two sources: the Longitudinal Integration Database for Health Insurance and Labour Market Studies (*Longitudinell integrationsdatabas för sjukförsäkrings- och arbetsmarknadsstudier, LISA*), provided by Statistics Sweden, and the Databases of Evolutionary Economic Geography in Lund (DEVIL), collected and compiled by researchers at the Department of Human Geography at Lund University.

LISA is an anonymised linked employer-employee database that aims at complementing traditional labour market statistics and providing a better description of the labour market and people's relationship to the world of work (SCB 2016b). It is a total-count individual register: all individuals registered in Sweden on December 31, are included in the population for the reference year. LISA is a longitudinal database, meaning that the data for the same person can be linked for all years, when she is included in the population.

LISA integrates the annual data from several registers, including education, income, employment, health insurance, and population registers. In that respect, it provides information on various dimensions of working life, connection to work, labour mobility, movement in and out of the labour market, etc. The connection of an employee to an employer is denoted by the identity number of the firm and the establishment where she has her main employment. The data contains also the detailed information on various individual variables, such as age, education, annual earnings, municipality of residence and employment, industry of employment, etc. Thus, the data

allows tracing employment histories of individuals and, for instance, constructing a very precise measure of worker reallocation flows. Annual data cover the period between 1985 and 2010.

One reservation should be made here. LISA is built around the data that comes from administrative registers and, thus, is primarily created for administrative purposes. That implies that the objects and variables can appear/disappear and change values as the purposes of the registers they comes from change over time (SCB 2016b). At the same time, being a high-quality dataset, it comes with clear explanations for the process of variable construction, compatibility of different variables, translation keys between different versions of variables, etc. This allows ensuring consistency of observations made for different years.

DEVIL is a longitudinal firm-level database, constructed from business registers of Statistics Sweden for studies of regional economic transformation. For the period between 1968 and 2008, it provides the information on a firm's municipality of registration, number of employees, value added, wage costs, etc. In that respect, it allows constructing measures of regional industrial profiles, industry concentration/diversification patterns, etc.

Finally, the research relies on data publicly available from Statistics Sweden.

5.1.2. Regional divisions

The principal spatial unit employed in the analysis is a local labour market (*lokal arbetsmarknad*), which represents a group of Swedish municipalities (*kommuner*). This definition of a region is chosen due to the specificity of its construct. The boundaries of local labour markets are defined by the statistics on commuting between municipalities in the way that maximises the self-containment of commuting flows. In other words, local labour market areas are defined in the way that maximises homogeneity of mobility patterns within a region while sustaining cross-regional heterogeneity.¹⁰

With regards to investigations of labour market dynamics, it implies that, for instance, there is a minimum chance to confuse commuters for individuals

¹⁰ For more detailed information regarding definition and construction of local labour markets refer to SCB (2010).

who truly relocate between regions. In that respect, operationalising spatial units as local labour markets is associated with minimising the risk that the labour market dynamics of a region will be affected by that of neighbouring regions.

Longitudinal character of investigations behind individual articles, included in this dissertation, posed some challenges with assigning municipalities to local labour markets. First, the Swedish municipality system underwent reforms over the considered time period. In 1985, there were 284 municipalities in Sweden. Two new municipalities were added in 1992 and in 1995. Also, one new municipality was added in 1999 and 2003. After 2003, the number of municipalities in Sweden remained the same: 290 municipalities.

Second, as a result of county reforms of 1997 (merging Malmöhus and Kristiandstad counties into Skåne county) and 1998 (merging Göteborg and Bohus, Älvsborg, and Skaraborg counties into Västra Götaland county), some municipalities changed their codes.

In cases of both new municipality additions and municipality code changes, division of municipalities into local labour markets was checked manually to make sure that it was consistent over time.

For the purpose of investigating the dynamics of industrial restructuring and labour market transformation at different levels of the regional hierarchy, local labour markets were also divided in regional groups. In Article I, seven groups are defined based on the population statistics, namely: Stockholm, Gothenburg, Malmö, large regions, medium-sized regions, small regions, and micro-regions. In Articles II-IV, instead, six regional families were introduced according to the methodology suggested by the Swedish Agency for Economic and Regional Growth (NUTEK 2004), namely: Stockholm, Gothenburg, Malmö, large regional centres, smaller regional centres, and peripheral regions. Criteria involved in the definition of these groups include population size and density, business dynamics, share of individuals with higher education as well as access to higher education institutions.

These two classifications correspond neatly to each other. Top three levels in both are the same – three Swedish metropolitan areas. Large regions in Article I match large regional centres in Articles II-IV, while medium-sized regions correspond to smaller regional centres. The only difference between two classification schemes is that small and micro-regions in the former are merged into the peripheral category in the latter.

5.1.3. Industry divisions

In LISA and DEVIL, the division of productive activities of firms into industries is performed according to the Swedish Standard Industrial Classification (*Standarden för svensk näringsgrensindelning, SNI*). SNI is primarily a statistical standard used to classify entities – that is, companies and workplaces – based on their primary economic activities. SNI is a part of an international system of economic classifications and is based on the Statistical classification of economic activities in the European Community (*Nomenclature statistique des activités économiques dans la Communauté européenne, NACE*). SNI and NACE are identical in the first four levels: sections, divisions, groups, and classes. In addition, SNI has a fifth level. In that respect, each workplace is characterised by a five-digit code representing its belonging to a certain industry.

Over the considered time period, three versions of SNI were provided in LISA and DEVIL: SNI69, SNI92, and SNI02. The summary of divisions into sections, groups, and classes according to these classification schemes is presented in Table 1.

	SNI6911	SNI92	SNI02	
Sections	9	17	17	
Divisions	33	60	62	
Groups	79	223	224	
Classes	185	505	514	
Sub-classes	328	755	774	

Table	1.	Versions	of SNI	

SNI92 and SNI2002 have a rather similar structure. Indeed, SNI02 was a result of a minor revision of SNI92. Therefore, it was no problem to establish unambiguous links between these two classification schemes in order to ensure the data consistency over time.

¹¹ SNI69 industry classification has six levels of industry classification. There are 368 sixdigit level classes.

At the same time, ensuring the compliancy of industry classifications between SNI69 and subsequent classification schemes was more problematic. It was, however, important since that was the only classification system for the period before 1990. As SNI69 is more aggregated than SNI92 (see Table 1), it was not possible to go back in time before 1990 with the same level of detailsation as for years after 1990 (after SNI92 was established). Therefore, only when data was required for broad categories of industries, was it possible to make necessary aggregations from both SNI69 and SNI92 to make them consistent over a long period of time.

5.2. Ontological and epistemological perspective

This dissertation is informed by the critical realist perspective that was originally suggested by Bhaskar (1975, 1979) and further developed in works of other authors (e.g., Sayer 1992; Archer et al. 1998; Sayer 2000). The basic notion behind this perspective is that there is an objective reality – physical processes and social phenomena – that exists independent of our knowledge of it but may be scrutinised by research efforts.

There is a fundamental division between the intransitive and transitive dimensions of science (Bhaskar 1975; Sayer 2000). The former relates to objects of science, that is physical processes and social phenomena. The latter refers to theories and discourses that are developed as means of understanding and explaining the objects of science. Two dimensions, thus, represent the existing world and researchers' subjective and limited experience of it. Changes in the transitive dimension do not necessarily imply that the intransitive dimension undergoes changes as well. While science is socially-influenced, investigated phenomena and explanatory mechanisms behind them exist independent of the scientific thought. Critical realism, thus, differs itself from relativism that rejects the existence of an objective reality and views the world as constructed in minds of individuals who experience it.

At the same time, critical realism rejects the notion of universal truth, suggested by proponents of positivist approaches, and underlines the temporary and revisable nature of theorising. Social phenomena, being outcomes of social interactions, are subject to transformation if the context around them, as well as the very interactions producing these phenomena, change. The theory explaining social phenomena should then change as well

(Bhaskar 1979). Therefore, the truth is conditional, related to recognisable forms of social organisation, and in the need of continuous reassessment (Pratt 1995).

The tension between the real world and limited ability of individuals to understand it is resolved through the stratified ontology: a distinction between the real, the actual, and the empirical (Bhaskar 1975). The real refers to whatever exists in the world – be it natural or social – independent of our understanding of it. It is the entirety of all objects of the world, including the structures and mechanisms governing their conduct, that are not directly observed. The actual refers to what happens if and when these mechanisms are activated and includes all events which can potentially be observed. Finally, the empirical is the domain of concrete experience of these events, which is accessible to a researcher through observation and empirical analysis. The objective of the critical realist research is then to identify the structures and mechanisms in the domain of the real and their impact on social interactions.

The critical realist methodology seeks to reconstruct the causal mechanisms behind social interactions and to understand when, and under what conditions, they are activated. Different from positivist approaches, where the causation is a model of regular successions of events, the causation in critical realism is not pre-determined and refers to potential rather than actual mechanisms (Yeung 1997).

What causes something to happen has nothing to do with the number of times we have observed it happening. Explanation depends instead on identifying causal mechanisms and how they work, and discovering if they have been activated and under what conditions. [...] [E]xplaining why a certain mechanism exists involved discovering the nature of the structure or object which possesses that mechanism or power. (Sayer 2000, p. 14)

The research process begins with identifying an empirical phenomenon to be investigated and proceeds from describing the phenomenon to something that causes or produces it (Yeung 1997). The empirical phenomenon is approached from a perspective of a certain theoretical framework. However, this framework does not fully pre-determine the analysis but is rather subject to the ongoing scrutiny, based on insights from the empirical world. In other words, while the theoretical framework provides an initial set up for research, empirical investigation provides the basis for its revision or even rejection in certain cases. In that respect, the central tool of the realist methodology is an abstract research (Sayer 1992, 2000). Abstraction is an adequate method to conceptualise social structures and causal mechanisms. To be able to address certain research issues, it is necessary to abstract from particular contingencies, that is to exclude those factors which do not have significant explanatory power in order to concentrate on those which do (Yeung 1997). Abstraction, though, should not be seen as a linear process, so that it only appears in the beginning of the research processes. Rather, it is an iterative and reciprocal movement between the abstract and concrete. In other words, one moves between pure descriptions of a phenomenon under consideration to abstractions of its possible causes.

Causal mechanisms and their properties are abstracted *a posteriori* as it is often impossible to realise *a priori* the existence of particular social interactions without experiencing some of their effects (Yeung 1997). Moreover, social phenomena are rarely as durable as natural science objects. In that respect, identified casual mechanisms are historical and contextual.

In particular, any empirical observation has spatial and temporal dimensions since social interactions are situated in particular conditions in time and space (Sayer 2000). At the same time, at any point in time space itself is constructed through the social relations and characteristics of objects occupying it. Thus, space and time provide contexts in which the empirical is observed. In that respect, from a critical realist perspective, explanations in social sciences should not be expected to remain stable over time and space.

5.2. Methodological perspective

The literature on research methods in the social science suggests that there is no predetermined connection between selection of a research method and a philosophical stance on which the research is based (Yeung 1997). In particular, choosing between qualitative and quantitative research strategies 'ought often to depend on the purposes and circumstances of the research, rather than being derived from methodological or philosophical commitments' (Hammersley 1992, p. 51). In this dissertation, quantitative research methods are employed that are sometimes referred to also as extensive methods (Sayer 1992).

Quantitative methods have been employed in a considerable body of research in economic geography since the mid-1950s, the period known as the quantitative revolution. At the time, however, this research was dominated by researchers who, explicitly or implicitly, accepted assumptions associated with positivism (Philip 1998). For that matter, they were criticised for commitment to the forms of economic modelling based on highly restrictive notions of rational behaviour, convexity, and perfect competition. Reductionist mathematical modelling may produce oversimplified explanations regarding the causal mechanisms for social interactions and nature of societal change. Particularly, if factors of context (e.g., spatial factors) are disregarded (Plummer 2003).

A considerable number of criticism towards quantitative methods in geography was expressed in years following the quantitative revolution. One of the claims being made was that quantitative geography did not address social, economic, and political challenges of the time, and dismissed research areas where objects of research were fuzzy, difficult to observe, classify and measure (Philip 1998).

[T]he radicalism of the quantitative revolution [...] did little if anything to loosen political perspectives or broaden the spectrum of political acceptability. Indeed it produced a form of selective objectivity, largely confining attention to previously 'acceptable' topics which could be squeezed to produce data amenable to quantification (Muir 1978, p. 323)

The 'cultural turn' in economic geography contributed further to the criticism of potential use of quantitative methods. Proponents of the cultural turn suggested that theorising in economic geography should abandon that of mainstream economics and search for economic knowledge elsewhere (Plummer and Sheppard 2001). Amin and Thrift (2000, p. 5) promoted a 'skill base depending upon the understanding of open systems, appreciation of context, and qualitative techniques'.

Cultural, social, and institutional processes, that are difficult to incorporate into quantitative explanatory frameworks, are, indeed, important for understanding the evolution of economic landscapes. One, however, cannot simply imply these processes are *the* explanation and economic factors should be squeezed out (Martin 2001). Sharing the position of cultural turn proponents that one of the tasks of economic geography is to be a critique of mainstream economics, one should realise that such critique should be also expressed in the theoretical language of mainstream economics (Plummer and Sheppard 2001).

Genuine refutation must penetrate the power of the opponent and meet him on the ground of his strength; the case is not won by attacking him somewhere else and defeating him where he is not (Adorno 1982, p. 5).

Employing quantitative tools of analysis, economic geographers can effectively criticise mainstream uses of quantitative regional science within their own terms of reference and develop alternative models aiming at understanding the dynamics of space economies (Plummer and Sheppard 2001). In that respect,

[q]uantitative methods can contribute by describing and analysing the broad contours of difference, by providing a basis for informed policy making and progressive political change, by identifying people and places for in-depth studies, and by situating qualitative research in a broader context (McLafferty 1995, p. 436).

One should bear in mind that quantitative research is not tied to positivist approaches in social inquiry *per se*, at least unless it is used to suggest universal laws and other generalisations. Besides, quantitative research may be thought of as 'a tool developed for use within the value position or the research question in hand. In itself it is neither a ready-made philosophy nor a value-free method of research' (Bennett 1985, p. 223).

Quantitative methods may, among other things, be useful for identifying and explaining external and contextual relations between objects (Yeung 1997). It should be noted, however, that observed empirical regularities should not be always treated as causal mechanisms. Rather, these should be treated as generalisations of empirical patterns. In this dissertation these generalisations are not considered universal, but rather contextual and valid at a specific spatio-temporal intersection.

5.4. Methods

Following the discussion in the previous sections, the co-evolutionary dynamics of patterns of industrial restructuring and worker reallocation was

studied in individual articles through the combination of abstraction and quantitative research methods.

Abstraction was the first step of research process underlying each individual article. The important component of the abstraction process is the previous research and theoretical frameworks relevant to the formulated research questions. Through reading of the relevant literature, the focus, with which empirical observations in the data were approached, was determined. This focus, in its turn, defined the operationalisations and selection of particular statistical tools and techniques that were employed in individual articles.

The first important aspect to be considered was to determine the way in which the diffusion of the GPT was operationalised in different papers. In Article I, the Perez' classification of leading industries¹², associated with the GPT diffusion, was revisited and empirically refined to analyse the role of locational patterns of these industries in shaping the inter-regional worker reallocation patterns. In Article II, the dynamics of labour sourcing in the GPT-related sector (i.e., the knowledge-intensive IT services sector) were analysed. Article III focused on the diffusion of GPT-related competences (operationalised as workers employed in IT occupations) across different industries and regions. Finally, in Article IV, entrepreneurship in the GPTrelated sector (again, the knowledge-intensive IT services) was considered as a mechanism of spatial diffusion of the GPT-related activities. In that respect, while different articles have rather different empirical foci, all of them address certain aspects of co-evolutionary mechanisms between the process of industrial restructuring and labour market dynamics during the GPT diffusion.

In that respect, the execution of empirical analyses in different articles required specifying different units of analysis, spatial and temporal divisions, as well as statistical tools and techniques that would, according to the author's/authors' view, suit best to answer posited research questions. Table 2 provides the summary of methodological issues connected to each of four articles.

In terms of primary units of analysis, the articles focused on individuals (Articles II and IV), firms (Article IV), industries and industry groups (Article III), as well as regions (Articles I and III). Flexible selection of the units of analyses allowed uncovering the mechanisms taking place at various levels

¹² Discussed in Section 2.2.2.

of the economy with a goal of identifying the meso-rules underlying these mechanisms.

	Article #1	Article #2	Article #3	Article #4
Data sources	LISA; DEVIL	LISA	LISA	LISA
Units of analysis	Regions	Individuals	Industry groups; regions	Entrepreneurs (individuals); entrepreneurial entries (firms)
Spatial units	Local labour markets; population- based regional groups	Local labour markets; regional families	Local labour markets; regional families	Local labour markets; metropolitan and peripheral areas
Time period	1985-2008	1991-2010	2001-2010	1991-2010
Time sub-periods	1985-1998; 1999-2008	1991-1993; 1994-1997; 1998-2001; 2001-2004; 2005-2007; 2008-2010	No	1991-2000; 2001-2010
Statistical tools and techniques	Exploratory data analysis; instrumental system-GMM regression (Arellano-Bover estimator)	Exploratory data analysis; OLS regression; logistic regression; Heckman-probit selection model	Exploratory data analysis	Exploratory data analysis; logistic regression; Cox proportional hazards model

Table 2. Summary of research methods

The principal spatial units – local labour markets – were the same in all four articles, even though the division of them into the levels of the regional hierarchy was performed differently in Article I as compared to Articles II-IV (see Section 5.1.3.).

Issues with data availability and ensuring the consistency of industry classifications over time made it impossible to cover the same time periods in different papers. For instance, Article I covers the longest time period (1985-2008). Articles II and IV cover the time period between 1991 and 2010. Finally, analyses in Article III are provided for the period 2001-2010.¹³

¹³ Analyses in Article III are based on the information about the occupation an individual is employed at, which is only available for the 2000s.

Besides, in all articles but Article III, the whole considered time period is divided into sub-periods according to different criteria (for the specifics, see methodology sections of the articles).

Finally, depending on the way the research questions in individual articles were posited, a wide range of statistical tools and techniques was employed. Some of them are widely used in quantitative economic geography research relatively and are relatively easy to implement and interpret (e.g., exploratory data analysis and OLS regression). Others are more specific (e.g., Arellano-Bover GMM estimation, Heckman-probit selection model, and Cox proportional hazards model). The selection of these techniques was dictated by the necessity to deal with particular features of the data at hand and inability of more conventional techniques to properly address the empirical issues. Simply put, the selection of statistical techniques, employed in individual articles, was driven primarily by their suitability to deal with the problem at hand, rather than simplicity of their application.

6. Current wave of technological change: Information technology as a GPT

There is no reason for any individual to have a computer in his home.

(Kenneth Olsen, CEO of Digital Equipment Corporation, at the 1977 World Future Society meeting in Boston, USA)

You can see the computer age everywhere but in the productivity statistics.

(Solow 1987, p. 36)

Two famous quotes above represent two early misconceptions regarding the role of the information technology (IT) for the economy and society. Nowadays, it is difficult to imagine our lives without computers and other electronic devices. However, some forty years ago even leaders of the world's largest manufacturers of computer equipment did not see the potential for using computers beyond specialised applications in the military (Kenneth Olsen's quote). Even later, when computers were already employed in a wide range of personal and business applications, economists cast doubt on benefits that the IT could bring in terms of generating productivity gains (reference to Solow).

One particular issue with the IT lies in identifying what exactly it embodies. Most often the IT is referred to generically. There are, however, more specific references to semiconductors (Bresnahan and Trajtenberg 1995), computers (David 1990), the Internet (Clarke et al. 2015), and others. However, even such delimitations are often not enough.

When we speak of the "internet", for example, what is paramount? Is it the hardware and software protocols such as TC/ICP, the http protocol, html markup language? Or is it the switches and routers, or the innovations that permitted multiplexing hundreds of bitstreams over a single fiber optic cable? (Field 2008, p. 11)

David and Wright (1999) considered the IT as a cluster of technologies that developed in certain stages. Each of these stages broadened the spectrum of potential uses of the IT and marked different phases of its diffusion.¹⁴

Early on, available technologies were purpose-built and task-specific (such as, monitoring material flows and delivery processes in manufacturing industries). At that stage, the narrow scope of applications of the IT made the mass production of personal computers neither appropriate nor robust.

Around the early 1990s improving capabilities of computers as network servers allowed companies to develop client-server data processing systems instead of using the centralised mainframe environment.¹⁵ Client-server architecture put the power of a mainframe into a server, whereas networks, connecting different servers, gave companies more flexibility with less powerful clients, thus, contributing to lowering costs required to implement IT solutions into business processes (ter Weel 2006). Bresnahan et al. (1996) used an analogy with Jonathan Swift's Gulliver's Travels to describe the advantage of client-server computing:

Before 1989 workstations and personal computers could no more replace mainframes than could the people of Lilliput wrestle Gulliver to the ground. Yet, like the Lilliputians ropes, networking cables created strength from numbers. Workstations, a technology originally intended to serve individuals, were deployed as servers. They did not need as much capacity as hosts, since PCs, deployed as clients, assumed some of the computing tasks (such as effectively interfacing with people). This technical opportunity produced large market and organisational change. (pp. 3-4)

Finally, a widespread propagation of the Internet opened up opportunities for a new class of data processing applications that enhanced the potential for

¹⁴ These stages are not related to the spread of computers for private use, but rather business applications of the IT.

¹⁵ A mainframe is a central data repository in the data processing centre of an organisation that is accessed by individual users through workstations and/or terminals. Mainframes are typical for a centralised form of computing, as opposed to a distributed (client-server) form of computing.

collective and cooperative forms of work organisation. Parallel to this, the elaboration of standards and increased compatibility of different software and hardware improved the ease of collaboration between workers within organisations as well as collaboration between organisations.

In what follows, the review of studies regarding the impact of IT on the economy is performed. Two dimensions are addressed. First, it is explored whether the IT lives up to the expectations of it as a driver of productivity gains and aggregate growth. In other words, can the IT be characterised as a GPT? Second, given the particular interest of this dissertation in labour market dynamics, some general aspects of the labour market impact of the IT are discussed.

6.1. Growth and productivity impact of the IT

This section summarises the empirical studies on the overall impact of the IT on the economy in three dimensions: determinants of IT adoption, productivity and growth implications of IT adoption, and mechanisms through which the IT generates productivity gains and aggregate growth.

When it comes the IT adoption, numerous studies have investigated various sets of traditional and non-traditional variables. It has been demonstrated that the probability of IT adoption co-varies with firms' age and size (Hollenstein 2004; Haller and Siedschlag 2011), share of skilled labour (Bayo-Moriones and Lera-López 2007; Haller and Siedschlag 2011), (local) competitive pressures (Fabiani et al. 2005; Bayo-Moriones and Lera-López 2007), export orientation and openness to international environment (Baliamoune-Lutz 2003; Haller and Siedschlag 2011), organisational structures (Caroli and Van Reenen 2001; Hollenstein 2004), expected performance benefits for more complex applications (Tsikriktsis et al. 2004), and others. Despite differences across firms with regards to mentioned variables, all these studies point to an impressive pace of the IT diffusion, pointing to the pervasive nature of the technology.

Many of adoption studies emphasise the division between basic and strategic adoption. The basic adoption is associated with generic applications such as establishing basic infrastructure (e.g., Internet access), e-mail use, passive document sharing. The strategic adoption involves technologies that change existing internal processes in firms and require substantial co-invention at the firm level to be used successfully.

The difference between basic and strategic adoptions manifests spatially (Forman et al. 2005). On the one hand, when it comes to the basic adoption, there is little variation in adoption rates between dynamic urban agglomerations and more peripheral regions. In other words, at least the basic IT applications are pervasive in the geographic space. On the other hand, when it comes to technology-intensive applications that require strategic adoption, firms located in larger regions tend to have a higher probability of adopting the IT (Galliano et al. 2011; Haller and Siedschlag 2011). These regions offer a combination of benefits – such as availability of complementary inputs, thicker labour markets, etc. – that make costs of the strategic adoption lower. What is interesting, however, the role of location of firms in the strategic IT adoption is secondary to that to which industry a firm belongs to (Forman et al. 2005). One can, therefore, infer that regional differences in IT adoption rates are primarily driven by the geographical distribution of IT-using industries.

The impact of the IT adoption on productivity has been investigated in numerous contributions that focused on different levels of analysis and employed various frameworks and definitions of the IT. Early studies demonstrated that prior to the 1990s the estimated contribution of IT to productivity growth was lower than expected (e.g., Oliner and Sichel 1994). This was the reason behind the productivity paradox formulated by Solow.

More recently, the positive productivity impact of the IT adoption has been repeatedly and extensively documented by many empirical works at firm, industry, regional, and national levels (see reviews by Draca et al. 2007; Stiroh 2010; Kretschmer 2012; Cardona et al. 2013; Stanley et al. 2015). The IT has become a driver of rising productivity through direct productivity increases in IT-producing sectors and through capital deepening and labour productivity improvement in IT-using sectors (Liao et al. 2016). In their turn, productivity gains in both IT-producing and IT-using industries have increasingly contributed to the aggregate growth of regional and national economies (Strohmaier and Rainer 2013).

What is the reason behind the Solow paradox then? Studies of the productivity impact of the IT demonstrate that, on the hand, a certain amount of the IT capital had had to be accumulated in the economy before its effect became visible, and, on the other hand, there were pronounced lags between

the IT investments made and productivity benefits reaped (Rincon et al. 2013; Strohmaier and Rainer 2013). For instance, productivity gains of IT investments in IT-using industries take 5 to 15 years to be realised (Basu and Fernald 2007). In that respect, the Solow productivity paradox was, to a great extent, an artefact of time and measurement:

The problem with "we see computers everywhere around us except in the productivity statistics" was not with productivity, but with looking at computers in economics departments rather than in firms. At the time of Solow's remark, the ICT capital stock was far too small to have (yet) created a growth boom, even though the private returns to use of computers were very substantial. (Bresnahan 2010, p. 789)

The mechanism of achieving productivity gains through IT investments lies primarily in fostering complementary innovation in IT-using industries. These complementarities offset potential diminishing returns to the IT itself and shift the demand curve for the IT capital further out (Basu and Fernald 2007). Vast amount of empirical literature investigated productivity-boosting spillovers that are the outcome of the IT diffusion and identified several domains of these spillovers.

First, the IT facilitates organisational change at the firm level through decentralisation of decision-making, improved team work, easier information sharing, etc. (Brynjolfsson and Hitt 2000; Bresnahan et al. 2002; Brynjolfsson and Hitt 2003). The organisational change is an outcome of learning-by-adopting and learning-by-doing processes (Rincon et al. 2013). In that respect, the IT capital is an input that is complementary to the organisational capital (Bertschek and Kaiser 2004; Belloc and Guerrieri 2015).

Second, the IT improves the efficiency of transactions among firms by, for example, reducing administrative and search costs, improving supply chain management, etc. (Rowlatt 2001; Criscuolo and Waldron 2003). In short, the IT makes it faster, easier and more efficient for firms to interact with each other (Brynjolfsson et al. 2002). This results in positive network externalities, efficiency of which increases with time as the IT is adopted by more users (Rincon et al. 2013).

The realisation of these spillovers has a temporal component in it (Rincon et al. 2013). Immediately after the investments in the IT capital are made in IT-using industries, intra-industry IT spillovers are negative because of the costly restructuring process in firms. Inter-industry IT spillovers – that is,

network externalities – are, however, directly positive. The total spillover effect is negative. The complementarity between the IT and R&D efforts at the firm level allows the most innovative firms to overcome the negative intra-industry spillovers relatively fast. With approximately five-year lag, intra-industry spillovers become positive for most (even the least innovative) firms. Therefore, the IT diffusion imposes periods of experimentation leading to delays between initial investments and performance improvements that also justify the lagged IT spillover effect (Brynjolfsson and Hitt 2003). In other words, just as with the direct effect of the IT on productivity, the indirect effect in the form of spillovers is lagged since IT investments, made by IT-using firms, need to accumulate before their positive effects are realised (Liao et al. 2016).

All in all, the empirical studies on the IT adoption and its effects demonstrate that the IT possesses features – such as, pervasive use, positive impact on productivity, ability to generate spillovers, the delay between the diffusion and its impact – that correspond neatly to the characteristics of a GPT discussed in Section 2.2.1.

6.2. Labour market impact of the IT

The division of labour between particular tasks is determined by a trade-off between specialisation benefits and costs of communication between individuals involved in various tasks (ter Weel 2006). Diffusion of the IT has improves communication and production opportunities, resulting in fundamental changes in the division of labour as well as the labour demand.

On the most obvious side, the IT diffusion as well as ongoing standardisation of certain tasks has led to the substantial growth in the number of processes the IT can automate and, thus, to the substitution of certain kinds of labour and decreasing labour intensity of production processes (Autor et al. 2003). According to the routinisation hypothesis (Autor et al. 2006)¹⁶, computers are particularly suited to perform routine, repetitive tasks that can be described by rules, while humans have a comparative advantage in non-routine tasks that involve creativity, social interaction, pattern recognition, etc. (Levy and

¹⁶ In the literature, related notions of task-biased (e.g., Adermon and Gustavsson 2015) and routine-biased technological change (e.g., Goos et al. 2014) are also employed.

Murnane 2004). The classical study by Autor et al. (2003), who investigated the replacement of intellectual skills of workers by computers in the assessment of mortgage applications, was followed by other studies that demonstrated the overall increasing demand for routine tasks – such as, book-keeping, middle-management and non-specialist blue collar occupations – but a declining demand for workers to perform these tasks (Autor et al. 2006; Spitz-Oener 2006; Goos and Manning 2007). The new generation of the IT allows automating even more complex tasks, such as driving, speech recognition, and basic motor skills (Brynjolfsson and McAfee 2014).

Does it mean that computers are eventually going to substitute all human labour through automation? Actually, not. The IT diffusion, while decreasing the demand for routine, repetitive skills, has boosted the demand for nonroutine analytical and interactive skills, leading to profound changes in the content of existing occupations and emergence of entirely new occupations (Vona and Consoli 2015). The increase in importance of non-routine tasks has increased the relative demand for highly educated workers who possess a comparative advantage in executing these tasks (Böckerman et al. 2016). The benefits of the IT can be enjoyed fully if human capital, accumulated at the organisation level, is upgraded accordingly resulting in better information collecting, processing and communicating (Caroli et al. 2001; Belloc and Guerrieri 2015). The IT diffusion has, therefore, been accompanied by investments in skilled labour (Caroli and Van Reenen 2001; Strohmaier and Rainer 2013). This tendency is particularly pronounced for industries, heavily investing in new technologies (Berger and Frey 2015). The decreasing relative demand for lower skilled labour and increasing relative demand for higher skilled labour, therefore, points to the fact that technological change is not only task-biased, but also skill-biased (Griliches 1969; Bresnahan et al. 2002; Autor et al. 2003).¹⁷

In particular, in a computerised work environment, advanced skills, related to information processing, become more valuable as there is a complementarity between the relative strength of the IT - such as, capacity for storage, communication and processing of information - and certain types of human

¹⁷ There are, however, some claims in the recent literature that as at least early domains of IT applications reach maturity, knowledge related to them becomes codified and widely diffused eroding the initial comparative advantage of highly skilled workers (Consoli and Rentocchini 2015; Vona and Consoli 2015).

judgement and decision making (Bresnahan et al. 2002). As a result, the demand for more educated and higher skilled workers increase; also, more investments are directed towards (re)training current workers. The information overload bottleneck is particularly severe for workers at higher levels of the organisational hierarchy since 'what information consumes is [...] the attention of its recipients' (Simon 1971, p. 40). As routinisation is one of the main mechanism producing polarisation in the labour market, this also implies increased wage differentials between low and highly skilled workers (Böckerman et al. 2016). Most prominent IT-using industries, particularly in the service sector, have seen these changes earlier (Wolff 2006) and to a greater extent (Autor et al. 1998) than others. And yet, all sectors of the economy were, in one or another way, affected by this process.

Task- and skill-biased character of IT-induced technological change has had important implications for the geography of job creation. The increased demand for skilled labour, employed in non-routine tasks, and the uneven distribution of technology-intensive industries, positing this demand, across regions has led to regions with higher skill endowments to become even more skill-endowed over time (Moretti 2012). In that respect, the diffusion of the IT, and the subsequent substitution of labour performing routine tasks by labour involved in more abstract tasks, have benefitted regions that were originally endowed with higher levels of human capital by boosting the job creation in these regions (Berger and Frey 2016).

7. Research setting: Sweden

7.1. Swedish economic growth and labour market trends, 1985-2010

Overall economic development of Sweden between 1985 and 2010 was characterised by a clear cyclical pattern as reflected in the GDP growth rate (Figure 4).



Figure 4. Economic growth in Sweden, 1985-2010 (annual GDP growth rate, fixed 2005 prices)

Over twenty-five years between 1985 and 2010, the Swedish economy experienced two periods of contraction. First, in the early 1990s, the Swedish demonstrated the negative growth rates for three years in a row (1991-1993) as an outcome of the crisis that started in the banking sector and spread out to

the whole economy. Second, in the late 2000s, when the Swedish economy was hit by the repercussions of the global financial crisis of 2007-2008 and subsequent Great Recession. Between these two crises, the Swedish economy demonstrated a rather steady economic growth, averaging at 3.3 per cent per annum between 1994 and 2007. As a result, the recovery from the early 1990s crisis took approximately two years. By 1995, the Swedish economy returned to the pre-crisis level of GDP.

The recovery of the labour market from the early 1990s crisis went much slower (Figure 5).



Figure 5. Total employment in Sweden, 1985-2010 (individuals aged 16-64)

Following the outburst of the crisis in the early 1990s, the total employment in Sweden contracted by more than 17 per cent. That is, almost each fifth job disappeared. The outcomes of job destruction were particularly pronounced in the manufacturing sector. Reaching the bottom level in 1994 and remaining there until 1997, the total employment demonstrated a steady recovery starting in 1998, when the employment growth rate averaged at approximately 1.5 per cent per year until 2008. It should be noted, however, the employment growth was slower than the expansion of the economy as a whole. In that respect, the GDP growth in the period between two major crises was driven, to a great extent, by the increases in labour productivity level. Nevertheless, the recovery of the labour market after the early 1990s crisis was never complete as the Swedish labour market never returned to the pre-1990s employment level.

At the same time, the aggregate labour market recovery statistics mask rather pronounced regional differences (Figure 6).



Figure 6. Employment growth across Swedish regions, 1985-2010 (1985=1)

The early 1990s crisis hit regions at all levels of the regional hierarchy. irrespective of the level of the regional hierarchy. However, effects of the crisis in more peripheral regions were more severe and longer lasting. The regional labour markets of Stockholm and Gothenburg labour markets began their recovery as early as 1995 and reached the pre-crisis employment by 2000 and 2003 respectively. In Malmö, the recovery process started later (1997), and the regional employment returned to the level of 1990 by 2006. While the recovery trend was observed in large regional centres, their labour markets never returned to the pre-crisis level. In smaller regional centres and most peripheral regions the employment stabilised after the fall and never recovered.

It is peculiar that by 2010 the performance of regions in terms of their recovery from the early 1990s crisis neatly corresponded to their position in the regional hierarchy. This manifested the divergence patterns across regional labour markets and continuing concentration of labour to the most

dynamics regions at the top of the regional hierarchy. In that respect, this underlines the importance of accounting for the level of a region in the regional hierarchy when analysing regional labour market dynamics.

7.2. The IT in Sweden

Linux, Skype, Spotify, ... The list of the Swedish IT start-ups, which shaped the evolution of the global IT, includes more widely-known names, and even more less famous ones. Over last 30 years, Sweden has become one of the world-leading knowledge economies with a strength in both providing and using the IT (Giertz et al. 2015), which is certified by the top positions of Sweden in various international rankings of the IT development (see Section 1.3.).

While it is rare in the world history that small economies are global leaders in the major sector, this is indeed the case with Sweden. In this section, a brief overview of the Swedish way towards the IT success as well as the current state of the IT sector in Sweden is performed.

The beginnings of the Swedish IT sector date back to the late 1940s.¹⁸ As in the case with many other countries, the IT research was initiated by the state and remained in its domain for a long time. In 1948, the Swedish government, being among the first in Europe, established the Board for Computing Machinery (*Matematikmaskinnämnden, MMN*), whose task was to develop and produce the first Swedish computers. Already in 1950, the first computer – Binary Arithmetic Relay Calculator (BARK) – was completed, followed by the Binary Electronic Sequence Calculator (BESK) in 1953. The latter was primarily used for the national security purposes, but also for some civil applications, such as meteorological studies.

In 1954, MMN signed the contract with Saab (car and military planes manufacturer) that regulated the collaboration for the further development of BESK. The major stakeholders from the state side were the Swedish military and the central Swedish government that both were early adopters of computers in various agencies and divisions they comprised. This marked the start of the public-private collaboration in the IT development in Sweden.

¹⁸ Historical overview of the Swedish IT sector relies heavily on Giertz (2015).

By 1976, more than twelve thousand computers from Saab (later Datasaab) were sold. The problem was, however, that the demand for Saab computers remained mostly domestic. Because of particularities of the data processing architecture, most customers worldwide preferred more wide-spread IT systems developed by IBM. The opportunities for expanding the market of Saab-manufactured computers were limited, to a great extent, by the size of the domestic market.

The next era in the Swedish computer manufacturing began in the late 1970s, when Luxor AB – Swedish home electronics manufacturer that was founded in 1923 – entered the market of personal computers. In August 1978, the first Advanced BASIC Computer 80 (ABC 80) was launched. It became a success on the domestic market. Being originally designed and used for industrial automation, office and education purposes, it also managed to grasp a substantial share of the emerging Swedish PC market. Following the expansion of the PC market, Luxor introduced the successor of ABC 80 -ABC 800 - in 1981. It was a computer aimed at the home use market primarily. With a great performance and relatively low price, it soon reached 60 per cent of the Swedish PC market. The problem of both ABC 80 and ABC 800 was that they still employed their own data processing standards, which were not compatible with US-manufactured computers that dominated the global market. The inconvenience of this incompatibility increased as computers were used in continuously expanding range of applications. As a result, the Swedish companies started to buy computers from abroad; and by the end of the 1980s the US-based companies took over the Swedish computer market completely.

Beginning in the 1990s, the deregulation of IT sector and declining state support (Zaring and Eriksson 2009; Giertz 2015) as well as the increased international standardisation of the IT equipment led to the contraction of the IT manufacturing in Sweden. At the same time, the ongoing digitalisation of production processes in a wide range of industries boosted substantially the demand for provision of software and other IT-related services. This resulted in a rapid expansion of the services part of the IT sector. To provide a perspective on the reverse tendencies in growth of the IT manufacturing and IT services, Table 3 reports employment numbers in these two parts of the IT sector in Sweden between 1985 and 2010.

	1985	1990	1995	2000	2005	2010
IT manufacturing	25369	27767	25150	28410	18545	17028
	57.87%	45.22%	38.13%	23.73%	16.72%	13.79%
IT services	18467	33632	40807	91336	92395	106431
	42.13%	54.78%	61.87%	76.27%	83.28%	86.21%
Total	43836	61399	65957	119746	110940	123459

Table 3. IT employment in Sweden

Between 1985 and 2010, the total employment in the Swedish IT sector increased almost three times - from 43,836 to 123,459 workers. This corresponded to the increase of the sector's share in the national employment from 1.0 per cent to 2.8 per cent.

This growth was predominantly driven by the expansion of the IT services part of the sector, where employment increased almost six time – from 18,467 workers in 1985 to 106,431 workers in 2010. At the same time, over the considered time period the employment in IT manufacturing contracted by more than 8,000 workers. The contraction proceeded at the particularly fast pace in the aftermath of the 2001 dot-com crash, the result of which was, among other things, the global restructuring of the IT sector.

These tendencies led to the increase of the IT services employment share in the Swedish IT employment from 42.1 per cent in 1985 to 86.2 per cent in 2010. In that respect, the expansion of the Swedish IT sector was paralleled by its functional reorientation – from being primarily manufacturing-driven in the beginning of the time period to predominantly service-driven in the end.

This expansion of employment in the Swedish IT sector was unequally distributed across regions at different levels of the Swedish regional hierarchy (Figure 7). For instance, three Swedish metropolitan areas – Stockholm, Gothenburg, and Malmö – concentrated by far the largest share of IT employment. Over the considered time period, around 70 per cent of IT workers (on average) were employed at the local labour markets of these three regions. Stockholm alone concentrated around 45 per cent of IT employment. Moreover, while the IT sector expansion (in absolute numbers) took place in all regions, metropolitan areas of Sweden demonstrated faster growth than non-metropolitan regions. This implies the further concentration of the IT activities to the most dynamic regions at the top of the regional hierarchy.



Figure 7. IT employment across Swedish regions, 1985-2010

The outpacing expansion of the IT sector in metropolitan areas, coupled with its increased servicification, points to the fact that most IT-producing firms locate their businesses in large regions because the demand for IT services in these regions is particularly high due to scale effects – as these regions host more potential client companies – and higher demand for advanced (or, strategic) IT solutions that, as it was discussed in Section 6.1., tends to be higher in large regions.

7.3. The 'Swedish model' of the labour market

The Swedish labour market between the late 1960s and the early 1990s was characterised by a situation of full employment. In the aftermath of the deep economic recession in the first half of the 1990s, unemployment rose dramatically to levels, which were not observed since the 1930s (Bengtsson

2012). Even the subsequent decrease in the unemployment rate over the late 1990s and the 2000s, the Swedish economy never returned to the situation of full employment. In 2015, the unemployment rate among the work age population was 7.4 per cent (SCB 2016a).

Besides, for couple of decades prior to the 1990s there was a decline in the wage inequality among Swedish workers, both with regards to the overall wage dispersion and to its particular components, such as relative earnings advantage of highly educated workers, wage differences between men and women, etc. However, the trend for declining wage inequality was also broken in the aftermath of the early 1990s crisis (Edin and Holmlund 1995; Domeij and Floden 2010).

These trends, as well as their temporal congruence, may be related to the rise and demise of the 'Swedish model' of the labour market that is characterised by the solidarity wage policy – realised through the centralised bargaining mechanism – and a vast array of active labour market policies aimed at improving the efficiency of labour matching. In what follows, basic features of the model as well as its consequences for the functioning of the Swedish labour market are discussed.

The post-World War II labour market policy of Sweden traces back to works of Gösta Rehn and Rudolf Meidner – both economists in the Swedish Confederation of Trade Unions (*Landsorganisationen i Sverige*) – who developed a model that underlined the crucial role of labour market policy for achieving economic growth and price stability, on the one hand, and full employment, wage levelling, and union solidarity, on the other hand. Rehn-Meidner-inspired policies aimed at promoting the labour market flexibility through improved income and employment security (van den Berg et al. 1997). These policies combined three components, namely: a strict fiscal policy, solidarity wage policy, and active labour market policies.

First, the Rehn-Meidner model aimed at building a balanced economy with low inflation. Their fiscal policy was based on Keynesian principles and assumed that the excess demand should be addressed through fiscal policy, while admitting the possibility of high short-term structural unemployment. For instance, a high-tax policy was suggested to replace redundancies in the private sector with public employment (Bengtsson 2012).

Second, the Rehn-Meidner model promoted the solidarity wage policy, that is, a rational differentiation of wages within and across sectors that would result in the wage compression (Lundh 2005). The effect of this policy was that firms and sectors with low productivity and profitability levels – unable to withstand the international competition – would face downsizing as they would not be able to pay wages that were centrally bargained. More productive firms and industries would then get more opportunities for expansion, which, in its turn, would boost national productivity levels and the aggregate growth. In that respect, the job protection was replaced by the employment protection. In other words, according to the the Rehn-Meidner model, it was the national employment that should be defended, rather than particular job matches (Magnusson 2007). The mechanism of wage dispersion – traditionally used for the effective labour allocation – was replaced by mechanisms of job, industry, and regional mobility.

Finally, as the increased flexibility of the labour market would generate higher unemployment in the short run, the Rehn-Meidner model offered a wide range of active labour market policies that addressed this short-term mismatch on the labour market. Training as well as allowances for the worker mobility across jobs, industries, and regions were suggested as policy instruments aimed to help people to find jobs in more productive parts (in both sectoral and regional dimensions) of the economy (Bengtsson 2012).

Throughout the 1950s-1960s, the Rehn-Meidner model was accepted by actors on both supply-side (trade unions and other worker organisations) and demand-side (employer unions) of the labour market. By the late 1970s, however, the model began to lose its legitimacy, primarily with employers but also with some trade unions and local interest groups (Lundh 2005). Several factors contributed to that.

First, the Swedish society changed. The expansion of the public sector and increased number of white-collar workers facilitated the development of a white-collar union movement. Wage demands posed by workers, participating in that movement, exceeded those considered normal within the centralised bargaining system (Lundh 2005). Also, the policy of employment protection, that replaced the job protection policy, was undermined by the adoption of the 1974 Swedish Employment Protection Act and the 1976 Act on Employee Consultation and Participation in Working Life (Bengtsson 2012). Besides, the focus of active labour market policies shifted in the 1970s and 1980s towards interventions on the demand-side of the labour market, e.g., job creation schemes in the public sector as an alternative to the private sector employment.

Second, the demise of the mass production system and the IT-induced economic transformation pointed to the need for a functioning local wage formation. As the IT-induced technological change was skill-biased, skilled workers became increasingly important for firms' competitive performance. High costs of employing and (re)training such workers made employers search for options to tie workers to their jobs through, for example, higher relative wages, other non-monetary benefits, etc. At the same time, an oversupply of lower skilled workers and undersupply of jobs, which these workers could occupy, undermined employment protection mechanisms and resulted in smaller incentives for unskilled workers to change jobs. In other words, the flexibility mechanisms of the Rehn-Meidner model did not pay off for unskilled workers anymore. This resulted in the shift of the overall focus of firms from the worker collective to individual workers. On the policy side, there is now a stronger emphasis on qualifications and activities of particular job seekers and the individual provision of the labour market policies (Bengtsson 2012).

Both these factors intensified the firms' need for wage flexibility as an instrument to recruit and retain attractive workers. Under such circumstances, it was important for firms to deviate from centralised wage bargaining within the solidarity wage policy. Developments in the 1980s-1990s resulted in more opportunities for firms to influence wages based on their own prerequisites. The union agreements were increasingly the basis that was complemented with further wage negotiations in firms. By the end of the 1990s, about half of the latter were distributed to the local level (Lundh 2005).

This is, however, not to say that the trade unions lost their role on the Swedish labour market. Compared to the weakening of the trade union movement in other European countries throughout the 1980s-1990s, the degree of the trade union organisation as well as the extent of coverage of collective agreements in Sweden remained high in the international comparison (Lundh 2005).

In that respect, the consequences of the Rehn-Meidner model implementation may still be observed in Sweden. Even though there is an upward trend in overall wage inequality since the early 1990s (Domeij and Floden 2010), Sweden is still characterised by one of the most equal income distributions (OECD 2011). With this in mind, it comes as no surprise that studies of internal migration demonstrate that reallocation of labour in Sweden is driven primarily by the patterns of job availability rather than wage differentials. (Westerlund 1997, 1998; Fredriksson 1999; Gärtner 2014). In other words, geographical reallocation of workers is directed first and foremost to regions characterised by higher rates of job creation, while the effect of regional wage differentials is relatively minor.

Fredriksson (1999) shows that mobility of workers in Sweden is more responsive to national economic conditions and region-specific shocks in the demand for labour than in any other European country. In other words, the Swedish labour market is more responsive to the shifts in the macroeconomic situation than that of other European countries. In that respect, it is interesting that switching a job is a major response to job destruction for Swedish workers, compared to becoming unemployed or leaving the labour market. This, among other things, results in the regional reallocation of workers in Sweden to remain high by European standards.

8. Conclusion

This chapter provides an overview of the major findings from the articles included in this dissertation. It continues by synthesising these findings and presenting general conclusions and emergent topics.

8.1. Summary of the articles

Article I addresses the gap in the literature with respect to identifying the leading industries associated with technological change at various times and measuring their effects on functioning of the economy. This is performed by revisiting and empirically refining Perez' (1983, 2010) classification of key industries driving the technology-induced transformation process into motive, carrier, and induced branches.

By analysing the patterns of inter-regional worker reallocation in Sweden between 1985 and 2010, the following research question is addressed:

In what ways do spatial patterns of distribution of leading industries associated with a current wave of technological change shape inter-regional labour reallocation flows?

It is hypothesised that a stronger regional presence of leading industries positively affects the ability of regions to attract and retain workers.

Results of the analysis confirm the hypothesis above. That is, the industrial structures of regions in general, and the presence of leading industries in particular, played an important role in shaping patterns of inter-regional reallocation of workers in Sweden over the considered time period. However, the decomposition of these effects in time demonstrated that there are several important qualifications to this result.

First, different groups of leading industries played different roles in attracting labour to and/or retaining labour in regions at different stages of the GPT diffusion process. The positive role of motive branches – comprising the activities that are paradigmatic to the recent wave of technological change – are only observed for the period between 1985 and 1998, but not at later stages of the GPT diffusion or over the whole time period. Rather, the locational patterns of carrier branches – the most visible and active users of the GPT – played the most pronounced role in shaping patterns of interregional worker reallocation. Induced branches were important in terms of generating gross flows of workers between regions, while their effect is not visible in the resulting net inter-regional labour reallocation.

Second, locational patterns of service branches are the major factor shaping the patterns of inter-regional labour mobility. Their role is particularly pronounced in terms of attracting workers to regions. The role of manufacturing branches is quite different. Their regional presence is a stabilising factor, helping regions to retain workers.

Third, the impact of diversification of regional employment mixes is different at different stages of the GPT diffusion process and, particularly, in regions at different levels of the regional hierarchy. At the early stages of GPTinduced transformation, the increased specialisation has a negative effect on the ability of regions to attract and retain labour. This effect becomes positive later. Negative effects of specialisation are especially strong in metropolitan areas. More peripheral regions benefit more from specialisation in terms of higher worker inflows, on the one hand. On the other hand, higher specialisation also generates larger worker outflows from these regions. This might explain why peripheral Swedish regions had problems with recovering from the early 1990s crisis with regards to labour market dynamics (see Section 7.1.).

All in all, the results of empirical analyses demonstrate that the presence of leading industries in regions has an impact on their ability to attract and retain labour. In that respect, the industrial restructuring, and the resulting structural change, induced by the diffusion of the IT as a GPT, affected the patterns of worker reallocation across regions. This provides the evidence for the role of the labour market as a response mechanism in the process of technology-induced adjustment. It is demonstrated, however, that the particular mechanisms of the labour market response change over time, as the GPT is

diffused across wider spectrum of sectors, and in regions at different levels of the regional hierarchy.

Article II combines the insights from the industry dynamics literature with the labour market studies to investigate how the evolution of the knowledgeintensive IT services sector in Sweden was reflected in the dynamics of its labour force building in the period between 1991 and 2010. Two research questions are posited:

How did the skill and experience composition among incoming staff vary as the IT services sector evolved over time?

How did previously acquired experience and skills affect entry wages of incoming workers, their propensity to stay in the sector and wage performance, reflecting the workers' measured value to the sector during different phases of its development?

It is demonstrated that the development of the IT services sector was unequally distributed across regions at different levels of the regional hierarchy. Not only did metropolitan areas concentrate around 75 per cent of the total employment in the sector, the regional patterns of sectoral employment growth strengthened the differences across regions in favour of major metropolitan areas over time.

The results of empirical analyses suggest that while the sector was rather turbulent with regards to its labour market dynamics, the expectations, requirements, and valuations of skills of new entrants into the sector consolidated fairly early in the process of its expansion. In fact, the evolution of the sector was reflected not so much in *how employers in the sector valued* skills of inflowing individuals, but in *what kind of skills* employers in the sector tended to attract over time.

The labour dynamics around the sector suggests the evolution of an increasingly specialised structure over the considered time period. In other words, the recruitment in the sector appeared to rely increasingly on inflows from related industries as it matured. For instance, individuals with prior experience in related industries enjoyed better entry conditions as well as more lasting employment prospects. Interestingly, however, the benefits accruing from having related work experience in the past vanish rather quickly, to the benefit of on-the-job training and formal education.
With regards to the latter, there is a strong evidence that the higher education became an increasingly important entrance ticket to jobs in the sector. While this trend is observed for labour force in Sweden in general, it is much more pronounced in the IT services sector. Also, it appears that the sectoral core competences are not embedded in one dominant educational or occupational group, but are rather distributed and made operational across a range of skill backgrounds in combination. In that respect, it provides an important lesson for the educational policy: educational programmes should be tailored to provide workers not only with core competences related to the specialisation of the programme, but also with a wide range of complementary skills.

One of the findings is that sectoral crises are important periods of evolution of specialised labour market structures around the sector, as well as of sectoral consolidation more generally. While the sectoral labour market dynamics is substantial during crises, it is of a different nature. Individuals getting jobs in the sector in these periods constitute a more homogeneous group than during the boom years. In other words, in crisis times sectoral recruitment patterns are shaped in the way of increased consolidation towards narrower skill profiles.

Finally, labour sourcing process manifests itself differently in regions at different levels of the regional hierarchy. Stockholm (and later Gothenburg and Malmö), being the hotspots of the sectoral employment, provided the highest wage benefits for workers relocating from other regions. What is interesting, while the wage premium for local entrants in metropolitan areas is realised through higher entry wages, while regional movers to metropolitan areas tend to enjoy better post-entry wage performance. Also, regional movers tend to be less entrenched in the sector as they demonstrate the lower probability to remain employed in the sector over time.

On a more general level, Article II suggests that the labour market dynamics in the core sector, associated with the GPT diffusion, develops towards a higher specialisation of its employment mix. While at the early stages of the GPT diffusion process the availability of relevant skills is limited, employers in the sector, nevertheless, develop the vision of relevant skill profiles rather early. This relevance is reflected in the entry conditions and employment prospects of individuals. Over time, as the availability of skills increases, these valuation structures do not change much. However, the 'content' of worker inflows with regards to skills embedded in workers increasingly resembles early defined relevant skill profiles. This implies that the availability, overall and particularly regional, of workers possessing certain skills is an important enabler for the evolution of GPT-related sectors.

The findings in Article II resonate nicely with those in **Article III** that investigates the distribution and reallocation of workers in the IT occupations across different industries and regions in Sweden over 2001-2010 with the aim of exploring the patterns of diffusion of IT-related competences in the Swedish economy. Particular emphasis in the article is made on the expansion of IT employment outside of the IT sector.

The research questions in the article were formulated in the following way:

To what extent does the distribution of the IT workforce between the IT and non-IT sectors reflect the quality of IT workers' human capital?

In what way does the position of a region in the regional hierarchy affect the regional distribution of the IT workforce between the IT and non-IT sectors?

How are the dynamics of the IT workers reallocation conditioned by sectoral and regional factors?

It is demonstrated that the labour market for IT workers expanded rapidly during the 2000s. What is interesting, this expansion was primarily driven by the outpacing growth of IT employment outside the IT sector. Out of 30,000 new IT jobs generated in the economy 75 per cent were created in non-IT sectors. This resulted in the decrease of a share of the IT sector in the total IT employment.

This expansion was coupled by the qualitative upgrade of the IT workforce in both IT and non-IT sectors. It realised through the increased share of workers with higher education and the substitution of technical IT labour by relatively higher skilled IT workers (professionals and managers).

In that respect, the diffusion of the IT manifested itself in the spread of (increasingly advanced) IT competences beyond the core IT sector. Thus, pervasiveness of the IT as a GPT implies not only the increased investment in the IT capital (computers and related equipment) but also the expansion of the labour market for IT workers within and beyond the core, GPT-related, sector.

It appears, however, that the pervasiveness is not universal in the sense that IT workers may easily switch IT jobs across different industries. There are rather rigid structures of worker reallocation as individuals switch IT jobs predominantly within the same industry they were previously employed in. This suggests that the pervasiveness characteristic of the IT as a GPT should be qualified further as specialised pervasiveness. That is, the IT, indeed, possesses the capacity to generate IT-related jobs in all sectors of the economy (thus, pervasiveness). At the same time, the existing clusters of worker reallocation in the industry space point to the increasing specialisation of IT workers that may limit employment opportunities for them in terms of finding an IT job in another industry (thus, specialised).

In that respect, the tendency towards an increasingly specialised labour market structure around the IT services sector, observed Article II, may also be extended to the case of the labour market for IT workers in non-IT sectors.

The results of analysis also point that the rapid expansion of the labour market for IT workers was accompanied by changes in the organisation of the market with respect to the reinforcement of regional hierarchies. For instance, compared to regions at lower levels of the regional hierarchy, metropolitan areas are characterised by larger, more IT-intensive, and more specialised labour markets for IT workers. The issue of regional availability of specialised IT competences is, therefore, highlighted. On the one hand, the share of non-IT sectors in regional IT employment mixes increased at lower levels of the regional hierarchy. A possible explanation to that is a limited supply of IT-related activities by specialised IT firms in regions at lower levels of the regional hierarchy which forces firms in these regions to develop IT competences internally. On the other hand, firms in more peripheral regions tend to employ workers with less specialised skill profile as well as workers from outside their immediate surroundings.

With regards to this dissertation, two important conclusions arise from Article III. First, the GPT diffusion requires not only investment in the GPT-related physical capital, but also hiring workers to perform GPT-related tasks. These workers are instrumental for making the best use of the GPT-related capital being rolled out. In that respect, hiring workers, possessing specialised human capital, is a mechanisms of establishing the innovation complementarities between the activities of firms and new technology being diffused.

Second, increasingly specialised patterns of worker reallocation imply that limits in availability of specialised human capital may constrain the opportunities for GPT-enabled development. Structures of supply of and demand for the GPT-related skills on regional labour markets may impose barriers to the successful diffusion of a new technology, particularly to more peripheral regions.

Finally, **Article IV** focuses on entrepreneurship as a potential mechanism of the IT diffusion from metropolitan areas to more peripheral regions. By investigating the spatial biographies of individuals, performing entrepreneurial entries in the IT sector in Swedish non-metropolitan regions between 1991 and 2010, the paper addresses two research questions:

In what way does prior locational behaviour of individuals influence their propensity to become entrepreneurs?

In what way is the post-entry performance of new firms affected by pre-entry locational behaviour of their founders?

The results of analysis suggest that entrepreneurship in the Swedish IT sector is characterised, to a great extent, by the locational inertia. Between 1991 and 2010, more than 75 per cent of entrepreneurial entries in non-metropolitan regions were performed by local individuals. Local individuals were more likely to prefer an entrepreneurial option to paid employment. Also, firms established by local entrants tended to survive longer.

At the same time, the results also suggest that the spatial origin of entrepreneurs should be considered beyond the local/non-local dichotomy. It is not just the location of individuals immediately prior to performing entrepreneurial entry, but also their extended spatial biographies that matter for their entrepreneurial activities. It is demonstrated that in certain cases individuals with broad spatial careers are more likely to start successful businesses than local entrants.

One of the more interesting results concerns the role of metropolitan areas in shaping the entrepreneurial dynamics in regions at lowers levels of the regional hierarchy. It is demonstrated that having prior work experience in metropolitan areas increases the probability of an individual to start a new firm, and a successful one, in a non-metropolitan region. In that respect, it is suggested that metropolitan areas are learning hotspots for individuals willing to obtain sector-specific knowledge in a dynamic urban context. This is particularly emphasized for individuals relocating to metropolitan areas for a while, but returning to their home regions to start a business. In this case, they

are both embedded locally in a region where they perform an entrepreneurial entry, but also embedded in broader knowledge networks at the sectoral level.

The effect of the metropolitan employment tenure is, however, mostly observed in the 1990s. This points that the relationship between spatial biographies of individuals and their entrepreneurial activities is related to the evolution of the sector itself and the environment, in which it operates.

For instance, the rapid expansion of IT-related activities in non-metropolitan areas, adaptation of educational systems to the digital paradigm, as well as the overall diffusion of IT as a GPT improved the opportunities for obtaining IT-related knowledge beyond the metropolitan areas' boundaries. This results in the decreasing importance of metropolitan areas as learning regions and source of IT entrepreneurs for more peripheral regions.

Article IV, therefore, suggests that the patterns of regional distribution of GPT-related activities in regions at different levels of the regional hierarchy have important implications for diffusion of the GPT over time. At the early stages of the GPT diffusion, the regions at the top of the regional hierarchy are hotspots for GPT-related activities, as it is predicted in the literature. Therefore, they act as learning regions for individuals willing to obtain work experience related to the GPT. As time goes by, the GPT-related activities diffuse across a wide range of regions, which manifests itself in the decreasing role of large regions as the only locomotives of GPT-induced transformation process.

All in all, Articles II-IV imply that considering the labour market only as a response mechanism to the GPT-induced structural change is an oversimplification. Availability of workers possessing certain skills has important implications for how, when, and where the technology-induced economic transformation takes place. In that respect, the labour market should also be seen as an enabler and/or restrictor for the opportunities of the economy to respond to transformation pressures generated by the introduction and diffusion of the GPT.

Table 4 below provides a brief overview of four articles with respect to their aim, theoretical framework and main findings.

les
tic
a
the
of
ew
Ξ.
Ne
\circ
4
le
b
Ë

Main findings	Stronger regional presence of leading industries, associated with the GPT- induced structural change, positively affects the ability of regions to attract and retain workers. Different functional groups of leading industries play different roles at different stages of the GPT diffusion process. Manufacturing branches act as a stabilising factor, helping regions to retain workers; service branches are the factor of attracting workers to regions. The degree of diversification of regional employment mixes has a different impact on the ability of regions to attract and retain workers: • in regions at different levels of the regional hierarchy; • at different stages of the GPT diffusion process.	Expectations, requirements, and valuations for skills of new entrants into the IT services sector consolidated fairly early in the process of its expansion The evolution of the sector is reflected not so much in how employers value the skills of incoming workers, but rather in what kind of skills individuals, hired into the sector, possess. For instance, the increasingly specialised structure with regards to importance, the increasingly specialised structure with regards to importance of industry experience evolved around the sector over the period of investigation. Also, the increased role of higher education as an entrance ticket to jobs in the sector is underlined. Crises are important periods for the sectoral consolidation and restructuring. The labour sourcing process has a strong geographical component, which is reflected in both entry conditions for workers becoming employed in the sector and the probability of them to remain employed in the sector and the probability of them to remain employed in the sector and the probability of them to remain employed in the sector.
Theoretical framework	Evolutionary economic geography Neo-Schumpeterian economics Labour economics	Evolutionary economic geography Industry studies Employer-employee matching theory Skill-relatedness
Aim	To assess the effects of technology-induced structural change on an ability of regions to attract and retain workers	To investigate structures of labour sourcing in the Swedish knowledge- intensive IT services sector during the period of its rapid expansion between 1991 and 2010
Article number and title	Article I. Technological Change and Geographical Reallocation of Labour: On the Role of Leading Industries	Article II. Labour force building in a rapidly expanding sector

articles
of the
erview
l). Ov
(continued
able 4

Pervasiveness of the IT as a GPT realises, among other things, through increased employment of IT workers in both IT providing and IT using sectors in regions at all levels of the regional hierarchy.	This pervasiveness is, however, specialised. That is, the IT has the abili generate IT-related jobs in all sectors of the economy but with an increa: degree of specialisation and subsequent segmentation of the labour markets for IT workers.	The degree of specialisation of the IT workforce is higher in regions at higher levels of the regional hierarchy.	 Entrepreneurs in the knowledge-intensive IT services sector exhibit locational inertia. That is, individuals tend to start IT firms in the region where they previously lived and worked. Also, entries by such individual tend to survive longer. However, in certain cases individuals with broad spatial biographies are more likely to start successful businesses than local entrants. The role of metropolitan areas goes beyond providing the most fertile ground for knowledge-intensive sectors. Having work experience in metropolitan areas increases the probability of an individual to start an firm (and a more successful one) in a non-metropolitan region. The importance of metropolitan work experience, however, decreases o time.
Evolutionary econorr geography GPT theorv			Evolutionary econom geography Entrepreneurship studies Population geograph
To explore the patterns of diffusion of IT-related competences in the Swedish economy in the period between 2001 and 2010			To investigate the extent to which spatial biographies of individuals, that is their histories of prior relocation, have an impact on their propensity to become entrepreneurs in the knowledge-intensive IT services sector and post-entry performance of firms they establish.
Article III. Specialised pervasiveness: Diffusion of T	competences in the Swedish economy		Article IV. Local or not? Spatial biographies of Swedish IT entrepreneurs

8.2. Synthesis of results

The aim of this dissertation is to investigate the intertwined dynamics of the industrial restructuring and labour market evolution induced by the introduction and diffusion of a GPT. It is hypothesised that the labour market played a dual role in this process: on the one hand, the response mechanism smoothing the transition to the new technological paradigm; and, on the other hand, a mechanism which itself enables or restricts the dynamics of industrial restructuring at different stages of the GPT diffusion process and across regions at different levels of the regional hierarchy. The particular focus is made on identifying the linkages between the micro-level processes of GPT-induced transformation with the macro-level determinants and outcomes of these processes through the meso-rules that ensure the translation of the former into the latter.

The analyses in four articles, included in this dissertation, provide a rather strong support for the hypothesis about the dual role of the labour market as it was discussed in the previous section. In what follows, the results of individual articles are synthesised and several emergent topics are presented.

The main finding of this dissertation is that the co-evolutionary dynamics of industrial restructuring and labour market evolution in the process of the GPT diffusion followed the meso-trajectory that may be described as the *transition from the GPT-centred story to the GPT-related story*.

At the early stages of the diffusion of the IT as a GPT, the core GPT-related sector – the knowledge-intensive IT services sector – played the most important role in generating the labour market dynamics across and within regional labour markets. For instance, regional presence of the sector was one of the main factors of attracting labour to regions (Article I). This period was characterised by the rapid expansion of the sector itself. The employment growth was fuelled by hiring workers with rather diverse backgrounds from a rather open pool of potential employees (Article II). At the same time, the concentration of the sector to large regions at the top of the regional hierarchy made these regions learning hotspots for individuals willing to obtain GPT-related skills. Employment experience in these regions provided individuals with superior learning opportunities which became a precondition for establishing successful businesses in the GPT-related story.

As time went by, there were notable changes in the development of the GPTrelated sector itself as well as in the labour market dynamics around it. For instance, the sector lost its role as the major factor of attraction of workers to regional labour markets. Rather, this role was delegated to the industries that became the most active and visible users of the GPT (Article I). Later stages of the GPT diffusion are characterised by the increased spread of GPT-related competences in a wide range of sectors and regions (Article III). At the same time, more specialised labour market structures evolved around both GPTproviding and GPT-using sectors with respect to sourcing GPT-related human capital (Articles II and III). This suggests that the GPT-related competences become less generic and more specific to the sector where they are applied. Also, large regions at the top of the regional hierarchy lost their monopoly as superior learning regions (Article IV). In that respect, the locus of GPT-induced transformation shifted both in industry and regional spaces to the wider range of industries and regions. Thus, the IT-related story.

These dynamics reflect nicely the predictions of the GPT and TEP literatures with their focus on the long-term aspect of the technology diffusion process. Indeed, the early GPT diffusion is an uncertain process of realising the potential of a new technology and making required complementary investments. At this stage, the GPT diffusion is conditional upon the dvnamics of GPT-related sector as well as behaviour of early adopters in other sectors. Therefore, the core GPT-related sectors is at the centre of the transformation story. With time, the GPT pervasiveness potential and emergence of innovation complementarities between old and new technologies across a wide range of sectors - located far from each other in the technological dimension - ensures that the benefits of the GPT are recognised by the increased number of economic agents in the economy. This is reflected in the shifting roles of various actors in the GPT-induced transformation. Thus, the identified meso-trajectory manifests in the increased role of the GPT as the emergent complement to already existing activities (Verspagen 2004) and the platform for transformation for the economy as a whole (Castellacci 2008).

The transition from the GPT-centred to the GPT-related story underlines the importance of considering the temporal dimension of the GPT diffusion process. It is no longer possible to assume that it proceeds linearly in time. To study the impact of technological change on the evolution of the economy, one should acknowledge the existence of particular points in time – some of

which may be driven by chance events – that characterise the shifts in the technology diffusion process.

Nobody can hope to understand the economic phenomena of any, including the present, epoch who has not an adequate command of the *historical facts* and an adequate amount of *historical sense* or of what may be described as *historical experience*. (Schumpeter 1954, pp. 12-13; emphasis in the original)

For instance, with regards to the diffusion of the IT as a GPT, the shifts in empirical patterns observed in individual articles took place simultaneously – around the turn of the millennium. Peculiarly, these changes coincided in time with the 2001 dot-com crash that was the crisis specific to the IT sector.

It is well possible that this crisis boosted the transition from the GPT-centred to the GPT-related story, or that it didn't. Nevertheless, it was the milestone that marked the beginning of the transformation of the IT sector itself (Article II) and of the increased accumulation of the IT competences by non-IT sectors (Article III), which resulted in the changing roles of various sectors in shaping the patterns of inter-regional worker reallocation (Article I) as well as shifting roles of regions at different levels of the regional hierarchy, at least with respect to their roles as learning regions (Article IV). Without accounting for these trends, and identifying the points in time when they occur, one cannot arrive at any satisfactory explanations of the impact of technology-induced transformation on the economy.

This is, however, not to say that any GPT will necessarily follow the same temporal patterns, or have as easily identifiable time points of fundamental shifts in the diffusion process. On the contrary, this points to the necessity to acknowledge that each GPT develops in its historical time and there is a myriad of micro-level processes and macro-level factors that interact in complex ways to shape its diffusion path.

Another important meso-rule that is identified through the analyses underlines the importance of spatial factors for the diffusion of the GPT. It may be formulated in the following way: a position of a region in the regional hierarchy determines both the micro-level processes of GPT-induced transformation at the regional level as well as the regional outcomes of this process and contribution of the region to the aggregate transformation dynamics at the macro-level of the economy.

Again and again, investigations in individual articles pointed that the extent of and opportunities for GPT-induced transformation, and its impact on regional labour market dynamics, are conditioned by the position of a region in the regional hierarchy.

On the one hand, it is demonstrated that large regions at the top of the regional hierarchy are characterised by larger scale of the GPT-related activities expansion, not only in absolute terms (total employment in GPT-related occupations) but also with regards to the degree of emerging complementarities to the GPT – reflected in, in the case of this dissertation, in the IT intensity of regional employment mixes (Articles II and III). Regions at lower levels of the regional hierarchy, even though demonstrating positive dynamics in the GPT-related activities over time, lagged far behind.

On the other hand, the importance of the position of a region in the regional hierarchy goes beyond the mere scale effects. It also defines the opportunities for GPT-induced transformation of regional labour markets. The ability of regional labour markets to adjust to the changing patterns of demand for labour is shown to be limited for regions at lower levels of the regional hierarchy (Article I). The extent of labour markets in such regions restricts the opportunities for hiring specialised labour possessing the GPT-related human capital (Article III). Also, the concentration of GPT-related activities to regions at the top of the regional hierarchy limits the ability of more peripheral regions to provide learning opportunities of the comparable qulity (Article IV). While the differences between regions at different levels of the regional hierarchy are, to some extent, mitigated over time – as the GPT is increasingly diffused to a wider range of industries and regions – they still remain quite persistent.

The significance of regional hierarchies in understanding the dynamics of technology-induced transformation has important implications for further research and regional policies. Traditionally, economic geography focused on understanding the uneven patterns of regional development. More recently this paradigm has all but disappeared due to 'an academic obsession [...] with the study of 'successful' regions' (Martin 2015, p. 21). It appears, however, that these successful regions are characterised by completely different dynamics than regions lagging behind (often found at lower levels of the regional hierarchy). Therefore, one cannot simply copy the same instruments which worked in the context of successful regions and expect that they will work in the same way in more problematic contexts. One-size-fits-all approach is likely to fail. In that respect, research in economic geography should revive its interest in the properties of regions lagging behind in

economic transformation to provide a more informed input to the design of regional development policies.

All in all, the dissertation contributes to the literature in economics, economic history, and economic geography by demonstrating that the dynamics of technological change as well as its implications for functioning of the economy should be approached through the prism of the technology diffusion process which *unfolds in time and space*. It suggests that investigations of technology-induced transformation dynamics should acknowledge four intertwined dimensions: *what* kind of technology is being diffused, *how* it is diffused, *when* it is diffused as well as *where* it is diffused at different points in time.

With regards to labour market dynamics, the dissertation suggests that the role of the labour market in the technology-induced transformation is far from the smoothly operating mechanism of adjustment to economy-wide economic shocks, as it is often assumed in mainstream approaches to economic theorising. By demonstrating that the technology-induced labour market transformation is not only characterised by emerging non-linearities and divergent trends across different regions, but also has repercussions for the technology diffusion process itself, the dissertation provides a reference point for studies of labour market evolution beyond the standard equilibrium approach.

References

- Abraham, K. G., & Katz, L. F. (1986). 'Cyclical Unemployment: Sectoral Shifts or Aggregate Disturbances?'. *Journal of Political Economy*, Vol. 94(3), pp. 507-522.
- Adermon, A., & Gustavsson, M. (2015). 'Job Polarization and Task-Biased Technological Change: Evidence from Sweden, 1975–2005'. *The Scandinavian Journal of Economics*, Vol. 117(3), pp. 878-917.
- Adorno, T. W. (1982). *Against epistemology: a meta-critique*. Oxford: Blackwell.
- Aghion, P., David, P. A., & Foray, D. (2006). Linking policy research and practice in 'STIG Systems': Many obstacles, but some ways forward. SIEPR Discussion Paper No. 06-09, Stanford Institute for Economic Policy Research, Stanford.
- Aghion, P., & Howitt, P. (1992). 'A Model of Growth Through Creative Destruction'. *Econometrica*, Vol. 60(2), pp. 323-351.
- Aldrich, H. E. (1999). Organizations Evolving. Thousand Oaks, CA: Sage.
- Ali-Yrkkö, J., Mattila, J., Pajarinen, M., & Rouvinen, P. (2015). A comparison of the Finnish and the Swedish ICT sector. In E. Giertz, A. Rickne, & P. Rouvinen (Eds.), *Small and Beautiful. The ICT success of Finland and Sweden*. Stockholm: VINNOVA.
- Amin, A., & Thrift, N. (2000). 'What Kind of Economic Theory for what Kind of Economic Geography?'. *Antipode*, Vol. 32(1), pp. 4-9.
- Anokhin, S., Wincent, J., & Ylinenpää, H. (2016). 'Technological Expansions, Catching-Up Innovations and Technological Shifts at the Regional Level: Conceptual Considerations and Empirical Illustration'. *Regional Studies*, Vol. 50(8), pp. 1433-1448.

- Antelius, J., & Lundberg, L. (2003). 'Competition, Market Structure and Job Turnover'. *Journal of Industry, Competition and Trade*, Vol. 3(3), pp. 211-226.
- Archer, M. S., Bhaskar, R., Collier, A., Lawson, T., & Norrie, A. (1998). *Critical Realism: Essential Readings*. London: Routledge.
- Arikan, A. T., & Knoben, J. (2014). 'Sources of Inter-Firm Heterogeneity in Accessing Knowledge-Creation Benefits Within Technology Clusters'. *Industry and Innovation*, Vol. 21(6), pp. 476-493.
- Arpaia, A., Kiss, A., Palvolgyi, B., & Turrini, A. (2015). Labour Mobility and Labour Market Adjustment in the EU. IZA Policy Paper No. 106. Bonn: Institute for the Study of Labour.
- Arrow, K. J. (1962). Economic welfare and the allocation of resources for inventions. In R. Nelson (Ed.), *The Rate and Direction of Inventive Activity: Economic and Social Factors*. Princeton, NJ: Princeton University Press, pp. 609-625.
- Asheim, B. T., & Coenen, L. (2005). 'Knowledge bases and regional innovation systems: Comparing Nordic clusters'. *Research Policy*, Vol. 34(8), pp. 1173-1190.
- Audretsch, D. B., & Feldman, M. P. (1996). 'R&D Spillovers and the Geography of Innovation and Production'. *The American Economic Review*, Vol. 86(3), pp. 630-640.
- Autor, D. H., Katz, L. F., & Kearney, M. S. (2006). 'The Polarization of the U.S. Labor Market'. *American Economic Review*, Vol. 96(2), pp. 189-194.
- Autor, D. H., Katz, L. F., & Krueger, A. B. (1998). 'Computing Inequality: Have Computers Changed the Labor Market?'. *The Quarterly Journal of Economics*, Vol. 113(4), pp. 1169-1213.
- Autor, D. H., Levy, F., & Murnane, R. J. (2003). 'The Skill Content of Recent Technological Change: An Empirical Exploration'. *The Quarterly Journal of Economics*, Vol. 118(4), pp. 1279-1333.
- Bachmann, R., & Burda, M. C. (2010). 'Sectoral Transformation, Turbulence and Labor Market Dynamics in Germany'. *German Economic Review*, Vol. 11(1), pp. 37-59.

- Baldwin, R. E. (1999). 'Agglomeration and endogenous capital'. *European Economic Review*, Vol. 43(2), pp. 253-280.
- Baliamoune-Lutz, M. (2003). 'An analysis of the determinants and effects of ICT diffusion in developing countries'. *Information Technology for Development*, Vol. 10(3), pp. 151-169.
- Baller, S., Dutta, S., & Lanvin, B. (2016). *The Global Information Technology Report 2016. Innovating in the Digital Economy.* Geneva: World Economic Forum.
- Baptista, R. (2003). Productivity and ddensity of regional clusters In J. Brocker, D. Dohse, & R. Soltwedel (Eds.), *Innovation Clusters and Interregional Competition*. Berlin: Springer, pp. 163-181.
- Baron, J. N. (2004). 'Employing identities in organizational ecology'. *Industrial and Corporate Change*, Vol. 13(1), pp. 3-32.
- Barro, R. J., & Sala-i-Martin, X. (2004). *Economic Growth* (2nd edition edition). Cambridge, MA: The MIT Press.
- Basu, S., & Fernald, J. (2007). 'Information and Communications Technology as a General-Purpose Technology: Evidence from US Industry Data'. *German Economic Review*, Vol. 8(2), pp. 146-173.
- Bathelt, H., Malmberg, A., & Maskell, P. (2004). 'Clusters and knowledge: local buzz, global pipelines and the process of knowledge creation'. *Progress in Human Geography*, Vol. 28(1), pp. 31-56.
- Bauer, T. K., & Bender, S. (2004). 'Technological change, organizational change, and job turnover'. *Labour Economics*, Vol. 11(3), pp. 265-291.
- Baumol, W. J. (2004). *The Free-Market Innovation Machine: Analyzing the Growth Miracle of Capitalism*. Princeton, NJ: Princeton University Press.
- Bayo-Moriones, A., & Lera-López, F. (2007). 'A firm-level analysis of determinants of ICT adoption in Spain'. *Technovation*, Vol. 27(6–7), pp. 352-366.
- Becker, G. (1964). Human Capital. Chicago, IL: University of Chicago Press.

- Belloc, M., & Guerrieri, P. (2015). 'Impact of ICT diffusion and adoption on sectoral industrial performance: evidence from a panel of European countries'. *Economia Politica*, Vol. 32(1), pp. 67-84.
- Bengtsson, M. (2012). Transformation of Labour Market Policies in the Nordic Countries: Towards a regime shift in Sweden and Denmark? Paper presented at the Paper presented at the ILERA World Congress 2012, 2-5 July 2012, Philadelphia, USA.
- Benhabib, J., & Spiegel, M. M. (2005). Chapter 13 Human Capital and Technology Diffusion. In A. Philippe & N. D. Steven (Eds.), *Handbook of Economic Growth (Vol. Volume 1, Part A)*: Elsevier, pp. 935-966.
- Bennett, R. J. (1985). Quantification and relevance. In R. J. Johnston (Ed.), *The Future of Geography*. London: Methuen, pp. 211-224.
- Berger, T., & Frey, C. B. (2015). 'Industrial Renewal in the 21st Century: Evidence from US Cities'. *Regional Studies*, pp. 1-10.
- Berger, T., & Frey, C. B. (2016). 'Did the Computer Revolution shift the fortunes of U.S. cities? Technology shocks and the geography of new jobs'. *Regional Science and Urban Economics*, Vol. 57, pp. 38-45.
- Bertschek, I., & Kaiser, U. (2004). 'Productivity Effects of Organizational Change: Microeconometric Evidence'. *Management Science*, Vol. 50(3), pp. 394-404.
- Bhaskar, R. (1975). A realist theory of science. Leeds: Leeds Books.
- Bhaskar, R. (1979). *The possibility of naturalism: a philosophical critique of the contemporary human sciences*. Brighton: Harvester.
- Bienkowska, D. (2007). Arbetskraftens rörlighet och klusterdynamik. En studie av IT- och telekomklustren i Kista och Mjärdevi. Geografiska Regionstudier 75. Uppsala: Uppsala Universitet.
- Böckerman, P., Laaksonen, S., & Vainiomäki, J. (2016). Are Jobs More Polarized in ICT Firms? IZA Discussion Paper No. 9851. Bonn: Institute for the Study of Labour.
- Boldrin, M., & Canova, F. (2003). Regional policies and EU enlargement. In B. Funck & L. Pizati (Eds.), *European Integration, Regional Policy* and Growth. Washington, DC: World Bank.

- Boldrin, M., Canova, F., Pischke, J.-S., & Puga, D. (2001). 'Inequality and Convergence in Europe's Regions: Reconsidering European Regional Policies'. *Economic Policy*, Vol. 16(32), pp. 207-253.
- Boschma, R., Eriksson, R., & Lindgren, U. (2009). 'How does labour mobility affect the performance of plants? The importance of relatedness and geographical proximity'. *Journal of Economic Geography*, Vol. 9(2), pp. 169-190.
- Boschma, R., & Frenken, K. (2011a). 'The emerging empirics of evolutionary economic geography'. *Journal of Economic Geography*, Vol. 11(2), pp. 295-307.
- Boschma, R., & Frenken, K. (2011b). Technological relatedness and regional branching. In H. Bathelt, M. P. Feldman, & K. D.F (Eds.), *Beyond Territory. Dynamic Geographies of Knowledge Creation, Diffusion and Innovation*. Oxford: Routledge, pp. 64-81.
- Boschma, R., & Martin, R. (2007). 'Editorial: Constructing an evolutionary economic geography'. *Journal of Economic Geography*, Vol. 7(5), pp. 537-548.
- Boschma, R. A., & Frenken, K. (2006). 'Why is economic geography not an evolutionary science? Towards an evolutionary economic geography'. *Journal of Economic Geography*, Vol. 6(3), pp. 273-302.
- Boschma, R. A., & Lambooy, J. G. (1999). 'Evolutionary economics and economic geography'. *Journal of Evolutionary Economics*, Vol. 9(4), pp. 411-429.
- Bosma, N., & Sternberg, R. (2014). 'Entrepreneurship as an Urban Event? Empirical Evidence from European Cities'. *Regional Studies*, Vol. 48(6), pp. 1016-1033.
- Breschi, S., Lenzi, C., Lissoni, F., & Vezzulli, A. (2010). The geography of knowledge spillovers: The role of inventors' mobility across firms and in space. In R. Boschma & R. Martin (Eds.), *The Handbook of Evolutionary Economic Geography*. Cheltenham and Northampton, MA: Edward Elgar, pp. 353-369.

- Breschi, S., & Lissoni, F. (2009). 'Mobility of skilled workers and coinvention networks: an anatomy of localized knowledge flows'. *Journal of Economic Geography*, Vol. 9(4), pp. 439-468.
- Bresnahan, T. (2010). Chapter 18 General Purpose Technologies. In B. Hall & N. Rosenberg (Eds.), *Handbook of the Economics of Innovation (Vol. 2)*. Oxford: Elsevier, pp. 761-791.
- Bresnahan, T., Greenstein, S., Brownstone, D., & Flamm, K. (1996). 'Technical Progress and Co-Invention in Computing and in the Uses of Computers'. *Brookings Papers on Economic Activity*. *Microeconomics*, Vol. 1996, pp. 1-83.
- Bresnahan, T. F., Brynjolfsson, E., & Hitt, L. M. (2002). 'Information Technology, Workplace Organization, and the Demand for Skilled Labor: Firm-Level Evidence'. *The Quarterly Journal of Economics*, Vol. 117(1), pp. 339-376.
- Bresnahan, T. F., & Trajtenberg, M. (1995). 'General purpose technologies 'Engines of growth'?'. *Journal of Econometrics*, Vol. 65(1), pp. 83-108.
- Brynjolfsson, E., Hitt, L., & Yang, S. (2002). 'Intangible Assets: How the Interaction of Computers and Organizational Structure Affects Stock Market Valuations'. *Brookings Papers on Economic Activity*, Vol. 33(1), pp. 137-198.
- Brynjolfsson, E., & Hitt, L. M. (2000). 'Beyond Computation: Information Technology, Organizational Transformation and Business Performance'. *Journal of Economic Perspectives*, Vol. 14(4), pp. 23-48.
- Brynjolfsson, E., & Hitt, L. M. (2003). 'Computing Productivity: Firm-Level Evidence'. *Review of Economics and Statistics*, Vol. 85(4), pp. 793-808.
- Brynjolfsson, E., & McAfee, A. (2014). *The Second Machine Age: Work, Progress, and Prosperity in a Time of Brilliant Technologies.* New York: Norton & Co.
- Cantner, U., & Vannuccini, S. (2012). *A New View of General Purpose Technologies*. Jena Economic Research Papers 2012-054. Jena: Friedrich-Schiller-University Jena.

- Capasso, S., Carillo, M. R., & De Siano, R. (2012). 'MIGRATION FLOWS, STRUCTURAL CHANGE AND GROWTH CONVERGENCE: A PANEL DATA ANALYSIS OF THE ITALIAN REGIONS'. *The Manchester School*, Vol. 80(4), pp. 468-498.
- Cardona, M., Kretschmer, T., & Strobel, T. (2013). 'ICT and productivity: conclusions from the empirical literature'. *Information Economics and Policy*, Vol. 25(3), pp. 109-125.
- Carlaw, K. I., & Lipsey, R. G. (2006). 'Gpt-Driven, Endogenous Growth*'. *The Economic Journal*, Vol. 116(508), pp. 155-174.
- Carlaw, K. I., & Lipsey, R. G. (2011). 'Sustained endogenous growth driven by structured and evolving general purpose technologies'. *Journal of Evolutionary Economics*, Vol. 21(4), pp. 563-593.
- Carlsson, B., & Eliasson, G. (2003). 'Industrial Dynamics and Endogenous Growth'. *Industry and Innovation*, Vol. 10(4), pp. 435-455.
- Carlsson, B., Jacobsson, S., Holmén, M., & Rickne, A. (2002). 'Innovation systems: analytical and methodological issues'. *Research Policy*, Vol. 31(2), pp. 233-245.
- Caroleo, F. E., & Pastore, F. (2010). Structural Change and Labour Reallocation Across Regions: A Review of the Literature. In F. E. Caroleo & F. Pastore (Eds.), *The Labour Market Impact of the EU Enlargement*. Heidelberg: Physica-Verlag HD, pp. 17-47.
- Caroli, E., Greenan, N., & Guellec, D. (2001). 'Organizational Change and Skill Accumulation'. *Industrial and Corporate Change*, Vol. 10(2), pp. 481-506.
- Caroli, E., & Van Reenen, J. (2001). 'Skill-Biased Organizational Change? Evidence from A Panel of British and French Establishments'. *The Quarterly Journal of Economics*, Vol. 116(4), pp. 1449-1492.
- Castellacci, F. (2007). 'EVOLUTIONARY AND NEW GROWTH THEORIES. ARE THEY CONVERGING?'. Journal of Economic Surveys, Vol. 21(3), pp. 585-627.
- Castellacci, F. (2008). 'Technological paradigms, regimes and trajectories: Manufacturing and service industries in a new taxonomy of sectoral patterns of innovation'. *Research Policy*, Vol. 37(6–7), pp. 978-994.

- Chakravorti, B., Tunnard, C., & Chaturvedi, R. S. (2014). *Digital Planet: Readying for the Rise of the e-Consumer*. A report on the state and trajectory of global digital revolution. Medford, MA: The Fletcher School, Tufts University.
- Chenery, H., Robinson, S., & Syrquin, M. (1986). *Industrialization and Growth: a Comparative Study*. Oxford, UK: Oxford University Press.
- Cimoli, M., Pereima, J. B., & Porcile, G. (2016). 'Introduction to the special issue SCED: Complexity and economic development'. *Structural Change and Economic Dynamics*, Vol. 38, pp. 1-2.
- Clarke, G. R. G., Qiang, C. Z., & Xu, L. C. (2015). 'The Internet as a generalpurpose technology: Firm-level evidence from around the world'. *Economics Letters*, Vol. 135, pp. 24-27.
- Coe, N. M., & Bunnell, T. G. (2003). "Spatializing' knowledge communities: towards a conceptualization of transnational innovation networks'. *Global Networks*, Vol. 3(4), pp. 437-456.
- Coff, R., & Kryscynski, D. (2011). 'Invited Editorial: Drilling for Micro-Foundations of Human Capital–Based Competitive Advantages'. *Journal of Management*, Vol. 37(5), pp. 1429-1443.
- Cohen, W. M., & Levinthal, D. A. (1990). 'Absorptive Capacity: A New Perspective on Learning and Innovation'. *Administrative Science Quarterly*, Vol. 35(1), pp. 128-152.
- Consoli, D., & Rentocchini, F. (2015). 'A taxonomy of multi-industry labour force skills'. *Research Policy*, Vol. 44(5), pp. 1116-1132.
- Cooke, P., Uranga, M. G., & Etxebarria, G. (1998). 'Regional Systems of Innovation: An Evolutionary Perspective'. *Environment and Planning A*, Vol. 30(9), pp. 1563-1584.
- Coulombe, S., & Tremblay, J.-F. (2009). 'Migration and Skills Disparities across the Canadian Provinces'. *Regional Studies*, Vol. 43(1), pp. 5-18.
- Criscuolo, C., & Waldron, K. (2003). 'E-commerce and firm productivity'. *Economic Trends*, Vol. 600, pp. 52-57.

- Dahlman, C. (2007). Technology, Globalization, and International Competitiveness: Challenges for Developing Countries. In UNDESA (Ed.), Industrial Development for the 21st Century: Sustainable Development Perspectives. New York, NY: United Nations.
- Dahmén, E. (1988). "Development blocks' in industrial economics'. Scandinavian Economic History Review, Vol. 36(1), pp. 3-14.
- David, P. A. (1985). 'Clio and the Economics of QWERTY'. *The American Economic Review,* Vol. 75(2), pp. 332-337.
- David, P. A. (1990). 'The Dynamo and the Computer: An Historical Perspective on the Modern Productivity Paradox'. *The American Economic Review*, Vol. 80(2), pp. 355-361.
- David, P. A., & Wright, G. (1999). General Purpose Technologies and Surges in Productivity: Historical Reflections on the Future of the ICT Revolution. Discussion Papers in Economic and Social History Number 31. Oxford: University of Oxford.
- Davis, S., Haltiwanger, J., & Schuh, S. (1996). *Job Creation and Job Destruction*. Cambridge, MA: The MIT Press.
- de Groot, B. (2006). *Essays on Economic Cycles*. Erasmus Research Institute of Management Ph.D. Series Research in Management 91. Rotterdam: Erasmus University Rotterdam.
- Diamond, A. M. (2010). Schumpeterian Labor Economics: The Labor Pains (and Labor Gains) from Creative Destruction Paper presented at the 2010 International Schumpeter Society Conference, Aalborg, Denmark.
- Dietbolt, C. (2005). *Long Cycles Revisited. An Essay in Econometric History*. Working Papers No. 05-05, Association Française de Cliométrie (AFC).
- Domeij, D., & Floden, M. (2010). 'Inequality trends in Sweden 1978–2004'. *Review of Economic Dynamics*, Vol. 13(1), pp. 179-208.
- Dopfer, K. (2007). The Pillars of Schumpeter's Economics: Micro, Meso, Macro. In H. Hanusch & A. Pyka (Eds.), *Elgar Companion to Neo-Schumpeterian Economics*. Cheltenham and Northampton, MA: Edward Elgar, pp. 65-77.

- Dopfer, K., Foster, J., & Potts, J. (2004). 'Micro-meso-macro'. *Journal of Evolutionary Economics*, Vol. 14(3), pp. 263-279.
- Dosi, G. (1982). 'Technological paradigms and technological trajectories: A suggested interpretation of the determinants and directions of technical change'. *Research Policy*, Vol. 11(3), pp. 147-162.
- Dosi, G., & Nelson, R. R. (2010). Technical change and industrial dynamics as evolutionary processes. In B. Hall & N. Rosenberg (Eds.), *Handbook of the Economics of Innovation (Vol. 1)*. Oxford: Elsevier.
- Dosi, G., & Winter, S. G. (2000). Interpreting Economic Change: Evolution, Structures and Games. LEM Working Paper Series No. 2000/08. Pisa: Laboratory of Economics and Management, Sant'Anna School of Advanced Studies.
- Draca, M., Sadun, R., & Van Reenen, J. (2007). Productivity and ICTs: A review of the evidence. In R. Mansell, D. Avgerou, D. Quah, & R. Silverstone (Eds.), Oxford Handbook of Information and Communication Technologies. Oxford: Oxford University Press, pp. 100-147.
- Edin, P.-A., & Holmlund, B. (1995). The Swedish wage structure: The rise and fall of solidarity wage policy? In R. Freeman & L. Katz (Eds.), *Differences and changes in wage structures*. Chicago: University of Chicago Press, pp. 307-344.
- Elliott, R. J., & Lindley, J. (2006). 'Skill Specificity and Labour Mobility: Occupational and Sectoral Dimensions'. *The Manchester School*, Vol. 74(3), pp. 389-413.
- Enflo, K., Henning, M., & Schön, L. (2014). Swedish regional GDP 1855-2000: Estimations and general trends in the Swedish regional system. In S. Wolcott & C. Hanes (Eds.), *Research in Economic History (Vol. 30)*: Emerald Group Publishing Limited, pp. 47-89.
- Enos, J. L. (1962). Invention and innovation in the petroleum refining industry. In NBER (Ed.), *The Rate and Direction of Inventive Activity*. Princeton, NJ: Princeton University Press.
- Epifani, P., & Gancia, G. A. (2005). 'Trade, migration and regional unemployment'. *Regional Science and Urban Economics*, Vol. 35(6), pp. 625-644.

- Essletzbichler, J., & Rigby, D. L. (2007). 'Exploring evolutionary economic geographies'. *Journal of Economic Geography*, Vol. 7(5), pp. 549-571.
- EU. (2016). Digital Economy and Society Index 2016. Country profile: Sweden. Accessed at <u>http://ec.europa.eu/newsroom/dae/document.cfm?action=display&d</u> <u>oc_id=14136</u> on October, 26, 2016.
- Fabiani, S., Schivardi, F., & Trento, S. (2005). 'ICT adoption in Italian manufacturing: firm-level evidence'. *Industrial and Corporate Change*, Vol. 14(2), pp. 225-249.
- Feldman, M. P., & Audretsch, D. B. (1999). 'Innovation in cities: Sciencebased diversity, specialization and localized competition'. *European Economic Review*, Vol. 43(2), pp. 409-429.
- Field, A. J. (2008). Does Economic History Need GPTs. SSRB eLibrary.
- Fine, B. (2000). 'Critical survey. Endogenous growth theory: a critical assessment'. *Cambridge Journal of Economics*, Vol. 24(2), pp. 245-265.
- Florida, R. (2014). *The rise of the creative class : revisited*. New York: Basic Books.
- Forman, C., Goldfarb, A., & Greenstein, S. (2005). 'Geographic location and the diffusion of Internet technology'. *Electronic Commerce Research and Applications*, Vol. 4(1), pp. 1-13.
- Forrester, J. W. (1981). 'Innovation and economic change'. *Futures*, Vol. 13(4), pp. 323-331.
- Foster, J. (2011). Evolutionary Macroeconomics: A Research Agenda. In A. Pyka & d. G. M. Derengowski Fonseca (Eds.), *Catching Up, Spillovers and Innovation Networks in a Schumpeterian Perspective*. Berlin, Heidelberg: Springer Berlin Heidelberg, pp. 7-30.
- Francis, J. (2009). 'Agglomeration, job flows and unemployment'. *The Annals of Regional Science,* Vol. 43(1), pp. 181-198.
- Fratesi, U. (2014). 'Editorial: The Mobility of High-Skilled Workers Causes and Consequences'. *Regional Studies*, Vol. 48(10), pp. 1587-1591.

- Fratesi, U., & Percoco, M. (2014). 'Selective Migration, Regional Growth and Convergence: Evidence from Italy'. *Regional Studies*, Vol. 48(10), pp. 1650-1668.
- Fratesi, U., & Riggi, M. R. (2007). 'DOES MIGRATION REDUCE REGIONAL DISPARITIES? THE ROLE OF SKILL-SELECTIVE FLOWS'. *Review of Urban & Regional Development Studies*, Vol. 19(1), pp. 78-102.
- Fratesi, U., & Senn, L. (2009). Regional growth, connections and economic modelling: an introduction. In U. Fratesi & L. Senn (Eds.), Growth and innovation of competitive regions: the role of internal and external connections. Berlin: Springer, pp. 3-28.
- Fredriksson, P. (1999). 'The dynamics of regional labor markets and active labor market policy: Swedish evidence'. Oxford Economic Papers, Vol. 51(4), pp. 623-648.
- Freeman, C. (1982). Innovation and Long Cycles of Economic Development. Paper presented at the Paper presented at International Seminar on Innovation and Development at the Industrial Sector, 25-27 August 1982, Campinas, Brazil.
- Freeman, C. (1995). 'The 'National System of Innovation' in historical perspective'. *Cambridge Journal of Economics*, Vol. 19(1), pp. 5-24.
- Freeman, C. (2007). A Schumpeterian renaissance? In H. Hanusch & A. Pyka (Eds.), *Elgar Companion to Neo-Schumpeterian Economics*. Cheltenham and Northampton, MA: Edward Elgar, pp. 130-141.
- Freeman, C., Clark, J., & Soete, L. (1982). Unemployment and Technical Innovation: a Study of Long Waves and Economic Development. Westport, CT: Greenwood Press.
- Freeman, C., & Louçã, F. (2001). *As time goes by. From the industrial revolutions to the information revolution.* Oxford: Oxford University Press.
- Freeman, C., & Perez, C. (1988). Structural crises of adjustment: business cycles and investment behaviour. In G. Dosi, C. Freeman, R. R. Nelson, L. Soete, & J. Silverberg (Eds.), *Technical Change and Economic Theory*. London: Pinter, pp. 38-66.

- Freeman, C., & Soete, L. (1997). *The Economics of Industrial Innovation* (Third edition). London and Washington: Pinter.
- Frenken, K., & Boschma, R. A. (2007). 'A theoretical framework for evolutionary economic geography: industrial dynamics and urban growth as a branching process'. *Journal of Economic Geography*, Vol. 7(5), pp. 635-649.
- Fujita, M., Krugman, P., & Venables, A. (1999). The Spatial Economy: Cities, Regions and International Trade. Cambridge, MA: The MIT Press.
- Funck, B., & Pizzati, L. E. (2002). Labour, Employment and Social Policy in the EU Enlargement Process. Changing Perspectives and Policy Options. Washington, DC: World Bank.
- Funck, B., & Pizzati, L. E. (2003). European Integration, Regional Policy and Growth. Washington, DC: World Bank.
- Galliano, D., Roux, P., & Soulié, N. (2011). 'ICT Intensity of Use and the Geography of Firms'. *Environment and Planning A*, Vol. 43(1), pp. 67-86.
- Gärtner, S. (2014). 'New Macroeconomic Evidence on Internal Migration in Sweden, 1967–2003'. *Regional Studies*, pp. 1-17.
- Gathmann, C., & Schönberg, U. (2010). 'How General Is Human Capital? A Task-Based Approach'. *Journal of Labor Economics*, Vol. 28(1), pp. 1-49.
- Gertler, M. S. (2003). 'Tacit knowledge and the economic geography of context, or The undefinable tacitness of being (there)'. *Journal of Economic Geography*, Vol. 3(1), pp. 75-99.
- Gertler, M. S. (2010). 'Rules of the Game: The Place of Institutions in Regional Economic Change'. *Regional Studies*, Vol. 44(1), pp. 1-15.
- Giannetti, M. (2003). 'On the mechanics of migration decisions: skill complementarities and endogenous price differentials'. *Journal of Development Economics*, Vol. 71(2), pp. 329-349.
- Giertz, E. (2015). Understandin the Swedish saga through an historical exposé. In E. Giertz, A. Rickne, & P. Rouvinen (Eds.), *Small and*

beautiful. The ICT success of Finland and Sweden. Stockholm: VINNOVA.

- Giertz, E., Rickne, A., & Rouvinen, P. (2015). *Small and beautiful. The ICT success of Finland & Sweden*. Stockholm: VINNOVA.
- Gittelman, M. (2007). 'Does Geography Matter for Science-Based Firms? Epistemic Communities and the Geography of Research and Patenting in Biotechnology'. *Organization Science*, Vol. 18(4), pp. 724-741.
- Goldstein, J. S. (1985). 'Kondratieff Waves as War Cycles'. *International Studies Quarterly*, Vol. 29(4), pp. 411-444.
- Goos, M., & Manning, A. (2007). 'Lousy and Lovely Jobs: The Rising Polarization of Work in Britain'. *Review of Economics and Statistics*, Vol. 89(1), pp. 118-133.
- Goos, M., Manning, A., & Salomons, A. (2014). 'Explaining Job Polarization: Routine-Biased Technological Change and Offshoring'. *American Economic Review*, Vol. 104(8), pp. 2509-2526.
- Göransson, B., & Söderberg, J. (2005). 'Long waves and information technologies—on the transition towards the information society'. *Technovation*, Vol. 25(3), pp. 203-211.
- Gordon, D. M. (1980). Stages of Accumulation and Long Economic Cycles. In T. K. Hopkins & I. M. Wallerstein (Eds.), *Processes of the World System*. Beverly Hills: Sage Publications.
- Gordon, I. R. (2015). 'Ambition, Human Capital Acquisition and the Metropolitan Escalator'. *Regional Studies*, Vol. 49(6), pp. 1042-1055.
- Graham, A. K., & Senge, P. M. (1980). 'A long-wave hypothesis of innovation'. *Technological Forecasting and Social Change*, Vol. 17(4), pp. 283-311.
- Grant, R. M. (1996). 'Toward a knowledge-based theory of the firm'. *Strategic Management Journal*, Vol. 17(S2), pp. 109-122.

- Greenaway, D., Haynes, M., & Milner, C. (2002). 'Adjustment, employment characteristics and intra-industry trade'. *Weltwirtschaftliches Archiv*, Vol. 138(2), pp. 254-276.
- Greenaway, D., Upward, R., & Wright, P. (2000). 'Sectoral transformation and labour-market flows'. *Oxford Review of Economic Policy*, Vol. 16(3), pp. 57-75.
- Griliches, Z. (1957). 'Hybrid Corn: An Exploration in the Economics of Technological Change'. *Econometrica*, Vol. 25(4), pp. 501-522.
- Griliches, Z. (1969). 'Capital-Skill Complementarity'. *The Review of Economics and Statistics*, Vol. 51(4), pp. 465-468.
- Grillitsch, M., Schubert, T., & Srholec, M. (2016). Knowledge diversity and firm growth: Searching for a missing link Papers in Innovation Studies No. 2016/13. Lund: Centre for Innovation, Research and Competence in the Learning Economy (CIRCLE), Lund University.
- Hägerstrand, T. (1967). *Innovation diffusion as a spatial process*. Chicago: Chicago University Press.
- Haller, S. A., & Siedschlag, I. (2011). 'Determinants of ICT adoption: evidence from firm-level data'. *Applied Economics*, Vol. 43(26), pp. 3775-3788.
- Hammersley, M. (1992). Deconstructing the qualitative-quantitative divide. In J. Brannen (Ed.), *Mixing methods: qualitative and quantitative research*. Aldershot: Avebury, pp. 39-55.
- Hansen, A. H. (1964). *Business Cycles and National Income* (Expanded edition). New York: Norton.
- Hanusch, H., & Pyka, A. (2007). "MANIFESTO" FOR COMPREHENSIVE NEO-SCHUMPETERIAN ECONOMICS'. *History of Economic Ideas*, Vol. 15(1), pp. 23-41.
- Harada, T. (2010). 'The division of labor in innovation between general purpose technology and special purpose technology'. *Journal of Evolutionary Economics*, Vol. 20(5), pp. 741-764.
- Hatch, N. W., & Dyer, J. H. (2004). 'Human capital and learning as a source of sustainable competitive advantage'. *Strategic Management Journal*, Vol. 25(12), pp. 1155-1178.

- Hauser, D. (2014). *Technology Shocks, Labour Mobility and Aggregate Fluctuations*. Working Paper/Document de travail 2014-4. Ottawa, Canada Bank of Canada.
- Haustein, H.-D., & Neuwirth, E. (1982). 'Long waves in world industrial production, energy consumption, innovations, inventions, and patents and their identification by spectral analysis'. *Technological Forecasting and Social Change*, Vol. 22(1), pp. 53-89.
- Helpman, E. (1998). *General Purpose Technologies and Economic Growth* (E. Helpman Ed.). Cambridge, MA: The MIT Press.
- Helpman, E., & Trajtenberg, M. (1998). The diffusion of general purpose technologies. In E. Helpman (Ed.), *General Purpose Technologies* and Economic Growth. Cambridge, MA: The MIT Press, pp. 85-119.
- Henning, M. (2009). Industrial Dynamics and Regional Structural Change: Geographical Perspectives on Economic Evolution. MEDDELANDEN FRÅN LUNDS UNIVERSITETS GEOGRAFISKA INSTITUTION. AVHANDLINGAR 181. Lund: Lund University.
- Henning, M., Enflo, K., & Andersson, F. N. G. (2011). 'Trends and cycles in regional economic growth: How spatial differences shaped the Swedish growth experience from 1860–2009'. *Explorations in Economic History*, Vol. 48(4), pp. 538-555.
- Henning, M., Lundquist, K.-J., & Olander, L.-O. (2016). Regional analysis and the process of economic development: changes in growth, employment and income. In J. Ljungberg (Ed.), *Structural Analysis* and the Process of Economic Development. Oxon and New York: Routledge, pp. 149-173.
- Hoisl, K. (2007). 'Tracing mobile inventors—The causality between inventor mobility and inventor productivity'. *Research Policy*, Vol. 36(5), pp. 619-636.
- Hoisl, K. (2009). 'Does mobility increase the productivity of inventors?'. *The Journal of Technology Transfer*, Vol. 34(2), pp. 212-225.
- Hollenstein, H. (2004). 'Determinants of the adoption of Information and Communication Technologies (ICT): An empirical analysis based on

firm-level data for the Swiss business sector'. *Structural Change and Economic Dynamics*, Vol. 15(3), pp. 315-342.

- Hung, S.-C., & Chu, Y.-Y. (2006). 'Stimulating new industries from emerging technologies: challenges for the public sector'. *Technovation*, Vol. 26(1), pp. 104-110.
- ITU. (2016). *ICT Development Index 2015*. Accessed at <u>http://www.itu.int/net4/ITU-D/idi/2015/</u> on October, 26, 2016.
- Jaffe, A. B., Trajtenberg, M., & Henderson, R. (1993). 'Geographic Localization of Knowledge Spillovers as Evidenced by Patent Citations'. *The Quarterly Journal of Economics*, Vol. 108(3), pp. 577-598.
- Jensen, M. B., Johnson, B., Lorenz, E., & Lundvall, B. Å. (2007). 'Forms of knowledge and modes of innovation'. *Research Policy*, Vol. 36(5), pp. 680-693.
- Jovanovic, B. (1979). 'Job Matching and the Theory of Turnover'. *Journal of Political Economy*, Vol. 87(5), pp. 972-990.
- Jovanovic, B., & Rousseau, P. L. (2005). Chapter 18 General Purpose Technologies. In A. Philippe & N. D. Steven (Eds.), *Handbook of Economic Growth (Vol. Volume 1, Part B)*: Elsevier, pp. 1181-1224.
- Kambourov, G., & Manovskii, I. (2009). 'Occupational Mobility and Wage Inequality'. *The Review of Economic Studies*, Vol. 76(2), pp. 731-759.
- Kanbur, R., & Rapoport, H. (2005). 'Migration selectivity and the evolution of spatial inequality'. *Journal of Economic Geography*, Vol. 5(1), pp. 43-57.
- Kleinknecht, A. (1987). Innovation Patterns in Crisis and Prosperity: Schumpeter's Long Cycle Reconsidered. London: Macmillan.
- Kogler, D. F. (2015). 'Editorial: Evolutionary Economic Geography Theoretical and Empirical Progress'. *Regional Studies*, Vol. 49(5), pp. 705-711.
- Kogut, B., & Zander, U. (1996). 'What Firms Do? Coordination, Identity, and Learning'. *Organization Science*, Vol. 7(5), pp. 502-518.

- Kondratiev, N. D. (1925). 'Bol'šie cikly konjunktury [The Major Economic Cycles]'. Voprosy konjunktury [Problems of Economic Conditions], Vol. 1, pp. 28-79.
- Kremer, M., & Maskin, E. (1996). Wage Inequality and Segregation by Skill. National Bureau of Economic Research Working Paper Series No. 5718.
- Kretschmer, T. (2012). Information and communication technologies and productivity growth: A survey of the literature. OECD Digital Economy Papers 195. Paris: OECD.
- Krugman, P. (1991). 'Increasing Returns and Economic Geography'. *Journal* of *Political Economy*, Vol. 99(3), pp. 483-499.
- Kuhn, T. (1962). *The structure of scientific revolutions*. Chicago: Chicago University Press.
- Lam, A. (2000). 'Tacit Knowledge, Organizational Learning and Societal Institutions: An Integrated Framework'. Organization Studies, Vol. 21(3), pp. 487-513.
- Landes, D. (1969). *The Unbound Prometheus: Technological Change and Industrial Development in Western Europe from 1750 to the Present.* Cambridge: Cambridge University Press.
- Landes, D. (1980). 'The Creation of Knowledge and Technique: Today's Task and Yesterday's Experience'. *Daedalus*, Vol. 109(1), pp. 111-120.
- Laudel, G. (2003). 'Studying the brain drain: Can bibliometric methods help?'. *Scientometrics,* Vol. 57(2), pp. 215-237.
- Lawson, C. (1999). 'Towards a competence theory of the region'. *Cambridge Journal of Economics*, Vol. 23(2), pp. 151-166.
- Lenzi, C. (2009). 'Patterns and determinants of skilled workers' mobility: evidence from a survey of Italian inventors'. *Economics of Innovation and New Technology*, Vol. 18(2), pp. 161-179.
- Levy, F., & Murnane, R. (2004). *The New Division of Labor: How Computers Are Creating the Next Job Market*. Princeton, NJ: Princeton University Press.

- Lewis, W. A. (1954). 'Economic Development with Unlimited Supplies of Labour'. *The Manchester School*, Vol. 22(2), pp. 139-191.
- Liao, H., Wang, B., Li, B., & Weyman-Jones, T. (2016). 'ICT as a generalpurpose technology: The productivity of ICT in the United States revisited'. *Information Economics and Policy*, pp. 1-16.
- Lilien, D. M. (1982). 'Sectoral Shifts and Cyclical Unemployment'. *Journal* of *Political Economy*, Vol. 90(4), pp. 777-793.
- Lindley, J., Upward, R., & Wright, P. (2002). Regional mobility and unemployment transitions in the UK and Spain. GEP Working Paper No. 02/19 Nottingham, UK: Leverhulme Centre for Research on Globalisation and Economic Policy, University of Nottingham.
- Lipsey, R. G., Carlaw, K., & Bekar, C. (2005). *Economic Transformations: General Purpose Technologies and Long-Term Economic Growth*. Oxford and New York: Oxford University Press.
- Liu, P.-W. (1986). 'Human capital, job matching and earnings growth between jobs: an empirical analysis'. *Applied Economics*, Vol. 18(10), pp. 1135-1147.
- Ljungberg, J. (2016). Introduction: structural analysis and the process of economic development. In J. Ljungberg (Ed.), *Structural Analysis and the Process of Economic Development*. Oxon and New York: Routledge, pp. 1-18.
- Lösch, A. (1954). *The Economics of Location*. New Haven and London: Yale University Press.
- Louçã, F. (2007). Long waves, the pulsation of modern capitalism. In H. Hanusch & A. Pyka (Eds.), *Elgar Companion to Neo-Schumpeterian Economics*. Cheltenham and Northampton, MA: Edward Elgar, pp. 800-819.
- Lund Vinding, A. (2006). 'Absorptive capacity and innovative performance: A human capital approach'. *Economics of Innovation and New Technology*, Vol. 15(4-5), pp. 507-517.
- Lundh, C. (2005). Wage Formation and Institutional Change in the Swedish Labour Market, 1850-2000. In C. Lundh, J. Olofsson, L. Schön, & L. Svensson (Eds.), Wage Formation, Labour Market Institutions and

Economic Transformation in Sweden 1860-2000. Stockholm: Almqvist & Wiksell International.

- Lundquist, K.-J., & Olander, L.-O. (2009). Tillväxt, cykler och kriser. Nationell omvandling och regional utvecking under tre decennier. Rapporter och Notiser 170. Lund: Institutionen för kulturgeografi och ekonomisk geografi, Lunds Universitet.
- Lundquist, K.-J., & Olander, L.-O. (2010). *Growth cycles: transformation and regional development.* Paper presented at the Paper presented at 50th Anniversary European Congress of the Regional Science Association International "Sustainable Regional Growth and Development in the Creative Knowledge Economy", 19-23 August 2010, Jönköping, Sweden.
- Lundquist, K.-J., Olander, L.-O., & Henning, M. (2008a). Creative destruction and economic welfare in Swedish regions: spatial dimensions of structural change, growth and employment. SRE-Discussion Paper No. 2008/03 Vienna: Institute for Regional Development and Environment.
- Lundquist, K.-J., Olander, L.-O., & Henning, M. (2008b). 'Decomposing the technology shift: evidence from the Swedish manufacturing sector'. *Tijdschrift voor economische en sociale geografie*, Vol. 99(2), pp. 145-159.
- Lundquist, K.-J., Olander, L.-O., & Henning, M. (2008c). 'Producer services: growth and roles in long-term economic development'. *The Service Industries Journal*, Vol. 28(4), pp. 463-477.
- Magnusson, L. (2007). *The Swedish Labour Market in a Globalised World*. Referat Westliche Industrieländer 1-2007. Stockholm: Friedrich Ebert Stiftung.
- Magrini, S. (2004). Chapter 62 Regional (di)convergence. In J. V. Henderson & T. Jacques-François (Eds.), *Handbook of Regional and Urban Economics (Vol. 4)*: Elsevier, pp. 2741-2796.
- Malerba, F. (2002). 'Sectoral systems of innovation and production'. *Research Policy*, Vol. 31(2), pp. 247-264.
- Maliranta, M., Mohnen, P., & Rouvinen, P. (2009). 'Is inter-firm labor mobility a channel of knowledge spillovers? Evidence from a linked

employer–employee panel'. *Industrial and Corporate Change*, Vol. 18(6), pp. 1161-1191.

- Mamede, R. (2002). Does innovation (really) matter for success? The case of an IT Consulting Firm. Munich Personal RePec Archive Paper No. 3999. Munich: Munich University Library, University of Munich.
- Mamede, R. (2009). 'Toward an integrated approach to industry dynamics and labor mobility'. *Industrial and Corporate Change*, Vol. 18(1), pp. 139-163.
- Mandel, E. (1975). Late Capitalism. London: New Left Books.
- Mandel, E. (1980). Long Waves of Capitalist Development: The Marxist Interpretation (Vol. Cambridge): Cambridge University Press.
- Mansfield, E. (1968). *Industrial Research and Technological Innovation*. New York: Norton.
- Marchetti, C. (1980). 'Society as a learning system: Discovery, invention, and innovation cycles revisited'. *Technological Forecasting and Social Change*, Vol. 18(4), pp. 267-282.
- Marshall, A. (1948). Principles of economics (8th Ed.). London: Macmillan.
- Martin, B. R. (1995). 'Foresight in science and technology'. *Technology* Analysis & Strategic Management, Vol. 7(2), pp. 139-168.
- Martin, R. (2001). 'Geography and public policy: the case of the missing agenda'. *Progress in Human Geography*, Vol. 25(2), pp. 189-210.
- Martin, R. (2015). 'Rebalancing the Spatial Economy: The Challenge for Regional Theory'. *Territory, Politics, Governance,* Vol. 3(3), pp. 235-272.
- Martin, R., & Sunley, P. (2006). 'Path dependence and regional economic evolution'. *Journal of Economic Geography*, Vol. 6(4), pp. 395-437.
- Martin, R., & Sunley, P. (2015). 'Towards a Developmental Turn in Evolutionary Economic Geography?'. *Regional Studies*, Vol. 49(5), pp. 712-732.
- Matzner, E., Schettkat, R., & Wagner, M. (1990). 'Labour market effects of new technology'. *Futures*, Vol. 22(7), pp. 687-709.

- McLafferty, S. L. (1995). 'Counting for women'. *The Professional Geographer*, Vol. 47(4), pp. 436-442.
- Metcalfe, J. S., Foster, J., & Ramlogan, R. (2006). 'Adaptive economic growth'. *Cambridge Journal of Economics*, Vol. 30(1), pp. 7-32.
- Metz, R. (1992). A Re-examination of Long Waves in Aggregate Production Series. In A. Kleinknecht, E. Mandel, & I. Wallerstein (Eds.), New Findings in Long-Wave Research. London: Palgrave Macmillan UK, pp. 80-119.
- Mokyr, J. (2002). *The Gifts of Athena: Historical Origins of the Knowledge Economy*. Princeton, NJ: Princeton University Press.
- Mokyr, J. (2005). 'The Intellectual Origins of Modern Economic Growth'. *The Journal of Economic History*, Vol. 65(2), pp. 285-351.
- Moretti, E. (2004). 'Estimating the social return to higher education: Evidence from cross-sectional and longitudinal data'. *Journal of Econometrics*, Vol. 121(1-2), pp. 175-212.
- Moretti, E. (2011). Chapter 14 Local Labor Markets. In C. David & A. Orley (Eds.), *Handbook of Labor Economics (Vol. Volume 4, Part B)*: Elsevier, pp. 1237-1313.
- Moretti, E. (2012). *The New Geography of Jobs*. Boston and New York: Houghton Mifflin Harcourt.
- Mortensen, D. T., & Pissarides, C. A. (1998). 'Technological Progress, Job Creation, and Job Destruction'. *Review of Economic Dynamics*, Vol. 1(4), pp. 733-753.
- Muir, R. (1978). 'Radical Geography or a New Orthodoxy?'. Area, Vol. 10(5), pp. 322-327.
- Mundell, R. A. (1961). 'A Theory of Optimum Currency Areas'. *The American Economic Review*, Vol. 51(4), pp. 657-665.
- Myrdal, G. (1957). *Economic theory and under-developed regions*. London: Duckworth.
- Neal, D. (1995). 'Industry-specific human capital: Evidence from displaced workers'. *Journal of Labor Economics*, Vol. 13(4), pp. 653-677.
- Neal, D. (1999). 'The Complexity of Job Mobility Among Young Men'. Journal of Labor Economics, Vol. 17(2), pp. 237-261.

- Neffke, F., & Henning, M. (2013). 'Skill relatedness and firm diversification'. *Strategic Management Journal*, Vol. 34(3), pp. 297-316.
- Neffke, F., Henning, M., Boschma, R., Lundquist, K.-J., & Olander, L.-O. (2011). 'The Dynamics of Agglomeration Externalities along the Life Cycle of Industries'. *Regional Studies*, Vol. 45(1), pp. 49-65.
- Neffke, F., & Nedelkoska, L. (2010). *Human capital mismatches along the career path*. Jena economic research papers No. 2010,051. Jena: Leibniz Information Centre for Economics.
- Neffke, F., Otto, A., & Weyh, A. (2016). *Inter-industry labor flows*. Available at <u>http://frankneffke.com/files/NeffkeOttoWeyh2016.pdf</u>.
- Nelson, R. R. (1995). 'Recent Evolutionary Theorizing About Economic Change'. *Journal of Economic Literature*, Vol. 33, pp. 48-90.
- Nelson, R. R. (1996). *The Sources of Economic Growth*. Cambridge, MA: Harvard University Press.
- Nelson, R. R., & Winter, S. G. (1974). 'Neoclassical vs. Evolutionary Theories of Economic Growth: Critique and Prospectus'. *The Economic Journal*, Vol. 84(336), pp. 886-905.
- Nelson, R. R., & Winter, S. G. (1977). 'In search of useful theory of innovation'. *Research Policy*, Vol. 6(1), pp. 36-76.
- Nelson, R. R., & Winter, S. G. (1982). *An Evolutionary Theory of Economic Change*. Cambridge, MA: Belknap Press.
- Neumann, G. R., & Topel, R. H. (1991). 'Employment Risk, Diversification, and Unemployment'. *The Quarterly Journal of Economics*, Vol. 106(4), pp. 1341-1365.
- Nickerson, J. A., & Olin, J. M. (2004). 'A Knowledge-Based Theory of the Firm—The Problem-Solving Perspective'. *Organization Science*, Vol. 15(6), pp. 617-632.
- NUTEK. (2004). Analyser för regionalt utvecklingsarbete En handbok med praktiska tips och metodexempel. Stockholm: Verket för näringslivsutveckling (NUTEK).
- O'Leary, E., & Webber, D. J. (2015). 'The Role of Structural Change in European Regional Productivity Growth'. *Regional Studies*, Vol. 49(9), pp. 1548-1560.
- O'Hara, P. A. (2001). Long Waves of Growth and Development. In P. A. O'Hara (Ed.), *Encyclopedia of Political Economy (Vol. 2)*. London and New York: Routledge, pp. 673-677.
- OECD. (2011). Divided We Stand: Why Inequality Keeps Rising. Paris

Organisation for Economic Co-operation and Development.

- Oliner, S., & Sichel, D. (1994). 'Computers and Output Growth Revisited: How Big Is the Puzzle?'. *Brookings Papers on Economic Activity*, Vol. 25(2), pp. 273-334.
- Overman, H. G., Puga, D., & Vandenbussche, H. (2002). 'Unemployment Clusters across Europe's Regions and Countries'. *Economic Policy*, Vol. 17(34), pp. 117-147.
- Owen-Smith, J., & Powell, W. W. (2004). 'Knowledge Networks as Channels and Conduits: The Effects of Spillovers in the Boston Biotechnology Community'. *Organization Science*, Vol. 15(1), pp. 5-21.
- Parent, D. (2000). 'Industry-specific capital and the wage profile: Evidence from the national longitudinal survey of youth and the panel study of income dynamics'. *Journal of Labor Economics*, Vol. 18(2), pp. 306-323.
- Pastore, F. (2013). *Primum Vivere ... Industrial Change, Job Destruction and the Geographical Distribution of Unemployment*. IZA Discussion Paper No. 7126. Bonn: Institute for the Study of Labour.
- Pavitt, K. (1984). 'Sectoral patterns of technical change: Towards a taxonomy and a theory'. *Research Policy*, Vol. 13(6), pp. 343-373.
- Penrose, E. T. (1989). History, the Social Sciences and Economic 'Theory', with Special Reference to Multinational Enterprise. In A. Teichova, M. Lévy-Leboyer, & H. Nussbaum (Eds.), *Historical Studies in International Corporate Business*. Cambridge: Cambridge University Press, pp. 7-14.
- Perez, C. (1983). 'Structural change and assimilation of new technologies in the economic and social systems'. *Futures*, Vol. 15(5), pp. 357-375.

- Perez, C. (2002). *Technological Revolutions and Financial Capital: The Dynamics of Bubbles and Golden Ages.* Cheltenham and Northampton, MA: Edward Elgar.
- Perez, C. (2004). Technological Revolutions, Paradigm Shifts and Socioinstitutional Change. In E. S. Reinert (Ed.), *Globalization, Economic Development and Inequality: An Alternative Perspective*. Cheltenham and Northampton, MA: Edward Elgar, pp. 217-242.
- Perez, C. (2010). 'Technological revolutions and techno-economic paradigms'. *Cambridge Journal of Economics*, Vol. 34(1), pp. 185-202.
- Petsas, I. (2003). 'The dynamic effects of general purpose technologies on Schumpeterian growth'. *Journal of Evolutionary Economics*, Vol. 13(5), pp. 577-605.
- Philip, L. J. (1998). 'Combining Quantitative and Qualitative Approaches to Social Research in Human Geography—An Impossible Mixture?'. *Environment and Planning A*, Vol. 30(2), pp. 261-276.
- Pike, A., Rodríguez-Pose, A., & Tomaney, J. (2016). 'Shifting horizons in local and regional development'. *Regional Studies*, pp. 1-12.
- Plummer, P., & Sheppard, E. (2001). 'Must Emancipatory Economic Geography be Qualitative?'. *Antipode*, Vol. 33(2), pp. 194-199.
- Plummer, P. S. (2003). The Modelling Tradition. In E. Sheppard & T. J. Barnes (Eds.), A Companion to Economic Geography. Oxford: Blackwell, pp. 27-40.
- Poletaev, M., & Robinson, C. (2008). 'Human capital specificity: evidence from the Dictionary of Occupational Titles and Displaced Worker Surveys, 1984–2000'. *Journal of Labor Economics*, Vol. 26(3), pp. 387-420.
- Power, D., & Lundmark, M. (2004). 'Working through Knowledge Pools: Labour Market Dynamics, the Transference of Knowledge and Ideas, and Industrial Clusters'. *Urban Studies*, Vol. 41(5-6), pp. 1025-1044.
- Pratt, A. C. (1995). 'Putting critical realism to work: the practical implications for geographical research'. *Progress in Human Geography*, Vol. 19(1), pp. 61-74.

- Puga, D. (2002). 'European regional policies in light of recent location theories'. *Journal of Economic Geography*, Vol. 2(4), pp. 373-406.
- Quatraro, F. (2012). *The Economics of Structural Change in Knowledge*. London and New York: Routledge.
- Reichlin, P., & Rustichini, A. (1998). 'Diverging patterns with endogenous labor migration'. *Journal of Economic Dynamics and Control*, Vol. 22(5), pp. 703-728.
- Rigby, D. L., & Essletzbichler, J. (1997). 'Evolution, Process Variety, and Regional Trajectories of Technological Change in U.S. Manufacturing'. *Economic Geography*, Vol. 73(3), pp. 269-284.
- Rincon, A., Vecchi, M., & Venturini, F. (2013). *ICT AS A GENERAL PURPOSE TECHNOLOGY: SPILLOVERS, ABSORPTIVE CAPACITY AND PRODUCTIVITY PERFORMANCE.* NIESR Discussion Paper No. 416. London: National Institute of Economic and Social Research.
- Robert, V., & Yoguel, G. (2016). 'Complexity paths in neo-Schumpeterian evolutionary economics, structural change and development policies'. *Structural Change and Economic Dynamics*, Vol. 38, pp. 3-14.
- Robson, M. (2009). 'Structural change, specialization and regional labour market performance: evidence for the UK'. *Applied Economics*, Vol. 41(3), pp. 275-293.
- Rose, A. (1941). 'Wars, Innovations and Long Cycles: A Brief Comment'. *The American Economic Review*, Vol. 31(1), pp. 105-107.
- Rosenberg, N., & Trajtenberg, M. (2004). 'fA General-Purpose Technology at Work: The Corliss Steam Engine in the Late-Nineteenth-Century United States'. *The Journal of Economic History*, Vol. 64(1), pp. 61-99.
- Rosenkopf, L., & Almeida, P. (2003). 'Overcoming Local Search Through Alliances and Mobility'. *Management Science*, Vol. 49(6), pp. 751-766.
- Rostow, W. W. (1985). 'The World Economy since 1945: A Stylized Historical Analysis†'. *The Economic History Review*, Vol. 38(2), pp. 252-275.

- Rowlatt, A. (2001). 'Measuring e-commerce: developments in the United Kingdom'. *Economic Trends*, Vol. 575, pp. 30-36.
- Saxenian, A. (1994). *Regional Advantage: Culture and Competition in Silicon Valley and Route 128*. Cambridge, MA: Harvard University Press.
- Sayer, A. (1992). *Method in social science: a realist approach* (Second edition). London: Routledge.
- Sayer, A. (2000). Realism in Social Science. London: Sage Publications.
- SCB. (2016a). Arbetsmarknadssituationen för hela befolkningen 15-74 år, AKU 2015. Statistiska meddelanden AM12 SM1501. Örebro: SCB.
- SCB. (2016b). Longitudinell integrationsdatabas för Sjukförsäkrings- och Arbetsmarknadsstudier (LISA) 1990-2013. Arbetsmarknad och Utbildning Bakgrundsfakta 2016:1. Örebro: SCB.
- Schamp, E. W. (2010). *On the notion of co-evolution in economic geography*. Cheltenham and Northampton, MA: Edward Elgar.
- Schön, L. (1998). Industrial Crises in a Model of Long Cycles; Sweden in an International Perspective. In T. Myllyntaus (Ed.), *Economic Crises* and Restructuring in History: Experiences of Small Countries. St. Katharinen: Scripta Mercaturae Verlag.
- Schön, L. (2010). Sweden's Road to Modernity. An Economic History. Stockholm: SNS Förlag.
- Schön, L. (2012). An Economic History of Modern Sweden. Oxon: Routledge.
- Schumpeter, J. A. (1939). Business Cycles: A Theoretical, Historical and Statistical Analysis of the Capitalist Process. 2 vols. New York: McGraw-Hill Press.
- Schumpeter, J. A. (1942). *Capitalism, socialism, and democracy*. London and New York: Harper & Brothers.
- Schumpeter, J. A. (1954). *History of Economic Analysis*. Oxford: Oxford University Press.
- Schumpeter, J. A. (1961). *The Theory of Economic Development*. New York: Oxford University Press.

- Shearmur, R. (2011). 'Innovation, Regions and Proximity: From Neo-Regionalism to Spatial Analysis'. *Regional Studies*, Vol. 45(9), pp. 1225-1243.
- Silverberg, J. (2007). Long waves: conceptual, empirical and modelling issues. In H. Hanusch & A. Pyka (Eds.), *Elgar Companion to Neo-Schumpeterian Economics*. Cheltenham and Northampton, MA: Edward Elgar, pp. 800-819.
- Simmie, J. (2001). Innovative Cities. London: Spon Press.
- Simon, H. A. (1971). Designing Organizations for an Information-Rich World. In M. Greenberger (Ed.), *Computers, Communication, and the Public Interest*. Baltimore, MD: The John Hopkins Press, pp. 38-73.
- Simonen, J., & McCann, P. (2008). 'Firm innovation: The influence of R&D cooperation and the geography of human capital inputs'. *Journal of Urban Economics*, Vol. 64(1), pp. 146-154.
- Singh, J., & Agrawal, A. (2011). 'Recruiting for Ideas: How Firms Exploit the Prior Inventions of New Hires'. *Management Science*, Vol. 57(1), pp. 129-150.
- Solow, R. (1987, July 12). We'd Better Watch Out. New York Times Book Review, p. 36.
- Song, J., Almeida, P., & Wu, G. (2003). 'Learning-by-Hiring: When Is Mobility More Likely to Facilitate Interfirm Knowledge Transfer?'. *Management Science*, Vol. 49(4), pp. 351-365.
- Sørensen, J. B. (2004). 'Recruitment-based competition between industries: a community ecology'. *Industrial and Corporate Change*, Vol. 13(1), pp. 149-170.
- Sorm, V., & Terrell, K. (2000). 'Sectoral Restructuring and Labor Mobility: A Comparative Look at the Czech Republic'. *Journal of Comparative Economics*, Vol. 28(3), pp. 431-455.
- Spitz-Oener, A. (2006). 'Technical Change, Job Tasks, and Rising Educational Demands: Looking outside the Wage Structure'. *Journal* of Labor Economics, Vol. 24(2), pp. 235-270.

- Scholec, M., & Verspagen, B. (2012). 'The Voyage of the Beagle into innovation: explorations on heterogeneity, selection, and sectors'. *Industrial and Corporate Change*.
- Stanley, T. D., Doucouliagos, C., & Steel, P. (2015). Does ICT Generate Economic Growth? A Meta-Regression Analysis. School Working Paper Economics Series 2015/9. Burwood, Australia: School of Accounting, Economics and Finance, Faculty of Business and Law, Deakin University.
- Sterman, J. D. (1985). 'An integrated theory of the economic long wave'. *Futures*, Vol. 17(2), pp. 104-131.
- Stiroh, K. (2010). Reassessing the impact of IT in the production function: A meta-analysis and sensitivity tests. In J. Mairesse & M. Trajtenberg (Eds.), *Contributions in memory of Zvi Griliches*. Cambridge, MA: NBER, pp. 529-561.
- Storper, M., & Walker, R. (1989). *The capitalist imperative : territory, technology, and industrial growth*. Oxford, UK and New York, NY: Basil Blackwell.
- Stoyanov, A., & Zubanov, N. (2014). 'The distribution of the gains from spillovers through worker mobility between workers and firms'. *European Economic Review*, Vol. 70, pp. 17-35.
- Strohmaier, R., & Rainer, A. (2013). On the Economic Purpose of General Purpose Technologies: A Combined Classical and Evolutionary Framework. Munich Personal RePec Archive Paper No. 45964. Munich: Munich University Library, University of Munich.
- Sullivan, P. (2010). 'Empirical evidence on occupation and industry specific human capital'. *Labour Economics*, Vol. 17(3), pp. 567-580.
- Szulanski, G. (1996). 'Exploring internal stickiness: Impediments to the transfer of best practice within the firm'. *Strategic Management Journal*, Vol. 17(S2), pp. 27-43.
- ter Weel, B. (2006). 'Feature: It Diffusion and Industry and Labour-Market Dynamics'. *The Economic Journal*, Vol. 116(509), pp. F1-F9.
- The World Bank. (2016). *World DataBank*. Retrieved from <u>http://databank.worldbank.org/data/reports.aspx?source=2&series=</u> <u>NY.GDP.MKTP.CD&country=#</u> on October 26, 2016.

- Thompson, W. R., & Zuk, L. G. (1982). 'War, Inflation, and the Kondratieff Long Wave'. *The Journal of Conflict Resolution*, Vol. 26(4), pp. 621-644.
- Tödtling, F., & Trippl, M. (2005). 'One size fits all?: Towards a differentiated regional innovation policy approach'. *Research Policy*, Vol. 34(8), pp. 1203-1219.
- Topel, R. H., & Ward, M. P. (1992). 'Job Mobility and the Careers of Young Men'. *The Quarterly Journal of Economics*, Vol. 107(2), pp. 439-479.
- Trajtenberg, M. (1990). *Economic Analysis of Product Innovation. The Case* of CT Scanners. Cambridge, MA: Harvard University Press.
- Trippl, M., & Maier, G. (2010). 'Knowledge spillover agents and regional development'. *Papers in Regional Science*, Vol. 89(2), pp. 229-233.
- Tsikriktsis, N., Lanzolla, G., & Frohlich, M. (2004). 'Adoption of e-Processes by Service Firms: An Empirical Study of Antecedents'. *Production and Operations Management*, Vol. 13(3), pp. 216-229.
- Usher, A. P. (1954). *A History of Mechanical Inventions* (Revised edition). Cambridge, MA: Harvard University Press.
- Uzawa, H. (1961). 'On a Two-Sector Model of Economic Growth'. *The Review of Economic Studies,* Vol. 29(1), pp. 40-47.
- Uzawa, H. (1963). 'On a Two-Sector Model of Economic Growth II'. *The Review of Economic Studies*, Vol. 30(2), pp. 105-118.
- van den Berg, A., Furåker, B., & Johansson, L. (1997). Labour Market Regimes and Patterns of Flexibility. A Sweden–Canada Comparison. Lund: Arkiv Förlag.
- van Duijn, J. J. (1983). *The long wave in economic life*. London: Allen and Unwin.
- van Ewijk, C. (1982). 'A SPECTRAL ANALYSIS OF THE KONDRATIEFF-CYCLE'. *Kyklos*, Vol. 35(3), pp. 468-499.
- van Zon, A., Fortune, E., & Kronenberg, T. (2003). *How to Sow and Reap as You Go: Simple Model of Cyclical Endogenous Growth*. MERIT-Infonomics Research Memorandum Series 2003-026. Maastricht

Economic Research Institute on Innovation and Technology and International Institute of Infonomics: Maastricht.

- Verspagen, B. (2004). 'Structural Change and Technology: A Long View'. *Revue économique*, Vol. 55(6), pp. 1099-1125.
- Verspagen, B. (2005). Innovation and Economic Growth. In J. Fagerberg, D. C. Mowery, & R. R. Nelson (Eds.), *The Oxford Handbook of Innovation*. Oxford: Oxford University Press.
- Volland, C. S. (1987). 'A comprehensive theory of long wave cycles'. *Technological Forecasting and Social Change*, Vol. 32(2), pp. 123-145.
- Vona, F., & Consoli, D. (2015). 'Innovation and skill dynamics: a life-cycle approach'. *Industrial and Corporate Change*, Vol. 24(6), pp. 1393-1415.
- Westerlund, O. (1997). 'Employment Opportunities, Wages and International Migration in Sweden 1970–1989'. *Journal of Regional Science*, Vol. 37(1), pp. 55-73.
- Westerlund, O. (1998). 'Internal Migration in Sweden: The Effects of Mobility Grants and Regional Labour Market Conditions'. *LABOUR*, Vol. 12(2), pp. 363-388.
- Witt, U. (2003). *The Evolving Economy. Essays on the Evolutionary Approach to Economics.* Cheltenham and Northampton, MA: Edward Elgar.
- Witt, U. (2008). Evolutionary economics. In S. N. Durlauf & L. E. Blume (Eds.), *The New Palgrave Dictionary of Economics*. Basingstoke: Palgrave Macmillan.
- Wolff, E. N. (2006). 'The growth of information workers in the US economy, 1950–2000: the role of technological change, computerization, and structural change'. *Economic Systems Research*, Vol. 18(3), pp. 221-255.
- Yeung, H. W.-c. (1997). 'Critical realism and realist research in human geography: a method or a philosophy in search of a method?'. *Progress in Human Geography*, Vol. 21(1), pp. 51-74.

- Zahra, S. A., & George, G. (2002). 'Absorptive Capacity: A Review, Reconceptualization, and Extension'. *The Academy of Management Review*, Vol. 27(2), pp. 185-203.
- Zaring, O., & Eriksson, C. M. (2009). 'The dynamics of rapid industrial growth: evidence from Sweden's information technology industry, 1990–2004'. *Industrial and Corporate Change*, Vol. 18(3), pp. 507-528.

Article I

Lund, November 2, 2016

To whom it may concern,

The authors hereby certify that the paper entitled 'Technological Change and Geographical Reallocation of Labour: On the Role of Leading Industries' is based on 75%/25% contributions by the respective authors Martynovich/Lundquist.

fli flimitycarich

Mikhail Martynovich

hope the lente

Karl-Johan Lundquist



Technological Change and Geographical Reallocation of Labour: On the Role of Leading Industries

MIKHAIL MARTYNOVICH* 6 and KARL-JOHAN LUNDQUIST

Department of Human Geography, Faculty of Social Sciences, Lund University, Sölvegatan 10, Kulturgeografi, Lunds Universitet, Lund SE-223 62, Sweden. Emails: mikhail.martynovich@keg.lu.se and karl-johan.lundquist@keg.lu.se

(Received May 2014; in revised form May 2015)

MARTYNOVICH M. and LUNDQUIST K.-J. Technological change and geographical reallocation of labour: on the role of leading industries, *Regional Studies*. This paper analyses inter-regional labour reallocation in Sweden over the period 1985–2008 and assesses the effects of technology-induced structural change on the ability of regions to attract and retain workers. The findings suggest that (1) the regional presence of leading industries associated with technological change has a significant effect; (2) the importance of leading industries is of dynamic character as various functional groups of leading industries play different roles at different stages of the technology-induced transformation process; and (3) while manufacturing branches act as a stabilizing factor, i.e. helping regions to retain workers, service industries drive labour reallocation in terms of attracting workers to regions.

Technological change Industrial restructuring Adjustment Labour reallocation Leading industries Sweden

MARTYNOVICH M. and LUNDQUIST K.-J. 技术变革与劳动的地理再配置:先导产业的角色,区域研究。本文分析瑞典 在1985年至2008年期间的跨区域劳动再配置,并评估由技术引导的结构变迂,对于区域吸引并留住劳工之能力的影响。研究结果显示:(1)与技术变革有关的先导产业存在于区域中,具有显着的效应:(2)先导产业的重要性,具 有强而有力的特徵、因先导产业的各种功能团体,在技术引导的转变过程的不同阶段中,扮演不同的角色:UAG (3) 当製造部门作为稳定的因素时,例如协助区域留住劳工,服务业则是在吸引劳工至该区域方面,驱动劳动的再配置。

技术变革 产业再结构 调节 劳动再配置 先导产业 瑞典

MARTYNOVICH M. et LUNDQUIST K.-J. La mutation technologique et la redistribution géographique du travail: à propos des industries phares, *Regional Studies*. Cet article cherche à analyser la redistribution interrégionale du travail pour la période allant de 1985 à 2008 et évalue l'impact de la mutation structurelle stimulée par la technologie sur la capacité des régions à recruter et à maintenir des travailleurs. Les résultats suggèrent que (1) la présence dans la région des industries phares associées à la mutation technologique a un impact non-négligeable; (2) l'importance des industries phares s'avère dynamique parce que divers groupes fonctionnels jouent des rôles différents à diverses étapes du processus de transformation qui est stimulé par la technologie; et (3) alors que les établissements industries levrent de facteur de stabilisation, à savoir en aidant les régions à maintenir l'emploi, les industries des services déterminent la redistribution du travail en termes de l'attrait des travailleurs vers les régions.

Mutation technologique Restructuration industrielle Ajustement Redistribution du travail Industries phares Suède

MARTYNOVICH M. und LUNDQUIST K.-J. Technischer Wandel und geografische Neuverteilung von Arbeitsplätzen: die Rolle der führenden Branchen, *Regional Studies*. In diesem Beitrag untersuchen wir die interregionale Neuverteilung von Arbeitsplätzen: von Regionen Zurtaum von 1985 bis 2008 und die Auswirkungen des technisch bedingten Strukturwandels auf die Fähigkeit von Regionen zur Anwerbung und Beibehaltung von Arbeitskräften. Aus den Ergebnissen geht hervor, dass (1) die regionale Präsenz von führenden mit dem technischen Wandel verknüpften Branchen eine signifikante Auswirkung hat; (2) die Bedeutung der führenden Branchen einen dynamischen Charakter aufweist, da verschiedene Funktionsgruppen der führenden Branchen in den verschiedenen Phasen des technisch bedingten Veränderungsprozesses unterschiedliche Rollen spielen; und (3) während produzierende Branchen die stabilisierender Faktor wirken, d. h. Regionen bei der Beibehaltung von Arbeitskräften unterstützen, Dienstleistungsbranchen die Neuverteilung von Arbeitskräften vorantreiben, indem sie Arbeitnehmer in Regionen anziehen.

Technischer Wandel Branchenumstrukturierung Anpassung Neuverteilung von Arbeitsplätzen Führende Branchen Schweden

*Corresponding author.

© 2015 Regional Studies Association http://www.regionalstudies.org MARTYNOVICH M. y LUNDQUIST K.-J. Cambio tecnológico y reasignación geográfica del trabajo: el papel de las industrias líderes, *Regional Studies*. En este artículo analizamos la reasignación laboral interregional en Suecia entre 1985 y 2008 y evaluamos los efectos del cambio estructural inducido por la tecnología en la capacidad de las regiones de atraer y mantener a los trabajadores. Los resultados indican que (1) la presencia regional de industrias líderes asociadas al cambio tecnológico tiene un efecto significativo; (2) la importancia de las industrias líderes tiene un carácter dinámico porque diferentes grupos funcionales de las industrias líderes desempeñan papeles diferentes en distintos estadios del proceso de transformación inducido por la tecnología; y (3) mientras las ramas manufactureras actúan como un factor estabilizador, es decir, ayudan a las regiones a conservar trabajadores, las industrias de servicios impulsan la reasignación laboral atravendo a los trabajadores a las regiones.

Cambio tecnológico Reestructuración industrial Ajuste Reasignación laboral Industrias líderes Suecia

JEL classifications: J61, N14, O33, R11

INTRODUCTION

This paper analyses inter-regional labour mobility patterns in Sweden over 1985–2008 with the aim of assessing effects of technology-induced structural change on a ability of regions to attract and retain workers. Provided the widely acknowledged role of human capital for regional growth and competitiveness (BARRO, 2001; BADINGER and TONDL, 2003; COHEN and SOTO, 2007), understanding the factors helping regions to attract workers is of high theoretical importance.

Aggregate outcomes of technological change are continually (re-)defined by changes in the industrial composition of the economy (SAVIOTTI and PYKA, 2004; METCALFE et al., 2006; HANUSCH and PYKA, 2007). In particular, heterogeneous response of regions to aggregate shocks is mediated, to a large extent, by their economic composition and industrial organization (ESSLETZBICHLER, 2007; ROBSON, 2009). In that respect, what has been missing in the literature is a method to identify leading sectors associated with technological change at various times and measure dynamically their effects on functioning of the economy (SILVERBERG, 2007). The paper addresses this gap by revisiting and empirically refining PEREZ's (1983, 2010) classification of key sectors driving technological change and analysing their role as factors of attracting and retaining workers in regions.

The following research question is aked: In what ways do spatial patterns of distribution of leading industries associated with a current wave of technological change shape inter-regional labour reallocation flows? Equivalently: In what way does a stronger presence of leading industries in a region contribute to its ability to attract and retain workers?

Previous attempts to identify and quantify the role of leading sectors in economic development (ROSTOW, 1960; THOMPSON, 1990) were confined to the national level. Literature on labour mobility as a response to structural change was primarily addressing the intersectoral reallocation dimension (GREENAWAY *et al.*, 2000; ELLIOTT and LINDLEY, 2006; BACHMANN and BURDA, 2010). Studies of regional industrial composition effects on the patterns of inter-regional labour reallocation (BÖCKERMAN *et al.*, 2004; ESSLETZBICHLER, 2007) have not taken technologyinduced structural change into the consideration.

The paper is built upon insights from macroevolutionary economic geography and labour economics' perspective on labour mobility as a response mechanism to economic shocks. Taking a macro-perspective allows the analysis of labour mobility patterns across all elements of a regional system while conceptualizing structural change in its time-dependent, regional and industrial dimensions.

The paper focuses on the latest wave of technological change induced by diffusion of information and communication technology (ICT)-related products and services. Using a unique combination of Swedish datasets 3403978 worker moves across 92 local labour markets over 1985–2008 are detected and analysed.

The paper is organized as follows. The second section summarizes the theoretical framework integrating the literature on technological change, its regional aspect as well as labour market response. The third section discusses employed data and methodology. It is followed by a summary of leading industries' development over time which provides estimates on the effects of their presence on regional labour markets. The fifth section concludes.

TECHNOLOGICAL CHANGE, LEADING INDUSTRIES AND REALLOCATION OF LABOUR

Dynamics of industrial restructuring and spatial transformation

Contemporary theorizing of economic growth is increasingly based on SCHUMPETER's (1939) idea that long-term fluctuations in the rhythm of economic growth are rooted in the discontinuities brought by the introduction of new interrelated and pervasive radical technologies (often referred as general purpose technologies), and subsequent structural adjustment of the economy to newly opening technological opportunities (VERSPAGEN, 2004; METCALFE et al., 2006; LOUÇA, 2007; SILVERBERG, 2007). This deep structural transformation has been referred to in the literature as a shift in the technological paradigm (DOSI, 1982), a technological revolution (PEREZ, 1983), or a technology shift (SCHÖN, 1998).

Although outcomes of this transformation are visible at the macro-level, they cannot be analysed at that level as sources of qualitative changes shaping the transition to the new technological paradigm are defined by industry dynamics at the meso-level (CARLSSON and ELIASSON, 2003; SAVIOTTI and PYKA, 2004; HANUSCH and PYKA, 2007). While industrial sectors, whose technological base is closely related to new general purpose technologies, are exposed to broader opportunities and experience more dynamic development, industries less connected to emerging technological paradigm may be forced to move along less dynamic paths (NELSON and WINTER, 1982; CASTELLACCI, 2008). Changes in the industrial composition of the economy induced by growth and decline in various sectors are then continually redefining the economy-wide relations between productivity, employment and output growth as well as the contributions that each industry makes to these aggregates (METCALFE et al., 2006).

At the same time, processes of an economy's adjustment to the new technological paradigm are not equally distributed across space. A contemporary view of regional economic growth builds upon the principle of increasing returns derived from internal- and external-scale effects, the size of a local market, transaction costs and the comparative advantages from production factors (KRUGMAN, 1991; FUJITA et al., 1999). The principle of increasing returns might be further connected to an endogenous capacity of regions to generate innovation as well as to receive and exploit ideas, innovations, technology, and market changes from the outside world (JAFFE et al., 1993; BOSCHMA and LAMBOOY, 1999). This capacity is dynamic in nature and defines the ability of a region to translate growth forces of the technology shift into competitive performance. As receiver capacities, roles and relations between regions are changing over time, it has been claimed that technology shifts are essentially geographical in character (LUNDQUIST et al., 2008a, 2008b; LUNDQUIST and OLANDER, 2009).

Spatial and sectorial outcomes of adjustment to newly emerging technological opportunities are then dependent on a range of interlinked mechanisms and channels connected to internal organizational structures of firms and industries, pure imitation by new and existing competitors, emerging complementarities between new and old parts of the economy, possibilities for decomposition and fragmentation of value chains, as well as opportunities grown from processes of standardization, price competition and falling transaction costs.

In particular, in the beginning of the adjustment process, industries making an early use of new technological opportunities are mainly located in markets at the top of a regional hierarchy as these provide sufficient economies of scale when transaction costs are high. Beginning gently and experimentally in these sites, new activities take time to become visible. Mature businesses initially dominating the scene gradually decrease in size and importance offering space and resources to newly emerging activities. Regions at lower levels of a regional hierarchy, lacking sufficient locational advantages and receiver capacity, cannot attract new industries resulting in a slower aggregate growth in output, productivity and employment.

With time, new general purpose technologies start shaping new production configurations outside industries of their origination through industrial interactions, demand pressures and technical competition (CANTNER and VANNUCCINI, 2012). This allows a wider spectrum of industries to benefit from opportunities offered by new technologies. At the same time, as transaction costs gradually fall and large regional markets become strained on a resource side, industries drawing upon new technologies or incorporating them into their core activities start to disseminate regionally leading to an accelerated growth in value added, productivity and employment in a wider set of regions.

All in all, the literature suggests that processes of adjustment following the introduction of new generalpurpose technologies are determined by complex patterns of changing roles of various industries and regions. In that respect, any attempt to analyse economic outcomes of shifts in technological paradigms requires an understanding of the underlying dynamics of industrial restructuring and spatial transformation.

Structural change and reallocation of labour

As people follow jobs (STORPER, 2013), patterns of labour mobility are driven by those of job availability. In that respect, in the economy experiencing an ongoing structural change, flexible labour markets smooth an adjustment to new industrial systems by ensuring that changing labour demand patterns between expanding and declining sectors meet a response in the form of worker reallocation (GREEN-AWAY et al., 2002; AGHION et al., 2006; BACHMANN and BURDA, 2010; PASTORE, 2013).

In particular, industrial restructuring in the economy (i.e., relative or absolute decline in some sectors as well as growth or emergence of others) implies changes in profitable opportunities across sectors resulting, among other things, in different rates of job creation and destruction between expanding and declining sectors (GREENAWAY et al., 2000). Old jobs are destroyed when new technologies arrive, but are replaced by updated employment opportunities in existing sectors or completely new jobs in emerging ones (MORTEN-SEN and PISSARIDES, 1998). The consequence of these processes is a reallocation of workers across occupations and sectors. In that respect, mobility of labour is a mechanism facilitating an adjustment to economic shocks.

There is an agreement in the literature that regions differ in their capacity to reabsorb redundant labour from declining industries into new sectors, which leads to different extents of spatial labour market imbalances (CAROLEO and PASTORE, 2010). Here, aggregate shocks have asymmetric effects across regions through their non-neutral effects on industries (NEUMANN and TOPEL, 1991). ROBSON (2009) distinguishes three sources of these effects: (1) in the short run, regions with concentration of employment in 'cyclically sensitive' industries are more vulnerable to aggregate labour demand shocks; (2) long-term shifts in labour demand across industries generate stronger reallocation pressure on workers in regions with a high concentration of declining industries; and (3) labour market performance is affected by a degree of specialization/ diversification with a more diverse industry employment mix providing regions with a higher capacity to reabsorb displaced workers through facilitating matching process in times of idiosyncratic shocks (MORETTI, 2011). As industrial development has uneven spatial patterns, inevitable emergence of imbalances across regional labour markets calls for redistribution of labour across not only sectors but also regions, which affects inter-regional labour mobility patterns.

Hypothesizing a role of leading industries in labour reallocation

Establishment of ICT as a cluster of new general purpose technologies leads to a vast reorganization in industrial structure of the Swedish economy through the emergence of new industries, the revitalization of growth in some existing industries, as well as a decline and phasing out of others. Such restructuring has a longterm impact on labour markets through shaping new demand for labour as there is an explosive growth in employment in new and renewed industries, while labour is displaced from declining ones. Changes in demand patterns are met with the reallocation of labour as a response mechanism to disparities across sectors. As there are uneven spatial patterns in location of growing and declining industries at various stages of the adjustment process, there are additional pressures for workers to be mobile not only across sectors but also regions.

In Sweden, labour mobility across regions was shown to be a major response to changes in economic conditions when compared with participating in labour market programmes, turning unemployed or leaving labour market (FREDRIKSSON, 1999). Besides, institutional factors such as centralized bargaining and the 'solidarity wage policy' have for a long time kept very compressed wage dispersion across regions (EDIN and HOLMLUND, 1995; ELIASSON et al., 2003). As a result, earlier studies of internal migration in Sweden (WESTERLUND, 1997, 1998; FREDRIKSSON, 1999) demonstrated that labour mobility was driven mainly by *job availability* so that workers migrated primarily to regions where employment growth took place with no or minor effects on regional wage differentials. GÄRTNER (2014) showed that, even controlling for wage differentials, increased job availability has a strong effect on inter-regional worker reallocation.

Locational patterns of leading industries associated with technological change are, therefore, hypothesized to play an important role in reshaping Swedish local labour markets in general, and in determining the direction of inter-regional worker flows, in particular. The reasons are twofold. First, these industries are expected to be ones with higher-than-average job creation rates and more pronounced growth in employment. In that respect, they act as employment anchors helping a region to attract and retain labour. Second, as these industries are not isolated and become embedded in existing regional industrial landscapes through, among other things, input-output linkages (CANTNER and VANNUCCINI, 2012; LIND, 2014), a stronger presence of leading industries should sustain/reinforce output and employment growth in other industries present in a region.

All in all, as previous studies show that internal mobility of workers in Sweden is primarily driven by job availability, and the latter is expected to be enhanced by locational patterns of leading industries, it is hypothesized that a stronger regional presence of leading industries results in an increased inflow of workers into and decreased outflow of workers from a region resulting in a positive rate of net employment change.

DATA AND METHODOLOGY

Data employed in the analysis are a compilation of two datasets: the Longitudinal Integration Database for Health Insurance and Labour Market Studies (LISA) provided by Statistics Sweden; and the Database of Evolutionary Economic Geography in Lund (DEVIL).

LISA is a linked employer–employee database covering all individuals registered in Sweden. For each individual it contains an in-depth set of background information including region, industry and firm of employment, which allows one to trace the employment history of an individual and construct a very precise measure of labour flows. Annual data are available for 1985–2011.

DEVIL is a longitudinal establishment-level database providing information on an establishment's municipality of registration, number of employees, value added, etc. It is used to model industrial profiles of Swedish regions with respect to regional employment mixes. Data cover the period between 1968 and 2008.

Data from LISA and DEVIL are merged at the municipality level and further aggregated to 92 local labour

1636

market areas (as of 1999). This specification of spatial units is chosen as it is defined in the way that maximizes homogeneity of within-region mobility patterns while sustaining cross-regional heterogeneity. The resulting dataset includes observations over 1985–2008 across 92 spatial units.

Identifying leading industries

PEREZ (1983, 2010) suggested that in the adjustment process following a technology shift key industries might be divided into groups with respect to their functional relation to technological transformation process, namely:

- Motive branches producing key factors and other inputs directly associated with them and creating, therefore, conditions for development of a new technological paradigm; growth of these industries is dependent on the rhythm of the penetration of new technologies to other industries.
- Carrier branches being the most visible and active users of new inputs and best adapted to a new organization of production; they induce a great variety of investment opportunities, thus having a great influence on a general rhythm of economic growth.
- Induced branches that may have existed before but which are modernized and take on a different role. Their development is both a consequence of and complementary to the growth of motive and carrier branches, and is indispensable for facilitating the maximum diffusion of new industries.

Identification of the leading sectors driving technological transformation of regional economies was performed in two steps. First, using the methodology suggested by LUNDQUIST et al. (2005, 2008a, 2008b), all industries present in the Swedish economy were divided into 30 actor branches with respect to productivity trajectories, price and volume development as well as their functional specialization. In the second step, most salient industries driving the transformation of the economy were identified with respect to total value-added growth patterns in each of 30 actor branches. Five manufacturing actor branches and eight service actor branches were then selected and further divided into three functional groups - motive, carrier and induced - following Perez's classification. For the more detailed information regarding the classification process, see Appendix A in the Supplemental data online.

Dependent variables: worker flow rates

Worker moves are observed when individuals switch region of employment between two consecutive years. In practice, the measure of worker flows was constructed according to the following procedure.

First, for each year *t* a sample of individuals aged 15–74 years was selected, which corresponds to the current

Statistics Sweden definition of the labour force.1 Second, an assumption of 'contracted' mobility - that is, labour mobility as an outcome of a successful job search process - was made. Given that all individuals of working age are potential job searchers in all regions at time t - 1, it implies that switching a region was considered as a worker move if: (1) an individual was employed at time t; and (2) a region of employment at time t was different from a region of residence (for previously unemployed) or employment (for previously employed) at time t - 1. Finally, to minimize the probability of including workers participating in short-term employment, who are less likely to contribute to the 'true' labour reallocation, the sample was further restricted to individuals who were employed directly at the moment of data collection (not just at some point over a year). As a result, 3403978 worker moves were identified.

Consequently, for each local labour market total worker inflows (WIFs) and outflows (WOFs) were calculated. Finally, to deal with potential scale issues (as bigger regions would always tend to have higher inand outflows), worker flows were transformed into worker flow rates:

- $WIFR_{jt} = WIF_{jt}/N_t \text{gross worker inflow rate for region } j$ at time $t^{[2]}$
- WOFR_{jt} = WOF_{jt}/ N_t gross worker outflow rate for region j at time t
- NETR_{jt} = WIFR_{jt} WOFR_{jt} net worker flow rate for region j at time t

Including gross worker flow rates into the analysis is motivated by two reasons: (1) net worker flow rates alone often mask substantial labour market dynamics behind net employment change (DAVIs and HALTI-WANGER, 1995; ESSLETZBICHLER, 2007); and (2) estimations for gross flow rates might demonstrate push and pull factors resulting in the net reallocation of workers.

Explanatory variables

Principal explanatory variables are regional employment shares in three groups of leading industries (separately for manufacturing and services):

$$LI_{fjt} = \frac{x_{fjt}}{\sum_{i=1}^{n} xi_{jt}}$$

where x_{fjt} is the employees in group f in region j at time t; x_{ijt} is the employees in industry i in region j at time t; and n is the number of industries. Changes in this variable are supposed to capture long-term shifts in demand for labour and, in that respect, reflect the effect of adjustment process following the technology shift.

As long-term shifts in labour demand are not the only source of variation in regional labour market performance (ROBSON, 2009), two more variables are introduced. First, annual growth in regional per capita value added in the productive sector (*GROWTH*) is included to capture effects of short-term fluctuations in business activity. Second, following WREN and TAYLOR (1999) and ROBSON (2009), the coefficient of absolute regional specialization (*CARS*) is added to control for possible effects of specialization/diversification in a regional employment mix:

$$CARS_{jt} = \frac{\left[\sum_{i=1}^{n} (x_{ijt} - \bar{x}_{jt})^2 / (n-1)\right]^{1/2}}{\bar{x}_{jt}}$$

where x_{ijt} is the employees in actor branch *i* in region *j* at time *t*; and *n* is the number of actor branches:

$$\bar{x}_{jt} = \sum_{i=1}^{n} x i_{jt} / n$$

CARS reflects an employment dispersion across actor branches present in a region. A higher value of CARS points to a more concentrated regional employment mix.³

As there are reasons to expect heterogeneous effects of specialization on labour market performance in regions of different size (LUNDQUIST and OLANDER, 2009), an additional estimation is performed, where CARS is interacted with dummy variables for various levels of the Swedish regional hierarchy. These levels are defined by conducting breaks analysis of a population distribution across regional labour markets. Seven hierarchy levels are then distinguished: three metropolitan areas (Stockholm, Gothenburg and Malmo) as well as large, medium-sized, small and micro-regions.⁴

Control variables include the size of regional labour markets, proxied by logged regional population (*POPULATION*), and average regional productivity (*PRODUCTIVITY*). Including the latter is motivated by findings according to which productivity level might be associated with a positive impact on employment in growing establishments so that highproductivity regions are able to create more jobs (BAR-TELSMAN and DOMS, 2000). Descriptive statistics for the variables considered in the analysis are provided in Appendix B in the Supplemental data online.

Estimation strategy

In the most general form, the estimated dynamic panel model is:

$$WFR_{jt} = \alpha_0 \sum_{k=1}^{p} \alpha_k WFR_{jt-k} + \sum_{k=1}^{p} \beta_k LI_{jt-k}$$
$$+ \sum_{k=1}^{p} \gamma_k CARS_{jt-k}$$
$$+ \sum_{k=1}^{p} \mu_k X_{jt-k} + \delta_j + \theta_t + u_{it}$$

where WFR_{jt} is the selected measure of worker flow rate; IJ_{it} is a vector of regional employment shares in leading industries; CARS_{jt} is the coefficient of absolute regional specialization; and X_{jt} is the control variables. The term δ_j accounts for regional fixed effects controlling for time-invariant unobservable regional characteristics; θ_t represents region-invariant unobservable time effects; and u_{it} captures the remaining disturbances.

A dynamic panel approach is used since it allows one to deal with (1) autocorrelation problems induced by including the lagged depending variable (making it possible to account for chain migration); and (2) simultaneity bias propelled by potential reverse causality relationships between dependent and independent variables (GARTNER, 2014). The equation is estimated by employing an instrumental system-generalized method of moments (GMM) estimator (ARELLANO and BOVER, 1995).

Such an approach requires distinguishing between exogenous and endogenous variables as they are instrumented with different lags (ROODMAN, 2009). In the specifications presented in the next section, regional shares of leading industries as well as per capita valueadded growth are treated as endogenous variables correlating with current errors. As the specification of instrumented endogenous variables and their lags adversely affects the test of over-identifying restrictions (ROODMAN, 2009), the number of instrumented lags is restricted. This allows one to deal with potentially poor performance of asymptotic results when the number of observed time periods is large (CAMERON and TRIVEDI, 2010).

RESULTS

Employment growth in leading industries

Table 1 summarizes employment growth across groups of leading industries and levels of the Swedish regional hierarchy. Growth patterns are indicated for the whole time period under consideration as well as for two sub-periods (1985–98 and 1999–2008) to demonstrate the changing roles of different groups in a long-term restructuring of the Swedish economy.

In both motive manufacturing and services (Table 1 (a)), employment growth was relatively faster in the first sub-period. Here, while growth in service branches was almost evenly spread across regions,⁵ employment generation in manufacturing was more pronounced in metropolitan areas. In the second sub-period, employment growth in manufacturing turned into a decline, which was also more visible in bigger regions (except Gothenburg). As for the service branches, employment growth continued in all but micro-regions, resulting in almost as many jobs created in the second sub-period (40723 jobs) as in the first one (41531 jobs).

Overall, between 1985 and 2008 motive branches were the only group where employment growth in

Technological Change and Geographical Reallocation of Labour

Table 1. Employment growth in motive, carrier and induced branches

(a) Motive branches

		Motive manufactur	ing		Motive services	
Annual employment growth rates	1985-98 (%)	1999-2008 (%)	1985-2008 (%)	1985-98 (%)	1999-2008 (%)	1985-2008 (%)
Stockholm	3.89	-3.26	0.78	9.99	5.58	8.07
Gothenburg	3.71	3.59	3.66	9.93	5.60	8.05
Malmo	6.75	-0.39	3.65	8.04	10.04	8.91
Large	1.23	-4.12	-1.10	11.91	3.96	8.46
Medium	1.38	0.26	0.89	9.20	3.34	6.65
Small	0.80	-0.86	0.07	8.64	5.25	7.17
Micro	1.84	0.86	1.41	21.75	-10.81	7.60
Whole Sweden	2.43	-1.72	0.62	9.92	5.37	7.94
Jobs created	31784	-18558	13226	41 531	40723	82 254

(b) Carrier branches

		Carrier manufactur	ing		Carrier services	
Annual employment growth rates	1985-98 (%)	1999-2008 (%)	1985-2008 (%)	1985-98 (%)	1999-2008 (%)	1985-2008 (%)
Stockholm	-2.50	2.22	-0.44	5.22	3.24	4.36
Gothenburg	0.39	1.35	0.81	6.46	3.95	5.37
Malmo	0.72	-0.90	0.02	6.44	5.04	5.83
Large	-0.53	0.37	-0.14	5.50	5.78	5.62
Medium	-0.63	0.27	-0.23	4.25	5.19	4.66
Small	-0.78	0.79	-0.10	6.34	5.77	6.09
Micro	-1.44	-2.09	-1.72	7.78	8.21	7.97
Whole Sweden	-0.69	0.59	-0.14	5.41	4.35	4.95
Jobs created	-24906	16049	-8857	89240	96175	185 415

(c) Induced branches

	I	Induced manufactu	ring		Induced services	
Annual employment growth rates	1985-98 (%)	1999-2008 (%)	1985-2008 (%)	1985-98 (%)	1999-2008 (%)	1985-2008 (%)
Stockholm	-2.53	-0.47	-1.63	0.68	-1.96	-0.47
Gothenburg	-0.95	-1.70	-1.28	1.44	0.20	0.90
Malmo	0.29	-3.32	-1.28	1.24	0.25	0.81
Large	-0.84	-1.79	-1.25	1.08	1.03	1.06
Medium	-1.33	-0.94	-1.16	0.74	0.81	0.77
Small	-1.12	-1.20	-1.15	1.79	1.07	1.48
Micro	-2.51	-1.28	-1.97	1.11	2.31	1.63
Whole Sweden	-1.19	-1.41	-1.29	0.95	-0.33	0.40
Jobs created	-28847	-22553	-51400	18786	-5183	13603

absolute numbers was observed in both manufacturing and services resulting in a total of 95 480 jobs created.

Carrier branches exhibit a different pattern (Table 1 (b)). In manufacturing, the first sub-period was characterized by job destruction with employment decreasing by almost 25000 jobs. The shift to increased employment in the second sub-period could not compensate for the decline in the first, resulting in almost 9000 jobs being lost over 1985–2008. In both sub-periods, metropolitan areas seem to outperform more peripheral regions.

Carrier services are characterized by fast employment growth in both sub-periods. In contrast to motive services which tended to grow faster in metropolitan areas, some convergence is observed in the second sub-period in carrier services as the latter grew somewhat faster in peripheral regions. Besides, contrary to motive services, carrier services demonstrated higher absolute employment growth in the second subperiod, potentially indicating that they intensified their development by adapting to new technological opportunities. In total, over 1985–2008 employment in carrier services increased by 185 415 workers.

Finally, induced manufacturing demonstrated negative developments in employment in both sub-periods resulting in 51400 jobs being lost (Table 1(c)). An overall increase in value added between 1985 and 2008, which classified them as one of the leading industries, came, therefore, mainly from increased productivity. Employment in induced services increased overall in the first sub-period. However, negative developments in Stockholm in the second sub-period outweighed growth in other regions to end up with 5183 jobs lost in 1999–2008. In total, over 1985–2008, increased employment in services did not compensate for the employment fall in manufacturing branches, resulting in 37797 job being lost in induced branches in total.

All in all, a total of 234241 jobs were generated in leading industries over 1985–2008, of which 281272 jobs were created in service branches and 47031 jobs were lost in manufacturing. This is of particular interest as the total employment growth in the economy over the considered time period equalled 220675 jobs. That implies that the remaining elements of the Swedish industrial structure were (in general) characterized by job destruction.

Effects of leading industries on inter-regional worker flow rates

The results of the estimation for worker in-, out- and net flow rates are presented for the whole time period (1985–2008) as well as for two sub-periods (1985–98 and 1999–2008) to capture the potentially varying effects of different groups of leading industries over time. In what follows, the regression output for each dependent variable is summarized, followed by a discussion of the results.

Worker inflow rate (WIFR)

The first striking feature in the regression output is that none of functional groups of manufacturing has a signifcant effect on the ability of regions to attract labour (Table 2). This holds for the whole time period as well as for both sub-periods. When it comes to service branches, estimated coefficients suggest that different industry groups play different roles in attracting workers to a region at various times.

In the first sub-period (models I and II), it is motive services that play the most important role in attracting workers, which underlines their early role in the transformation of the economy. There is additional (although, weak) evidence for regions having higher inflows due to wider representation of induced services. This effect, however, disappears as soon as specialization at various levels of the regional hierarchy is included (model II). When it comes to effects of specialization/ diversification in the regional employment mix, while CARS is included as it is, it shows no significant impact on WIF. However, controlling for levels of the regional hierarchy suggests that specialization might be beneficial (or, at least, less harmful) for more peripheral regions, but potentially harmful for bigger ones (negative, but insignificant coefficient for general CARS).

An interesting observation comes from considering the interplay between CARS and regional population. Model I – with no spatial differentiation in CARS – returns a strongly significant (though, surprisingly negative) coefficient for population variable. Accounting for differences across regional levels in model II makes the population variable turn insignificant. This result suggests that industrial structure of a region with respect to its specialization/diversification is a more important explanatory factor for region's ability to attracting labour than its size.

In the second sub-period (models III and IV), the roles are changing: motive and induced services no longer demonstrate a significant impact, while carrier services become highly significant and positive. These switching roles between various functional groups underlines once again the fact that outcomes of technology-induced structural change are defined by industrial dynamics. Specialization is now demonstrated to have a positive effect on the ability of regions to attract workers irrespective of position in the regional hierarchy. This is, to a great extent, predicted by theory: as new technologies are spread through the whole industrial system and developed into standardized applications, it is possible to expect more benefits coming from Marshall-Arrow-Romer (MAR)-type agglomeration externalities (LUNDQUIST and OLANDER, 2009).

The resulting pattern for the whole time period (models V and VI) is that while motive services do not have any significant effect on inflow rates, positive and significant coefficients for carrier and induced service branches imply that their stronger presence in a region increases its attractiveness for potential employees manifesting in higher WIF. No evidence for motive branches to make a region more competitive in terms of attracting labour might point to the fact that their role is confined to an initial spark in the first sub-period, while, in general, the role of economy transformers is delegated to carrier branches, which becomes particularly visible in the second sub-period. This is confirmed by the fact that over 1985–2008 there were more than twice as many jobs generated in carrier than in motive services.

The insignificant coefficient for CARS with no differentiation across regional levels (model V) suggests that it is not possible to make any reasonable conclusions regarding its effect on the WIF rate. However, introducing spatial differentiation (model VI) returns an interesting result. Large regions, in particular the metropolitan areas of Stockholm, Gothenburg and Malmo, tend to benefit more from diversification in their employment mix (negative and significant coefficient for Stockholm, with no significant differences for Gothenburg and Malmo tiers). In that respect, the urbanization externalities thesis is supported. At the same time, in smaller regions this effect is mitigated and eventually turns positive for regions at the very bottom of regional hierarchy, suggesting that small regions may benefit from becoming more specialized and forming clusters of particular industries.

1640

			Worker ir	iflow rate					Worker ou	tflow rate		
	1985	5-98	1999-	-2008	1985-	-2008	1985	86	1999-	2008	1985-	2008
Variable	-	п	Ш	IV	Λ	IV	ПЛ	IIIA	IX	x	IX	ШΧ
$Dependent_{i-1}$ Dependent2	0.49***	0.45***	0.31^{***}	0.57***	0.620^{***}	0.581^{***}	0.29*** 0.20**	0.30^{***} 0.20^{***}	***09"0	0.50***	0.552*** 0.181***	0.496*** 0.147***
Motive man ⁺	31.21	-73.33	-207.05	-224.76	19.737	16.883	134.72	217.88	5.72	123.62	110.973	114.71
Motive serv ⁺	4986.28**	4924.97**	-228.9	-132.46	242.035	270.369	-1812.73	-2905.64	756.91	637.56	-98.4	-42.854
Carrier man ⁺	93.45	-73.33	105.87	-109.65	-70.424	-88.956	73.72	104.89	-136.29	-248.66^{**}	-178.525^{***}	-205.800^{***}
Carrier serv ⁺	-748.99	-1212.49	1052.21^{***}	625.95*	190.214*	200.041^{*}	35.23	-188.44	-74.56	-151.29	-245.214^{**}	-201.902*
Induced man ⁺	69.34	30.07	74.03	-137.02	21.466	16.598	-414.71^{**}	-406.97^{**}	-151.93	-226.39^{**}	-131.827*	-71.954
Induced serv $^+_{r-1}$	1371.07*	999.3	-706.04	-90.12	570.113^{***}	576.694^{***}	647.35	-69.3	451.29*	684.56	180.394	332.955^{***}
CARS _{i-1}	32.8	-204.95	72.44***	119.89^{***}	17.017	-250.218^{**}	29.12	32.1	56.58**	66.80^{**}	48.829***	-32.275
CARS_Gothenburg, 1		80.81^{*}		12.93		188.112		0.45		46.54		45.061
CARS_Malmö _{r-1}		73.82		15.53		195.53		19.21		35.1		35.769
CARS_Large ₁₋₁		119.41^{*}		-9.58		254.051**		0.71		28.11		67.768**
CARS_Mediumr-1		113.93		-45.3		217.574**		29.71		-24.9		68.366^{**}
CARS_Small _{f-1}		163.91^{*}		-55.46		253.239**		41.43		-32.13		76.163^{**}
CARS_Micro _{f-1}		227.30^{**}		11.33		281.490^{**}		102.89		10		93.914^{*}
Population _{t-1}	-69.18^{***}	-19.61	3.16	-66.06^{**}	-29.547^{***}	-7.896	-49.10^{*}	-9.39	-66.96***	-87.41^{***}	-10.643*	-7.366
Productivity ₁₋₁	-0.14	0.03	-0.03	0.12	0.045	0.265	-0.07	-0.18	0.01	0.18^{*}	-0.113	-0.036
$\operatorname{Growth}_{r-1}^+$	33.14	8.4	-98.29	-101.21	18.502	36.792	-34.63	-23.45	83.41	23.1	-1.49	1.516
SARGAN	0.79	0.605	0.842	0.794	0.246	0.18	0.162	0.138	0.401	0.796	0.104	0.049
AR(2)	0.478	0.582	0.876	0.9	0.755	0.854	0.15	0.13	0.526	0.404	0.543	0.763
WALD	0	0	0	0	0	0	0	0	0	0	0	0
N of observations	1104	1104	920	920	2024	2024	1012	1012	920	920	1932	1932
N of instruments	91	67	85	91	296	302	88	94	85	91	290	296
<i>Notes:</i> Results correspon AR(2) is a statistic for thu	id to one-step e: e presence of au	stimates. The su to correlation in	perscript '+' ind the level equati	licates that a va- ion. WALD is a	riable is generali a test for the joii	ized method of 1 nt significance o	noments (GMN f explanatory va	 style instrume riables. All test st 	nted. SARGAI atistics are repo	N is a test for th rted as <i>p</i> -value:	te over-identifyii s. Region and ye	ng restrictions. ar dumnies as

Table 2. Estimated effects on worker inflow and outflow rates

Technological Change and Geographical Reallocation of Labour

Worker outflow rate (WOFR)

Estimating the same models for WOFRs produces quite different results, suggesting that the presence of various groups of leading industries has different effects as push and pull factors (Table 2).

In the first sub-period (models VII and VIII), induced manufacturing industries seem to contribute to retaining workers in the region. In this turbulent period of competition between old and new parts of the economy these industries might, therefore, play a role of stabilization factor. As manufacturing facilities are difficult to relocate, their ongoing functioning in a region might contribute to a lower job destruction rate, increasing the ability of a region to keep workers. Neither other functional groups of leading industries nor diversification/specialization patterns have any significant effect on WOFRs in this sub-period.

The situation changes completely in the second subperiod (models IX and X). While the models are quite unstable, there is some evidence that induced manufacturing still contributes to retaining workers. But this effect is now supported by the presence of carrier manufacturing as well. In that respect, there is a further evidence that manufacturing branches might play a stabilizing role by helping regions to retain workers.

A quite surprising observation here is that a higher share of induced services in regional employment tends to increase the WOF (positive and significant coefficient in model IX). One possible explanation is that this effect might reflect a strong connection of induced services to other parts of regional economies. LUNDQUIST and OLANDER (2009) claimed that actor branches constituting the induced services functional group had strong elements of industry-related activities. This implies that the captured effect might be indicative of developments in those sectors of the economy that are not considered in the model. Another possible explanation is a high turnover of workers in these industries, which might result in a positive impact on the outflow rate.

Regions with higher specialization in their employment mix tend to experience a higher WOF (positive and significant coefficients for CARS), irrespective of their position in the regional hierarchy. Besides, bigger regions seem to be more successful in retaining workers.

Models for the whole time period (XI and XII), in general, confirm results obtained for sub-periods: there is a negative effect from the stronger presence of induced and carrier manufacturing on the WOFRs and some evidence for a positive effect of induced services. What is different, however, is that now the effect of carrier services becomes visible: a stronger regional presence of these branches tends to decrease the outflow of workers. What is also observed is that specialization tends to increase the WOF (model XI), but this effect is more pronounced in smaller regions (model XII). When it comes to the interplay between CARS and regional size, the same pattern is observed for the whole time period as in the case of WIF models, that is, specialization/diversification in the regional employment mix seems to provide a better explanation for the WOF than a size of the region.

Net rate of employment change (NETR)

The dynamic effects of various functional groups of leading industries on gross worker flow rates are visible in models for the net worker flow rates, illustrating the outcome of worker reallocation across regions (Table 3).

In the first sub-period (models I and II), motive and induced services have a positive effect on the net flow, which is a manifestation of their role in attracting workers to a region (see inflow rate models). At the same time, induced manufacturing has a positive effect through its ability to help regions to retain labour (see outflow models). Puzzling is the significant and negative coefficient for carrier services. One possible explanation is that the sector's strong growth could be a result of outsourcing/spin-off trends from manufacturing (BAINES et al., 2009). Services included in this functional group (marketing, design, advertising, security, and research and development (R&D) services) were mainly produced in-house in manufacturing firms before, but are now outsourced to independent service units that probably employ the same labour force as before and are still located close to the old 'parent' firm. This might produce a negative effect on the WIF rates as this process is likely to take place within the region. Specialization is shown to have a negative effect irrespective of the regional hierarchy level.

In the second sub-period (models III and IV), it is carrier services that exhibit a significant and positive effect, which confirms their role in attracting labour in WIF models. This might indicate a switch from employment growth in the carrier through spin-offs/outsourcing to self-generated growth within these sectors. There is some evidence that induced services have a negative impact. Potential explanations for that were discussed when presenting results for WOF models. The second period shows a shift from a positive role of diversification to a positive role of specialization for the ability to attract a labour force.

All in all, for the whole time period (models V and VI) there is a strong positive effect coming from a presence of carrier service industries. This is not surprising as carrier services is the only industry group which over the whole time period provided a consistent positive influence on regional labour markets in terms of attracting (inflow models) and retaining (outflow models) labour. Coefficients for other functional groups of industries are not significant.

When it comes to the effect of specialization/diversification in regional employment structure, the increased degree of specialization makes a region unable to attract

			Net worke	r flow rate		
	198	5-98	1999-	-2008	1985	-2008
Variable	I	II	III	IV	V	VI
Dependent _{t-1}	-0.05	-0.06	0.31***	0.24**	0.050	0.049
Motive man+	-296.58	-335.90	-207.05	-46.22	30.126	29.747
Motive serv+	3632.52*	4199.13*	-228.90	-789.84	-37.925	-151.128
Carrier man+	-8.64	-20.64	105.87	18.14	94.345	91.440
Carrier serv $_{t-1}^{+}$	-1423.74*	-1703.77**	1052.21**	941.78**	354.100**	315.042**
Induced man+	448.20***	428.81***	74.03	38.52	121.854	115.355
Induced serv+	1336.18*	1084.17	-706.04	-987.39**	-286.345	-285.913
CARS _{t-1}	-73.84**	-91.71**	72.45**	135.28***	-31.340*	-77.777**
CARS_Gothenburg _{t-1}		32.89		-34.94		-0.520
CARS_Malmö _{t-1}		8.93		-29.44		6.052
CARS_Large _{t-1}		41.75		-81.35		-5.131
CARS_Medium _{t-1}		48.29		-75.75		12.714
CARS_Small _{t-1}		86.35		-97.38		30.309
CARS_Micro _{t-1}		107.06		-51.67		49.021
Population _{t-1}	12.67	41.38	3.16	-0.87	8.177	22.640***
Productivity,-1	0.05	0.05	-0.03	-0.09	0.014	-0.003
Growth_{t-1}^+	74.84	74.53	-98.29	-73.17	59.541	65.213
SARGAN	0.245	0.230	0.842	0.739	0.165	0.171
AR.(2)	0.140	0.127	0.876	0.928	0.451	0.450
WALD	0.000	0.000	0.000	0.000	0.000	0.000
N of observations	1104	1104	920	920	2024	2024
N of instruments	139	145	85	91	296	302

Table 3. Estimated effects on net worker flow rate

Note: See Table 2.

labour, which is reflected in negative signs of estimated coefficients for CARS. Moreover, as indicated by model II, there are no significant differences in this effect across regions of different size. Contrary to gross flows models, accounting for differences in specialization/diversification effects across regions, the effect of population remains significant and positive, implying that larger regions are beneficiaries of the inter-regional labour reallocation.

DISCUSSION

In general, the results suggest that an industrial structure of a region, and, particularly, a presence of leading industries, associated with structural change induced by the introduction of ICT, played an important role in shaping parterns of inter-regional labour mobility in Sweden since 1985. More particular results are discussed below.

First, by distinguishing between manufacturing and services within each of the leading industry functional groups, it is demonstrated that service branches seem to be a major driver behind shaping patterns of labour reallocation, while manufacturing plays a different role, mostly by helping to retain workers in a region. This suggests that only service branches had job creation rates high enough to be a factor in attracting workers into a region. This is supported by findings in Table 1 regarding a regional distribution of employment growth in leading industries. This is fully in line with the idea of the increasing service orientation of advanced economies, which is, more or less, taken for granted in the literature (PENEDER *et al.*, 2003; SCHETTKAT and YOCARINI, 2006).

Second, it is shown that there are important differences across functional groups of leading industries, that is, motive, carrier and induced branches, when it comes to their effect on attracting labour into and/or retaining labour in a region in different phases of technology-induced transformation process.

In particular, the role of motive industries, which are expected to generate the initial impulses for transition to new industrial structures, is only visible in the beginning of the adjustment process, but not in the later stages or for the whole time period. This relates to the findings of VERSPAGEN (2004) who claimed that while ICTrelated industries (which constitute the motive branches) have substantial effects on the structure of the economy, ICT is an important complement to rather than a substitute for older technologies, which continue to play an important role in the economy. Therefore, rather than being primary drivers of change, these sectors provide a growth platform for other industries as a new technological paradigm is emerging (CASTELLACCI, 2008).

In that respect, the role of major mobility-shaping actors is 'delegated' to carrier branches, which by incorporating ICT into their existing routines become the most visible and active users of new inputs. These industries are demonstrated to be the largest group when it comes to total employment as well as a group with the most jobs created over the considered time period. Therefore, it comes as no surprise that carrier branches (particularly, services) consistently generated significant effects on labour mobility across most of the model specifications. Induced branches are demonstrated to be important actors in terms of generating gross flows of workers, while their effect is not visible in the resulting net reallocation of workers across regions.

Third, specialization/diversification in the regional employment mix tends to exhibit various effects in time and across space. When comparing evidence from models for two sub-periods it is observed that specialization seems to have a negative impact in the beginning of the adjustment process, while is more positive in the later stages. It points to the fact that in the early stages of adjustment regions benefit more from diversified environment as it allows for a recombination of knowledge across old and new parts of the economy. As the new technologies are standardized and spread widely through a wider set of industries, it becomes possible for regions to benefit from more specialized structures. Models for the whole period, however, still point to the negative effect of an increased specialization in the long run.

When it comes to regional variation, increased specialization is shown to have negative effects in metropolitan areas and large regions supporting, in that respect, the urbanization externalities thesis and the importance of diversification related to Jacobs' externalities. Regions at lower levels of the regional hierarchy, however, seem to benefit more (in terms of inflow of workers), but, at the same time, lose more (increasing outflow of workers) from specialization and MAR externalities. As a result, these effects for smaller regions cancel each other out, leading to the same negative influence of specialization as in metropolitan areas, which are demonstrated in net flow models. This leaves particularly regions lower down in the hierarchy in a tricky dilemma: a way to increase inflow of labour seems to be increased specialization, but this will at the same time generate even higher outflows of workers generating a negative cumulative process of shrinking local labour markets, increasing specialization and fewer possibilities to diversify into related industries. Processes of this kind could provide an additional explanation about why small Swedish regions had such severe problems in catching up with other regions after the latest technology shift (LUNDQUIST and OLANDER, 2009).

Finally, some evidence has been obtained that an employment structure in a region plays a greater role than its size in terms of attracting/retaining workers in a region. In particular, accounting for regional differences in specialization/diversification in the whole-period gross-flow models made the regional size variable become insignificant. This might be because the interpretation of gross in- and outflow rates is qualitatively different from that for a net flow rate. Particularly, while the net rate is indicative of true labour reallocation, gross flow rates are more a measure of labour market flexibility, that is, how employers adjust their demand for labour, and employees – their labour supply (GREENAWAY *et al.*, 2000, 2002; ELLIOTT and LINDLEY, 2006). Since there is no reason to believe that it is the size of a labour market per se that has an effect on its flexibility but rather its employment mix, such an interplay between employment structure and regional population variables seems to be reasonable. Net flow models, in their turn, reflect the resulting reallocation of labour across regions. Here, even when accounting for regional differences in employment structure, a regional size variable is still significant, which may be attached to the tendency of larger regions to have higher net job-generation rates, and thus attracting more workers from outside of a region.

CONCLUSIONS

This paper aimed at analysing the role of leading industries associated with technology shifts as factors of attracting and retaining workers in regions experiencing an ongoing structural change. This was performed by revisiting and empirically refining PERE2's (1983, 2010) classification of key sectors driving technological change into motive, carrier and induced branches. It was hypothesized that a stronger regional presence of leading industries would positively affect the ability of a region to attract and retain labour.

All in all, the results provide evidence supporting the hypothesis, that is, leading industries are shown to affect WIF positively and to have a negative impact on WOFs. However, the underlining dynamics suggest that this effect is far from linear. It has been demonstrated that various functional groups of leading industries: (1) played different roles at different stages of technologyinduced transformation; and (2) differently affected the ability of regions to attract labour when compared with an ability to retain labour. Also, the role of manufacturing and service branches has been shown to be different. While the former primarily act as a stabilizing factor, helping regions to retain workers, the latter are shown to be real drivers behind worker reallocation in terms of attracting workers to a region.

In order to strengthen the claim of the importance of leading industries for local labour market dynamics, however, some other issues might be addressed in further analysis. First, the influence of leading industries might be not confined to themselves: while 234 241 new jobs were created in these branches, the total amount of inter-regional worker moves exceeded 3.4 million. It would be interesting to analyse further whether this is a multiplicative effect of leading industries establishing their position on regional labour markets or individual developments in these industries.

Apart from labour mobility across regions it is also important to analyse the role of leading industries in generating labour market dynamics within regions. Besides, there are reasons to believe that labour mobility is not only a response mechanism, which follows the development of new industries (and, thus, spreading of general purpose technologies associated with the current growth cycle), but also a mechanism that shapes and enhances these developments. In that respect, analysing cross-sectoral mobility patterns (with respect to how, at what pace and in which form do leading industries enter existing industrial structures) would make it possible to look at mechanisms facilitating the establishment and regional diffusion of a new general purpose technology as a core driver of economic growth. Finally, the role of leading industries in labour reallocation (in both regional and sectoral dimensions) might be considered from the point of view of labour market outcomes, such as income and unemployment convergence/divergence patterns as well as regional resilience.

Acknowledgements – The authors would like to thank two anonymous reviewers, whose comments and suggestions made helped greatly to improve the quality of the paper. Besides, they are grateful for the constructive advice and critical comments made on earlier versions of this paper by Mariti Henning, Frank Nerffke, Lars-Olof Olander and David Rigby. The paper also benefited from fruitful discussions conducted at the Geography of Innovation Conference held in Utrecht, the Netherlands, on 23–25 January 2014; and at the Annual Meeting of the American Association of Geographers held in Tampa, Florida, USA, on 8–12 April 2014. The usual disclaimer applies.

Disclosure statement – No potential conflict of interest was reported by the authors.

Supplemental data – Supplemental data for this article can be accessed at http://dx.doi.org/10.1080/00343404. 2015.1052062

ORCID

Mikhail Martynovich bhttp://orcid.org/0000-0002-0776-480X

NOTES

- Sensitivity checks were performed by considering individuals aged 16–64 (to exclude those in retirement age) and 20–64 (to exclude those in the 'career shopping' stage of their lives and/or are likely to relocate with their parents). As estimations based on these samples produced the same output, the broadest sample was included in the final version of the paper.
- 2. Where $N_t = E_{jt} + E_{jt-1}/2$, where E_{jt} is employment in region *j* at time *t*.
- 3. CÂRS was preferred as a measure of concentration to the more conventional Hirschman-Herfindahl index (HHI) since the latter tends to overemphasize the presence of highly concentrated industries. As the Swedish regional system includes some highly specialized regions (e.g., small regions in northerm parts of the country, where up to 70% of employment is concentrated in one industry), using the HHI might have induced estimation bias for those regions.
- 4. Using a demographic criterion for differentiating between levels of the regional hierarchy is motivated by findings in the literature demonstrating that regional size is an important predictor of regional development (ERIKSSON and HANSEN, 2013). In the case of this paper, it also appeared that regional population is highly correlated with patterns of industry penetration across regions. That is, the lower a region is in the regional hierarchy, the fewer industries tend to be represented.
- The outstanding growth rate for micro-regions (21.75% on average over 1985–98) can be explained by the low base effect because in 1985 there were only 23 employees in motive services at this level of the regional hierarchy.

REFERENCES

- AGHION P., DAVID P. A. and FORAY D. (2006) Linking Policy Research and Practice in 'STIG Systems': Many Obstades, but Some Ways Forward. SIEPR Discussion Paper No. 06–09. Stanford Institute for Economic Policy Research (SIEPR), Stanford, CA.
- ARELLANO M. and BOVER O. (1995) Another look at the instrumental variable estimation of error-components models, Journal of Econometrics 68, 29–51. doi:10.1016/0304-4076(94)01642-D

BACHMANN R. and BURDA M. C. (2010) Sectoral transformation, turbulence and labor market dynamics in Germany, German Economic Review 11, 37–59.

BADINGER H. and TONDL G. (2003) Trade, human capital and innovation: the engines of European regional growth in the 1990s, in FINGLETON B. (Ed.) European Regional Growth, pp. 215–240. Springer, New York, NY.

BAINES T. S., LIGHTFOOT H. W., BENEDETTINI O., KAY J. M., BAINES T. S., LIGHTFOOT H. W., BENEDETTINI O. and KAY J. M. (2009) The servitization of manufacturing: a review of literature and reflection on future challenges, *Journal of Manufacturing Technology Management* 20, 547–567. doi:10.1108/17410380910960984

BARRO R. J. (2001) Human capital and growth, American Economic Review 91, 12-17. doi:10.1257/aer.91.2.12

BARTELSMAN E. J. and DOMS M. (2000) Understanding productivity: lessons from longitudinal microdata, Journal of Economic Literature 38, 569–594. doi:10.1257/jel.38.3.569

BÖCKERMAN P., HÄMÄLÄINEN K. and MALIRANTA M. (2004) Sources of job and worker flows: evidence from a panel of regions, *Labour* 18, 105–129. doi:10.1111/j.1121-7081.2004.00260.x

BOSCHMA R. A. and LAMBOOY J. G. (1999) Evolutionary economics and economic geography, Journal of Evolutionary Economics 9, 411–429. doi:10.1007/s001910050089

CAMERON A. C. and TRIVEDI P. K. (2010) Microeconometrics Using Stata. Stata Press, College Station, TX.

CANTNER U. and VANNUCCINI S. (2012) A New View of General Purpose Technologies. Jena Economic Research Papers No. 2012– 054. Friedrich-Schiller-University Jena, Max-Planck-Institute of Economics, Jena.

CARLSSON B. and ELIASSON G. (2003) Industrial dynamics and endogenous growth, Industry and Innovation 10, 435–455. doi:10. 1080/1366271032000163676

CAROLEO F. E. and PASTORE F. (2010) Structural change and labour reallocation across regions: a review of the literature, in CAROLEO F. E. and PASTORE F. (Eds) *The Labour Market Impact of the EU Enlargement*, pp. 17–47. Physica HD, Heidelberg.

CASTELLACCI F. (2008) Technological paradigms, regimes and trajectories: manufacturing and service industries in a new taxonomy of sectoral patterns of innovation, *Research Policy* **37**, 978–994.

COHEN D. and SOTO M. (2007) Growth and human capital: good data, good results, Journal of Economic Growth 12, 51–76. doi:10. 1007/s10887-007-9011-5

DAVIS S. J. and HALTIWANGER J. (1995) Measuring Gross Worker and Job Flows. National Bureau of Economic Research (NBER) Working Paper Series No. 5133. NBER, Cambridge, MA.

Dost G. (1982) Technological paradigms and technological trajectories: a suggested interpretation of the determinants and directions of technical change, *Research Policy* 11, 147–162. doi:10.1016/0048-7333(82)90016-6

EDIN P.-A. and HOLMLUND B. (1995) The Swedish wage structure: the rise and fall of solidarity wage policy?, in FREEMAN R. and KATZ L. (Eds) Differences and Changes in Wage Structures, pp. 307–344. University of Chicago Press, Chicago, IL.

ELIASSON K., LINDGREN U. and WESTERLUND O. (2003) Geographical labour mobility: migration or commuting?, Regional Studies 37, 827–837. doi:10.1080/0034340032000128749

ELLIOTT R. J. and LINDLEY J. (2006) Skill specificity and labour mobility: occupational and sectoral dimensions, *Manchester School* 74, 389–413. doi:10.1111/j.1467-9957.2006.00500.x

ERIKSSON R. H. and HANSEN H. K. (2013) Industries, skills, and human capital: how does regional size affect uneven development?, Environment and Planning A 45, 593–613. doi:10.1068/a45186

ESSLETZBICHLER J. (2007) The geography of gross employment flows in British manufacturing, European Urban and Regional Studies 14, 7–26. doi:10.1177/0969776406068589

FREDRIKSSON P. (1999) The dynamics of regional labor markets and active labor market policy: Swedish evidence, Oxford Economic Papers 51, 623–648. doi:10.1093/oep/51.4.623

FUJITA M., KRUGMAN P. and VENABLES A. (1999) The Spatial Economy: Cities, Regions and International Trade. MIT Press, Cambridge, MA.

GÄRTNER S. (2014) New macroeconomic evidence on internal migration in Sweden, 1967-2003, Regional Studies, 1-17.

GREENAWAY D., HAYNES M. and MILNER C. (2002) Adjustment, employment characteristics and intra-industry trade, Weltwirtschaftliches Archiv 138, 254–276.

GREENAWAY D., UPWARD R. and WRIGHT P. (2000) Sectoral transformation and labour-market flows, Oxford Review of Economic Policy 16, 57–75. doi:10.1093/oxrep/16.3.57

HANUSCH H. and РУКА A. (2007) 'Manifesto' for comprehensive neo-Schumpeterian economics, History of Economic Ideas 15, 23–41.

JAFFE A. B., TRAJTENBERG M. and HENDERSON R. (1993) Geographic localization of knowledge spillovers as evidenced by patent citations, Quarterly Journal of Economics 108, 577–598.

KRUGMAN P. (1991) Increasing returns and economic geography, Journal of Political Economy 99, 483–499. doi:10.1086/261763

LIND D. (2014) Value Creation and Structural Change During the Third Industrial Revolution. The Swedish Economy from a Vertical Perspective. Lund Studies in Economic History No. 64. Department of Economic History, Lund University School of Economics and Management, Lund.

LOUCA F. (2007) Long waves, the pulsation of modern capitalism, in HANUSCH H. and PYKA A. (Eds) Elgar Companion to Neo-Schumpeterian Economics, pp. 800–819. Edward Elgar, Cheltenham.

LUNDQUIST K.-J. and OLANDER L.-O. (2009) Tillväxt, cykler och kriser. Nationell omvandling och regional utvecking under tre decennier. Rapporter och Notiser No. 170. Institutionen för kulturgeografi och ekonomisk geografi, Lunds Universitet, Lund.

LUNDQUIST K.-J., OLANDER L.-O. and HENNING M. (2005) Renewal and Obsolescence. An Evolutionary Perspective on Industry Growth and Regional Development in Sweden 1968–2002. Rapporter och Notiser No. 166. Institutionen för kulturgeografi och ekonomisk geografi. Lunds Universitet, Lund.

LUNDQUIST K.-J., OLANDER L.-O. and HENNING M. (2008a) Creative Destruction and Economic Welfare in Swedish Regions: Spatial Dimensions of Structural Change, Growth and Employment. SRE Discussion Paper No. 2008/03. Institute for Regional Development and Environment, Vienna.

LUNDQUIST K.-J., OLANDER L.-O. and HENNING M. S. (2008b) Decomposing the technology shift: evidence from the Swedish manufacturing sector, Tijdschrift voor Economische en Sociale Geografie 99, 145–159. doi:10.1111/j.1467-9663.2008.00450.x

METCALFE J. S., FOSTER J. and RAMLOGAN R. (2006) Adaptive economic growth, Cambridge Journal of Economics 30, 7–32. doi:10. 1093/cje/bei055

MORETTI E. (2011) Local labor markets, in DAVID C. and ORLEY A. (Eds) Handbook of Labor Economics, pp. 1237–1313. Elsevier, Amsterdam.

MORTENSEN D. T. and PISSARIDES C. A. (1998) Technological progress, job creation, and job destruction, *Review of Economic Dynamics* 1, 733–753. doi:10.1006/redy.1998.0030

NELSON R. R. and WINTER S. G. (1982) An Evolutionary Theory of Economic Change. Belknap, Cambridge, MA.

NEUMANN G. R. and TOPEL R. H. (1991) Employment risk, diversification, and unemployment, Quarterly Journal of Economics 106, 1341–1365. doi:10.2307/2937967 PASTORE F. (2013) Primum Vivere ... Industrial Change, Job Destruction and the Geographical Distribution of Unemployment. IZA Discussion Paper No. 7126. Institute for the Study of Labor (IZA), Bonn.

PENEDER M., KANIOVSKI S. and DACHS B. (2003) What follows tertiarisation? Structural change and the role of knowledge-based services, Service Industries Journal 23, 47–66. doi:10.1080/02642060412331300882

- PEREZ C. (1983) Structural change and assimilation of new technologies in the economic and social systems, Futures 15, 357–375. doi:10.1016/0016-3287(83)90050-2
- PEREZ C. (2010) Technological revolutions and techno-economic paradigms, Cambridge Journal of Economics 34, 185–202. doi:10. 1093/cje/bep051
- ROBSON M. (2009) Structural change, specialization and regional labour market performance: evidence for the UK, Applied Economics 41, 275–293.

ROODMAN D. (2009) How to do xtabond2: an introduction to difference and system GMM in Stata, Stata Journal 9, 86-136.

- ROSTOW W. W. (1960) The Stages of Economic Growth: A Non-Communist Manifesto. Cambridge University Press, Cambridge, MA. SAVIOTTI P. P. and PYKA A. (2004) Economic development by the creation of new sectors, Journal of Evolutionary Economics 14,
- 1–35. doi:10.107/s00191-003-0179-3 SCHETTKAT R. and YOCARINI L. (2006) The shift to services employment: a review of the literature, *Structural Change and Economic Dynamics* 17, 127–147. doi:10.1016/j.strucco.2005.04.002

SCHOR L. (1998) Industrial crises in a model of long cycles; Sweden in an international perspective, in MYLLYNTAUS T. (Ed.) *Economic Crises and Restructuring in History: Experiences of Small Countries*, pp. 397–413. Scripta Mercaturae, St. Katharinen.

SCHUMPETER J. A. (1939) Business Cycles: A Theoretical, Historical and Statistical Analysis of the Capitalist Process, 2 vols. McGraw-Hill, New York, NY.

SILVERBERG J. (2007) Long waves: conceptual, empirical and modelling issues, in HANUSCH H. and PYKA A. (Eds) Elgar Companion to Neo-Schumpeterian Economics, pp. 800–819. Edward Elgar, Cheltenham.

STORPER M. (2013) Keys to the City: How Economics, Institutions, Social Interaction, and Politics Shape Development. Princeton University Press, Woodstock.

THOMPSON W. R. (1990) Long waves, technological innovation, and relative decline, International Organization 44, 201–233. doi:10.1017/S0020818300035256

VERSPAGEN B. (2004) Structural change and technology: a long view, Revue Économique 55, 1099-1125.

WESTERLUND O. (1997) Employment opportunities, wages and international migration in Sweden 1970–1989, Journal of Regional Science 37, 55–73.

WESTERLUND O. (1998) Internal migration in Sweden: the effects of mobility grants and regional labour market conditions, Labour 12, 363–388. doi:10.1111/1467-9914.00072

WREN C. and TAYLOR J. (1999) Industrial restructuring and regional policy, Oxford Economic Papers 51, 487–516. doi:10.1093/oep/ 51.3.487

APPENDIX A. ACTOR BRANCHES. SUPPLY- AND DEMAND-DRIVEN INDUSTRIES

Division of manufacturing industries into actor branches is performed in three steps (LUNDQUIST et al., 2005, 2008b).

First, industry groups exhibiting similar patterns in their value added growth in different temporal phases of the growth cycle are identified. Second, industry groups are divided into subgroups based on similarities in labour productivity growth patterns. A matrix with cells corresponding to various combinations of value added and productivity growth patterns is constructed.

After that, two theoretically stylised industry groups are introduced: supply- and demand-driven industries. Supply-driven industries are the first ones to be affected by transformation as they adopt new technologies before other industries in order to create new products (thus, forming the new demand in the economy). Demand-driven industries are affected by transformation later and act as providers of raw materials and intermediate goods for supply-driven industries (therefore, expanding the new demand in the economy).

In the third stage, relative price and relative volume development for industries is used to distinguish between those within the cells/subgroups that could be assumed to be more supplydriven in their development and those that could be assumed to be more demand-driven. Finally, theoretically stylised industry groups are linked to those matrix cells that contain actual industries. Groups of industries in each cell then form the so called actor branches reflecting their assumed roles in the technology shift process. In the end, nine actor branches are identified. Their division between supply- and demand-drive group is presented below.

Supply-driven manufacturing industries	Demand-driven manufacturing industries
1. new/renewed	5. induced I
2. transformed	6. induced II
3. early followers	7. contracting
4. late followers	8. obsolete I
	9. obsolete II

For the service sector a slightly different logic is applied (LUNDQUIST et al., 2008a). In the first stage, service industries are divided into three groups based mainly on their user orientation. In the second stage, these user-oriented groups are classified into two sub-groups according to value added development (but also controlling for productivity dynamics): 'strong to medium growth'; and 'medium to weak growth'. Because of the absence of the relative price series for service industries for the time period under consideration, service industries within the sub-groups cannot be classified into supply- and demand-driven following the price/volume rationale. It is, therefore, the growth and productivity dynamics together with the user-orientation of an industry that determine whether it should be classified as supply- or demand-driven. The final classification of service industries is presented below.

Produce	r services
Supply-driven	Demand-driven
10. ICT services	14. financial and legal services
11. MAD services	15. leasing of manufacturing equipment
12. R&D laboratories	16. technical and engineering consultancy
13. security services	17. industry-related wholesale
Consumer and	general services
Mainly der	nand-driven
18. retail of food	25. transport and communication
19. supermarkets and shopping malls	26. hotels and restaurants
20. retail of consumer durables	27. cleaning, waste management, recycling
21. retail-directed wholesale	28. retail/services of cars and fuel
22. recreation, culture and sports	29. electricity, gas, water
23. other consumer services	30. construction
24. specialised retail sale	

Aggregated growth can thereby be dismantled and followed for industry groups affected by the technology shift in various ways. This classification is used as a guide for selection of those industries that could be used as indicators of the economic transformation, when it comes to estimate its impact on labour markets.

After identifying 30 actor branches, most salient industries driving the transformation of the economy were identified with respect to total value added growth patterns. Five manufacturing actor branches and eight service actor branches were then selected and further divided into three functional groups – motive, carrier and induced – following (PEREZ, 1983) classification. Division of selected actor branches across functional groups is presented in the Table below.

Leading industries	LUNDQUIST and OLANDI	LR (2009, 2010) actor branches
functional groups	Manufacturing	Services
Motive	New/renewed industries	ICT services
		MAD ¹ services
Comion	Transformed industries	R&D laboratories
Carrier	Early followers	Security services
		Technical consulting
		Financial/legal services
Induced	Induced I industries	Leasing of manufacturing
Induced	Induced II industries	equipment
		Industry-related wholesale

ading industries LUNDQUIST and OLANDER (2009, 2010) actor branches

Note: It is necessary to mention that to some extent PEREZ' (1983, 2010) functional groups are used rather as a useful metaphor than direct guidance for classification. While she considers motive branches to include only producers of cheap inputs with pervasive applicability (semiconductors in our case), in present classification these also include paradigmatic industries associated with the current structural change cycle (ICT/electronics manufacturing and ICT-services). Otherwise, the classification employed in this paper corresponds to a great extent follows that of Perez.

¹ Marketing, advertisement, and design

	Description iross worker inflow rate iross worker outflow rate it worker flow rate (net rate of employment change) hare of motive manufacturing industries in a regional employment mix hare of motive service industries in a regional employment mix hare of carrier manufacturing industries in a regional employment mix hare of induced manufacturing industries in a regional employment mix hare of induced manufacturing industries in a regional employment mix	Mean 460,4512 499,855 -39,40384 0,0358588 0,0101056 0,1663791 0,045262 0,045262 0,0918433 0,0395723	St dev. 153,7388 168,1501 109,3795 0,0489737 0,0152764 0,0152764 0,01221209 0,0356546 0,0356546 0,0240426 0,0240426	Min 172,5905 148,9822 -1533,351 -1533,351 0 0 0,0026212 0 0,0026212	Max 1381,108 1840,748 480,7586 0,355844 0,1545564 0,1545564 0,1545566 0,3396501 0,4387363 0,4387363 0,3301414
Soeffic	ient of absolute regional specialisation	1,487538	0,4821554	0,8409394	_
ogari Avera,	thm of regional population ge productivity of a worker in a productive sector I growth in per capita value added in a productive sector	10,45615 512,0748 0,0182784	1,327333 100,044 0,0993165	7,913155 157,2072 -0,5661153	14,70 1010, 1,041

APPENDIX B. VARIABLES DESCRIPTION AND STATISTICS

Article II

Lund/Göteborg, November 1, 2016

To whom it may concern,

The authors hereby certify that the paper entitled 'Labour force building in a rapidly expanding sector' is based on 60%/40% contributions by the respective authors Martynovich/Henning.

hartynard

Mikhail Martynovich

Martin Henning
Labour force building in a rapidly expanding sector

Mikhail Martynovich

Department of Human Geography, Lund University

Martin Henning

Centre for Regional Analysis/Economic Geography, Gothenburg University

Abstract: Between 1991 and 2010, employment in the knowledge-intensive IT services sector in Sweden increased from about 30,000 to 104,000 workers. Building on recent theoretical insights suggesting that the skill composition of worker inflows is an indicator of knowledge relevant to employers, we investigate the labour inflows into the sector. Who were people getting jobs in this expanding sector? And, how were their skills valued by employers as the sector evolved? Our findings suggest that the sectoral evolution was reflected not in *how the skills* of incoming workers *were valued*, but rather *who was hired* into the sector. The paper suggests that the analysis of worker inflows is a tool for investigating the evolution of sectors and their knowledge bases. It provides some lessons for industrial and educational policies regarding technologically turbulent industries, and takes the first step towards developing an approach integrating industry dynamics with labour force sourcing and evolution.

Keywords: industry dynamics; evolution; labour sourcing; labour mobility; human capital

JEL codes: J23, J24, J44, L86

1. Introduction

In 1995 five guys in southern Sweden diverged from their pastime interest in role-play gaming and started a web design consultancy business. Catching the wave of skyrocketing demand for its services, *Framtidsfabriken* (The Future Factory) expanded, nationally and internationally, to around 3,000 employees five years after its foundation. However, with the dot-com crash in 2001 fortunes changed quickly, and the share price dropped by 98% bringing the company on the verge of bankruptcy. In modified form and drastically reduced in size, it survived and eventually merged with others (Åsenlund 2001).

Framtidsfabriken's story is spectacular, but neatly illustrates the dramatic expansion of the IT services sector in the 1990s. It heralded a new digital era of development and manifested an increasingly important role of IT firms as providers of vital inputs necessary to enhance performance of firms in other sectors of the economy (Miozzo and Grimshaw 2005; Doloreux and Muller 2007). In Sweden, considering the knowledge-intensive IT services only, employment increased from about 30,000 workers in 1990 to over 104,000 in 2010. In the aftermath of the dot-com crash, sectoral employment shrank by 11 per cent, or 10,000 workers, only to return to the steady growth path afterwards. These developments were not unique to Sweden, but reflected global employment trends in the IT sector (OECD 2010).

The history, growth, international success and transformation of the Swedish IT sector has been extensively analysed from various perspectives elsewhere (e.g., Johansson 2001; Dahlin 2007; Giertz et al. 2015). To our knowledge, however, no study has systematically investigated the labour sourcing aspect of the 'boom and bust' pattern described above. Who were the people entering the IT services sector? Was it all about inexperienced enthusiasts rushing in, or did the sector draw also from the competences already existing in the economy? Who enjoyed lasting employment prospects? And did the sectoral labour market calm down eventually to become that of just any other sector?

Answers to these questions could carry important implications, going beyond the labour market dynamics of one particular sector. In the labour market literature, there is a plethora of studies on workers displaced from declining industries (Pinch and Mason 1991; Bruce C. Fallick 1996; Shuttleworth et al. 2005; Oesch and Baumann 2015). Although the complementary question of how new and rapidly expanding industries build their labour force is not

entirely novel (Freeman et al. 1982), only a few empirical studies on labour sourcing in such industries have been conducted (Bruce Chelimsky Fallick 1996; Neumark and Reed 2004; Warhurst et al. 2006; Mukkala and Tohmo 2013).

Recent theoretical arguments suggest that this should be changed. As labour flows are fundamentally an indication of skills and experiences carried to new applications (Neffke and Henning 2013), how industries build their labour force is a prime determinant of their knowledge formation as well as a manifestation of the relationship between new knowledge requirements and formerly accumulated competences. Evolving skill structures of industries may be seen as both causing and being affected by changing industrial regimes based on the generation, adaptation and diffusion of useful knowledge (Consoli and Rentocchini 2015). Therefore, an inflow of labour into an industry at any point in time is indicative of required competences, and how these are valued and put into productive use. How this plays out in surging industries is of vital importance to the renewal of the economy.

Using data on the inflows of labour into the knowledge-intensive IT services sector in Sweden between 1991 and 2010, this paper focuses on entry conditions and post-entry performance of workers in the sector addressing, in particular, the questions of:

- how the skill and experience composition among incoming staff varied as the sector evolved over time;
- how previously acquired experience and skills affected entry wages of incoming workers, their propensity to stay in the sector and wage performance, reflecting the workers' value to the sector during different phases of its development.

Even though the IT services sector was turbulent over the considered time period, the paper indicates that expectations, requirements and valuations of skills of new entrants into the sector consolidated fairly early. In that respect, evolution of the sector was reflected not so much in *how employers valued* the skills of inflowing individuals over time, but change in *who was hired*.

Perhaps most notably, higher education was becoming an increasingly important 'entrance ticket' to the sector. Also, the results emphasize the striking importance of workers educated with a main profile in the social sciences. Besides, we find strong evidence that, over time, an increasingly specialised labour market was growing around the sector, in related sectors, and that the collapse of the dot-com bubble played a particularly pronounced role for sectoral consolidation and restructuring.

The remainder of the paper is organised as follows. Section 2 summarises the research design behind the paper. In Section 3, a brief overview of the development of the Swedish knowledge-intensive IT services sector over 1991-2010 is provided, with a particular focus on the labour market aspect of its expansion. Section 4 presents the results of regression analyses. Section 5 discusses the observed labour market dynamics through the prism of the sector's evolution. Finally, Section 6 concludes.

2. Industry dynamics and labour inflows: research design

2.1. Data and sampling

The data employed in this paper comes from the Longitudinal Integration Database for Health Insurance and Labour Market Studies provided by Statistics Sweden. For each individual registered in Sweden it contains a set of personal attributes as well as a linkage to establishment of her main employment for which a type of employment, location, and industry affiliation are known.

Our sampling frame is the complete set of individuals and firms in the knowledge-intensive IT services sector. According to the Swedish Standard Industrial Classification, it comprises five 5-digit industries under the activity code 72 'computer and related activities' which we include in our study: (1) hardware consultancy, (2) software consultancy, (3) software supply, (4) data processing, and (5) database activities.¹

¹ We exclude two 5-digit industries also classified as computer and related activities, namely (1) maintenance and repair of office, accounting and computing machinery, because this industry has a different functional focus from our selected industries, and (2) other computer related activities, because this industry collects a vast number of heterogeneous activities. Besides, contrary to other 5-digit industries, the latter industry changed its classification code when the industrial classification system in Sweden changed in 2002, which would make it difficult to follow it up over the whole period that we study.

For each year between 1991 and 2010 a sample of individuals employed in the sector was derived. It was restricted to individuals in the full working age, 18-64 years old. Besides, to minimize the probability of including workers participating in short-term or project employment, we restricted the sample further to individuals who were employed directly at the moment of data collection (not just at some point over a year). All in all, 1.41 million workerfirm-year matches were identified, of which 365,552 were associated with an industry switch and were included in the final dataset.

2.2. Dependent variables and estimation strategy

Three dependent variables are included in the empirical analyses. First, we define entry conditions of an individual getting a job in the sector as her entry wage relative to the median industry wage at the year of entry:

$$wage_ind_{ij}^t = \frac{wage_{ij}^t}{med_wage_i^t}$$

where *t* is the year of entry, wage^t – yearly wage of an individual *i* in a 5-digit industry *j* at year *t*; median_wage^t – median yearly wage in a 5-digit industry *j* at year *t*. The relationship between entry conditions and individuals' backgrounds is then modelled using pooled OLS regression.

The estimation output is reported for the year of entry $(wage_ind_{ij}^t)$ and the succeeding year $(wage_ind_{ij}^{t+1})$. We do so for two reasons. First, we control for a possible situation when a large part of the registered income during the year of entry accrues from the previous job. Second, this allows us to check whether estimations for the year of entry reflect the effects of individuals' backgrounds accurately, or also include some temporary negotiation component in the recruitment process.

Second, as we are interested in the stability of job matches, we estimate the probability of individuals to remain employed in the sector three years after the entry. Formally, we define a binary variable:

$$Stay_i^{t+3} = \begin{cases} 1, \text{ if an individual is employed in the IT services sector at } t+3\\ 0, \text{ otherwise} \end{cases}$$

where *t* is the year of entry. The propensity of individuals to remain employed in the sector is then estimated as discrete-time event history models using

logistic regression. Such specification is expected to expose differential patterns of individuals' post-entry performance dependent on the overall industry performance at the moment of entry as well as observed stay/exit. While estimating this model, we additionally restrict our sample to individuals aged 18-61 at the year of entry to avoid including individuals who are likely to retire at the age of 65. One reservation needs to be made here. As it is possible that individuals deliberately self-select out of the sector – e.g., leave for in-house IT positions in other industries – estimated effects should be treated as correlations rather than causalities.

To add another perspective on the post-entry performance of individuals, we also estimate a model assessing the probability of a worker to perform better in terms of wage growth than an industry average over three years after the entry. The dependent variable is defined in the following way:

$$Wage_growth_{ij}^{t+3} = \begin{cases} 1, \text{ if } \Delta \text{wage}_{ij} \geq \Delta \text{med}_\text{wage}_{j} \\ 0, \text{ if } \Delta \text{wage}_{ij} < \Delta \text{med}_\text{wage}_{j} \end{cases}$$

where *t* is the year of entry; $\Delta wage_{ij} = \ln(wage_i^{t+3}) - \ln(wage_i^t)$; $wage_i^t - yearly$ wage of an individual *i* employed at an industry *j* at year *t*; $\Delta med_wage_j = \ln(med_wage_j^{t+3}) - \ln(med_wage_j^t)$; $med_wage_j^t - median$ yearly wage in a 5-digit industry *j* at year *t*.

As the wage performance is only observed for the subset of individuals, who are employed in the sector at t+3, estimation results may be inconsistent if staying in the sector and having a faster wage growth are not independent phenomena. Indeed, individuals with more suitable backgrounds are less likely to be systematically selected out of the sector. To correct for this selection bias, a Heckman-probit model is specified.

Formally, this model includes an outcome (wage growth) equation and a selection (stay) equation, as follows:

$$Wage_growth_{i}^{t+3} = \begin{cases} \beta_{o}I_{i} + \beta_{1}C_{i} + \varepsilon_{i}, & \text{if } y_{i}^{*} > 0\\ \text{not identified, if } y_{i}^{*} \leq 0 \end{cases}$$
$$Stay_{i}^{t+3} = \begin{cases} 1, & \text{if } y_{i}^{*} > 0\\ 0, & \text{if } y_{i}^{*} \leq 0 \end{cases}$$

where $y_i^* = \beta_o I_i + \beta_1 C_i + \beta_2 S_i + \theta_i$ represents a latent variable measuring the differential in new entrants' utility between staying or leaving the sector; I_i – vector of individuals' background characteristics; C_i – vector of control

variables; S_i – vector of selection variables.² By allowing the error terms ε_i and θ_i to be correlated, it is possible to correct for non-randomness of the sample of individuals remaining employed in the sector. While such specification doesn't allow us to control fully for self-selection of individuals out of the sector, it helps to come closer in terms of establishing causal relationships between individuals' backgrounds and their performance. As in the case of estimating models for a propensity to stay in the sector, the sample is restricted to individuals aged 18-61.

2.3. Independent variables: Operationalising individuals' backgrounds

We develop three principal background indicators operationalising knowledge, skills and experiences that individuals bring with them when becoming employed in one of IT service industries: entry mode, educational background, and regional experience.

For the entry mode, we distinguish between job-to-job, unemployment-to-job, and non-participation-to-job transitions.³ Such distinction allows tracking the value and quality of previous experiences flowing into the sector.

Individuals, employed at a year before getting a job in one of IT service industries, are registered as performing direct job-to-job switches. Such transitions are likely consequences of active decisions to participate in a new industry (March and Simon 1958). They signal, in that respect, a 'perceived desirability of movement' and reflect an attractiveness of individuals on the labour market. Given that, the industry background of individuals may also be expected to matter. Neffke and Henning (2013) argue that switching jobs is easier between industries in which activities draw upon similar skills (or, skill-related industries) as in that case individuals can re-use parts of their previously acquired human capital. In that respect, individuals performing job-to-job switches from skill-related industries will be of a higher value for destination industries than individuals switching directly from just any job.

² Our selection variable is the sectoral employment growth over three years after the entry. We expect it be positively associated with a probability of an individual to remain employed in the sector.

³ An entry into one of IT service industries is observed at a particular point in time, implying that individuals can appear several times in the data set during the period.

To detect such skill transferability, we distinguish between job-to-job entries from other IT service industries⁴ and from outside the IT services sector.

An individual is considered entering from non-participation if she was not connected to any firm by employment for three years prior to the entry, or was below full working age the year before the entry. In practice, such entries fall into two categories: young people entering labour market directly from education, and those older who for some reason have been inactive on the labour market and then get a job in the sector. To capture this distinction, we divide the group of entrants from non-participation into those older than 30 years old, and those younger than 30 years old. Although we cannot ascertain this with completely accuracy given the structure of our data, we find it likely that many of the individuals in the latter group are fresh graduates.

Individuals of working age, not categorised into the groups above, are considered entering from unemployment. Previous research has repeatedly found that spells of unemployment are detrimental to individuals' position on the labour market, particularly in the longer run. This may be related to the deterioration or loss of human capital, and the fact that employers may take previous performance as a sign of the future (Eliason and Storrie 2006). At the same time, individuals may engage in training programs during unemployment to upgrade or even re-direct their human capital, which may have a positive impact on their future performance (Kluve 2010). We believe that the industry background of individuals entering from unemployment may play a role just as in the case of direct job-to-job switches. We, therefore, distinguish between those who enter with experience in the IT services sector prior to unemployment, and those without.

Formally the entry mode variable is defined in the following way:

 $Entry_mode_{i}^{t} = \begin{cases} Fresh \ graduate, \text{ if NTJ switch, individual younger than 30 y. o.} \\ Non - participant, \text{ if NTJ switch, individual older than 30 y. o.} \\ Unrelated \ UTJ, \text{ if UTJ switch from an unrelated industry} \\ Related \ UTJ, \text{ if UTJ switch from a related industry} \\ Unrelated \ JTJ, \text{ if JTJ switch from an unrelated industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ JTJ, \text{ if JTJ switch from a related industry} \\ Related \ Relat$

where NTJ – a non-participation-to-job switch; UTJ – an unemployment-to-job switch; JTJ – a job-to-job switch.

⁴ 5-digit industries at *t*-1 and *t* should be different.

The second experience variable we construct is educational background. As a baseline, we expect a traditional positive correlation between education length and on-the-job performance (Card 1994). In young industries, in particular, demand may be strong for skills that did not exist to a large extent in the labour force before (Freeman et al. 1982). In other words, we can expect that there will be a shortage of people with formal skills directed towards those industries as it takes time for educational systems to respond to newly emerging skill demands (Vona and Consoli 2015). Therefore, firms will be prone to recruit people with a broader set of general, or 'conventional' skills (reflected by education level) in dramatic expansion phases, and will be willing to pay rather well to do so (Freeman et al. 1982). Formally, our education level variable (*Edu_level*) includes five categories: (1) pregymnasial education; (2) gymnasial education; (3) post-gymnasial education shorter than 2 years; (4) post-gymnasial education longer than 2 years; (5) doctoral education.⁵

Over time, however, adaptation of the educational system may to an increasing extent be expected to provide industries with more tailor-suited competences. Thus, apart from the education level, we also record the education track (Edu_track), where we specifically distinguish between educational background in (1) social sciences, (2) natural sciences, mathematics, data and computing, and (3) engineering and technology. Although this division does not capture all the subtleties of educational specialisation (for example, social science tracks may include some technical courses, while engineering degrees may include courses in management or marketing), it does point to a core focus of an educational program pursued by an individual.

Finally, based on the previous location affiliation of an individual's workplace (for previously employed) or municipality of residence (for previously unemployed/non-participating) we record whether getting a job in one of the IT service industries is associated with a regional move (*Regional move*).⁶ The literature points to two possible effects. On the one hand, recruiting an individual from outside the immediate surrounding may be

⁵ Post-gymnasial education is predominantly university education, but could also include other forms. Gymnasium is the highest level of secondary education in Sweden.

⁶ Here, regions are local labour markets defined by patterns of commuting between municipalities in the way that maximises homogeneity of mobility patterns within a region while sustaining cross-regional heterogeneity. This definition of a region allows distinguishing between 'real' movers and commuters in the most efficient way.

beneficial for firms in terms of obtaining knowledge and skills which are not locally available (Boschma et al. 2009). Besides, when firms are in the need of highly-skilled individuals they tend to expand their spatial search (Russo et al. 1996). From that perspective, a regional move may be associated with better employment prospects for an individual. On the other hand, moving implies that an individual has to give up some of the localised personal networks that she has accumulated, which leads to monetary and social costs (Eriksson et al. 2014). Besides, specific practices or routines develop in regions over time (e.g., Rigby and Essletzbichler (1997) demonstrated regional variations in production technologies). Therefore, a regional move may cause an individual to become 'out of place' until sufficient knowledge about regional routines is accumulated and networks are built again (Eriksson et al. 2014), which may lead to a negative impact on individual's value to employers as well as her employment prospects.

Besides the experience variables discussed above, we include a variable for the size of a firm in which an individual becomes employed as the literature reports that not only entry wages and post-entry performances (Fackler et al. 2015), but also responses to cyclical dynamics (Fort et al. 2013) are different for firms of different sizes. Also we control for the demographic variables (age and sex). Appendix A1 provides a detailed description of the variables presented above.

3. The knowledge-intensive IT services sector in Sweden

3.1. Expansion of the sector between 1991 and 2010

Over two decades between 1991 and 2010 employment in the knowledgeintensive IT services increased from 30,287 to 104,059 employees (Figure 1), corresponding to more than a threefold increase of a sector's share in the national employment (from 0.68 per cent to 2.37 per cent).

Over this period, however, employment growth patterns varied considerably. Stagnant period in the early 1990s, *1991-1993*, marks our first sub-period. Explosive growth of employment was particularly pronounced in the 1990s and culminated in 2001. We divide this into two sub-periods, *1994-1997* and

1998-2001. During this period, the expansion of small firms was essential. Characterizing the growth if IT firms in Sweden between 1993 and 1998, Johansson (2004) found that growth emanated nearly exclusively from small and medium-sized firms. Following this expansion, the dot-com crash resulted in a dramatic consolidation of the IT services sector as the number of workers in the sector reduced by more than 10,000 until 2004. Separate estimates are provided for this period. Thereafter, employment growth recovered, but demonstrated a steadier pace during the last two sub-periods, *2005-2007* and *2008-2010.* In fact, by 2007 number of employees exceeded the peak of 2001.



Figure 1. Employment in the knowledge-intensive IT services sector

Notably, the worker inflow into the sector appears to be correlated to its growth phases. Overall, the periods of rapid expansion corresponded to higher inflows (and vice versa) implying their pro-cyclical nature. Nevertheless, even in the period of contraction, there were still around 20,000 workers getting a job in the sector every year.

3.2. Regional aspects of IT services expansion

We know that some regions act as hotspots for early industry development, serving as places of attraction for labour from other regions (e.g., Mukkala

and Tohmo 2013). Large, dense regions offer advantages in terms of knowledge flows and knowledge spillovers as they combine the localisation of clusters in specific industries with industrial diversity (Karlsson et al. 2010). In particular, the literature has frequently emphasised that the role of knowledge-intensive business services is particularly pronounced in metropolitan regions (Simmie and Strambach 2006; Chadwick et al. 2008). Figure 2 clearly shows that this is the case also for Sweden.



Stockholm	Gothenburg	Malmö	Large regional centres	smaller regional centres	Periphery	Metro	Non- metro
242,77%	277,49%	397,88%	202,38%	126,85%	132,57%	265,81%	184,80%

Figure 2. Regional distribution of the knowledge-intensive IT services employment

Metropolitan areas (Stockholm, Gothenburg and Malmö)⁷ concentrated around 75 per cent of IT services employment throughout the observed time

⁷ We introduce six regional groups (or families) according to the methodology suggested by the Swedish Agency for Economic and Regional Growth (NUTEK 2004): metropolitan areas (Stockholm, Gothenburg and Malmö), large regional centres, smaller regional centres and peripheral regions. Criteria involved in the definitions of these groups include population size and density, business dynamics, share of individuals with higher education as well as access to higher education institutions.

period. This location tendency goes not only for the selection of industries reported here, but also for other ICT-related industries. For a larger selection of ICT industries, Giertz (2015) found that nearly half of all employees could be found in Stockholm.

Moreover, while Lundmark (1995) documented a tendency for spatial decentralisation of Swedish IT services employment in the 1980s, the metropolitan regions outperformed non-metropolitan areas in terms of employment growth dynamics during the period that we study. The former managed to combine internal sources of growth with attracting workers from more peripheral regions. This supports previous claims that growth in knowledge-intensive services is likely to strengthen differences across regions in favour of major metropolitan areas (Wood 2005; Corrocher and Cusmano 2012).

3.3. Labour inflows into the knowledge-intensive IT services sector

Returning to the core questions of this paper, Figure 3 summarises the distribution of worker inflows into the sector by the entry mode.



By far, the highest number of entries was performed directly from a previous job, but from outside the sector. Their share, however, decreased over time: from 66.0 per cent in 1991 to 51.6 per cent in 2010. On the contrary, the fraction of job-to-job switches within the sector increased from 15.0 per cent to 19.4 per cent respectively. Nevertheless, direct switches as a whole lost their importance in favour of other modes of entry: entries from unemployment reached 22.5 per cent of inflows in 2010 (17.5 per cent in 1991), from non-participation - 3.7 per cent (0.9 per cent in 1991), while fresh graduates increased their share from 0.5 per cent in 1991 to 2.8 per cent in 2010.

Two conclusions can be made based on these trends. On the one hand, there are indications of an increasingly dynamic labour market *within* the sector. This recalls findings of Maliranta and Nikulainen (2008) who documented worker inflows into the Finnish IT service industries and emphasized an increasingly important role of worker reallocations within the IT sector. Partly, this might be attributed to the dramatic expansion of the sector over time, which improved the availability of workers with the specialised human capital. At the same time, this might also point to the increased value that employers in the sector put on the relevant prior experience.

On the other hand, there is evidence for a diversification of sectoral recruitment strategies through employment of individuals unemployed/nonparticipating prior to the hire. This should not be interpreted as a falling demand for skilled labour possessing specialised human capital. First, unemployed individuals do not necessarily possess less skills than those performing a direct job-to-job switch. Second, even if the latter holds true, the recruitment of less skilled and/or specialised workers surged during the years of the most rapid expansion of the sector when the labour market was running scarce of relevant competences and employers in the sector were forced to expand the search for suitable workers. An increasing share of unrelated unemployment-to-job entries during the 1990s as well as its decrease in the aftermath of the dot-com crash are a clear indication of that. Third, a surge in the share of fresh graduates and non-participants may point to a growing interest of firms in the sector for individuals who can be hired at a relatively low cost and trained on-the-job, alternatively for individuals instructed in the latest applications of knowledge relevant to the sector.

With regards to the education level of workers (Figure 4), one can observe a great surge in the share of entrants with higher education⁸ over time: from 37.4 per cent in 1991 to 53.9 per cent in 2010. While this trend reflects a general tendency towards an upgrade of the education level in the Swedish workforce (for instance, the corresponding numbers for employees in service sectors (excluding IT) are 25.4 and 37.6 per cent respectively), the absolute numbers suggest that having a higher education is a particularly strong prerequisite for IT jobs. Other education level categories lost their shares considerably, pointing to an increasing role of higher education as an 'entrance ticket' to the sector.



Figure 4. Worker inflows by education level

The similar tendency is observed by Warhurst et al. (2006) who claim that a higher education, relevant experience, and computing interest have increasingly become important factors for getting advanced IT jobs. While this may be assigned to a growing competition for jobs and an increased complexity of tasks in the process of sectoral evolution, it would probably not be possible without an adaptation of the educational sector through an

⁸ Here, defined as having at least two years of post-gymnasial education (including individuals with doctoral degrees).

improved availability of specialised educational programs providing individuals with knowledge and skills relevant to the sector.

In that respect, it is interesting that education in data and computing not only fell behind education in social sciences and technology and engineering over the whole time period, but also somewhat lost its share over time: from 19.4 per cent in 1991 to 16.4 per cent in 2010 (Figure 5).



Figure 5. Worker inflows by education track

While this might seem surprising at first, one should consider the specificity of the sector with regards to services it provides. The latter involve an intensive interaction between providers of services and their clients (Strambach 2001). In order to match clients' needs, employees in the sector are required, beyond their technical tasks, to obtain an in-depth understanding of clients' business and production processes (Valenduc 2008), which increases the demand for education in social sciences and engineering. More generally, recent developments in the demand for IT workers emphasize that they should embody a mixture of technical, organisational, business, and management skills (Hawk et al. 2012). In that respect, a representation of an IT worker as a 'technical nerd' has become misleading over time, resulting among other things in a widening variety of education tracks opening doors to IT jobs.

Given the descriptive evidence presented above, an intriguing question is whether overall background profiles of incoming workers demonstrated any tendency towards convergence as the sector matured. On the one hand, evolution of the sector could lead to an increased need for broader skill portfolios as sectoral niches emerged as well as a range of possible applications widened. On the other hand, standardisation of a sectoral technological core, routinisation of tasks and a maturing educational system around the sector could increase the demand for rather narrow skill profiles of incoming individuals. To investigate this, we have calculated an entropy measure combining workers' background characteristics over time (Figure 6).⁹



Figure 6. Dispersion of incoming workers' background characteristics (3-year moving average)

Background profiles of incoming workers appeared to be narrower when the sector faced difficult times: during the Swedish banking crisis in the early 1990s, in the aftermath of the dot-com crash in the early 2000s, and, to some extent, during the outbreak of the global economic downturn in the late 2000s.

⁹ An entropy measure is calculated as a joint entropy of four characteristics: education level, education track, type of the job switch (job-to-job, unemployment-to-job, non-participation-to-job), and previous background in the sector: $H(X_1, X_2, X_3, X_4) = -\sum_{X_1} \sum_{X_2} \sum_{X_3} \sum_{X_4} p(x_1, x_2, x_3, x_4) \log p(x_1, x_2, x_3, x_4)$, where $p(x_1, x_2, x_3, x_4)$ is the observed probability of each possible combination of background characteristics. Higher values of an entropy index imply higher dispersion of background characteristics.

In these periods, firms within the sector tended to hire individuals with narrower background profiles, which points to the consolidation of their expectations regarding new employees. Alternatively, this might point to narrowing of the scope of firms' operations to those which are crucial for firms' survival as compared to times of sectoral expansion.

In that respect, while during the crisis periods the number of individuals getting jobs in the sector still remained high (see Figure 1), they constituted a more homogeneous group than during the boom years. During the latter, however, firms might be expected to generate higher profits, making them care less about cost minimisation and be able to hire less experienced individuals with broader skill profiles to provide them with the relevant training and experience at the workplace.

4. Regression analysis

4.1. Entry conditions

Table 1 reports the results of estimating pooled OLS regressions relating entry conditions of individuals becoming employed in the sector to their background characteristics. Appendix A2 provides a sample summary as well as a correlation table for variables included in the analysis.

One striking feature of the estimation results is that, overall, the relationship between various aspects of workers' backgrounds and their entry wage remained fairly stable over time. Nevertheless, there are some notable changes which are discussed below.

We find a well-documented positive association between the education level of a worker and her wage. There are, however, two qualifications to that result. First, the education premium was decreasing along the evolution of the sector as a substantial compression of coefficients' magnitude is observed in later sub-periods. A possible explanation to that is an increased availability of workers possessing specialised human capital over time. Indeed, during the early stages of sectoral expansion one can expect a limited supply of workers with relevant experience as compared to the demand for them. In such situation, education level of a worker is a signal about her ability level to a potential employer, which would then be ready and willing to offer higher

Wage index;	1991.	-2010	1991-	-1993	1994-	1997	1998.	-2001	2002-	2004	2005-	2007	2008-	2010
Pooled OLS	Wage_i _t	Wage_i+1	Wage_i,	Wage_i++1	Wage_i _t	Wage_i _{t+1}								
Edu_level (pre-gym)														
Gym	0.0354***	0.0513***	0.0817***	0.0847***	0.0642***	0.0799***	0.0327***	0.0337***	0.0198*	0.0426***	0.0074	0.0215	-0.0009	0.0108
After-gym (<2 years)	0.1037***	0.1434***	0.1085***	0.1408***	0.0925***	0.1565***	0.1086***	0.1288***	0.1241***	0.1539***	0.0781***	0.1006***	0.0602***	0.0686***
After-gym (≥2 years)	0.1831***	0.2179***	0.2293***	0.2627***	0.2297****	0.2563***	0.2179***	0.2225***	0.1839***	0.2409***	0.1199***	0.1477***	0.0863***	0.1070***
Doctoral	0.2785***	0.3268***	0.3764***	0.4449***	0.3699***	0.3611***	0.3156***	0.3051***	0.3340***	0.3807***	0.1443***	0.2663***	0.1634***	0.2020***
Edu_track (other)														
Social science	0.0863***	0.0883***	0.0790***	0.0715***	0.0643***	0.0738***	0.0917***	0.0928***	0.0827***	0.0837***	0.0974***	0.1149***	0.1093***	0.1041***
Natural science, data	-0.0078*	-0.0047	0.0380***	0.0013	0.0100	0.0148	0.0017	0.0249***	-0.0169*	-0.0278***	-0.0242**	-0.0205*	-0.0013	-0.0157
Technology	0.0666***	0.0567***	0.0690***	0.0581***	0.0652***	0.0570***	0.0733***	0.0719***	0.0530***	0.0467***	0.0589***	0.0561***	0.0800***	0.0495***
Entry (unrelated JTJ)														
Fresh graduate	-0.2855***	-0.1065***	-0.2678***	-0.1327*	-0.3075***	-0.1400***	-0.3318***	-0.1316***	-0.2627***	-0.0815***	-0.2541***	-0.0542***	-0.2344***	-0.0749***
Non-participant	-0.4913***	-0.2829***	-0.4728***	-0.2518***	-0.5120***	-0.3177***	-0.6027***	-0.3656***	-0.3828***	-0.2211***	-0.4488***	-0.2197***	-0.4506***	-0.2401***
Unrelated UTJ	-0.3182***	-0.2001***	-0.2498***	-0.1632***	-0.2421***	-0.1782***	-0.3060***	-0.1918***	-0.3815***	-0.2325***	-0.3476***	-0.2053***	-0.3271***	-0.2061***
Related UTJ	-0.3657***	-0.1626***	-0.2928***	-0.1443***	-0.3551***	-0.1599***	-0.4101***	-0.1832***	-0.3632***	-0.1807***	-0.3386***	-0.1325***	-0.3184***	-0.1325***
Related JTJ	0.0931***	***6690.0	0.0832***	0.0327**	0.0931***	0.0636***	0.0716***	0.0456***	0.0615***	0.0613***	0.1432***	0.1113***	0.0989***	0.0685***
Regional move	0.0293***	0.0432***	-0.0128	0.0595***	0.0051	0.0330***	0.0330***	0.0434***	0.0626***	0.0624***	0.0355***	0.0256*	0.0307**	0.0414***
Reg. family (large reg.														
Stockholm	0.1939***	0.2110***	0.1236***	0.1591***	0.1500***	0.1695***	0.2192***	0.2225***	0.2286***	0.2483***	0.2028***	0.2171***	0.1794***	0.2130***
Gothenburg	0.0557***	0.0637***	-0.0125	-0.0179	0.0582***	0.0684***	0.0684***	0.0723***	0.0714***	0.1009***	0.0868***	0.0643***	0.0304***	0.0490***
Malmö	0.0582***	0.0667***	-0.0115	-0.0045	0.0040	0.0120	0.0419***	0.0319***	0.0858***	0.1169***	0.0890***	0.0790***	0.0791***	0.1168***
Smaller reg. centres	-0.0491***	-0.0673***	-0.0494***	-0.0670***	-0.0442***	-0.0855***	-0.0633***	-0.0803***	-0.0109	0.0048	-0.0855***	-0.1170***	-0.0625***	-0.0809***
Periphery	-0.0349***	-0.0583***	-0.0856***	-0.1045***	-0.0659***	-0.0435*	-0.0373***	-0.0758***	0.0192	-0.0094	-0.0230	-0.0671***	-0.0423*	-0.0305***
Reg. move*Reg. family (large reg. centres)														
Stockholm	-0.1421***	-0.0953***	-0.0759***	-0.0851***	-0.0961***	-0.0636***	-0.1426***	-0.0881***	-0.1970***	-0.1489***	-0.1468***	-0.0885***	-0.1481***	-0.1065***
Gothenburg	-0.0087	-0.0038	0.0944*	0.0019	0.0066	0.0156	-0.0139	-0.0075	-0.0100	0.0066	-0.0628***	-0.0252	0.0001	-0.0137
Malmö	0.0051	0.0088	0.1508***	0.0522	0.0300	0.0213	0.0264	0.0389**	-0.0416	-0.0340	-0.0100	0.0292	-0.0259	-0.0608*
Smaller reg. centre	0.0068	0.0168	0.0589	0.0313	0.0158	0.0304	0.0061	0.0151	-0.0332	-0.0372	0.0539*	0.0623**	-0.0160	0.0272
Periphery	-0.0243*	-0.0202	-0.0521	-0.0496	0.0309	0.0360	-0.0218	-0.0262	-0.0853***	-0.0536*	-0.0177	-0.0107	-0.0140	-0.0208
Firm size (2-10 emp)														
1 emp	-0.4174***	-0.5027***	-0.3300***	-0.3540***	-0.3755***	-0.4533***	-0.4091***	-0.5066***	-0.4353***	-0.5199***	-0.4284***	-0.5284***	-0.4131***	-0.4936***
> 10 emp	0.1535***	0.1724***	0.1092***	0.1070***	0.1330***	0.1615***	0.1139***	0.1354***	0.2121***	0.2386***	0.1876***	0.1857***	0.1717***	0.2024***
R ²	0.3584	0.3027	0.4109	0.2976	0.4372	0.3287	0.3465	0.2894	0.3464	0.3326	0.3612	0.2919	0.3493	0.3166
z	351982	256832	18633	13895	51828	41595	103637	76560	54242	41499	67728	52704	55914	30579
Note: Demographic contr ***/** */ indicate circuition	rols (sex and	age), industr	y and year du	immies as we	il as a consta.	nt are include	ed in the mod	els but not rep	oorted. Stand	ard errors are	robust and c	lustered at th	e individual le	evel.
(,) indicate significat	ICE BL 0. 1% (170, 370) IEVE	_											

Table 1. Workers' backgrounds and entry conditions

wages to those with higher education level. With the development of a more specialised labour market and educational system around the sector, education level *per se* might be expected to lose in importance in comparison to other background characteristics of workers, such as education specialisation and relevant work experience. This, in its turn, would have a downward pressure on the wage premium from education level.

Second, in the aftermath of the dot-com crash the education premium remains observable only for individuals with at least some education at the postgymnasial level. This supports the claim made above that higher education was becoming an 'entrance ticket' to good (at least, better paid) jobs in the sector.

Another interesting observation is that in all time periods the magnitude of coefficients for education level increases somewhat a year after the entry. One implication is that the association between entry conditions and education level remains quite persistent, that is, individuals with higher levels of education not only enjoy better entry conditions, but also faster wage growth, at least over the first year of employment.

With regards to the education track, it is the background in social sciences which tends to yield the highest wage premium in entry wage closely followed by that in engineering and technology. Surprisingly, education in natural science, computing and data did not have any significant impact on entry conditions in most time periods. What might be behind this is the relative ease with which general skills in programming can be obtained (Østergaard and Dalum 2011) and a rapid obsolescence of technical IT skills (Prabhakar et al. 2005). This would make it less strategically valuable to hire workers with purely technical education, and increase the value of education tracks which include complementary applied skills. Moreover, the premium for education in social sciences and engineering and technology tends to increase over time, which supports the explanation provided for compression of coefficients for education level.

The association between entry conditions and the entry mode is very consistent over the whole time period. The best entry conditions are, unsurprisingly, experienced by individuals performing a direct switch from a job in another IT service industry. These are closely followed individuals performing a job-to-job switch from outside the sector. Among the rest, it is fresh graduates who might expect better entry conditions. Thereafter follow entrants from unemployment with only minor difference between those with and without experience in the sector.

It is interesting that performing an unrelated job-to-job switch tends to yield better entry conditions than having employment history in IT services but being unemployed before getting a new job. This might have two, potentially complementary, explanations. On the one hand, employers in the sector may consider unemployment periods detrimental to individuals' human capital and, thus, to their potential performance. On the other hand, individuals performing a direct switch, even from outside the sector, might have a better position for negotiating entry wages while previously unemployed individuals might tend to accept lower ones.

Another peculiar observation is that, contrary to education level and track, the magnitude of coefficients for various entry modes reduces considerably a year after the entry, which feasibly points to a faster wage growth for those with worse entry conditions. Potential explanation is a very fast accumulation of specialised human capital through on-the-job learning process. If the latter claim is correct, the importance of general human capital (reflected by education level and track) would be more persistent than that of having relevant work experience.

Individuals becoming employed in metropolitan areas (Stockholm, Gothenburg and Malmö), unsurprisingly, may expect higher entry wages than individuals getting jobs in more peripheral regions, which is predicted by the literature on the urban wage premium (Glaeser and Maré 2001; Yankow 2006). This wage premium increases somewhat over time, peaking in the aftermath of the dot-com crash.

At the same time, there is a difference between metropolitan areas with regards to entry conditions for individuals, for whom a job switch is associated with a regional move. In particular, individuals relocating to Stockholm tend to find themselves in a worse position than those performing a local job switch. Potential explanation to that is that incoming workers may not be acquainted with a relevant wages level in Stockholm, which pushes their expectations lower than those of local workers. This is partly confirmed by the fact that the year after entry the difference between local entrants and region switchers is somewhat mitigated (but does not completely disappear).

Individuals performing a regional move to regions other than Stockholm appear to enjoy a slight wage premium over local entrants, which is not significantly different between secondary metropolitan areas of Sweden and more peripheral regions. Possible explanation to that is that the entry conditions of regional switchers are pushed up by individuals relocating from Stockholm. Indeed, once the regional movers from Stockholm are excluded, the association between entry conditions and regional relocation becomes insignificant.¹⁰

Finally, larger firms tend to provide better entry conditions. This tendency is more pronounced in the aftermath of the dot-com crash. Particularly in the period of sector's contraction, one can expect the reallocation of workers to the most productive and resistant firms, with a higher probability of survival, which are likely to be the largest firms in the sector.

4.2. Probability to stay in the sector

Table 2 reports the estimation output of logistic regressions relating individuals' probability to remain employed in the sector three years after the entry to their background characteristics. Appendix A3 provides a sample summary as well as a correlation table for variables included in the analysis.

With exception of individuals with a doctoral degree, higher level of education tends to be positively associated with the propensity of individuals to stay employed in the sector when the whole time period is concerned. There is, however, a huge variation in the coefficients for different subperiods. In the periods, when the stay/exit is observed in the times of rapid expansion (entry in 1991-1993 and 1994-1997), there is no systematic tendency in the coefficients. However, when the stay/exit is observed during and in the aftermath of the dot-com crash (remaining sub-periods), importance of education tends to be more pronounced. As in the case with entry conditions, by the end of the considered time period it is only higher education which tends to yield a higher probability to stay in the sector. This puts another perspective on the importance of education level as a background characteristic.

Education track tends to be related with the propensity of individuals to remain employed in the sector. However, the observed magnitude of coefficients is completely reversed compared to entry conditions regressions.

¹⁰ These models are not reported but are available from the authors upon request.

regression	1991-2010	1991-1993	1994-1997	1998-2001	2002-2004	2005-2007
Edu_level (pre-gym)						
Gym	0.0758***	0.0258	-0.0124	0.1566***	0.0617	0.1054
After-gym (<2 years)	0.2171***	0.2001*	0.1140*	0.3034***	0.2338***	0.1015
After-gym (≥2 years)	0.2827***	0.2270**	0.0914	0.3356***	0.3529***	0.3389***
Doctoral	0.0111	0.1805	-0.1713	0.0804	0.0240	0.0689
Edu_track (other)						
Social science	0.0667***	-0.0689	0.0839**	0.0986***	0.0242	0.0202
Natural science, data	0.5917***	0.6728***	0.6307***	0.5881***	0.4770***	0.5762***
Technology	0.2999***	0.1923***	0.2318***	0.3121***	0.2632***	0.4386***
Entry (unrelated JTJ)						
Fresh graduate	-0.0398	-0.2435	-0.0771	-0.1126	0.1083	0.0106
Non-participant	-0.2410***	-0.4784*	-0.2556**	-0.3611***	-0.1572	-0.0916
Unrelated UTJ	0.1171***	0.0224	0.1437***	0.1006***	0.1127**	0.0568
Related UTJ	-0.0693***	-0.1310**	-0.0762**	-0.0819***	-0.1032***	-0.0696*
Related JTJ	0.6025***	0.4132***	0.5011***	0.4810***	0.6694***	0.7119***
Regional move	-0.0476*	-0.1157	-0.0894*	0.0916**	-0.2209***	-0.3161***
Reg. family (large reg.						
Stockholm	0.0087	-0.0251	-0.2188***	0.1639***	-0.1128***	-0.1536***
Gothenburg	0.2152***	0.0880	0.0588	0.2580***	0.3540***	-0.1029*
Malmö	0.0901***	-0.2570***	-0.1003*	0.3170***	0.0996***	-0.1994***
Smaller reg. centres	0.0691**	0.1149	0.0570	0.3715***	-0.3629***	0.0897
Periphery	-0.1790***	-0.0492	-0.1663	-0.0043	-0.1299	-0.6949***
Reg. move*Reg. family (large reg. centres)						
Stockholm	-0.0641*	0.0302	0.0348	-0.2432***	0.1100	0.1974**
Gothenburg	-0.1369***	-0.3156*	-0.1078	-0.1638**	-0.1688	0.1903*
Malmö	-0.1229***	-0.1745	-0.0095	-0.3839***	0.0821	0.3684***
Smaller reg. centre	-0.1321**	-0.3437	-0.2165*	-0.3443***	0.2010	-0.0413
Periphery	-0.2194***	-0.9347***	-0.0868	-0.3576***	-0.0792	0.2508
Firm size (2-10 emp)						
1 emp	0.0327	0.0995	0.0745	0.1284***	-0.0324	-0.0861*
> 10 emp	0.2114***	0.3440***	0.3699***	0.1666***	0.0933***	0.2430***
Pseudo-R ²	0.0491	0.0512	0.0393	0.0364	0.0578	0.0639
z	294020	18514	51605	103185	53812	47788
Note: Demographic controls (sex a ***(**,*) indicate significance at 0.1	and age), industry and % (1%, 5%) level	year dummies as well as a cor	stant are included in the mod	els but not reported. Standard	errors are robust and clustered	I at the individual level.

in the sector	
o stay	•
pensity t	
d pro	
unds and	
backgro)
Workers'	
Table 2.	

That is, having more technical education is strongly and positively associated with the probability of staying (being the highest for individuals with education in natural sciences, data and computing). Coefficients for education in social sciences are quite unstable in different time periods. One possible explanation is that individuals with education in social sciences (and, to some extent, engineering) perform tasks and develop skills which are easier to transfer to other sectors, while individuals with education in natural sciences, data and computing have a higher tendency to be 'locked-in' in the sector.

Also, entry mode co-varies extensively with the probability to remain employed in the sector. For instance, it is the highest for individuals performing related job-to-job moves followed by individuals performing unrelated direct switches and fresh graduates. The lowest propensity is observed for entrants from non-participation. Quite surprisingly, individuals performing unrelated unemployment-to-job switches are more likely to stay in the sector than those entering from unemployment with their last job in another IT service industry. Possible explanation is that the latter make several attempts to find a job in the sector, and, failing, move on to jobs elsewhere.

There is, however, an interesting time dynamics. First, the tendency for individuals performing related job-to-job switches to stay longer in the sector becomes stronger over time, pointing to the increased stability of job matches for this category of workers. Second, the difference between other groups of workers becomes less pronounced with the evolution of the sector, virtually disappearing in the latest observed sub-period.

Regional relocation is negatively associated with the propensity to stay in the sector in all regions when the whole time period is taken into consideration. With exception of large regional centres, this tendency is particularly pronounced for individuals, for whom stay/exit is observed right in the aftermath of the dot-com crash. One possible explanation to that is that relocating individuals might find it difficult to fit into labour markets of destination regions. In that respect, it is interesting that probability of remaining employed in the sector is in most sub-periods the lowest in the most peripheral regions. Another possible reason is that some workers could use a job in the sector as the 'entry platform' to a new region (as there would be lots of job opportunities due to sectoral expansion), being ready to move to jobs in other sectors as they become more embedded in regional labour markets.

With regards to the firm size, the largest probability to stay in the sector is observed for individuals becoming employed in large firms. This tendency, however, becomes weaker during and after the dot-com crash. What might be behind this is that, as the sector matured, its labour market was becoming less turbulent (reinforced by the consolidation of the sector in the aftermath of the dot-com crash), leading to the mitigation of differences between firms of different sizes in terms of providing stable employment opportunities. While it is only a speculation at this point, the dot-com crash may also have put severe pressure on relatively large firms in the sectors that we study, just as it did in other ICT sectors in Sweden. For example, Ericsson had over 40,000 employees in Sweden in 2000, decreasing to about 21,000 in 2004 (Giertz 2015).

4.3. Wage growth after the entry

Table 3 reports the results of estimating the Heckman-probit selection model for the wage performance of individuals provided that they stay employed in the sector three years after the entry. Appendix A4 provides a sample summary as well as a correlation table for variables included in the analysis. Selection equation estimates are reported in Appendix A5.

As in the models discussed above, education level has a positive and strictly hierarchical association with a probability of individuals' wage growth above industry average over the whole time period. Education premium with respect to wage growth is particularly pronounced in years of sectoral expansion (1994-1997 and 2005-2007). What's interesting here, while individuals with doctoral education do not exhibit any outstanding pattern in propensity to remain employed in the sector, they are the ones to expect the fastest wage growth provided that they stay.

Individuals with social sciences education background appear to not only enjoy the best entry conditions, but may also expect the highest wage growth when the whole time period is considered. This tendency is particularly observed after the dot-com crash. Individuals with more technical education perform worse. This points once again to the importance of 'soft' skills for performance in the sector. Besides, as individuals with technical education might have a higher risk to be 'locked-in' in the sector (as suggested by the propensity to stay regressions) they might also be in a relatively worse position when it comes to wage negotiations.

Wage growth; Heckman probit	1991-2010	1991-1993	1994-1997	1998-2001	2002-2004	2005-2007
Edu_level (pre-gym)						
Gym	0.0741***	-0.0875	0.0836*	0.0505	0.0454	0.0991*
After-gym (<2 years)	0.1221***	-0.0148	0.2341***	0.0921*	-0.0430	0.0865
After-gym (≥2 years)	0.1239***	0.0459	0.1511***	0.0646	0.0614	0.1141*
Doctoral	0.2323***	0.0320	0.3043**	0.1582**	0.2068**	0.2279**
Edu_track (other)						
Social science	0.0309**	-0.0087	0.0111	0.0336	0.0113	0.0706**
Natural science, data	-0.0214	-0.0493	0.0170	-0.0436	-0.1301***	-0.0054
Technology	-0.0343**	0.0668	-0.0286	-0.0576**	-0.0491*	-0.0735**
Entry (unrelated JTJ)						
Fresh graduate	0.4255***	0.0453	0.2103*	0.6891***	0.0581	0.4556***
Non-participant	0.4450***	0.7120**	0.4457***	0.7200***	0.0514	0.4849***
Unrelated UTJ	0.2938***	0.2286***	0.2230***	0.2853***	0.3266***	0.3075***
Related UTJ	0.5766***	0.3234***	0.4783***	0.6719***	0.4364***	0.6071***
Related JTJ	-0.1117***	-0.0770*	-0.0881***	-0.1228***	-0.1573***	-0.1818***
Regional move	-0.0012	0.0967	0.0473	0.0232	0.0062	-0.0491
Reg. family (large reg. centres)						
Stockholm	-0.0237*	-0.0466	0.0316	-0.0531**	-0.0296	-0.0132
Gothenburg	-0.0570***	-0.0212	-0.0630*	-0.0787***	-0.0666*	-0.0614*
Malmö	-0.0286*	0.0074	0.0267	-0.0136	0.0086	-0.1080***
Smaller reg. centres	-0.0637**	-0.1101	-0.2029***	0.0006	0.0842	-0.0945*
Periphery	-0.0440	-0.0513	0.0320	-0.0010	-0.1466*	0.1510
Reg. move*Reg. family (large reg. centres)						
Stockholm	0.1746***	0.1979*	0.1343*	0.1624***	0.1385**	0.1613**
Gothenburg	0.0876**	-0.0309	0.0445	0.0478	0.0492	0.1683*
Malmö	0.0611	-0.0462	-0.0377	-0.0198	0.0391	0.1464*
Smaller reg. centre	0.0316	0.0023	0.0550	-0.0001	0.0792	-0.0618
Periphery	0.0609	0.0748	-0.0717	0.0264	0.1941	-0.1169
Firm size (2-10 emp)						
1 emp	-0.7355***	-0.5241***	-0.7359***	-0.6747***	-0.7155***	-0.7191***
> 10 emp	0.1049***	0.1283***	0.1678***	0.1380***	0.0107	0.0469*
Prob. (p=0)	0,0000	0,0055	0,0000	0,0000	0,0001	0,0000
N censored	128167	7942	19978	51407	22872	17324
N uncensored	165853	10572	31627	51778	30940	30464
Note: p≠0 indicates that are included in the model	selection and outcome equatic s but not reported. Standard er	ns are dependent (i.e., that He rors are robust and clustered a	ckman specification is correct) it the individual level. ***(**,*) ii	 Demographic controls (sex a ndicate significance at 0.1% (1 	nd age), industry and year dum %, 5%) level	imies as well as a constant

performance
wage
oost-entry
1d p
unds ai
backgro
^
rkers
0
5
\geq
e.
Ĭ
ą
<u>_</u>

With regards to the entry mode, it is somewhat surprising that individuals performing a direct switch from employment within the sector are less likely to enjoy a better-than-industry-average performance when compared to other modes of entry. Possible explanation is that these workers have the best negotiation position for the entry wage, so that a better performance of individuals with other modes of entry may just be a realisation of the 'lowbase' effect. If the latter is true, then it comes as no surprise that while initial wage premium for direct switchers increases over time, their post-entry performance in terms of wage growth becomes worse over time.

While performing a regional move is negatively associated with a probability to remain employed in the sector three years after entry, it has no significant impact on individuals' wage performance. There are, however, regional differences. That is, individuals moving to and remaining employed in Stockholm (in the later sub-periods, also Gothenburg and Malmö) enjoy a significant wage growth premium.

Employees in large firms tend to have the best wage performance as compared to medium-sized firms and individual consultancies. However, this wage growth premium decreases over time, pointing once again to the increasing importance of medium-sized firms in the sector.

5. Discussion

Due to the international success of the Swedish ICT sector, especially in the 1990s, previous investigations covered a range of issues concerning its historic growth, firm structure, and innovation system characteristics over time. Yet, to our best knowledge, no studies have taken an explicit interest in the sources of labour supply for these expanding activities. This is not particular to Sweden, or to the IT industries. In the international literature, few efforts have been devoted to a fruitful integration of the literatures on industry dynamics and labour market dynamics (Mamede 2009).

We think, however, there are a couple of reasons to pursue this integration. From the perspective of industry dynamics, economic sectors could be regarded as constantly evolving bundles of activities, execution of which entails the application of specific knowledge. Sectoral evolution is, therefore, necessarily shaped by the dynamics of knowledge creation (Krafft et al. 2014). This knowledge is brought into and formed within industries by

employees. Changes in the workforce composition of an industry over time is, in that respect, a reflection of its evolution. The study of labour sourcing processes may provide insights into how various types of knowledge are formed, combined and valued in an industry over time, also in particular localities. It may also shed light on relationships with and knowledge diffusion between other related industries (Neffke and Henning 2013), indicating how independent expanding industries are, or not, from previously established structures in the economy.

From the perspective of the labour market dynamics, employment structures and worker turnover can be considered as coordinating devices which ensure a coherence between what is required from the workforce, and the pool of skills and capabilities that are available in the economy. The skill composition of a workforce is an indicator of knowledge relevant to employers in an industry at any point in time (Consoli and Rentocchini 2015). Understanding the labour sourcing processes over time may, therefore, provide insights into which types of labour demand, as well as opportunities for labour, are created in industries at different stages of maturity. Such understanding may then inform training initiatives, education policies and human capital policies, aiming at promoting growth and development of both new and maturing sectors.

We have observed several tendencies in the labour sourcing strategies of employers in the knowledge-intensive IT services sector in Sweden, which may be manifestations of evolution of the sector as a whole, as well as its demand for labour in particular. First, dynamics of worker reallocation around the sector suggest the evolution of an increasingly specialised structure over the period that we study. In other words, the more mature the sector turns, the more it relies on the internal labour market dynamics. In particular, individuals moving into an IT services job from a related industry tend to increase their share among labour inflows, enjoy better entry conditions as well as more lasting employment prospects throughout the whole period under consideration. In that respect, one can point out that the knowledge relevant to an industry is not only tenured in the industry itself, but also in other, related, industries. This provides important qualifications to recent claims that industries cannot be meaningfully studied as isolated units. Rather, their evolution is shaped in a mutual exchange with related industries (Frenken and Boschma 2007; Neffke et al. 2011). But while people moving from closely related sectors are attractive and initially rewarded, they quickly need to adapt to new circumstances and contexts to sustain their perishable

mover advantages. The benefits accruing from work experience in related industries vanish rather quickly, to the benefit of experience (on-the-job training) and formal education.

Second, the results strongly suggest that higher education becomes an increasingly important 'entrance ticket' to jobs in the IT service industries. In that respect, job opportunities become increasingly restricted over time (Warhurst et al. 2006), at least for 'better jobs' in the sector. While this is a tendency which the IT services sector shares with the Swedish labour market in general, this is also in line with Østergaard and Dalum (2011) who argue that even if the human capital entry barriers have been low historically for IT, they seem to have increased. Higher education level also tends to be associated with more lasting employment prospects in the sector as well as better post-entry wage performance. Even though we are inclined to argue that the growth of specialised IT education tracks in Swedish higher education institutions has contributed to pushing the 'entrance ticket' upwards, we cannot rule out that fiercer competition for IT service jobs was behind this. Nevertheless, evolution of the sector seems instead to lead to an increasing need for, and in particular reward to, the most educated workers.

Third, complementary to the latter finding, our investigations also indicate that qualified industry experience, or even industry-specific skills (Neffke and Henning 2013), of great value to the sector can be attained by people with very diverse educational backgrounds. Sectoral core competences seem not to be embedded in one dominant educational or occupational group, but are rather distributed and made operational across a range of skill backgrounds in combination. An example in our investigation is the striking importance of workers educated with a main core in social sciences. We believe that this reflects the broad scope of competences needed in today's advanced service industries. These workers are also highly appreciated in the sector, and the status of their work is recognised in the wage. However, they seem to have a lower tendency to stay on, compared to workers with specialised technical backgrounds. The latter rather seem to become 'locked-in' in their existing positions and conditions.

Fourth, one of our most interesting findings is how sectoral crises, in our case the dot-com crash, become important periods for both building linkages to other industries (Eriksson and Boschma 2013), and general sectoral consolidation and restructuring. During severe crises, sectoral labour dynamics is substantial but of a very different character. That is, while labour inflows into industries remain rather high, the group of workers getting jobs in the sector in these periods is more homogeneous than in times of rapid expansion. The implication is that dedicated specialists have an easier time in periods of crisis, and the ones hurt are those who do not fit the narrower skill profile. In other words, crises are periods of an increased consolidation of sectoral activities as reflected, among other things, in the sectoral demand for labour. Because crises also seem to be dynamic periods, we would find it interesting to further study how labour reallocation during crisis affects industry evolution in the longer run.

Finally, labour sourcing processes have important geographical components. The hotspot for the knowledge-intensive IT services sector in Sweden has always been the capital region of Stockholm, which not only concentrates around half of the sectoral employment but also hosts employers providing the best conditions for their workers in terms of wages. Stockholm's particular strength is also reflected in the wage benefits for workers relocating to Stockholm from other regions. With time, two other Swedish metropolitan areas (Gothenburg and Malmö) caught up on Stockholm to demonstrate the similar developments. In that respect, it is interesting that different types of spatial entries are differently associated the entry conditions and post-entry performance of workers in the sector. While for local entrants in metropolitan areas urban wage premium is realised through higher entry wages, regional movers tend to demonstrate better post-entry wage performance. One tentative explanation for this is that (at least some of) regional movers are quite footloose, and, therefore, more inclined to move on to other jobs unless they enjoy a solid wage growth over the first couple of years of employment. Indeed, complementary to that finding, it appears that regional movers are less entrenched in the sector as they demonstrate lower probability to remain employed in the sector. What might be behind this is that such workers may use an entry to a job in expanding industries as stepping stones to other types of work in a region of arrival. One important thing to mention is that, even though the sector demonstrated the fastest expansion in metropolitan areas, it also enjoyed substantial dynamics even in the most peripheral regions of Sweden

6. Conclusion

This paper investigated structures of labour sourcing in the Swedish knowledge-intensive IT services sector during the period of its rapid expansion between 1991 and 2010. Complementing the spectacular examples of entrepreneurs with no apparently relevant education and experience making fortunes in the early phase of IT services expansion, our results suggest that the expectations, requirements and valuations for skills of new entrants into the sector consolidated fairly early, in line with what could be found for most industries (Bruce Chelimsky Fallick 1996). Backgrounds of incoming workers had substantial impacts on compensation levels and propensity to stay, but did not differ much from what was traditionally observed for larger sets of industries with regards to importance of education (Oreopoulos and Petronijevic 2013) and relevant work experience (Parent 2000; Sullivan 2010). Taken together with the fact that the industries that we study have clear historical antecedents in the Swedish economy (see Johansson 2001), our results may call into question how 'new' even very dynamic sectors actually are in terms of the knowledge they employ.

In particular, our findings suggest that the evolution of the knowledgeintensive IT services sector was reflected not so much in *how it valued* skills of inflowing individuals, but in *which kind of skills* employers in the sector tended to attract over time. This could be illustrated by the idea of an 'entrance ticket' for labour into the sector, and how this changed as the sector was evolving.

While we focus on a particular sector, many of our findings may be relevant, in particular those sectors where production processes are heavily relying on knowledge and underlying technological base changes fast over time, such as technology-based KIBS (e.g. engineering services, cf. Miles et al. 1996) and other R&D-intensive industries. In such sectors, for instance, fast technological change ensures that technical skills, relevant to the sector, become obsolete very fast, which in its turn means that sectoral core competences are not strongly embedded in one dominating educational group at a particular point in time. Fast on-the-job training and high labour market turbulence leave more persistent effects from education as compared to experience in the sector *per se*, even if the latter provides workers with initial benefits at the moment of the hire. Also, formation of clusters of related industries in terms of labour market dynamics would be relevant for most sectors. Again, especially so for sectors which rely on inputs of knowledge

embodied in workers as that creates particularly fruitful preconditions for forming linkages between industries. We believe, however, that further research need to investigate more thoroughly how those structures evolve over time, especially in rapidly changing industries.

While the Framtidsfabriken story in the beginning of this paper was seen by some as the manifestation of the growth dynamics of new industries, our aggregate numbers suggest that the IT services in general developed along a less spectacular, but much more powerful trajectory. Indeed, the herald industries of the 'new economy' seem to form, keep and value their labour force much like everyone else in the end, and have consistently developed towards a clearly distinguishable set of industries, forming a vital part of today's economy.

Acknowledgements

The authors are grateful for constructive advice and critical comments on earlier versions of this paper by Emelie Hane-Weijman, Karl-Johan Lundquist, David Rigby, Olof Zaring, Christian Østergaard, Rikard Eriksson, and Agnes Andersson Djurfeldt. The paper benefited from fruitful discussions at the Workshop on Industry Relatedness and Regional Change held in Umeå, Sweden, on October 29-31, 2014, the Annual Meeting of the American Association of Geographers held in Chicago, IL, USA, on April 21-25, 2015, and the Workshop on Evolutionary Approaches Informing Research on Entrepreneurship and Regional Development held in Gothenburg, Sweden, on December 10-11, 2015.

References

- Åsenlund, J. (2001). *Framfabs säregna kultur*. (Student thesis), Department of Sociology, Lund University, Lund.
- Boschma, R., Eriksson, R., & Lindgren, U. (2009). 'How does labour mobility affect the performance of plants? The importance of relatedness and geographical proximity'. *Journal of Economic Geography*, Vol. 9(2), pp. 169-190.
- Card, D. (1994). Earnings, Schooling, and Ability Revisited. National Bureau of Economic Research Working Paper Series No. 4832.
- Chadwick, A., Glasson, J., & Smith, H. L. (2008). 'Employment Growth in Knowledge-Intensive Business Services in Great Britain during the 1990s — Variations at the Regional and Sub-Regional Level'. *Local Economy*, Vol. 23(1), pp. 6-18.
- Consoli, D., & Rentocchini, F. (2015). 'A taxonomy of multi-industry labour force skills'. *Research Policy*, Vol. 44(5), pp. 1116-1132.
- Corrocher, N., & Cusmano, L. (2012). 'The 'KIBS Engine' of Regional Innovation Systems: Empirical Evidence from European Regions'. *Regional Studies*, Vol. 48(7), pp. 1212-1226.
- Dahlin, P. (2007). Turbulence in Business Networks. A Longitudinal study of Mergers, Acquisitions and Bankruptcies Involving Swedish IT-companies. (Doctoral Dissertation, Mälardalen University Press Dissertations No. 53), Mälardalen University, Västerås, Sweden.
- Doloreux, D., & Muller, E. (2007). The Key Dimensions of Knowledge-Intensive Business Services (KIBS) Analysis. A Decade of Evolution. Working Paper on Firms and Regions Number U1/2007. Karlsruhe, Germany: Fraunhofer-Institut für System- und Innovationsforschung (ISI).
- Eliason, M., & Storrie, D. (2006). 'Lasting or Latent Scars? Swedish Evidence on the Long-Term Effects of Job Displacement'. *Journal of Labor Economics*, Vol. 24(4), pp. 831-856.
- Eriksson, R., & Boschma, R. (2013). *Mapping Industry Space: (How) does skillrelatedness matter over the industry life-cycle?* Paper presented at the Workshop on Evolutionary Economic Geography, Aalborg, Denmark.

- Eriksson, R., Henning, M., & Otto, A. (2014). Regional and industrial mobility of workers leaving mature industries. A study of individuals who exit the Swedish shipbuilding industry 1970-2000. Papers in Evolutionary Economic Geography (PEEG) 1415. Utrecht, the Netherlands: Utrecht University, Section of Economic Geography.
- Fackler, D., Schank, T., & Schnabel, C. (2015). 'Does the plant size-wage differential increase with tenure? Affirming evidence from German panel data'. *Economics Letters*, Vol. 135, pp. 9-11.
- Fallick, B. C. (1996). 'The hiring of new labor by expanding industries'. Labour Economics, Vol. 3(1), pp. 25-42.
- Fallick, B. C. (1996). 'A Review of the Recent Empirical Literature on Displaced Workers'. *Industrial and Labor Relations Review*, Vol. 50(1), pp. 5-16.
- Fort, T. C., Haltiwanger, J., Jarmin, R. S., & Miranda, J. (2013). 'How Firms Respond to Business Cycles: The Role of Firm Age and Firm Size'. *IMF Economic Review*, Vol. 61(3), pp. 520-559.
- Freeman, C., Clark, J., & Soete, L. (1982). Unemployment and Technical Innovation: a Study of Long Waves and Economic Development. Westport, CT: Greenwood Press.
- Frenken, K., & Boschma, R. A. (2007). 'A theoretical framework for evolutionary economic geography: industrial dynamics and urban growth as a branching process'. *Journal of Economic Geography*, Vol. 7(5), pp. 635-649.
- Giertz, E. (2015). The Swedish ICT sector today. In E. Giertz, A. Rickne, & P. Rouvinen (Eds.), *Small and beautiful. The ICT success of Finland & Sweden*. Stockholm: VINNOVA.
- Giertz, E., Rickne, A., & Rouvinen, P. (2015). *Small and beautiful. The ICT success of Finland & Sweden*. Stockholm: VINNOVA.
- Glaeser, E. L., & Maré, D. C. (2001). 'Cities and Skills'. Journal of Labor Economics, Vol. 19(2), pp. 316-342.
- Hawk, S., Kaiser, K. M., Goles, T., Bullen, C. V., Simon, J. C., Beath, C. M., . . . Frampton, K. (2012). 'The Information Technology Workforce: A Comparison of Critical Skills of Clients and Service Providers'. *Information Systems Management*, Vol. 29(1), pp. 2-12.

- Johansson, D. (2001). *The Dynamics of Firm and Industry Growth. The Swedish Computing and Communications Industry.* (Doctoral dissertation), Royal Institute of Technology (KTH), Stockholm, Sweden, Stockholm.
- Johansson, D. (2004). 'Is small beautiful? The case of the Swedish IT industry'. *Entrepreneurship & Regional Development*, Vol. 16(4), pp. 271-287.
- Karlsson, C., Maier, G., Trippl, M., Siedschlag, I., Owen, R., & Murphy, G. (2010). *IT and Regional Economic Dynamics: A Literature Review*. EUR 24510 EN. Luxembourg: Publications Office of the European Union.
- Kluve, J. (2010). 'The effectiveness of European active labor market programs'. *Labour Economics*, Vol. 17(6), pp. 904-918.
- Krafft, J., Lechevalier, S., Quatraro, F., & Storz, C. (2014). 'Emergence and evolution of new industries: The path-dependent dynamics of knowledge creation. An introduction to the special section'. *Research Policy*, Vol. 43(10), pp. 1663-1665.
- Lundmark, M. (1995). 'Computer Services in Sweden: Markets, Labour Qualifications and Patterns of Location'. *Geografiska Annaler. Series B, Human Geography*, Vol. 77(2), pp. 125-139.
- Maliranta, M., & Nikulainen, T. (2008). Labour Force Paths as Industry Linkages: A Perspective on Clusters and Industry Life Cycles. Discussion papers No. 1168. Helsinki, Finland: The Research Institute of the Finnish Economy.
- Mamede, R. (2009). 'Toward an integrated approach to industry dynamics and labor mobility'. *Industrial and Corporate Change*, Vol. 18(1), pp. 139-163.
- March, J., & Simon, H. (1958). Organizations. New York: John Wiley and Sons, Inc.
- Miles, I., Kastrinos, N., Bilderbeek, R., Hertog, P. d., Flanagan, K., Huntink, W., & Bouman, M. (1996). *Knowledge-intensive business services: users, carriers* and sources of innovation. Retrieved from Brussels, Belgium:
- Miozzo, M., & Grimshaw, D. (2005). *Does EDS Add Value? The Expansion of IT Outsourcing and the Nature and Role of Computer Services Firms.* Paper presented at the DRUID Tenth Anniversary Summer Conference 2005, Copenhagen, Denmark.
- Mukkala, K., & Tohmo, T. (2013). 'Inter-industry job mobility in the knowledge economy in Finland'. *International Journal of Manpower*, Vol. 34(8), pp. 918-938.
- Neffke, F., & Henning, M. (2013). 'Skill relatedness and firm diversification'. *Strategic Management Journal*, Vol. 34(3), pp. 297-316.
- Neffke, F., Henning, M., Boschma, R., Lundquist, K.-J., & Olander, L.-O. (2011). 'The Dynamics of Agglomeration Externalities along the Life Cycle of Industries'. *Regional Studies*, Vol. 45(1), pp. 49-65.
- Neumark, D., & Reed, D. (2004). 'Employment relationships in the new economy'. *Labour Economics*, Vol. 11(1), pp. 1-31.
- NUTEK. (2004). Analyser för regionalt utvecklingsarbete En handbok med praktiska tips och metodexempel. Stockholm: Verket för näringslivsutveckling (NUTEK).
- OECD. (2010). OECD Information Technology Outlook 2010. Paris: OECD Publishing.
- Oesch, D., & Baumann, I. (2015). 'Smooth transition or permanent exit? Evidence on job prospects of displaced industrial workers'. *Socio-Economic Review*, Vol. 13(1), pp. 101-123.
- Oreopoulos, P., & Petronijevic, U. (2013). 'Making College Worth It: A Review of Research on the Returns to Higher Education'. *National Bureau of Economic Research Working Paper Series*, Vol. No. 19053.
- Østergaard, C. R., & Dalum, B. (2011). Evolution of the geographical concentration pattern of the Danish IT sector. In H. Bathelt, M. P. Feldman, & D. F. Kogler (Eds.), *Beyond Territory: Dynamic Geographies of Knowledge Creation*, *Diffusion, and Innovation*. New York: Routledge.
- Parent, D. (2000). 'Industry-specific capital and the wage profile: Evidence from the national longitudinal survey of youth and the panel study of income dynamics'. *Journal of Labor Economics*, Vol. 18(2), pp. 306-323.
- Pinch, S., & Mason, C. (1991). 'Redundancy in an Expanding Labour Market: A Case-study of Displaced Workers from Two Manufacturing Plants in Southampton'. Urban Studies, Vol. 28(5), pp. 735-757.
- Prabhakar, B., Litecky, C. R., & Arnett, K. (2005). 'IT skills in a tough job market'. *Communications of the ACM*, Vol. 48(10), pp. 91-94.
- Rigby, D. L., & Essletzbichler, J. (1997). 'Evolution, Process Variety, and Regional Trajectories of Technological Change in U.S. Manufacturing'. *Economic Geography*, Vol. 73(3), pp. 269-284.

- Russo, G., Rietveld, P., Nijkamp, P., & Gorter, C. (1996). 'Spatial aspects of recruitment behaviour of firms: an empirical investigation'. *Environment* and Planning A, Vol. 28(6), pp. 1077-1093.
- Shuttleworth, I., Tyler, P., & McKinstry, D. (2005). 'Redundancy, Readjustment, and Employability: What Can We Learn from the 2000 Harland & Wolff Redundancy?'. *Environment and Planning A*, Vol. 37(9), pp. 1651-1668.
- Simmie, J., & Strambach, S. (2006). 'The contribution of KIBS to innovation in cities: an evolutionary and institutional perspective'. *Journal of Knowledge Management*, Vol. 10(5), pp. 26-40.
- Strambach, S. (2001). Innovation Processes and the Role of Knowledge-Intensive Business Services (KIBS). In K. Koschatzky, M. Kulicke, & A. Zenker (Eds.), *Innovation Networks (Vol. 12)*: Physica-Verlag HD, pp. 53-68.
- Sullivan, P. (2010). 'Empirical evidence on occupation and industry specific human capital'. *Labour Economics*, Vol. 17(3), pp. 567-580.
- Valenduc, G. (2008). Occupational monograph IT professionals in software services. In G. Valenduc, P. Vendramin, B.-J. Krings, & L. Nierling (Eds.), *How restructuring is changing occupations. Case study evidence from knowledge-intensive, manufacturing and service occupations*. Leuven: KU Leuven.
- Vona, F., & Consoli, D. (2015). 'Innovation and skill dynamics: a life-cycle approach'. *Industrial and Corporate Change*, Vol. 24(6), pp. 1393-1415.
- Warhurst, C., Lockyer, C., & Dutton, E. (2006). 'IT jobs: opportunities for all?'. New Technology, Work and Employment, Vol. 21(1), pp. 75-88.
- Wood, P. (2005). 'A service-informed approach to regional innovation or adaptation?'. *The Service Industries Journal*, Vol. 25(4), pp. 429-445.
- Yankow, J. J. (2006). 'Why do cities pay more? An empirical examination of some competing theories of the urban wage premium'. *Journal of Urban Economics*, Vol. 60(2), pp. 139-161.

Appendices

A1. Description of the variables

Variable name	Variable description	Values
Wage index	Entry wage relative to the median wage in the industry	Continuous
Stay	Probability of an individual to remain employed in the sector three years after entry	1 – an individual is employed in the sector at $t+3$ 0 – otherwise
Wage growth	Probability of an individual to perform better than an industry average in terms of wage growth	1 – an individual enjoyed faster wage growth than an industry average 0 – otherwise
Sex	Sex of an individual	1 – male individual 0 – female individual
Age	Age of an individual	Continuous
Age squared	Squared age of an individual	Continuous
Edu_level	Highest achieved education level of an individual	1 – pre-gymnasial 2 – gymnasial 3 – after-gymnasial (<2 years) 4 – after-gymnasial (≥2 years) 5 – doctoral
Edu_track	Main specialisation of individual's education	 1 – social sciences 2 – natural sciences, data and computing 3 – technology and engineering 9 – other
Entry mode	Entry mode of an individual to a job in one of the IT services industries	 1 – fresh graduates 2 – non-participants 3 – related unemployment-to-job switches 4 – unrelated unemployment-to-job switches 5 – related job-to-job switches 6 – unrelated job-to-job switches
Reg. move	Indicator variable for regional move	1 – job switch associated with a regional move 0 - otherwise
Reg. family	Regional family of a region in which an individual becomes employed	 Stockholm Gothenburg Malmö Large regional centres Smaller regional centres Periphery
Firm size	Categorical variable for a size of a firm in which individual becomes employed	 1 – 1 employee (individual consultancy) 2 – 2-10 employees 3 – more than 10 employees

A2. Descriptive statistics and correlation table. Entry conditions models

	z	Mean	Std. Dev.	٢	2	3	4	5	9	7	8	6	10
Wage index	351982	0.8993	0.6566	۲									
Sex	351982	0.7117	0.4530	0.1598	-								
Age	351982	35.0872	9.9179	0.3301	0.0033	-							
Age squared	351982	1329.4730	772.1719	0.2935	0.0051	0.9889	-						
Edu_level	351982	3.0288	0.9979	0.1607	0.0727	0.0128	-0.0128	-					
Edu_track	351982	3.8016	3.0058	-0.1571	-0.0916	-0.0621	-0.0407	-0.4182	-				
Entry mode	351982	5.2239	1.0970	0.3015	-0.0140	0.1773	0.1476	0.0215	-0.0438	~			
Reg. move	351982	0.1684	0.3742	-0.0808	0.0486	-0.1142	-0.1068	0.0492	-0.0198	-0.0512	-		
Reg. family	351982	2.1863	1.4379	-0.1711	0.0175	-0.0421	-0.0354	-0.0418	0.0158	-0.0571	0.1771	-	
Firm size	351982	2.6916	0.5988	0.2576	-0.0838	-0.0453	-0.0543	0.0413	-0.0709	0.1072	0.0170	-0.0495	~

A3. Descriptive statistics and correlation table. Probability to stay in the sector models

-	_												
	z	Mean	Std. Dev.	1	2	3	4	5	9	7	8	6	10
Stay	294020	0.5642	0.4959	-									
Sex	294020	0.7071	0.4551	0.0919	-								
Age	294020	34.7854	9.6274	0.0545	0.0075	-							
Age squared	294020	1302.7120	738.0420	0.0434	0.0094	0.9898	-						
Edu_level	294020	3.0021	0.9933	0.1096	0.0770	0.0222	-0.0027	-					
Edu_track	294020	3.7687	2.9990	-0.0967	-0.0885	-0.0580	-0.0380	-0.4146	-				
Entry mode	294020	4.2282	1.0878	0.0044	-0.0104	0.1858	0.1584	0.0243	-0.0435	-			
Reg. move	294020	0.1670	0.3730	-0.0249	0.0494	-0.1173	-0.1104	0.0534	-0.0233	-0.0572	-		
Reg. familv	294020	2.1977	1.4469	-0.0136	0.0147	-0.0463	-0.0396	-0.0432	0.0173	-0.0646	0.1760	۲	
Firm size	294020	2.7103	0.5803	0.0402	-0.0812	-0.0341	-0.0397	0.0416	-0.0667	0.0988	0.0188	-0.0436	~

A4. Descriptive statistics and correlation table. Post-entry wage performance (output equations)

	z	Mean	Std. Dev.	٢	2	3	4	5	9	7	8	6	10
Wage growth	165853	0.6217	0.4850	-									
Sex	165853	0.7439	0.4365	0.0133	-								
Age	165853	35.2475	9.4092	-0.2222	-0.0111	-							
Age squared	165853	1330.9170	727.0425	-0.2181	-0.0075	0.9907	-						
Edu_level	165853	3.0977	0.9656	0.0415	0.0648	-0.0285	-0.0454	-					
Edu_track	165853	3.5139	2.7831	-0.0282	-0.0554	-0.0144	-0.0014	-0.4041	4				
Entry mode	165853	4.2325	1.0460	-0.1209	-0.0113	0.1722	0.1464	0.0051	-0.0245	-			
Reg. move	165853	0.1588	0.3655	0.0653	0.0482	-0.1355	-0.1270	0.0636	-0.0305	-0.0619	-		
Reg. familv	165853	2.1803	1.4139	0.0188	0.0189	-0.0502	-0.0444	-0.0152	-0.0017	-0.0664	0.1719	-	
Firm size	165853	2.7308	0.5674	0.1209	-0.1002	-0.0293	-0.0356	0.043	-0.0685	0.0946	0.0222	-0.0404	-

	-	-	- -		_	
Heckman probit	1991-2010	1991-1993	1994-1997	1998-2001	2002-2004	2005-2007
Edu_level (pre-gym)						
Gym	0.0474***	0.0173	-0.0081	0.0967***	0.0359	0.0654*
After-gym (<2 years)	0.1350***	0.1274**	0.0701**	0.1877***	0.1424***	0.0641*
After-gym (≥2 years)	0.1749***	0.1413***	0.0560*	0.2077***	0.2153***	0.2080***
Doctoral	0.0088	0.1061	-0.1026	0.0493	0.0120	0.0495
Edu_track (other)						
Social science	0.0418***	-0.0422	0.0525***	0.0613***	0.0159	0.0128
Natural science, data	0.3630***	0.4094***	0.3837***	0.3654***	0.2931***	0.3471***
Technology	0.1862***	0.1197***	0.1438***	0.1941***	0.1651***	0.2676***
Entry (unrelated JTJ)						
Fresh graduate	-0.0257	-0.1467	-0.0484	-0.0704*	0.0568	0.0010
Non-participant	-0.1501***	-0.2949**	-0.1587***	-0.2244***	-0.1010*	-0.0609
Unrelated UTJ	0.0715***	0.0107	0.0879***	0.0617***	0.0659***	0.0306
Related UTJ	-0.0435***	-0.0821***	-0.0467***	-0.0514***	-0.0710***	-0.0457**
Related JTJ	0.3675***	0.2465***	0.3019***	0.2970***	0.4101***	0.4229***
Regional move	-0.0304**	-0.0707	-0.0556**	0.0571***	-0.1400***	-0.1913***
Region (large reg. centres)						
Stockholm	0.0040	-0.0155	-0.1343***	0.1014***	-0.0731***	-0.0940***
Gothenburg	0.1325***	0.0547	0.0372	0.1613***	0.2169***	-0.0650**
Malmö	0.0542***	-0.1583***	-0.0631**	0.1963***	0.0559**	-0.1203***
Smaller reg. centres	0.0424***	0.0681	0.0349	0.2312***	-0.2229***	0.0511
Periphery	-0.1113***	-0.0257	-0.1015*	-0.0021	-0.0786	-0.4324***
Reg. move*Region (large reg. centres)						
Stockholm	-0.0378***	0.0183	0.0220	-0.1514***	0.0755*	0.1206***
Gothenburg	-0.0837***	-0.1952**	-0.0652	-0.1029***	-0.1004*	0.1182**
Malmö	-0.0738***	-0.1106	-0.0048	-0.2379***	0.0600	0.2240***
Smaller reg. centre	-0.0795***	-0.2105*	-0.1298**	-0.2144***	0.1271*	-0.0188
Periphery	-0.1350***	-0.5752***	-0.0546	-0.2215***	-0.0477	0.1570*
Firm size (2-10 emp)						
1 emp	0.0192*	0.0620	0.0459*	0.0794***	-0.0210	-0.0552**
> 10 emp	0.1303***	0.2109***	0.2275***	0.1033***	0.0583***	0.1471***
National sectoral growth	3.5428***	0.1537*	1.4777***	2.0964***	0.5968***	2.4760***
Note: Demographic conti ***(**,*) indicate significar	ols (sex and age), industry and nce at 1% (5%, 10%) level	l year dummies as well as a co	nstant are included in the moc	lels but not reported. Standard	l errors are robust and clustered	l at the individual level.

equations)
(selection
performance
5. Post-entry wage
Y

Article III

Specialised pervasiveness: Diffusion of IT competences in the Swedish economy

Mikhail Martynovich

Department of Human Geography, Lund University

Abstract: The paper investigates the diffusion of IT competences in the Swedish economy between 2001 and 2010 by looking at patterns of employment and reallocation of individuals in IT-related occupations. Particular emphasis is made on the expansion of IT employment outside of the IT sector. The results indicate that the labour market for IT workers was expanding rapidly during the period. At the same time, there were changes in its organisation with respect to emergence of clusters in the industry space as well as reinforcement of regional hierarchies. With regards to the IT as a general purpose technology, a concept of specialised pervasiveness – the ability of the IT to generate IT-related jobs in all sectors of the economy but with an increasing degree of specialisation and subsequent segmentation of the labour markets for IT workers – was proposed to capture this phenomenon.

Keywords: human capital, competences, diffusion, labour market, general purpose technology, IT

JEL codes: J24, J44, J61, J62

1. Introduction

This paper investigates the distribution and reallocation of workers in the information technology (IT) occupations across industries and regions in Sweden over 2001-2010 with the aim of exploring the patterns of diffusion of IT-related competences in the Swedish economy.

The introduction and spreading of the IT has left a deep structural impact on the modern society leading to, among other things, a vast reorganisation of economic systems, which allowed to characterise the IT as a general purpose technology (Bresnahan and Trajtenberg 1995; Bresnahan 2010). Multiple studies have demonstrated that the IT transformed modern production processes and generated significant productivity gains even in industries and sectors that do not produce with a high IT intensity (for literature reviews see Draca et al. 2007; Cardona et al. 2013).

More specifically, the access to the IT resources has been demonstrated to affect positively the performance of firms in terms of increasing product quality (Thatcher and Pingry 2004), improving workflow (Buhler and Vidal 2005), enhancing flexibility in responding to customer needs (Gunasekaran and Ngai 2004), improving communication with customers and suppliers (Fiala 2005) as well as increasing innovative capacity (Gordon and Tarafdar 2007). In that respect, the availability of the IT resources is an important factor for maintaining competitiveness and achieving sustained growth at a firm level.

At the same time, the diffusion of the IT in the economy generated the demand for new skills and led to the emergence of new (IT-related) occupations and new forms of workplace organisation (Bresnahan et al. 2002; López-Bassols 2002). Nevertheless, the theoretical and empirical discussion in the literature has so far largely focused on investments in the IT capital at the firm and industry levels, changes in and impact of the IT literacy of workers on firm productivity, as well as on the routinisation of certain tasks as an outcome of the IT diffusion and the subsequent substitution of relatively lower skilled workers by highly skilled ones – phenomena known as task-biased and skill-biased technological change (Autor et al. 2003; Adermon and Gustavsson 2015). Much less attention, however, has been devoted to the specialised IT human capital, that is, workers in IT occupations (Hagsten and Sabadash 2014).

This gap in the literature seems to be quite unmotivated though. IT workers play a fundamental role in the process of diffusion and adoption of the IT that fuels technological developments in IT producing sectors and enables the actual realisation of productivity gains in IT using industries (Peneder 2003). The specialised IT human capital allows making better use of the IT capital inputs, which means that a deficit in specialised IT skills not only restricts the IT adoption itself, but also limits potential benefits of adoption at a firm level (Forth and Mason 2004). It is, therefore, important to gain a better understanding of the extent and quality of IT-related labour resources available in the economy.

While some research efforts have been made to describe the patterns of employment and human capital formation in the IT sector (Lundmark 1995; Bienkowska 2007; Hawk et al. 2012; Martynovich and Henning 2016), it should be noted that the IT workforce is not equivalent to employees in the IT sector. A large share of the software development and other IT-related activities take place in many different sectors (Østergaard and Dalum 2007). Thus, focusing solely on employment in the IT sector would underestimate the availability and spread of IT competences in the economy.

The paper builds around three (sets of) research questions. The first question is formulated in the following way: *To what extent does the distribution of the IT workforce between the IT and non-IT sectors reflect the quality of IT workers' human capital?* This question aims at (1) estimating the degree to which the IT workforce is concentrated to the IT sector that is 'paradigmatic' to the IT diffusion, and (2) understanding whether there are differences in skill profiles of IT workers employed in the IT and non-IT sectors.

The second question concerns the regional patterns of IT employment: *In what way does the position of a region in the regional hierarchy affect the regional distribution of the IT workforce between the IT and non-IT sectors?* The motivation for this question is twofold. On the one hand, IT producing industries are characterised by highly uneven locational patterns. It has been demonstrated that the specialised IT activities tend to concentrate in dynamic regions at the top of the regional hierarchy (Lundmark 1995; Martynovich and Henning 2016). On the other hand, the IT adoption by user industries is also known to be spatially biased towards large urban centres (Forman et al. 2005). In that respect, the interaction of supply and demand factors may result in a very different distribution of IT labour between the IT and non-IT sectors at different layers of the regional hierarchy.

While the first two questions have a rather static, or snapshot, perspective on the distribution of IT workers in industrial and geographic spaces, there is also an important dynamic issue. That is, *how are the dynamics of the IT workers reallocation conditioned by sectoral and regional factors?* In other words, are there any differences in how firms in different industries and different regions source IT workers? This is the third question to be addressed in the paper. Traditionally it was assumed that firms in the IT and non-IT sectors compete for the same pool of labour, even if some firms hire IT workers with certain niche skills, while others – with a broader skill portfolio (Hawk et al. 2012). The paper questions this assumption and aims at investigating whether the demand for IT workers is generic or sector-specific, and how this plays out in different regions.

The remainder of the paper is organised as follows. Section 2 provides a brief literature review on the issue of IT as a general purpose technology with a particular emphasis on the impact of IT on the labour markets. In Section 3, data and methodological considerations behind the empirical analysis are reviewed. Sections 4 and 5 present the empirical results with respect to the sectoral and regional patterns of IT workers distribution as well as reallocation of IT workers respectively. Section 6 synthesises the results through the prism of the literature on the general purpose technologies. Finally, Section 7 provides concluding remarks and suggestions for further research.

2. Diffusion of the IT and specialised human capital: A literature review

The diffusion of the IT initiated the processes of industrial restructuring – that is, changes in the production processes within industries – and structural change – that is, changes in the role of different industries in the development of the economy as a whole. It has been demonstrated that the IT possesses the characteristics of a general purpose technology, such as (Bresnahan and Trajtenberg 1995; Bresnahan 2010):

1. *pervasiveness* – having potential applications across a wide variety of products, processes and industries;

- 2. *technological dynamism* providing a wide scope for technological improvement and elaboration;
- 3. *innovation spawning* facilitating complementarities with existing technologies, fostering complementary innovation in user sectors, and bringing about sustained and pervasive productivity gains.

Indeed, the IT has been demonstrated to be adopted by a wide range of firms in different industries and to be a source of productivity growth at the firm, industry, and national levels (for reviews of the literature, see Draca et al. 2007; Stiroh 2010; Kretschmer 2012; Cardona et al. 2013; Stanley et al. 2015).

The IT diffusion has also left a deep impact on the functioning of the labour market as the latter is the mediator that ensures the adjustment to the new industrial systems (Aghion et al. 2006; Bachmann and Burda 2010). Not least, it led to the restructuring of regional labour markets and changing roles of different regions in the technology-induced economic transformation (Martynovich and Lundquist 2016).

To date, quite many studies focused on the general impact of the IT diffusion on the labour market. For instance, the research focused on the automation of routine tasks and resulting substitution of workers, involved in these tasks, by workers employed in occupations with more abstract task contents – the phenomenon known as the task-biased technological change (Spitz-Oener 2006; Goos and Manning 2007). As abstract tasks put demands on workers' human capital, this substitution resulted in higher relative demand for highly educated workers who possess a comparative advantage in executing these tasks (Böckerman et al. 2016). This, in its turn, boosted labour productivity and output growth in firms and industries (Belloc and Guerrieri 2015).

At the same time, productivity growth depends on the capacity of firms to learn from the new technology, also known as the absorptive capacity (Cohen and Levinthal 1990). Highly skilled labour is, indeed, an important component of this capacity (Liao et al. 2016). But it is not just skilled labour *per se.* Specialised IT labour is a crucial enabler of IT-induced change at the firm level (Forth and Mason 2004). IT workers – who are referred as 'paradigmatic' for the new, digital economy (Ensmenger 2003) – are positioned at the cutting edge of technological development, work with the IT itself, and are at the core of the transformation of production processes at the firm level (Adams and Demaiter 2008). In that respect, it comes as no

surprise that the impact of employing IT workers on productivity gains is larger than that coming from generically skilled workers (Hagsten and Sabadash 2014).

One may claim that certain IT tasks may be outsourced to specialised providers of IT services. However, organisational learning has for a long time been seen as a by-product of activities performed at a firm level (Nelson and Winter 1982). For instance, firms learn the value of IT applications and further development opportunities by actually using them (Earl 1996). In that respect, the outsourcing of IT activities leads to the erosion of the IT-related knowledge in firms (McCray and Clark 1999; Willcocks et al. 1999). Therefore, discontinuing in-house IT functions may lead to losing the ability to benefit from future developments in the IT (King and Malhotra 2000).

In that respect, hiring IT workers is instrumental for firms to benefit from opportunities offered by the IT. While the diffusion of IT capital inputs is widely accepted as a showcase of pervasiveness feature of IT, it is the availability of IT workers which ensures realisation of its technological dynamism and innovation spawning features. Therefore, the availability of the IT labour is important for reaping full benefits of the IT-induced transformation.

3. Data and definitions

3.1. Data

Data employed in this paper comes from the linked employer-employee data set – the Longitudinal Integration Database for Health Insurance and Labour Market Studies (LISA) – provided by Statistics Sweden. For each individual registered in Sweden it contains a wide range of personal attributes including age, education, region of residence, etc. Individuals are linked to the establishments of their main employment for which location and industry affiliation, as well as occupations that they are employed at, are known. The occupation data is available from 2001 till 2010, while certain indicators can be traced back to 1990.

For each year between 2001 and 2010 a sample of individuals in the full working age (18-64) was derived. It was further restricted to individuals

employed at the moment of data collection in each reference year, which corresponds to the official employment status *gainfully employed* in the data set.¹ The latter restriction was imposed to minimise the probability of including workers participating in short-term/project employment in the sample.

3.2. Defining IT occupations

According to the Swedish Standard Classification of Occupations (SCB 1998), there are five occupations that are directly related to performing IT tasks, namely: (1) computing services managers, (2) computer systems designers analysts and programmers, (3) computing professionals not elsewhere classified, (4) computer assistants, and (5) computer equipment operators. Of these, occupation (1) belongs to the category of Legislators, senior officials and managers, occupations (2) and (3) – Professionals, and occupations (4) and (5) – Technicians and associate professionals. Occupational codes, titles and main tasks descriptions for these are summarised in Appendix A1.

For the purposes of this paper, these five occupations were groups into three occupational categories, namely:

- 1. *IT managers* occupation (1) above;
- 2. IT professionals occupations (2) and (3); and,
- 3. IT technicians occupations (4) and (5).

These occupational categories are employed in the remainder of the paper.

For each year between 2001 and 2010 a sample of individuals, who were employed in one of these occupations, was derive. The resulting data set included 238,272 unique individuals and 409,904 unique employer-employee matches.²

¹ Data in LISA is collected once a year (on December 31). The employment status of an individual reports whether she is employed at the moment of data collection, was employed at some point over the reference year but not at the moment of data collection, or was not at all employed over the reference year.

² A match here is defined as an employment relation between a particular worker and a particular establishment. Employment lasting for several years is considered as one match.

3.3. The IT and non-IT sectors

In this paper, the IT sector is defined as a sector providing IT services to non-IT firms. According to the Swedish Standard Industrial Classification (SCB 2003), it comprises seven five-digit industries under the activity code 72 'computer and related activities', that is: (1) hardware consultancy, (2) software consultancy, (3) software supply, (4) data processing, (5) database activities, (6) maintenance and repair of office, accounting and computing machinery, and (7) other computer related activities. Individuals are then classified as employed in the IT sector if an establishment, in which they are employed, operate in one of these industries.

Individuals in IT occupations, not employed in one of the IT service industries, are classified as employed in non-IT sectors. The latter are divided into seven industry groups based on the section they belong to in the Swedish Standard Industrial Classification, namely: (1) agriculture, forestry and fishing, (2) mining and quarrying, (3) manufacturing, (4) electricity, gas and water supply, (5) construction, (6) private services (excluding IT sector)³, and (7) public services. These industry groups are used in the empirical analysis to investigate the employment patterns of IT workers outside the IT sector.

One reservation should be made here. There is, of course, the heterogeneity in employment patterns of IT workers in individual five-digit industries within these industry groups. However, the within-group variation in these patterns is much smaller than the between-group variation. Therefore, since the primary focus is on finding differences between the IT and non-IT sectors, it was decided to stick to the rather aggregated level of industry divisions.

3.4. Regional divisions

The division of individual regions into regional groups was performed in two steps. First, 290 Swedish municipalities were divided into the so-called local labour markets according to the methodology suggested by Statistics Sweden (SCB 2010). These are based on the statistics of commuting between

³ Individuals employed in firms belonging to five-digit industries coded 74501 (Labour placement agencies) and 74502 (Temporary employment agencies) were excluded from the sample. The reason is that being formally employed in one of these industries might be non-indicative: a person classified as employed there might, in reality, work for or be lent to a firm in another industry.

municipalities in the way that maximises homogeneity of commuting patterns within a region while sustaining cross-regional heterogeneity. This level of regional divisions is used to identify individuals for whom becoming employed in an IT occupation is associated with changing a region of employment.

Second, local labour markets are aggregated into the so-called regional families that are used to capture potential hierarchies in the labour market for IT workers. These are defined according to the methodology suggested by the Swedish Agency for Economic and Regional Growth that involves criteria, such as the population size and density, regional business dynamics, the share of individuals with higher education as well as the access to higher education institutions (NUTEK 2004). Six regional families are defined. The first three layers in the regional hierarchy are represented by three Swedish metropolitan areas – Stockholm, Gothenburg, and Malmö. Remaining local labour markets are divided into the large regional centres, smaller regional centres, and peripheral regions.

4. The labour market for IT workers in Sweden

4.1. Industrial segmentation

Over a decade between 2001 and 2010, the total employment in IT occupations in Sweden increased from 113,239 to 143,401 workers, equivalent to 26.64 per cent expansion (Figure 1). With the average annual employment growth rate of 2.66 per cent, this group of occupations expanded more than twice faster than the total employment in Sweden over the same time period (the average annual growth rate of 1.21 per cent). This resulted in the increase of its share in the national employment mix from 2.95 per cent in 2001 to 3.35 per cent in 2010.



Figure 1. Workers in IT occupations, 2001-2010

The expansion of the labour market for IT workers was, however, unequally distributed between the IT and non-IT sectors: the IT employment growth in the latter was almost three times faster than in the former (the average annual growth rates of 3.67 per cent and 1.45 per cent respectively). Out of 30 thousand new IT jobs generated between 2001 and 2010⁴, 75 per cent were created in non-IT sectors. This resulted in the increase of their share in the national IT employment mix from 52.26 per cent in 2001 to 57.10 per cent in 2010.

Within this period, the IT sector was quite severely affected by consequences of the 2001 dot-com crash. Between 2001 and 2003, the sector lost almost 6 per cent of IT employees as a result of the consolidation of the sectoral demand for workers (Martynovich and Henning 2016). At the same time, the IT employment in non-IT sectors continued to grow during this period, even though at a slower pace. In that respect, the collapse of the dot-com bubble was, to a great extent, driven by the financial factors, implying that its severe effects were confined to the IT sector itself. The demand for IT resources, and IT labour in particular, in the economy as a whole did not diminish even in the aftermath of the dot-com crash.

⁴ In terms of the net employment growth, not the gross job creation.

It should be noted here that these developments were, by no means, unique to Sweden. The similar tendencies have also been reported for, for example, the US (Currid and Stolarick 2010), the UK (IES 2002), Canada (ICTC 2015) and Denmark (Østergaard and Dalum 2006).

While non-IT sectors, in general, demonstrated an outpacing growth dynamics, there was a huge heterogeneity between various industry groups within, both when it comes to absolute numbers of IT workers and growth in IT employment (Figure 2).



Figure 2. Major employers of IT workers outside the IT sector⁵

Firms in private services industries were, by far, the major IT workforce employers among non-IT sectors, concentrating around 60 per cent of workers throughout the considered time period. This industry group was followed, with a significant gap, by public services providers (21.12 per cent of IT employment in non-IT sectors in 2010) and firms in manufacturing industries (15.21 per cent respectively).

The IT employment in absolute terms increased in all industry groups between 2001 and 2010. This included manufacturing industries, where the

⁵ IT employment in two industry groups – namely, agriculture and forestry as well as mining and quarrying – is not reported in the figure as their contribution over the whole time period was negligible (between 183 workers, combined, in 2001 and 400 workers in 2010).

total employment shrank in the same period. When it comes to the relative growth, public services demonstrated the fastest expansion in IT employment among non-IT sectors with the average annual growth rate of 6.55 per cent and total growth of more than 80 per cent. To a great extent, this expansion was driven by the digitalisation of routines in public organisations and the increase in the number of public services accessible online. By 2009, 100 per cent of basic public services for business were fully provided online in Sweden; the corresponding value for public services for citizens was 92 per cent (European Commission 2010).

Even though the IT employment increased in all industry groups in absolute terms, it is interesting to check to what extent its expansion reflected the total employment growth in these groups. In other words, whether the number of IT workers increased to the same extent as the total employment. Figure 3 reports the changes in the IT intensity⁶ of the employment mix in non-IT sectors.



Figure 3. IT intensity of employment mix in non-IT sectors⁷

⁶ IT intensity is defined as the share of IT workers in the total employment mix in an industry group.

⁷ Here and further, a superscript * indicates a difference between 2010 and 2001 that is statistically significant at the 5 per cent level.

From this perspective, the picture becomes quite the opposite. Despite the fastest growth in the IT employment growth, public services providers remained one of the least IT-intensive industry groups with 1.21 per cent of workers being employed in IT occupations in 2010. Electricity, gas and water supply industries remained the leaders in the IT intensity of their employment mix, even though the share of IT workers in the sector remained more or less fixed between 3.4 and 3.5 per cent throughout the whole time period. All other industry groups demonstrated, to a varying degree, an increase in IT intensity.⁸ Of course, the absolute numbers are not even closely comparable to the IT intensity of the IT sector that was around 60 per cent. Nevertheless, the tendency towards an increasing share of IT workers in the employment mix points towards the recognition of the importance of employing more IT workers in-house.

All in all, the labour market for IT workers expanded fast throughout the 2000s. Somewhat surprisingly, its expansion was particularly pronounced outside the IT sector. Not only industries in non-IT sectors demonstrated faster employment growth in absolute terms, but also almost all of them increased significantly the IT intensity of their employment mixes. These tendencies indicate that the ongoing IT adoption by firms manifested itself not only in investments in the IT capital but also in hiring IT workers. More generally, this points to the diffusion of IT competences in the Swedish economy beyond the core IT sector.

4.2. Human capital quality of the IT workforce

An important question is whether the quantitative expansion of IT labour market and the diffusion of IT competences beyond the IT sector was associated with changes in the quality of human capital of IT workers. Also, it is interesting to check whether there are any differences between the IT and non-IT sectors with respect to the latter. These questions are addressed in this section by looking at the education level and occupation structure of IT workers in different industries.

With regards to the education level (Figure 4), one can observe the increase in the share of highly educated workers in both IT and non-IT sectors. By

⁸ Including agriculture, fishing, and mining sectors which are not reported here.

2010, the share of IT workers with at least some post-gymnasial education⁹ reached 72.23 per cent and 60.01 per cent in the IT and non-IT sectors respectively. While this tendency reflects a general trend towards an upgrade of the education level in the Swedish workforce – the corresponding share among all Swedish workers increased from 31.70 per cent in 2001 to 38.56 per cent in 2010 – the absolute numbers suggest that having at least some education at the post-gymnasial level is a particularly strong prerequisite for IT jobs, both within and outside the IT sector.



Figure 4. Education level of IT workers

In that respect, advanced education has increasingly become an important factor for becoming employed in an IT position (Warhurst et al. 2006). One possible explanation is that when it comes to IT occupations, the demand for knowledge and skills seem to be frequently shifting as the ongoing technological change renders today's knowledge obsolete tomorrow (Adams and Demaiter 2008). In this situation, the importance of higher levels of education is in signalling to employers that an individual has an ability to learn, and may adapt to changing knowledge requirements in the technologically dynamic field.

⁹ Gymnasium is the highest level of the secondary education in Sweden. The post-gymnasial education is predominantly the university education, but could also include other forms. Here, it includes having courses at the university level, having completed higher education, as well as obtaining a PhD degree.

When it comes to occupational divisions on the labour market for IT workers (Figure 5), it is IT professionals (highly skilled workers not involved in managerial tasks) that dominate the employment mix in both IT and non-IT sectors. While in absolute terms the labour market for this occupational group became of the equal size in the IT and non-IT sectors in 2010, the IT professionals represent a much higher share of IT workers in the IT sector (77.76 per cent against 58.46 per cent in non-IT sectors).



Figure 5. IT workers by occupation

What should be pointed, the relative demand for IT technicians – relatively lower skilled workers – decreased in non-IT sectors over the considered time period. Their share in the IT employment mix decreased by almost four percentage points. Coupled with almost the five-time expansion (in absolute terms) of the IT managers occupational group¹⁰, this points to an increased complexity of tasks in which IT workers in non-IT sectors are involved.

Changing occupational structures reflect the evolution of IT activities in which IT workers are involved. Decentralisation of IT activities, driven by growing importance of networks in the 1990s, increased the demand for business and organisational skills in the IT workforce (Lee et al. 1995; Todd

¹⁰ The number of IT managers in non-IT sectors increased from 691 to 3354.

et al. 1995). Developments in the 2000s (most importantly, the explosion of the Internet use for business purposes) has further emphasised that IT workers should embody a mixture of technical, organisational, business, and management skills (Bassellier and Benbasat 2004; Byrd et al. 2006). This trend was reinforced by the rapid obsolescence of technical IT skills making it less strategically valuable to hire technical IT workers (Prabhakar et al. 2005).

All in all, there are two major points to be made with respect to the quality of human capital of IT workers. First, the increased level of education as well as the evolution of occupational structures imply that the quantitative expansion of the labour market for IT workers was complemented with the qualitative upgrade of the human capital in the IT workforce. Second, while non-IT sectors tend to lag behind somewhat with respect to both education level and share of workers in occupations requiring more advanced skills, some convergence is observed between the IT and non-IT sectors. In that respect, the skill requirements of IT prover and IT user sectors are becoming more alike than would be expected (Hawk et al. 2012). This provides another important qualification to the diffusion of IT competences in the Swedish economy.

4.3. Regional distribution of the IT employment

Figure 6 reports the regional distribution of the IT employment (irrespective of the sector) across different levels of the Swedish regional hierarchy. Not surprisingly, Stockholm remained, by far, the major regional labour market for IT workers throughout the whole time period. However, a slight trend towards decentralisation is observed, at least beyond the Stockholm region – from 49.97 per cent of the national IT employment in 2001 to just above 46 per cent in 2010. It was primarily driven by a fast expansion of the labour market for IT workers in Malmö that increased its share by almost 3 percentage points over the 2000s. This corresponds neatly to the period of transformation of Malmö into a 'knowledge city' that focuses on attracting technology- and knowledge-intensive activities, particularly in service sectors (Anderson 2014). Taken together though, three Swedish metropolitan areas concentrated around 70 per cent of IT workers throughout the whole time period, with only minor fluctuations over time.

2001	49.97%	12.33%	6 7.86%	22.77%	
2002	49.63%	12.28%	8.05%	22.67%	
2003	48.40%	12.17%	8.43%	23.43%	
2004	47.01%	12.16%	8.90%	24.28%	
2005	46.67%	12.40%	9.40%	24.01%	
2006	46.28%	12.48%	10.05%	23.88%	
2007	46.62%	12.50%	10.30%	23.52%	
2008	46.14%	12.36%	10.45%	23.92%	
2009	45.82%	12.43%	10.54%	23.99%	
2010	46.27%	12.53%	10.41%	23.71%	

Stockholm Gothenburg Malmö Large reg centres Smaller reg centres Periphery

Figure 6. Regional distribution of the IT employment

It was demonstrated elsewhere that over the considered time period the employment in the IT sector was increasingly concentrating to Swedish metropolitan areas (Martynovich and Henning 2016). However, as their share in the total IT employment in these regions remained stable throughout the 2000s, this may indicate that the expansion of the IT employment in non-IT sectors proceeded at a faster pace in non-metropolitan regions. Figure 7 demonstrates that this is, indeed, the case.



Figure 7. Share of non-IT sectors in the total IT employment

Two observations can be made here. First, the national trend an outpacing expansion of the IT employment in non-IT sectors is observed at all levels of the regional hierarchy. By 2010, in all regions but Gothenburg more than a half of IT workers were employed in non-IT sectors.

Second, there is a very strict hierarchy of regions with respect to the share of non-IT sectors in the regional IT employment. In fact, this hierarchy strengthened significantly over time. That is, the lower a region is located in the regional hierarchy, the higher is the share of non-IT sectors in the regional IT employment mix. In the most peripheral regions, this share reached more than 84 per cent in 2010.

This hierarchy may be driven by the tendency of knowledge-intensive activities to locate in the most dynamic regions at the top of the regional hierarchy, particularly metropolitan areas (Wood 2005; Chadwick et al. 2008). Apart from being regions with the highest potential demand, these regions offer advantages in terms of knowledge flows and knowledge spillovers by combining the localisation of clusters in specific industries with the industrial diversity (Karlsson et al. 2010). In that respect, firms in more peripheral regions may be prone to build up the IT capacity internally as the local supply of IT services is limited. This manifests itself in the higher shares of non-IT sectors in the regional IT employment mix.

Irrespective of sectoral divisions, there is also a rather strict hierarchy with respect to the IT intensity of regional labour markets (Figure 8).



Figure 8. IT intensity of regional labour markets

All regions increased significantly the IT intensity of their employment mixes over the considered time period. However, there is a huge regional variation. For instance, the IT intensity of Stockholm's labour market reached 5.54 per cent by 2010. This number declines with each level of the regional hierarchy, being just 1.14 per cent in the most peripheral regions.

All in all, two conclusions emerge out of the discussion of regional divisions in the labour market for IT workers. First, even though metropolitan areas dominate the scene, the expansion of the IT employment proceeded at the same pace in regions at all level of the Swedish regional hierarchy. All regions increased both the absolute numbers of IT employment and the IT intensity of their employment mixes. Second, the underlying mechanisms of this expansion, however, were rather different. In more peripheral regions the growth in the number of IT workers was fuelled primarily by the expansion of IT employment in non-IT sectors, while it was more balanced in metropolitan areas. This points to strong sorting mechanisms involved in the regional distribution of the IT workforce and, consequently, the spatial diffusion of IT competences in the Swedish economy.

5. IT workers reallocation dynamics

This section focuses on the patterns of IT workforce sourcing across different industries and regions. Figure 9 reports the division of the IT worker stocks in the IT and non-IT sectors for each year between 2001 and 2010 with respect to their employment status in the previous year. In that respect, the distinction is made between workers who stay in the same job between years t-1 and t, workers becoming employed directly from previous employment (JTJ), from unemployment (UTJ), and from non-participation (NTJ). Here, the differentiation between switches from unemployment and non-participation is performed in the following way: if a person was unemployed at t-1 and t-3, she is considered a non-participant. The remaining workers, unemployed at t-1, are regarded as becoming employed from unemployment.



(a) non-IT sectors



(b) IT sector

Figure 9. IT workers by their previous employment status

It appears that the job matches of IT workers become increasingly stable as there is an increasing tendency for IT workers to remain employed in the same firm over time. This tendency is particularly pronounced in non-IT sectors. There is some cyclical variation also. Reallocation of workers across jobs has been repeatedly shown to demonstrate pro-cyclical patterns, that is, the number of job switchers increases during the upswing of the business cycle and decreases in the downswing (Bjelland et al. 2011; Hyatt and McEntarfer 2012). An increase on the share of workers remaining employed in the same job in the early 2000s and around 2008-2009 corresponds neatly to the periods of economic downturn in the Swedish economy (reinforced, in case of the IT sector, by the effects of the 2001 dot-com crash). In-between these two periods a slight decrease in the job stability is observed.

If workers do change jobs, however, they do so predominantly within the industry¹¹ of their previous employment (Figure 10).¹² The first striking feature here is that, in both the IT and non-IT sectors, most IT workers switch jobs within the industry of their previous employment. In particular, around 80 per cent of job switches happened with the same industry. Additionally, between 10 and 14 per cent of job switches (depending on a year) took place between firms belonging to the same industry group. This left less than 10 per cent of job switches across industry groups, and, thus, across unrelated employment.

This pattern of worker reallocation points to the rather rigid skill structures in the demand for IT workers. In other words, there is a strong component of the human capital specificity involved. Skilled workers possess highly specialised human capital which they acquire through education and on-thejob experience and which allows them to carry out certain tasks (Neffke et al. 2016). Empirical work has claimed that such human capital specificity is observed in the domains of a firm (Becker 1964), an industry of employment (Parent 2000; Sullivan 2010) or to an occupation (Poletaev and Robinson 2008; Kambourov and Manovskii 2009). The tendency workers to switch jobs within the same industry points to the fact, even though the evidence is indirect, that even within the same occupational category industry experience tends to be valuable for employers. One potential explanation to that lies in the nature of tasks that IT workers perform. As it was discussed above, there is an increased demand for IT workers to possess organisational, business,

¹¹ An industry here is meant as a five-digit industry in the Swedish Standard Industrial Classification.

¹² Here, workers performing a direct switch from previous employment, as well as from unemployment are considered. For the latter, it is the industry of the last employment before becoming unemployed that is accounted for.

and management skills that are likely to be specific to an industry as compared to more technical IT skills which are rather generic.



(a) non-IT sectors



Figure 10. Job switchers by the source industry

There are however some notable regional differences in the reallocation of IT workers across industries (Figure 11). As in the case of other indicators discussed above there is a clearly observable hierarchy of regions. That is, the share of within-industry reallocations of IT workers decreases from the highest to the lowest levels of the regional hierarchy. In Stockholm, 89 per

cent of IT workers switch jobs within the same industry. This share falls with each level of the regional hierarchy to 51.94 per cent in the most peripheral regions.



Figure 11. Job switchers by source industry and region (average, 2001-2010)¹³

There are three possible explanations for this tendency. First, Timmermans and Boschma (2013) find indications in the Danish case that firms in large regions need to be more specialised to sustain their competitiveness. In that respect, industry experience of workers would be of a higher value to employers in these regions. Second, and related, explanation is that firms in large regions differ in their IT adoption strategies from firms in smaller regions (Forman et al. 2005). For instance, the latter tend to implement less sophisticated IT applications into their production processes. In that respect, the tasks that IT workers perform may be more generic, which makes the importance of industry experience less. Finally, it is likely to be the issue of the availability of industry-specific skills at the regional level. As the labour market for IT workers is much larger in metropolitan areas, one may expect it to be much easier to find workers with narrowly defined skill profiles (e.g., with a particular industry experience) simply because the pools of available

¹³ As there are no significant differences across years as well as between the IT and non-IT sectors, the shares are reported as averages over the whole time period and across all industry groups.

workers is larger. In more peripheral regions, where the availability of IT workers is more limited, the demand for industry-specific skills might outpace the supply of these skills leading to regional skill shortages. This might force firms to employ workers who are (potentially) less suitable for jobs they are hired for, at least with respect to their industry experience.

Summing up the discussion, the industrial decomposition of IT workers job switches provides a strong evidence for existence of clusters of worker reallocation. That is, workers tend to switch jobs within the same industry that they have been employed before. With respect to the diffusion of IT competences, it implies that it is not just generic IT skills that are being spread, but rather the process of diffusion is conditional upon accumulation of industry-specific human capital. The regional decomposition, however, suggests that this process is dependent on the nature of IT activities performed in a region and the availability of workers at the regional level.

If it is the availability issue that explains the regional differences, one might expect that employers might search for workers outside of a region of they are not available locally. To check for this possibility, Figure 12 reports the share of workers for whom a job switch is associated with changing a region of employment.



Figure 12. Share of regional movers among job switchers

Yet again, a clear hierarchy of regions is observed here. The more peripheral the region is, the higher is the share of inter-regional movers among job switchers. It varies between 12 and 20 per cent of IT workers becoming employed in Stockholm to more than a half – in the most peripheral regions. The distribution holds the same for both IT and non-IT sectors. This hierarchical distribution points, from another perspective, to the importance of regional availability of IT workers. It has been demonstrated that regional labour markets in more peripheral regions are smaller and have a lower tendency for job-to-job reallocations within the same sector. Here, one can also add that the local availability of IT workers.

That is, in dynamic labour markets at higher levels of the regional hierarchy, larger pools of IT workers allow employers to hire local workers who have relevant work experience that allows them to re-use significant portions of their accumulated human capital. When it comes to regions at lower levels of the regional hierarchy, there seems to be higher pressure to hire workers with less suitable backgrounds or from outside the immediate surrounding. With respect to the latter, it has been shown in the literature that worker reallocation across regions is a mechanism of knowledge dissemination across space (Fratesi and Senn 2009; Breschi et al. 2010). Therefore, for firms in more peripheral regions, where the supply of IT workers is rather small, hiring workers from outside the region may help bringing the relevant knowledge. Regional worker reallocation is then an important channel of diffusion of IT competences across regions.

6. Discussion

The results of the empirical investigation in this paper allow generating some interesting conclusions in the light of the literature on general purpose technologies. Scholars in this tradition predicted that the IT, being a pervasive technology, would increasingly penetrate production systems in a wide range of industries through investments in the IT capital, which would eventually lead to generalised productivity gains (David and Wright 1999; Bresnahan 2010). This paper demonstrated that the pervasiveness of the IT realises also through the increased employment of IT workers in both IT providing and IT user sectors.

Industrial decompositions of the labour market for IT workers in Sweden indicated that over the 2000s the IT employment grew rapidly in all sectors, both in absolute terms, but also in terms of the increased IT intensity of employment mixes of different sectors. In particular, 30 thousand new IT jobs were created during the considered time period. Of these, 75 per cent were generated in non-IT sectors, resulting in the decrease of a share of the IT sector in the total IT employment.

This rapid expansion was coupled by the qualitative upgrade of the IT workforce in both IT and non-IT sectors. On the one hand, the share of highly educated individuals was increasing over time at a faster pace than for the economy as a whole. On the other hand, particularly in non-IT sectors, the substitution of technical IT labour by relatively higher skilled IT workers (professionals and managers) was observed. In that respect, not only more IT workers were to be found outside the IT sector, but they became also increasingly skilled and involved in more advanced tasks. In that respect, the diffusion of the IT manifested itself in the ongoing diffusion of (increasingly advanced) IT competences in the Swedish economy beyond the core IT sector. In that respect, the IT demonstrated its pervasiveness also with respect to the labour market dynamics.

At the same time, there are signs of the labour market for IT workers to become increasingly stable over time. The share of individuals remaining employed in the same job between two consecutive years followed the downward trend over the 2000s, despite demonstrating some cyclical variation. However, if individuals did switch jobs, they did so predominantly within the same industry, both in IT and non-IT sectors. This points to the increasing importance of accumulating the industry specific human capital. In other words, the labour market of IT workers was characterised by clusters of worker reallocations in the industry space.

This suggests that the pervasiveness characteristic of the IT as a general purpose technology should be qualified further, at least with respect to the labour market dynamics it generates. Instead of talking of pervasiveness *per se*, one should talk about *specialised pervasiveness*. That is, the IT, indeed, possesses the capacity to generate IT-related jobs in all sectors of the economy (thus, pervasiveness). At the same time, the segmentation of the labour market for IT workers point to the increasing specialisation of IT workers that may limit employment opportunities for them in terms of finding an IT job in another industry (thus, specialised).

The development of the labour market for IT workers was also differentiated spatially as regions at different levels of the Swedish regional hierarchy demonstrated rather different patterns of organisation of the IT labour market. For instance, metropolitan areas played a major role as regional labour markets for IT workers as compared to more peripheral regions, which resembles the situation observed for the IT sector itself (Martynovich and Henning 2016). Nevertheless, despite the strict hierarchies, the expansion of IT employment took place in all regions, both in absolute terms and through increasing the IT intensity of their regional employment mixes. In that respect, the diffusion of IT competences realised not only across industries but also across regions.

Regional decompositions of the labour market for IT workers suggest that an important factor of development of regional IT labour markets is the local availability of the IT workforce. On the one hand, the share of non-IT sectors in regional IT employment mixes increased with each level of the regional hierarchy. It is known that firms involved in knowledge-intensive activities tend to locate in large regions (Wood 2005) due to the combination of the supply and demand factors. In that respect, the increased share of non-IT sectors in the regional IT employment in regions at lower level of the regional hierarchy may be explained by the fact that, in a situation of the limited supply of IT-related activities, firms in these regions are forced to develop IT competences in the Swedish economy were fuelled more by non-IT sectors in more peripheral regions, while the expansion was more balanced in metropolitan areas.

At the same time, firms located in more peripheral regions tend to employ individuals with less specific skill profile as compared to firms located in large regions. While this might be attributed to the specificity of IT activities in which IT workers are involved in different regions, it might also be the issue of smaller pools of IT workers in the periphery, and thus, availability. In this situation, firms located in smaller regions tend to hire workers from outside their immediate surroundings. This creates a mechanism for IT competences diffusion in space.

With respect to the IT as a general purpose technology, this implies that regional labour markets, with associated structures of supply of and demand for the GPT-related skills, may impose barriers to the successful diffusion of a new technology, particularly to more peripheral regions. Even under
assumption that capital markets are absolutely fluid and that investments are made in cutting edge IT capital in peripheral regions, it is the specialised IT labour that is instrumental for making the best use of IT capital being rolled out (Forth and Mason 2004; Hagsten and Sabadash 2014). Limits in availability of specialised human capital may then constrain the possibilities for IT-enabled development. More generally, ensuring the diffusion of the general purpose technology requires not only investments in related capital assets but also corresponding accumulation of the specific human capital at the regional level.

7. Conclusion

This paper investigated the diffusion of IT competences in the Swedish economy by looking at the patterns of IT workers employment and reallocation between 2001 and 2010. The results of the study suggest that the rapid expansion of the labour market for IT workers was accompanied by changes in the organisation of the market with respect to emergence of clusters in the industry space as well as reinforcement of regional hierarchies. It is suggested that with regards to the labour market dynamics the pervasiveness of the IT as a general purpose technology is not universal, but rather specialised.

Analyses performed in this paper are associated with certain limitations which open up space for further research on the topic. First, data restrictions made it not possible to address earlier periods of the IT diffusion. There are indications that the expansion of the labour market for IT workers slowed down in the 2000s (Martynovich and Henning 2016). Therefore, considering the period before 2001 might add an important qualification to the tendencies observed in this paper. Second, more thorough analysis of importance of the industry-specific human capital may be performed. More research on income dynamics and employment prospects of IT workers possessing such human capital in different sectors is desirable. Finally, there is an emerging empirical evidence for firms to backsource IT activities, that is pulling back in-house activities that were previously outsourced (Kern et al. 2002). There are indications that backsourcing becomes one of the key strategies in improving the access to the IT resources at the firm level (for a review, see Dibbern et al. 2004). This makes it interesting to investigate to what extent the outpacing

expansion of IT employment outside the IT sector is driven by the trend toward backsourcing, or whether it is a complementary development.

Acknowledgments

The author is grateful for constructive advice and critical comments on earlier versions of this paper by Karl-Johan Lundquist, Martin Henning, Rikard Eriksson, and Agnes Andersson Djurfeldt.

References

- Adams, T. L., & Demaiter, E. I. (2008). 'Skill, education and credentials in the new economy: the case of information technology workers'. *Work, Employment* & Society, Vol. 22(2), pp. 351-362.
- Adermon, A., & Gustavsson, M. (2015). 'Job Polarization and Task-Biased Technological Change: Evidence from Sweden, 1975–2005'. *The Scandinavian Journal of Economics*, Vol. 117(3), pp. 878-917.
- Aghion, P., David, P. A., & Foray, D. (2006). Linking policy research and practice in 'STIG Systems': Many obstacles, but some ways forward. SIEPR Discussion Paper No. 06-09, Stanford Institute for Economic Policy Research, Stanford.
- Anderson, T. (2014). 'Malmo: A city in transition'. Cities, Vol. 39, pp. 10-20.
- Autor, D. H., Levy, F., & Murnane, R. J. (2003). 'The Skill Content of Recent Technological Change: An Empirical Exploration'. *The Quarterly Journal* of Economics, Vol. 118(4), pp. 1279-1333.
- Bachmann, R., & Burda, M. C. (2010). 'Sectoral Transformation, Turbulence and Labor Market Dynamics in Germany'. *German Economic Review*, Vol. 11(1), pp. 37-59.
- Bassellier, G., & Benbasat, I. (2004). 'Business Competence of Information Technology Professionals: Conceptual Development and Influence on IT-Business Partnerships'. *MIS Quarterly*, Vol. 28(4), pp. 673-694.
- Becker, G. (1964). Human Capital. Chicago, IL: University of Chicago Press.
- Belloc, M., & Guerrieri, P. (2015). 'Impact of ICT diffusion and adoption on sectoral industrial performance: evidence from a panel of European countries'. *Economia Politica*, Vol. 32(1), pp. 67-84.
- Bienkowska, D. (2007). Arbetskraftens rörlighet och klusterdynamik. En studie av IT- och telekomklustren i Kista och Mjärdevi. Geografiska Regionstudier 75. Uppsala: Uppsala Universitet.
- Böckerman, P., Laaksonen, S., & Vainiomäki, J. (2016). Are Jobs More Polarized in ICT Firms? IZA Discussion Paper No. 9851. Bonn: Institute for the Study of Labour.

- Breschi, S., Lenzi, C., Lissoni, F., & Vezzulli, A. (2010). The geography of knowledge spillovers: The role of inventors' mobility across firms and in space. In R. Boschma & R. Martin (Eds.), *The Handbook of Evolutionary Economic Geography*. Cheltenham and Northampton, MA: Edward Elgar, pp. 353-369.
- Bresnahan, T. (2010). Chapter 18 General Purpose Technologies. In B. Hall & N. Rosenberg (Eds.), *Handbook of the Economics of Innovation (Vol. 2)*. Oxford: Elsevier, pp. 761-791.
- Bresnahan, T. F., Brynjolfsson, E., & Hitt, L. M. (2002). 'Information Technology, Workplace Organization, and the Demand for Skilled Labor: Firm-Level Evidence'. *The Quarterly Journal of Economics*, Vol. 117(1), pp. 339-376.
- Bresnahan, T. F., & Trajtenberg, M. (1995). 'General purpose technologies 'Engines of growth'?'. *Journal of Econometrics,* Vol. 65(1), pp. 83-108.
- Buhler, P. A., & Vidal, J. M. (2005). 'Towards Adaptive Workflow Enactment Using Multiagent Systems'. *Information Technology and Management*, Vol. 6(1), pp. 61-87.
- Byrd, T. A., Lewis, B. R., & Turner, D. E. (2006). The Impact of IT Personnel Skills on IS Infrastructure and Competitive IS. In K.-P. Mehdi (Ed.), Advanced Topics in Information Resources Management, Volume 5. Hershey, PA, USA: IGI Global, pp. 63-91.
- Cardona, M., Kretschmer, T., & Strobel, T. (2013). 'ICT and productivity: conclusions from the empirical literature'. *Information Economics and Policy*, Vol. 25(3), pp. 109-125.
- Chadwick, A., Glasson, J., & Smith, H. L. (2008). 'Employment Growth in Knowledge-Intensive Business Services in Great Britain during the 1990s
 — Variations at the Regional and Sub-Regional Level'. *Local Economy*, Vol. 23(1), pp. 6-18.
- Cohen, W. M., & Levinthal, D. A. (1990). 'Absorptive Capacity: A New Perspective on Learning and Innovation'. *Administrative Science Quarterly*, Vol. 35(1), pp. 128-152.
- Currid, E., & Stolarick, K. (2010). 'The Occupation—Industry Mismatch: New Trajectories for Regional Cluster Analysis and Economic Development'. Urban Studies, Vol. 47(2), pp. 337-362.

- David, P. A., & Wright, G. (1999). General Purpose Technologies and Surges in Productivity: Historical Reflections on the Future of the ICT Revolution. Discussion Papers in Economic and Social History Number 31. Oxford: University of Oxford.
- Dibbern, J., Goles, T., Hirschheim, R., & Jayatilaka, B. (2004). 'Information systems outsourcing: a survey and analysis of the literature'. *SIGMIS Database*, Vol. 35(4), pp. 6-102.
- Draca, M., Sadun, R., & Van Reenen, J. (2007). Productivity and ICTs: A review of the evidence. In R. Mansell, D. Avgerou, D. Quah, & R. Silverstone (Eds.), Oxford Handbook of Information and Communication Technologies. Oxford: Oxford University Press, pp. 100-147.
- Earl, M. (1996). 'The Risks of Outsourcing IT'. *Sloan Management Review*, Vol. 37(3), pp. 26-32.
- Ensmenger, N. L. (2003). 'Letting the 'Computer Boys' Take Over: Technology and the Politics of Organizational Transformation'. *International Review* of Social History, Vol. 48(Supplement), pp. 153-180.
- European Commission. (2010). Europe's Digital Competitiveness Report. Comission Staff Working Document, SEC (2010) 627. Brussels: European Comission.
- Fiala, P. (2005). 'Information sharing in supply chains'. *Omega*, Vol. 33(5), pp. 419-423.
- Forman, C., Goldfarb, A., & Greenstein, S. (2005). 'Geographic location and the diffusion of Internet technology'. *Electronic Commerce Research and Applications*, Vol. 4(1), pp. 1-13.
- Forth, J., & Mason, G. (2004). Information and Communication Technology (ICT) Adoption and Utilisation, Skill Constraints and FirmLevel Performance: Evidence from UK Benchmarking Surveys. NIESR Discussion Paper No. 234. London: National Institute of Economic and Social Research.
- Fratesi, U., & Senn, L. (2009). Regional growth, connections and economic modelling: an introduction. In U. Fratesi & L. Senn (Eds.), Growth and innovation of competitive regions: the role of internal and external connections. Berlin: Springer, pp. 3-28.

- Goos, M., & Manning, A. (2007). 'Lousy and Lovely Jobs: The Rising Polarization of Work in Britain'. *Review of Economics and Statistics*, Vol. 89(1), pp. 118-133.
- Gordon, S. R., & Tarafdar, M. (2007). 'How do a company's information technology competences influence its ability to innovate?'. *Journal of Enterprise Information Management*, Vol. 20(3), pp. 271-290.
- Gunasekaran, A., & Ngai, E. W. T. (2004). 'Information systems in supply chain integration and management'. *European Journal of Operational Research*, Vol. 159(2), pp. 269-295.
- Hagsten, E., & Sabadash, A. (2014). The Impact of Highly-skilled ICT labour on Firm Performance: Empirical evidence from six countries. Digital Economy Working Paper 2014/02, Institute for Prospective Technological Studies, Joint Research Centre, European Comission.
- Hawk, S., Kaiser, K. M., Goles, T., Bullen, C. V., Simon, J. C., Beath, C. M., ... Frampton, K. (2012). 'The Information Technology Workforce: A Comparison of Critical Skills of Clients and Service Providers'. *Information Systems Management*, Vol. 29(1), pp. 2-12.
- ICTC. (2015). *Digital Economy Annual Review 2015*. Ottawa: The Information and Communications Technology Council.
- IES. (2002). An Assessment of Skill Needs in Information and Communication Technology. Nottingham: Institute for Employment Studies.
- Kambourov, G., & Manovskii, I. (2009). 'Occupational Mobility and Wage Inequality'. *The Review of Economic Studies*, Vol. 76(2), pp. 731-759.
- Karlsson, C., Maier, G., Trippl, M., Siedschlag, I., Owen, R., & Murphy, G. (2010). *IT and Regional Economic Dynamics: A Literature Review*. EUR 24510 EN. Luxembourg: Publications Office of the European Union.
- Kern, T., Willcocks, L. P., & van Heck, E. (2002). 'The Winner's Curse in IT Outsourcing: Strategies for Avoiding Relational Trauma'. *California Management Review*, Vol. 44(2), pp. 47-69.
- King, W. R., & Malhotra, Y. (2000). 'Developing a framework for analyzing IS sourcing'. *Information & Management*, Vol. 37(6), pp. 323-334.

- Kretschmer, T. (2012). *Information and communication technologies and productivity growth: A survey of the literature*. OECD Digital Economy Papers 195. Paris: OECD.
- Lee, D. M. S., Trauth, E. M., & Farwell, D. (1995). 'Critical Skills and Knowledge Requirements of IS Professionals: A Joint Academic/Industry Investigation'. *MIS Quarterly*, Vol. 19(3), pp. 313-340.
- Liao, H., Wang, B., Li, B., & Weyman-Jones, T. (2016). 'ICT as a general-purpose technology: The productivity of ICT in the United States revisited'. *Information Economics and Policy*, pp. 1-16.
- López-Bassols, V. (2002). ICT Skills and Employment: OECD Publishing.
- Lundmark, M. (1995). 'Computer Services in Sweden: Markets, Labour Qualifications and Patterns of Location'. *Geografiska Annaler. Series B, Human Geography*, Vol. 77(2), pp. 125-139.
- Martynovich, M., & Henning, M. (2016). LABOUR FORCE BUILDING IN A RAPIDLY EXPANDING SECTOR. CRA Working Paper 2016:3.
 Gothenburg, Sweden: Centre for Regional Analysis, School of Business, Economics and Law, Gothenburg University
- Martynovich, M., & Lundquist, K.-J. (2016). 'Technological Change and Geographical Reallocation of Labour: On the Role of Leading Industries'. *Regional Studies*, Vol. 50(10), pp. 1633-1647.
- McCray, G. E., & Clark, T. D. (1999). 'Using system dynamics to anticipate the organizational impacts of outsourcing'. *System Dynamics Review*, Vol. 15(4), pp. 345-373.
- Neffke, F., Otto, A., & Weyh, A. (2016). *Inter-industry labor flows*. Available at <u>http://frankneffke.com/files/NeffkeOttoWeyh2016.pdf</u>.
- Nelson, R. R., & Winter, S. G. (1982). An Evolutionary Theory of Economic Change. Cambridge, MA: Belknap Press.
- NUTEK. (2004). Analyser för regionalt utvecklingsarbete En handbok med praktiska tips och metodexempel. Stockholm: Verket för näringslivsutveckling (NUTEK).
- Østergaard, C., & Dalum, B. (2006). GROWTH AND EVOLUTION OF THE DANISH IT SECTOR: GEOGRAPHICAL CONCENTRATION,

SPECIALISATION, AND DIVERSITY. Paper presented at the DRUID Summer Conference 2006, Copenhagen, Denmark.

- Parent, D. (2000). 'Industry-specific capital and the wage profile: Evidence from the national longitudinal survey of youth and the panel study of income dynamics'. *Journal of Labor Economics*, Vol. 18(2), pp. 306-323.
- Peneder, M. (2003). 'THE EMPLOYMENT OF IT PERSONNEL'. National Institute Economic Review, Vol. 184, pp. 74-85.
- Poletaev, M., & Robinson, C. (2008). 'Human capital specificity: evidence from the Dictionary of Occupational Titles and Displaced Worker Surveys, 1984– 2000'. Journal of Labor Economics, Vol. 26(3), pp. 387-420.
- Prabhakar, B., Litecky, C. R., & Arnett, K. (2005). 'IT skills in a tough job market'. *Communications of the ACM*, Vol. 48(10), pp. 91-94.
- SCB. (1998). Standard för svensk yrkesklassificering. Meddelanden i samordningsfrågor för Sveriges officiella statistik MIS1998:3. Örebro, Sweden: SCB-Tryck.
- SCB. (2003). Standard för svensk näringsgrensindelning 2002. Meddelanden i samordningsfrågor för Sveriges officiella statistik MIS2003:2. Örebro, Sweden: SCB-Tryck.
- SCB. (2010). Lokala arbetsmarknader egenskaper, utveckling och funktion. Örebro, Sweden: SCB.
- Spitz-Oener, A. (2006). 'Technical Change, Job Tasks, and Rising Educational Demands: Looking outside the Wage Structure'. *Journal of Labor Economics*, Vol. 24(2), pp. 235-270.
- Stanley, T. D., Doucouliagos, C., & Steel, P. (2015). Does ICT Generate Economic Growth? A Meta-Regression Analysis. School Working Paper Economics Series 2015/9. Burwood, Australia: School of Accounting, Economics and Finance, Faculty of Business and Law, Deakin University.
- Stiroh, K. (2010). Reassessing the impact of IT in the production function: A metaanalysis and sensitivity tests. In J. Mairesse & M. Trajtenberg (Eds.), *Contributions in memory of Zvi Griliches*. Cambridge, MA: NBER, pp. 529-561.
- Sullivan, P. (2010). 'Empirical evidence on occupation and industry specific human capital'. *Labour Economics*, Vol. 17(3), pp. 567-580.

- Thatcher, M. E., & Pingry, D. E. (2004). 'Understanding the Business Value of Information Technology Investments: Theoretical Evidence from Alternative Market and Cost Structures'. *Journal of Management Information Systems*, Vol. 21(2), pp. 61-85.
- Timmermans, B., & Boschma, R. (2013). 'The effect of intra- and inter-regional labour mobility on plant performance in Denmark: the significance of related labour inflows'. *Journal of Economic Geography*, Vol. 14(2), pp. 289-311.
- Todd, P. A., McKeen, J. D., & Gallupe, R. B. (1995). 'The Evolution of IS Job Skills: A Content Analysis of IS Job Advertisements from 1970 to 1990'. *MIS Quarterly*, Vol. 19(1), pp. 1-27.
- Warhurst, C., Lockyer, C., & Dutton, E. (2006). 'IT jobs: opportunities for all?'. *New Technology, Work and Employment,* Vol. 21(1), pp. 75-88.
- Willcocks, L. P., Lacity, M. C., & Kern, T. (1999). 'Risk mitigation in IT outsourcing strategy revisited: longitudinal case research at LISA'. *The Journal of Strategic Information Systems*, Vol. 8(3), pp. 285-314.
- Wood, P. (2005). 'A service-informed approach to regional innovation or adaptation?'. *The Service Industries Journal*, Vol. 25(4), pp. 429-445.

Article IV

Local or not? Spatial biographies of Swedish IT entrepreneurs

Mikhail Martynovich

Department of Human Geography, Lund University

Abstract: The paper investigates whether spatial biographies of individuals have an impact of their probability to become entrepreneurs as well as postentry success of firms they establish. In particular, the analysis aims at examining the extent of locational inertia of entrepreneurs, traditionally assumed in the literature, as well as a potential trade-off between individuals' local embeddedness and exposure to diverse knowledge accumulation opportunities across different geographical settings. It is claimed that locational inertia of entrepreneurs may be overestimated. The results of analysis suggest that employment histories of individuals have more information to offer for explaining entrepreneurial behavior than simply the contents and location of the most recent employment.

Keywords: entrepreneurship, mobility, locational behavior, spatial biographies, local embeddedness, external knowledge

JEL codes: J61, L26, L86, M13

1. Introduction

As it is often claimed in the literature, entrepreneurship is predominantly a local event. That is, individuals tend to start businesses in regions where they previously have lived and worked (Dahl and Sorenson 2009; Audretsch et al. 2012). This locational inertia of entrepreneurs has been explained as driven by personal factors (Lafuente et al. 2010; Curran et al. 2016) as well as by opportunities to leverage their social capital which is hardly transferrable to other locations (Varga and Schalk 2004; Kalnins and Chung 2006). Moreover, firms established by local founders tend to enjoy a better postentry performance (Dahl and Sorenson 2012).

Yet, the locational inertia of entrepreneurs may be overemphasized. Studying the technology-based cluster in Ottawa, Harrison et al. (2004) demonstrated that while most individuals started new firms nearby their immediate past employer, many of them had relocated there to accumulate human and social capital prior to performing an entrepreneurial entry. Investigating an international in- and out-migration of individuals in the Silicon Valley, Saxenian (2006) went as far as characterizing entrepreneurs as the modern-day Argonauts roving the world in pursuit of their own golden fleeces. Frederiksen et al. (2016) demonstrated that the number of geographical relocations an individual performs prior to starting a business has an impact on its subsequent performance.

Indeed, knowledge-based theories of entrepreneurship suggest that individual mobility, being an important mechanism of knowledge transfer across firms, industries and locations, boosts generation, diffusion and exploitation of entrepreneurial opportunities in different markets (Agarwal et al. 2007; Wright 2011). Entrepreneurs draw on the experience and networks established at different firms and places during their entire career; and these condition the spatial organization of activities they initiate with their businesses (Stam 2010). In that respect, benefits of entrepreneur's local embeddedness, while being supportive to a new firm, may be outweighed by benefits of acquiring diverse knowledge and building networks in different geographical settings. However, spatial biographies of individuals, i.e. histories of spatial relocation, and their impact on entrepreneurial activities have so far received an insufficient attention (Frederiksen et al. 2016). This paper is a step in this direction.

Two research questions are posited. The first question regards the probability of individuals to perform an entrepreneurial entry conditional upon their spatial biographies: *In what way does prior locational behavior of individuals influence their propensity to become entrepreneurs?* This question aims at revisiting the issue of locational inertia and investigating the extent of this phenomenon.

The second question addresses the relationship between spatial biographies of individuals and the performance of new firms they establish: *In what way is the post-entry performance of new firms affected by pre-entry locational behavior of their founders?* This question targets at investigating whether individuals' local embeddedness or their exposure to diverse knowledge accumulation opportunities across different geographical settings matters more for performance of their firms.

To address these questions, the paper focuses on spatial biographies of entrepreneurs in the Swedish knowledge-intensive IT services sector between 1991 and 2010. The sector is of an interest for several reasons. First, being a prominent example of the new entrepreneurial economy (Audretsch and Thurik 2001), the sector has experienced a remarkable growth in terms of new firm formation, employment and value added (Lasch et al. 2013). Second, knowledge-intensive services are increasingly identified as key nodes in knowledge networks fostering processes of knowledge creation and diffusion (Doloreux and Muller 2007). The sector, therefore, serves as a suitable research setting for analyzing the relationship between individual mobility (and associated knowledge flows), on the one hand, and entrepreneurial activity, on the other. Third, while IT services are often characterized as 'footloose' activities (Audretsch et al. 2011), the empirical evidence to date suggests that the spatial pattern of the sector remains citydominated (Martynovich and Henning 2016). In that respect, it is interesting to investigate the locational behavior of entrepreneurs in the sector with supposedly limited spatial constraints.

Further, the analysis is restricted to entrepreneurial entries performed outside of the Swedish metropolitan areas for the following reasons. First, while many studies focus on entrepreneurship in urban contexts, less is known about entrepreneurial behavior in more peripheral regions (Bürcher et al. 2016). Second, since the knowledge-intensive IT services is a city-dominated sector, peripheral regions provide limited opportunities for acquiring sectorspecific knowledge necessary to perform a successful entrepreneurial entry. At the same time, smaller regions are characterized by more tightly woven social fabric leading to better local networking opportunities (Atterton 2007). Therefore, peripheral regions provide a great setting for analyzing a possible trade-off between local embeddedness and deploying external knowledge. In that respect, a particular emphasis is made on whether spatial biographies of entrepreneurs involved an employment experience in metropolitan areas which are the hotspots for knowledge accumulation in the sector.

Provided the setting discussed above then, the paper contributes empirically to the discussion on locational behavior of entrepreneurs by suggesting a methodology for constructing detailed spatial biographies of individuals and analyzing their impact on propensity of individuals for entrepreneurship and post-entry success of their businesses.

Conceptual contribution of the paper lies in demonstrating that the employment histories of individuals, not least their spatial biographies, have more to offer than the contents and location of the most recent employment. It is demonstrated that individuals with broad spatial biographies are, in certain cases, more likely to become entrepreneurs, and more successful ones for that matter, than local individuals. In that respect, locational inertia of entrepreneurs is, to a certain extent, overestimated.

The remainder of the paper is organized as follows. In Section 2, a brief review of the relevant literature is performed, while research questions formulated in the introduction are developed into a series of hypotheses to be tested. Section 3 provides a detailed discussion of the data, definitions, and methods employed. Section 4 presents a descriptive evidence on entrepreneurship in the Swedish knowledge-intensive IT services sector. Modeling results are presented in Section 5. Section 6 positions the major findings through the lens of current debates in the literature and provides concluding remarks.

2. Locational behavior of individuals and entrepreneurship

Process of entrepreneurship is spatially uneven (Stam 2010). Empirical evidence suggests that entrepreneurship is, to a great extent, a local event (Sorenson and Audia 2000; Audretsch et al. 2012). That is, new firms tend to

be established and stay in regions where their founders have previously lived and worked (Pellenbarg et al. 2002; Michelacci and Silva 2007). In that respect, the question for an entrepreneur is often what kind of firm to start in a given location, rather than selecting a location for a given firm (Stam 2007).

This locational inertia of entrepreneurs might, on the one hand, be attached to their willingness to remain close to family and friends (Gimeno et al. 1997; Dahl and Sorenson 2012) and/or attain a certain quality of life (Marcketti et al. 2006; Lafuente et al. 2010). Such personal motivations often outweigh market-related locational factors (Curran et al. 2016).

On the other hand, entrepreneurs are economically and relationally embedded in places (Dahl and Sorenson 2009). This embeddedness stems from the location-specific social capital in the form of localized network externalities which are tangible assets with a limited transferability across space (Varga and Schalk 2004; Kalnins and Chung 2006). It improves the access to information and financial resources at a pre-entry stage (Shane and Stuart 2002; Nijkamp 2003) and plays an important role in continuing operations of new firms as their development is conditional upon establishing linkages to customers and suppliers as well as hiring employees. These are easier when an entrepreneur has an extended local network (Westlund and Bolton 2003). Local embeddedness may be particularly important in more peripheral regions as these are characterized by stronger informal networks (Atterton 2007; Dahl and Sorenson 2012).

In that respect, embeddedness in local social networks may be expected to lead to an increased quality of local knowledge exchange (Bürcher et al. 2016) as well as reduced risks associated with entrepreneurial entries (Westlund and Bolton 2003). This leads to the first two hypotheses to be tested in this paper:

Hypothesis 1a: Individuals are more likely to perform local entrepreneurial entries

Hypothesis 1b: Local entrepreneurial entries enjoy better post-entry performance

At the same time, entrepreneurial entries benefit from tapping into a variety of knowledge sources across space (Lorentzen 2008; Sacchetti 2009). It provides the access to new non-redundant knowledge (Maskell 2014) and allows avoiding risks of spatial lock-in and short-sightedness of learning (Liu et al. 2010). In that respect, an individual mobility across locations has been

proposed as a channel of knowledge flows to newly founded firms (Saxenian 2006; Agarwal et al. 2007). These flows are related to entrepreneurial entries and their performance (Helfat and Lieberman 2002; Frederiksen et al. 2016).

Previous studies of locational behavior of entrepreneurs suggest that their locational inertia may be overestimated. It is possible that individuals are not so much rooted in place but rather have histories of spatial relocation in their pre-entry careers through which they accumulate knowledge necessary for establishing a new firm (Harrison et al. 2004; Reuschke 2014). Moreover, mobile individuals, acquiring knowledge in different geographical settings, may make the most promising entrepreneurs (Dahl and Sorenson 2012; Frederiksen et al. 2016).

Indeed, geographical mobility exposes individuals to new opportunities, which may be recognized and exploited, in different regional markets (Wright 2011). Besides, a broad spatial experience of entrepreneurs allows them to expand their social network with potential customers, suppliers, etc. In that respect, broad networks, and particularly their spatial configurations (Bahlmann 2015), provide individuals with a better position for discovering and exploiting entrepreneurial opportunities. Yet again, because of limited knowledge exchange opportunities, new firms in more peripheral regions are particularly dependent on external knowledge sources (Bürcher et al. 2016).

Therefore, systematic differences may be expected in individuals' propensity to become entrepreneurs as well as in performance of their firms between those with and without histories of spatial relocation. This provides the basis for the next two hypotheses.

Hypothesis 2a: Individuals' histories of spatial relocation affect their propensity for entrepreneurship

Hypothesis 2b: Post-entry performance of new firms is conditional upon spatial biographies of their founders

Provided that a history of spatial relocation matters for knowledge acquisition and building social networks, it may be important not only whether an individual relocated but also *where* she relocated. Prospects of learning depend on the combination of individuals' capacities and motivations, on the one hand, and an access to jobs offering opportunities for acquiring marketable assets, such as tacit knowledge and social networks, on the other (Gordon et al. 2015). In that respect, metropolitan areas may play a particularly pronounced role. Metropolitan regions, characterized by high densities of human capital and knowledge spillovers, provide potential entrepreneurs with superior opportunities for learning and building social ties that may constitute the basis for a new firm (Acs et al. 2009; Gordon 2015). In that respect, these areas may act as attraction hotspots for individuals willing to work in the sector, acquire sector-specific knowledge, and more generally 'get on' in their careers (Glaeser and Saiz 2003). Relocation to metropolitan areas is then a precondition for if not a mechanism of a rapid achievement of their career oriented goals (Mulder and van Ham 2005). This allows characterizing such regions as 'escalator regions' (Fielding 1992).¹

Having acquired relevant knowledge, individuals often relocate to more peripheral regions, where learning opportunities are fewer, but accumulated knowledge assets can be deployed in a more efficient way (Gordon et al. 2015). What is interesting, stepping off the 'metropolitan escalator' is often associated with performing an entrepreneurial entry in a destination region (Findlay et al. 2000; Reuschke 2015). Fielding (1989) referred to this as an export of the metropolitan 'entrepreneurial culture'.

In that respect, employment experience in a metropolitan area, and superior opportunities for learning and networking associated with that, may compensate for the absence of social capital in a region where entrepreneurial entry is performed. Provided that an individual has obtained advanced knowledge, she will be in a better position to identify and exploit entrepreneurial opportunities in a destination region. This suggests the third set of hypotheses to be tested.

Hypothesis 3a: Metropolitan employment experience increases the probability of an individual to perform an entrepreneurial entry in a peripheral region

Hypothesis 3b: Post-entry performance of entrepreneurial entries is positively associated with employment experience of firm founders in metropolitan areas

¹ Originally, the idea of a 'metropolitan escalator' applied to global cities (e.g., London). More recent research suggests that escalator effects are observed also for second-order metropolitan areas (Champion et al. 2014).

3. Data and methodological considerations

3.1. Data and definitions

The data employed in this paper comes from the Longitudinal Integration Database for Health Insurance and Labor Market Studies provided by Statistics Sweden. For each individual registered in Sweden it contains a set of personal attributes as well as linkage to establishment of her main employment for which a type of employment, location and industry affiliation are known.

Location of individuals and firms is recorded at the municipality level. For purposes of the analysis, 290 Swedish municipalities are aggregated to 90 local labor markets (LLMs)². This allows minimizing the risk of recording changing commuting patterns between municipalities as a spatial relocation. Metropolitan areas are then defined as LLMs of three largest Swedish cities, namely Stockholm, Gothenburg and Malmö. Remaining 87 local labor markets are considered as non-metropolitan regions.

The sampling frame of the database is the complete set of individuals and firms in the knowledge-intensive IT services sector. The latter includes five 5-digit industries under the activity code 72 'computer and related activities' according to the Swedish Standard Industrial Classification: (1) hardware consultancy; (2) software consultancy; (3) software supply; (4) data processing; (5) database activities.

Entrepreneurial entries are defined as new legal organizations (Frederiksen et al. 2016) and were identified in two steps. First, for each year between 1991 and 2010 the sample of all establishments and firms (establishments belonging to the same parent firm are linked through a shared firm identifier) operating in the knowledge-intensive IT services sector was derived. Second, for each year *t* new entries were identified according to the following rule: (1) an identifier for an establishment existed at year *t* but not year *t*-1, and (2) an identifier of a firm, to which an establishment was linked, existed for year *t* but not year *t*-1. The second condition ensured that establishments entering

² LLMs are defined by the statistics on commuting between municipalities in the way that maximizes the self-containment of commuting flows.

through incumbent firms' expansion were excluded. As a result, 25457 entrepreneurial entries were identified over the considered time period.

Based on the income information distinguishing between incomes derived from wages and from business activities, founders behind each entry were identified. An entrepreneur is, therefore, defined as a person establishing a new firm. The sample was further restricted to entries with one founder. While new firms with multiple founders represent an interesting case from the perspective of the knowledge-based view of entrepreneurship – they might benefit from recombination of knowledge and social capital accumulated by all firm founders – considering them would require composing combined measures of their founders' backgrounds, bringing unnecessary complications into the analysis. 1586 entries were, thus, excluded from the sample. The resulting dataset included 23871 entries, of which 7492 (31.4 per cent) were performed in non-metropolitan regions.

3.2. Dependent variables and modelling issues

The first dependent variable, a propensity of an individual to become an entrepreneur, is defined as a binary variable indicating whether an individual becomes a founder of a new firm at year t, which did not exist at year t-1. In order to account for the possibility that individuals' spatial histories may be related to any type of job switches (entrepreneurial entry vs. getting a paid job), a reference group to individuals becoming entrepreneurs at time tconsists of individuals getting a paid job in the sector at time t (Nanda and Sørensen 2010). Individuals, who perform a change in their job situation at certain time, may be characterized by different opportunity structures and risk attitudes as compared to individuals, who do not (Sørensen and Sharkey 2014). At the very least, the latter individuals are more likely to be satisfied with their current job situation, meaning that starting a business at a certain point in time is not in the list of their options. Besides, as the data at hand does not allow to define the motivation behind starting a business, it is at least possible to consider that as partly driven by a decision to change a current job situation. Given these considerations, it was job switchers, and not the whole working population, which was decided as a reference group.

Formally, the first dependent variable is operationalized in the following way:

Entrepreneur = $\begin{cases} 1, \text{ if an individual starts a new firm in IT services at time } t \\ 0, \text{ if an individual gets a new job in IT services at time } t \end{cases}$

The impact of spatial biographies of individuals on their propensity for entrepreneurship is then estimated as discrete-time event history models using logistic regression (Nanda and Sørensen 2010).

The second dependent variable, a post-entry performance of a new firm, is operationalized as its survival time and measured as the number of years the firm exists after entry. Firm survival rates are modelled using the semi-parametric Cox proportional hazard model (Cox 1972). The advantage of this widely used approach in survival analysis lies in its flexibility and robustness: neither does the model necessitate any assumptions about the distribution of survival rates, so that no restrictions are imposed on a shape of the hazard function, nor does it require estimating the baseline hazard.

Formally, the model is specified in the following way:

$$h(t) = h_o(t) * e^{X\beta}$$

where h(t) – is a hazard function at time *t* after entry, $h_o(t)$ – a baseline hazard function (an unspecified hazard function reflecting the risk of failure provided that a firm survived until time *t*), *X* – a vector of covariates, and β – a vector of coefficients. Parameters in the vector β are estimated using the partial likelihood method, which focuses on the ordering of establishment exits and calculates the conditional probability of failure for firms that are observed to exit.

As the data is available until 2010, all firms that still exist at this year are right censored at that time point but are included to the analysis prior to this date.

3.3. Operationalizing spatial biographies of individuals

The spatial organization of entrepreneurial activities, initiated by individuals, are defined by their employment biographies (Stam 2010). It was, therefore, necessary to come up with an operationalization of individuals' experience prior to starting a new firm. This was performed by generating six-year employment histories for all individuals employed in the sector between 1991 and 2010³. These included an employment status, regions of employment and residence, an industry of employment, etc. Based on this, several

³ The annual data is available from 1985, which allows going back maximum six years from the first year in the considered time period (1991).

variables capturing biographies of individuals were specified. These included three variables describing their spatial biographies.

The first, *Entry location*, describes whether an individual switched regions immediately prior performing an entrepreneurial entry. It is formally defined in the following way:

 $\text{Entry location} = \begin{cases} 0, \text{ if } region_t = region_{t-1} \\ 1, \text{ if } region_t \neq region_{t-1} \text{ and } region_{t-1} \text{ is a metropolitan region} \\ 2, \text{ if } region_t \neq region_{t-1} \text{ and } region_{t-1} \text{ is a non - metropolitan region} \end{cases}$

where $region_t - local$ labor market region where an entry is performed at year t, $region_{t-1} - local$ labor market of employment (for employed individuals) or residence (for unemployed individuals) at year t-l. Three possible situations are described by the following qualitative categories: local entrants (if *Entry location=*0), movers from metropolitan regions (if *Entry location=*1), and movers from non-metropolitan regions (if *Entry location=*2).

The second spatial biography variable, *Prior relocation*, captures individuals' relocation patterns over six years of recorded employment history. Four relocation patterns are defined:

- 1. *stayers* individuals who remained in the same region;
- 2. *one-time movers* individuals who switched employment regions once;
- 3. *onward movers* individuals who switched employment regions several times, without returning to the region of employment at the first year of recorded employment history;
- 4. *returners* individuals who switched employment regions but returned to the region of employment at the first year of recorded employment history.

This variable is an extended measure of spatial biographies of individuals.

Finally, the third spatial biography variable, *Metropolitan employment tenure*, is simply the number of years (within six years of recorded employment history) that an individual spent being employed in a metropolitan area. Appendix A1 illustrates possible interactions between three spatial biography variables.

One important reservation should be made with regards to the spatial biographies of individuals and their impact on entrepreneurial behavior. The nature of the data does not allow distinguishing between joint motivations of individuals to relocate and to start a business. There are two possible situations. First, a decision to relocate might come first (e.g., for personal reasons). It is then possible that the decision to start a new firm may be driven by, for example, an absence of suitable employment opportunities in a destination region. In other words, starting a new business is an outcome of spatial relocation. On the other hand, an individual might have identified an entrepreneurial opportunity in a certain region prior to making a decision to relocate there. That is, an entrepreneurial opportunity is driving a relocation decision. Therefore, the association between spatial biographies of individuals and their entrepreneurial activities should be interpreted with caution.

3.4. Control variables

The extent to which mobility of individuals can induce entrepreneurial activities may vary with individuals' capacity to coordinate new knowledge and their knowledge acquisition efficiency (Qian and Acs 2013). Prior experiences expose individuals to different opportunity costs structures (Gimeno et al. 1997) and affect their potential to capitalize on knowledge flows (Cohen and Levinthal 1990). Thus, they influence the probability of recognizing and exploiting entrepreneurial opportunities (Shane 2000). Therefore, several control variables were included to account for effects previously demonstrated in the literature to have an impact on entrepreneurial behavior.

First, a dummy variable (*Higher education*) is added to control whether an individual has been involved in higher education. Second, as previous entrepreneurial experience has an impact on an ability to identify and exploit entrepreneurial opportunities (Audretsch et al. 2006), a dummy variable (*Entrepreneurial experience*) is introduced to capture whether an individual has established a firm within six year of recorded employment history (in other words, whether an individual possesses an entrepreneurial capital). Third, knowledge about technology, competitors, customer needs, etc. can often be learned only through industry participation (Klepper 2001). Besides, individuals with industry experience can make use of social networks from

their prior settings (Colombo and Grilli 2005). To control for possible effects of prior employment in the sector, a variable measuring years of employment experience in the knowledge-intensive IT services (*Sector experience*) is included in the analysis⁴. Finally, length of unemployment (in years) prior to starting a firm or getting a job in the sector is recorded.

Besides, previous research highlighted the impact of demographic characteristics of individuals on their entrepreneurial decisions. Therefore, gender effects are captured with a dummy variable (Sørensen 2007). Also, potential linear and non-linear effects of age are captured by additional control variables (Dunn and Holtz-Eakin 2000).

In survival models controls for employment in a new firm at the year of entry as well as total employment in the IT services sector in a region of entry are included.

4. Entrepreneurial activity in the Swedish knowledge-intensive IT services sector

This section presents a descriptive evidence on the extent of entrepreneurship in the Swedish knowledge-intensive IT services sector between 1991 and 2010. To start with, Figure 1 illustrates the total amount of entrepreneurial entries in the sector projected against the sectoral employment.

There is a clear upward trend in both total employment and number of new entries. The latter increased from 196 in 1991 to 2389 in 2010. Quite pronounced is the rapid increase in the number of new entries between 2001 and 2004. Interestingly, this surge in entrepreneurial activity coincided with an overall contraction of the sector in the aftermath of the 2001 dotcom crash. Possible explanation to this may be that adverse labor market conditions forced individuals into entrepreneurship since other employment options were not readily available (Bergmann and Sternberg 2007; Hessels et al. 2008). For individuals, laid off their jobs as the dotcom crash hit the sector, starting a business could be the only opportunity to remain in the sector.

⁴ As this variable is available in the original dataset, it captures the total employment experience, not only employment experience over six years of recorded employment history.



Figure 1. Entrepreneurial entries and employment in the Swedish knowledgeintensive IT services sector

Distribution of entrepreneurial entries across Swedish regions was, however, highly unequal as demonstrated in Figure 2.



Figure 2. Entrepreneurial entries within and beyond metropolitan areas

Between 1991 and 2010, 68.5 per cent of new entries were performed in metropolitan areas. The share of such entries increased from 58.9 per cent in

1991 to 71.4 per cent in 2010. This corresponds to the previously documented pattern of an increased concentration of the knowledge-intensive IT services to metropolitan areas (Martynovich and Henning 2016). It should be noted, however, that the number of entrepreneurial entries in non-metropolitan regions was on the constant rise, increasing from 83 in 1991 to 684 in 2010.

With respect to entry location, a stark difference is observed between metropolitan and non-metropolitan regions (Figure 3).



Figure 3. Entrepreneurial entries by non-local individuals

The share of entries by non-local individuals (that is, individuals who were not employed or residing in a region of entry at *t-1*) was significantly lower in metropolitan regions. In particular, between 1991 and 2010, around 7.2 per cent of entries in metropolitan areas were performed by individuals who relocated at the year of entry. In contrast to that, almost a quarter (22.9 per cent) of new firms in non-metropolitan regions were established by non-local individuals. The slight upward trend in the share of non-local founders may be observed over time. Nevertheless, it is clear that performing an entrepreneurial entry is predominantly a local event.

Finally, with respect to entrepreneurial entries in non-metropolitan areas, which are of the main interest in this paper, distribution of new firms according to the origin of their founders is presented in Figure 4.



Figure 4. Entrepreneurial entries in non-metropolitan regions

Between 1991 and 2010, 14 per cent of entrepreneurial entries were performed by individuals relocating from metropolitan areas. Such entries were more common that those associated with a relocation between non-metropolitan regions (8.9 per cent of all entries). This difference has slightly increased over time in favor of entries from metropolitan areas.

In what follows, the descriptive evidence on relocation patterns of entrepreneurs in the knowledge-intensive IT sector is complemented by the analysis of the impact of individuals' spatial biographies on their propensity to become entrepreneurs as well as post-entry success of new firms.

5. Regression analysis

5.1. Transitions into entrepreneurship

Table 1 reports the results of estimating propensity of individuals for entrepreneurship conditional upon their spatial biographies. Appendix A2 provides a sample summary as well as a correlation table for variables included in the analysis.

Logistic regression	Model 1 Model 2		
Entry location (Local entrant)			
Mover from metro	0.7273***	0.5761***	
	(0.0287)	0.3209***	
Mover from non-metro	(0.0209)	(0.0246)	
Prior relocation (One-time mover)	0.000 /////	0 700 5444	
Stayer	0.8624*** (0.0272)	0.7885*** (0.0260)	
	1.0735	0.9856	
Onward mover	(0.0423)	(0.0540)	
Returner	1.3268*** (0.0500)	0.9757 (0.0497)	
	(0.0000)	(0.0.01)	
(One-time mover)			
Onward mover from metro		1.2692*	
		1.5135***	
Onward mover from non-metro		(0.1615)	
Returner from metro		1.7881***	
		(0.1072) 2 6428***	
Returner from non-metro		(0.2807)	
	0.0000	0.000 (1111	
Higher education (no)	0.8830*** (0.0225)	0.8831*** (0.0225)	
0	0.9114***	0.9104***	
Sector experience	(0.0061)	(0.0061)	
Unemployment spell	1.0743**** (0.0072)	1.0719*** (0.0072)	
	4.3958***	4.3774***	
Entrepreneur experience (no)	(0.1347)	(0.1343)	
Individual controls (age, age ² , sex)	Yes	Yes	
Industry dummies	Yes Yes		
Year dummies	Yes	Yes	
Region dummies	Yes	Yes	
Pseudo-R ²	0.1235	0.1252	
Ν	101789 101789		

Odds ratios are reported. Values above 1 indicate an increased probability of becoming an entrepreneur. Robust standard errors clustered at the individual level are reported in brackets. ***(**,*,¹) indicate significance at 0.1% (1%, 5%, 10%) level. Since odds ratios are reported, a significant value indicates a significant difference from 1. For categorical variables and dummies values in the brackets are reference values. (Fielding 1989)

Table 1. Transitions to entrepreneurship

Local individuals in non-metropolitan regions have the highest propensity to become entrepreneurs (*Entry location*). Probability to start a new firm as compared to obtaining a paid employment position is 27 per cent lower for individuals moving from metropolitan areas to a non-metropolitan region, and 51 per cent lower for movers between non-metropolitan regions (Model 1). In that respect, while individuals relocating from metropolitan areas are more likely to become entrepreneurs than other movers, they are still not as entrepreneurial as their local counterparts.

Considering more detailed spatial biographies of individuals (*Prior relocation*), however, adds an important qualification to this result. Individuals, remaining employed in the same region over six years of recorded employment history (stayers), tend to have the lowest probability to become entrepreneurs (14 per cent less likely than one-time movers). The most likely entrepreneurs are returners: they have 33 per cent higher probability to choose an entrepreneurship option for paid employment than one-time movers (and, 54 per cent higher than stayers). In that respect, while individuals tend to start new firms in regions where they were employed or resided at least a year prior to an entry, it is important to account for history of their previous relocations to be able to estimate more accurately the extent to which being local matters for probability of becoming an entrepreneur.

Moreover, an entry location tends to have significantly different effect on the propensity for entrepreneurship conditional upon individuals' prior relocations history (interaction term in Model 2). Returners from metropolitan areas have 79 per cent higher propensity to start a new firm than one-time movers from metropolitan areas, while it is 27 per cent higher for onward movers with a last relocation from a metropolitan area. Same order of effects is observed for movers between non-metropolitan regions. In that respect, when extended spatial biographies are taken into the consideration, a difference in the propensity for entrepreneurship between local and non-local individuals narrows down, and even disappears for certain groups of movers. For instance, returners from metropolitan areas are 3 per cent more likely to become entrepreneurs than local individuals.

When it comes to control variables, it appears that more educated (*Higher education*) and more experienced (*Sector experience*) individuals have lower propensity for entrepreneurship. In particular, having higher education tends to decrease the probability to become an entrepreneur by 12 per cent, while each additional year of work experience in the sector decreases it by around

9 per cent. This tendency might be driven by the sectoral specificity: low physical capital requirements as well as relative easiness of obtaining general skills in programming (Østergaard and Dalum 2011) might lead to an overrepresentation of young individuals among entrepreneurs in the sector⁵, while more experienced workers would tend to go for more stable and secure employment positions. Another potential explanation is that individuals without official certification may have a hard time to get a paid employment position leaving them with the choice to start their own firms if they wish to work in the sector. Each year of unemployment (*Unemployment spell*) increases the propensity for entrepreneurship by 7 per cent. Finally, individuals with prior entrepreneurship option for paid employment position than those without such experience.

Modelling the propensity for entrepreneurship with regards to spatial biographies of individuals sheds some new light on the locational inertia of entrepreneurs. On the one hand, it is, indeed, the case that local individuals in non-metropolitan regions have a higher tendency to become entrepreneurs compared to relocating individuals. This result provides support for Hypothesis 1a. On the other hand, it is not only a relocation immediately prior to the entry which is important to look at, but also an extended spatial biography of individuals. Different relocation patterns are associated with significantly different propensities for entrepreneurship, which provides an evidence in favor of Hypothesis 2a. Besides, it appears that a relocation direction – from a metropolitan area or from another non-metropolitan region - matters. Individuals moving from the former are more likely to prefer an entrepreneurship option to paid employment than those moving from the latter (but still not as likely as individuals performing a local entry). Moreover, when two measures of spatial biographies are combined, it appears that, for instance, returners from metropolitan areas are as, if not more, likely to start a business as local individuals. In that respect, a mixed evidence is also obtained for Hypothesis 3a. Indeed, metropolitan experience has a positive impact, but only with respect to particular relocation pattern.

All in all, the results of modelling propensity of individuals for entrepreneurship suggest that the locational inertia of individuals takes place, but is, to a certain extent, overestimated. If, however, the spatial biography of

⁵ This is also confirmed by the U-shaped relationship between age and propensity for entrepreneurship (not reported, but available upon request).

an individual matters for her decision to become an entrepreneur, the next question is whether it also matters for a post-entry success of her business. This issue is addressed further.

5.2. Post-entry survival

Table 3 reports the results of estimating Cox proportional hazards model for exit probabilities of new firms with regards to spatial biographies of their founders. Appendix A3 provides a sample summary as well as a correlation table for variables included in the analysis.

As in the case of propensity for entrepreneurship, entrepreneurs' pre-entry relocation patterns matter for survival of firms that they establish (*Prior relocation*). In particular, entries performed by individuals, staying in the same non-metropolitan region over six years of observed employment history, have the highest survival probability. The exit probability is higher for individuals with a history of spatial relocation: around 12-13 per cent higher chance of failure for one-time movers and returners, and almost 30 per cent higher – for onward movers (Model 1). In that respect, prolonged exposure to the regional environment (or, regional tenure), which provides an individual with opportunities for building local social capital, tends to improve survival chances of new entries. In that respect, the locational inertia of individuals is associated not only with their propensity for entrepreneurship, but also on the post-entry success of their businesses.

At the same time, the work experience in a metropolitan area tends to compensate for disadvantages associated with an absence of a regional tenure (*Metropolitan emp. tenure*). Each year of firm founder's employment in a metropolitan area is associated with a 2 per cent decrease in the exit probability (even though this effect is only significant at a 10 per cent level).

The importance of the latter is, however, conditional upon patterns of entrepreneurs' prior spatial relocation patterns (interaction term in Model 2). Returners tend to enjoy the biggest benefits from their metropolitan experience -7 per cent decrease in the exit probability for each year of employment in a metropolitan area. For other movers, this effect becomes somewhat smaller. That returners are the main beneficiaries of the metropolitan employment experience is not really surprising, given that they have an opportunity to take the best of both worlds: having work experience

in the most dynamic IT services clusters in metropolitan areas, providing them with superior learning opportunities, and being able to capitalize on the social capital they have accumulated in their pre-metropolitan lives.

Cox proportional hazards model	Model 1 Model 2		
Prior relocation (Stayer)			
One-time mover	1.1234** 1.1109** (0.0436) (0.0444)		
Onward mover	1.2996*** 1.3005*** (0.0597) (0.0691)		
Returner	1.1284** 1.1735*** (0.0481) (0.0560)		
Metropolitan emp. tenure	0.9809 [⊥] 0.9344* (0.0111) (0.0274)		
Prior relocation * Metropolitan emp. tenure (Returner)			
One-time mover		1.0617 [⊥] (0.0341)	
Onward mover		1.0496 (0.0379)	
Higher education (no)	0.9658 (0.0287)	0.9640 (0.0287)	
Sector experience	0.9512*** (0.0082)	0.9515*** (0.0082)	
Unemployment spell	1.0185* (0.0081)	1.0182* (0.0081)	
Entrepreneur experience (no)	$\begin{array}{ccc} 0.8967^{\bot} & 0.8955^{*} \\ (0.0505) & (0.0504) \end{array}$		
Firm employment at entry	0.7669*** (0.0340)	0.7671*** (0.0340)	
Regional IT employment at entry (log)	0.9962 0.9964 (0.0191) (0.0191)		
Individual controls (age, age ² , sex)	Yes	Yes	
Industry dummies	Yes	Yes	
Year dummies	Yes Yes		
Region dummies	Yes	Yes	
N of subjects	6580	6580	
N of failures	4820	4820	
Log-likelihood	-38519.55	-38517.75	

Hazard ratios are reported. Values above 1 indicate an increased probability of exit. Robust standard errors are reported in brackets. ***(**,*,^L) indicate significance at 0.1% (1%, 5%, 10%) level. Since hazard ratios are reported, a significant value indicates a significant difference from 1. For categorical variables and dummies values in the brackets are reference values. (Fielding 1989)

Table 2. Post-entry survival

Having work experience in the sector tends to decrease the exit probability by around 5 per cent for each year. This points to the increased importance of the specific human capital, which can only be obtained through learning-bydoing. Interestingly, this comes in contrast with results of Martynovich and Henning (2016), who demonstrated that a level of education (which is not significant here) has a more persistent effect on a wage performance of employees in the sector than industry experience. While it might be the outcome of using different measures of success for workers and entrepreneurs in the sector, this might also point to various mechanisms of learning and aspects of relevance of human capital between employees and entrepreneurs.

While longer spells of unemployment tend to increase the propensity of individuals for entrepreneurship, they tend to negatively affect survival chances of such entries. Having previous entrepreneurial experience helps an individual is associated with longer surviving entries, even though this effect is weakly significant.

Starting a firm with more employees tends to increase survival probability, which has two complementary aspects to it. On the one hand, number of employees at the moment of entry might be an indication for the seriousness of intentions behind the decision to start a firm. On the other hand, such effect qualifies the difference between firms which start with an intention to grow from firms which are established as individual IT consultancies.

Modelling survival of new entrepreneurial entries provides support to Hypotheses 2a-2c, even though to a different extent. On the one hand, new firms established by stayers tend to have the lowest exit probability compared to firms established by movers. On the other hand, different histories of entrepreneurs' prior relocations are associated with varying survival chances, which provides support for Hypothesis 2b. Firm founders' employment experience in metropolitan areas tends to have a positive impact on their businesses' survival in non-metropolitan regions. In that respect, the metropolitan employment tenure might be a substitute for employment tenure in a region where entrepreneurial entry is performed. The extent of substitution is, however, conditional upon entrepreneurs' prior relocation patterns: while returners enjoy the largest benefit from employment experience in metropolitan areas, it is less pronounced for one-time movers and, particularly, onward movers (who are, potentially, adventurers travelling the country in search of opportunities and ready for further relocation if their business is not living up to expectations). In that respect, only a limited support for Hypothesis 3b is obtained.

5.3. Dynamics over time

A surge in a number of entrepreneurial entries (Figures 1 and 2) as well as an increased mobility of entrepreneurs (Figure 3) after 2000 points to a potential shift in the dynamics of entrepreneurial activities in the knowledge-intensive IT services sector. Peculiarly, this shift coincided with the 2001 dotcom crash, which played an important role in the restructuring of employment opportunities in the sector (Warhurst et al. 2006; Martynovich and Henning 2016). This makes it interesting to check whether it was reflected also in the impact of spatial biographies of individuals on their entrepreneurial activities. Tables 3 and 4 present the results of re-estimating the models presented in Tables 1 and 2 for two sub-periods: 1991-2000 and 2001-2010⁶.

When it comes to the propensity for entrepreneurship (panel (a) of Table 3), estimation results confirm the outcome for the whole time period: local entrants are most likely to become entrepreneurs in both sub-periods (Entry location). However, the difference between local entrants and movers, however, is mitigated over time. In the first sub-period movers from metropolitan areas were 45 per cent less likely to become entrepreneurs, while movers from non-metropolitan regions – 56 per cent less likely. In the second sub-period, the corresponding numbers are 21 per cent and 48 per cent respectively (Models 1 and 3). Notably, movers from metropolitan areas demonstrated a particularly fast catch-up to local entrants in the propensity for entrepreneurship. The negative impact of staying in the same region over six years of recorded employment history, as compared to individuals with a history of spatial relocation, becomes stronger in the second sub-period (Prior relocation). Besides, returners (particularly from metropolitan areas), which were the most likely entrepreneurs among movers over the whole time period, are become even more likely to choose an entrepreneurship option for paid employment over time (Models 2 and 4). For instance, individuals, returning from metropolitan areas, are 9 per cent more likely to become entrepreneurs compared to local entrants (3 per cent over the whole time

⁶ Only variables representing spatial biographies of individuals are reported. Other estimates are available upon request.

period, and negative in the first sub-period). All in all, it appears that individuals' spatial biographies have differential effect on their propensity for entrepreneurship at different times. That is, in the second sub-period the positive association between being local and starting a firm is reduced, while benefits from relocation (in particular, from metropolitan areas) become larger.

Logistic regression	1991-2000		2001-2010	
	Model 1	Model 2	Model 3	Model 4
Entry location (Local entrant)				
Mover from metro	0.5531*** (0.0445)	0.3843*** (0.0539)	0.7929*** (0.0363)	0.6509*** (0.0471)
Mover from non-metro	0.4349*** (0.0341)	0.2145*** (0.0332)	0.5212*** (0.0264)	0.3751*** (0.0334)
Prior relocation (One-time mover)				
Stayer	0.9433 (0.0545)	0.8357** (0.0495)	0.8191*** (0.0310)	0.7591*** (0.0302)
Onward mover	1.0726 (0.0811)	0.9587 (0.0960)	1.0672 (0.0496)	0.9930 (0.0655)
Returner	1.5989*** (0.1099)	1.0740 (0.0969)	1.2201*** (0.0550)	0.9360 (0.0578)
Entry location*Prior relocation (One-time mover)				. ,
Onward mover from metro		1.5067* (0.3083)		1.2075 (0.1350)
Onward mover from non-metro		1.9080** (0.4030)		1.3857** (0.1730)
Returner from metro		2.2091*** (0.4238)		1.6714*** (0.1815)
Returner from non-metro		4.4199*** (0.8849)		2.1278*** (0.2694)
Pseudo-R ²	0.0957	0.0993	0.1172	0.1184
Ν	45830	45830	55959	55959

Odds ratios are reported. Robust standard errors clustered at the individual level are reported in brackets. ****(**,*) indicate significance at 0.1% (1%, 5%) level. Since odds ratios are reported, a significant value indicates a significant difference from 1. For categorical variables and dummies values in the brackets are reference values. (Fielding 1989)

Table 3.	Propensity for	r entrepreneurship	. Dynamics	over time
----------	----------------	--------------------	------------	-----------

What my lay behind this dynamics is technological advances, such as cloud computing, high-speed broadband internet and Wi-Fi, which enabled greater worker and entrepreneur mobility (Lafuente et al. 2010). Increasingly over time, where there was a signal, there was a potential business opportunity (Rigby 2008). In that respect, an increased opportunity for mobility would also lead to discovery of new entrepreneurial opportunities beyond immediate

location of individuals, which would, in its turn, lead to an increased propensity of mobile individuals for entrepreneurship. Besides, the diffusion of IT as a platform for development of business activities in all regions would generate an increased demand for IT services beyond metropolitan areas, implying more potential business opportunities and more firms established to serve that demand. In that situation of an increased demand, on the one hand, an increased competition, individuals from metropolitan areas would be more likely to identify entrepreneurial opportunities as they were used to working in a highly competitive environment.

An impact of spatial biographies of individuals on survival of their businesses also changes over time (Table 4). For instance, the benefits of staying in a region over six years of recorded employment history, documented for the whole time period, become more pronounced as time goes by (*Prior relocation*). Indeed, in the 1990s there is no significant difference between stayers and one-time movers and only a weakly significant difference between returners and stayers, while these effects become stronger in the 2000s. The overall hierarchy of effects, however, remains more or less the same: onward movers appear to be the most disadvantaged ones.

Cox proportional hazards model	1991-2000		2001-2010	
	Model 1	Model 2	Model 3	Model 4
Prior relocation (Stayer)				
One-time mover	1.0917 (0.0725)	1.0747 (0.0735)	1.1359** (0.0552)	1.1266* (0.0563)
Onward mover	1.3576*** (0.1112)	1.3026** (0.1223)	1.2801*** (0.0719)	1.3003*** (0.0842)
Returner	1.1268 [⊥] (0.0778)	1.2238** (0.0930)	1.1357** (0.0603)	1.1496* (0.0706)
Metropolitan emp. tenure	0.9590* (0.0208)	0.8678** (0.0412)	0.9891 (0.0130)	0.9741 (0.0355)
Prior relocation*Metropolitan emp. tenure (Returner)		. ,		
One-time mover		1.1293* (0.0617)		1.0226 (0.0403)
Onward mover		1.1483* (0.0715)		1.0046 (0.0443)
N of subjects	1999	1999	4581	4581
N of failures	1778	1778	3042	3042
Log-likelihood	-12093.39	-12090.29	-23445.96	-23445.69

Hazard ratios are reported. Robust standard errors are reported in brackets. ***(**,*, $^{\perp}$) indicate significance at 0.1% (1%, 5%, 10%) level. Since hazard ratios are reported, a significant value indicates a significant difference from 1. For categorical variables and dummies values in the brackets are reference values. (Fielding 1989)

Table 4. Post-entry survival. Dynamics over time
With regards to the metropolitan employment tenure (*Metropolitan emp. tenure*), its effect is only observable in the 1990s (and quite strongly; for instance, returners enjoy a 13 per cent decrease in failure probability for each year of employment in a metropolitan region). It disappears in the 2000s.

This might be the other side of the story of an increased penetration of IT service firms to non-metropolitan regions. Indeed, the development of the IT services sector beyond metropolitan areas boundaries (not least, due to an increased mobility of entrepreneurs) might have improved the opportunities for learning and obtaining relevant knowledge outside the latter. This, coupled with the adaptation of educational systems to the digital paradigm (Vona and Consoli 2015), could lead to a decreased importance of metropolitan areas as learning hotspots, even if they still concentrated most of firms and IT employment opportunities. In its turn, this would decrease the advantage of employment experience in these regions for obtaining knowledge necessary to start a longer-surviving firm.

All in all, considering, how the relationship between spatial biographies of individuals and their entrepreneurial activities changes over time, suggests that this relationship develops as the sector matures and the environment around it changes.

6. Discussion and conclusion

The paper aimed at investigating the extent to which spatial biographies of individuals, that is their histories of prior relocation, have an impact on their propensity to become entrepreneurs and post-entry performance of firms they establish. The departing point was the notion of locational inertia of entrepreneurs (Audretsch et al. 2012; Dahl and Sorenson 2012), implying that most individuals start new firms in regions where they have previously lived and worked. It was hypothesized that the locational inertia, driven by the local embeddedness of entrepreneurs (Dahl and Sorenson 2009), may be offset by deployment of diverse knowledge from different geographical settings accumulated by mobile individuals (Agarwal et al. 2007; Frederiksen et al. 2016), particularly those with employment experience in metropolitan areas. Empirically the paper was set around entrepreneurial entries in the knowledge-intensive IT services sector in Swedish non-metropolitan regions between 1991 and 2010.

The results of analysis demonstrate that the locational inertia of entrepreneurs, indeed, takes place. Between 1991 and 2010, more than three quarters of entrepreneurial entries in non-metropolitan regions were performed locally. Besides, these individuals were demonstrated to have higher propensity to choose entrepreneurship option for paid employment. Firms established by such individuals may also be expected to survive longer.

At the same time, empirical investigation suggests that there is more complexity beyond the local/non-local dichotomy of entrepreneurs' spatial origin. It was demonstrated that it is not just the location of individuals immediately prior to performing entrepreneurial entry, but also their extended spatial biographies that matter for their entrepreneurial activities. Distinguishing between several patterns of relocation (stayers, one-time movers, onward movers, and returners) as well as directions of relocation (from a metropolitan area or a non-metropolitan region), it was shown that in certain cases individuals with broad spatial careers may be more likely to start a successful business than local entrants.

The analysis suggests that while the traditional role of metropolitan areas is in providing the most fertile ground for knowledge-intensive sectors (Simmie and Strambach 2006; Karlsson et al. 2010), their importance goes beyond that. Having work experience in metropolitan areas was demonstrated to increase the probability of an individual to start a new firm, and a successful one. This might be attributes to the role of metropolitan areas as learning hotspots for individuals willing to obtain sector-specific knowledge in a dynamic urban context. This is particularly emphasized for individuals relocating to metropolitan areas for a while, but returning to their home regions to start a business. In this case, they become both embedded locally in a region of entrepreneurial entry, but also embedded in broader knowledge networks at the sectoral level. This allows drawing parallels between such individuals and Saxenian's (2006) New Argonauts: just as the latter travel to the Silicon Valley to work in the dynamic learning context and started successful businesses upon return to their home countries, the former tended to become the most promising entrepreneurs upon their return to their home regions.

The relationship between spatial biographies of individuals and entrepreneurial activity was shown to be changing over time, which points to the fact that the latter is related to the sectoral evolution (Audretsch and Thurik 2003). Indeed, as the industry matures and develops, so do the constraints and opportunity structures for potential entrepreneurs (Aldrich and Baker 2001). In that respect, it is important to account for how the evolution of the sector as well as the development of the environment, in which it operates, is reflected in the entrepreneurial behavior, not least in its locational aspect.

All in all, the results of analysis indicate that spatial biographies of individuals are related to the probability of entrepreneurial entry and its subsequent performance. In that respect, employment histories of individuals offer more information than simply the contents and location of the most recent employment, which should be incorporated in future analyses of entrepreneurial activities.

6.1. Limitations and further research

The study comes with limitations which provide the space for further investigation. First, while it was demonstrated that sectoral context matters for the changing relationship between spatial biographies of individuals and their entrepreneurial behavior, it is quite possible that the observed dynamics may be unique to the IT services sector. Therefore, other sectors should be considered in future analyses (particularly, other knowledge-intensive industries, where mobility of entrepreneurs may be an important channel of knowledge diffusion).

Second, the analysis of post-entry performance of new firms should be extended to incorporate not only survival of new entries but also their employment and/or sales growth. This is important since previous research suggests the existence of underperforming firms which manage to survive despite the low performance (Gimeno et al. 1997). Accounting for other performance indicators would allow connecting the entrepreneurial performance to development of regions, particularly in terms of upgrading their knowledge base (Neffke et al. 2014), as well as the role of entrepreneurs in the spatial diffusion of industries.

Finally, the data employed in this paper did not allow saying much about individuals' motivations, neither with regards the decision to start a firm, nor the decision to relocate. In that respect, deeper qualitative studies of the relationship between spatial biographies of individuals and their entrepreneurial behavior might offer a more detailed perspective on the issue.

Acknowledgments

The author is grateful for constructive advice and critical comments on earlier versions of this paper by Emelie Hane-Weijman, Karl-Johan Lundquist, David Rigby, Christian Østergaard, Ron Boschma, Martin Henning, Rikard Eriksson, and Agnes Andersson Djurfeldt. The paper benefited from fruitful discussions at the Workshop on Industry Relatedness and Regional Change held in Umeå, Sweden, on October 29-31, 2014, the DRUID Academy Conference 2015 held in Aalborg, Denmark, on January 18-20, 2015, and the Annual Meeting of the American Association of Geographers held in Chicago, IL, USA, on April 21-25, 2015.

References

- Acs, Z. J., Braunerhjelm, P., Audretsch, D. B., & Carlsson, B. (2009). 'The knowledge spillover theory of entrepreneurship'. *Small Business Economics*, Vol. 32(1), pp. 15-30.
- Agarwal, R., Audretsch, D., & Sarkar, M. B. (2007). 'The process of creative construction: knowledge spillovers, entrepreneurship, and economic growth'. *Strategic Entrepreneurship Journal*, Vol. 1(3-4), pp. 263-286.
- Aldrich, H. E., & Baker, T. (2001). Learning and Legitimacy: Entrepreneurial Responses to Constraints on the Emergence of New Populations and Organizations. In C. B. Schoonhoven & E. Romanelli (Eds.), *The entrepreneurship dynamic: origins of entrepreneurship and the evolution of industries.* Stanford, CA: Stanford University Press, pp. 207-235.
- Atterton, J. (2007). 'The 'Strength of Weak Ties': Social Networking by Business Owners in the Highlands and Islands of Scotland'. *Sociologia Ruralis*, Vol. 47(3), pp. 228-245.
- Audretsch, D. B., Falck, O., Feldman, M. P., & Heblich, S. (2012). 'Local Entrepreneurship in Context'. *Regional Studies*, Vol. 46(3), pp. 379-389.
- Audretsch, D. B., Grilo, I., & Thurik, A. R. (2011). Globalization, entrepreneurship and the region. In M. Fritsch (Ed.), *Handbook of Research on Entrepreneurship and Regional Development*. Cheltenham and Northampton, MA: Edward Elgar, pp. 11-32.
- Audretsch, D. B., Keilbach, M. C., & Lehmann, E. E. (2006). Entrepreneurship and economic growth. New York: Oxford University Press.
- Audretsch, D. B., & Thurik, R. A. (2001). 'What's New about the New Economy? Sources of Growth in the Managed and Entrepreneurial Economies'. *Industrial and Corporate Change*, Vol. 10(1), pp. 267-315.
- Audretsch, D. B., & Thurik, R. A. (2003). ENTREPRENEURSHIP, INDUSTRY EVOLUTION AND ECONOMIC GROWTH. In R. Koppl, Jack Birner, & P. Kurrild-Klitgaard (Eds.), *Austrian Economics and Entrepreneurial Studies (Advances in Austrian Economics, Volume 6)* pp. 39-56.
- Bahlmann, M. D. (2015). 'Finding value in geographic diversity through prior experience and knowledge integration: a study of ventures' innovative performance'. *Industrial and Corporate Change*.

- Bergmann, H., & Sternberg, R. (2007). 'The Changing Face of Entrepreneurship in Germany'. *Small Business Economics*, Vol. 28(2), pp. 205-221.
- Bürcher, S., Habersetzer, A., & Mayer, H. (2016). Entrepreneurship in peripheral regions: A relational perspective. In E. A. Mack & H. Qian (Eds.), *Geographies of Entrepreneurship*. London and New York: Routledge, pp. 143-164.
- Champion, T., Coombes, M., & Gordon, I. (2014). 'How Far do England's Second-Order Cities Emulate London as Human-Capital 'Escalators'?'. *Population, Space and Place,* Vol. 20(5), pp. 421-433.
- Cohen, W. M., & Levinthal, D. A. (1990). 'Absorptive Capacity: A New Perspective on Learning and Innovation'. *Administrative Science Quarterly*, Vol. 35(1), pp. 128-152.
- Colombo, M. G., & Grilli, L. (2005). 'Founders' human capital and the growth of new technology-based firms: A competence-based view'. *Research Policy*, Vol. 34(6), pp. 795-816.
- Cox, D. R. (1972). 'Regression Models and Life-Tables'. Journal of the Royal Statistical Society. Series B (Methodological), Vol. 34(2), pp. 187-220.
- Curran, D., Lynn, T., & O'Gorman, C. (2016). 'The Role of Personal Factors in the Location Decision of Software Services Start-up Firms'. *European Planning Studies*, Vol. 24(3), pp. 551-567.
- Dahl, M., & Sorenson, O. (2009). 'The embedded entrepreneur'. European Management Review, Vol. 6(3), pp. 172-181.
- Dahl, M., & Sorenson, O. (2012). 'Home Sweet Home: Entrepreneurs' Location Choices and the Performance of Their Ventures'. *Management Science*, Vol. 58(6), pp. 1059-1071.
- Doloreux, D., & Muller, E. (2007). The Key Dimensions of Knowledge-Intensive Business Services (KIBS) Analysis. A Decade of Evolution. Working Paper on Firms and Regions Number U1/2007. Karlsruhe, Germany: Fraunhofer-Institut für System- und Innovationsforschung (ISI).
- Dunn, T., & Holtz-Eakin, D. (2000). 'Financial Capital, Human Capital, and the Transition to Self-Employment: Evidence from Intergenerational Links'. *Journal of Labor Economics*, Vol. 18(2), pp. 282-305.

- Fielding, A. J. (1989). 'Inter-Regional Migration and Social Change: A Study of South East England Based upon Data from the Longitudinal Study'. *Transactions of the Institute of British Geographers*, Vol. 14(1), pp. 24-36.
- Fielding, A. J. (1992). 'Migration and Social Mobility: South East England as an Escalator Region'. *Regional Studies*, Vol. 26(1), pp. 1-15.
- Findlay, A. M., Short, D., & Stockdale, A. (2000). 'The labour-market impact of migration to rural areas'. *Applied Geography*, Vol. 20(4), pp. 333-348.
- Frederiksen, L., Wennberg, K., & Balachandran, C. (2016). 'Mobility and Entrepreneurship: Evaluating the Scope of Knowledge-Based Theories of Entrepreneurship'. *Entrepreneurship Theory and Practice*, Vol. 40(2), pp. 359-380.
- Gimeno, J., Folta, T. B., Cooper, A. C., & Woo, C. Y. (1997). 'Survival of the Fittest? Entrepreneurial Human Capital and the Persistence of Underperforming Firms'. *Administrative Science Quarterly*, Vol. 42(4), pp. 750-783.
- Glaeser, E. L., & Saiz, A. (2003). *The Rise of the Skilled City*. National Bureau of Economic Research Working Paper Series No. 10191.
- Gordon, I. R. (2015). 'Ambition, Human Capital Acquisition and the Metropolitan Escalator'. *Regional Studies*, Vol. 49(6), pp. 1042-1055.
- Gordon, I. R., Champion, T., & Coombes, M. (2015). 'Urban Escalators and Interregional Elevators: The Difference that Location, Mobility, and Sectoral Specialisation Make to Occupational Progression'. *Environment and Planning A*, Vol. 47(3), pp. 588-606.
- Harrison, R. T., Cooper, S. Y., & Mason, C. M. (2004). 'Entrepreneurial Activity and the Dynamics of Technology-based Cluster Development: The Case of Ottawa'. Urban Studies, Vol. 41(5-6), pp. 1045-1070.
- Helfat, C. E., & Lieberman, M. B. (2002). 'The birth of capabilities: market entry and the importance of pre-history'. *Industrial and Corporate Change*, Vol. 11(4), pp. 725-760.
- Hessels, J., Gelderen, M., & Thurik, R. (2008). 'Entrepreneurial aspirations, motivations, and their drivers'. *Small Business Economics*, Vol. 31(3), pp. 323-339.

- Kalnins, A., & Chung, W. (2006). 'Social Capital, Geography, and Survival: Gujarati Immigrant Entrepreneurs in the U.S. Lodging Industry'. *Management Science*, Vol. 52(2), pp. 233-247.
- Karlsson, C., Maier, G., Trippl, M., Siedschlag, I., Owen, R., & Murphy, G. (2010). *IT and Regional Economic Dynamics: A Literature Review*. EUR 24510 EN. Luxembourg: Publications Office of the European Union.
- Klepper, S. (2001). 'Employee Startups in High-Tech Industries'. *Industrial and Corporate Change*, Vol. 10(3), pp. 639-674.
- Lafuente, E., Vaillant, Y., & Serarols, C. (2010). 'Location decisions of knowledgebased entrepreneurs: Why some Catalan KISAs choose to be rural?'. *Technovation*, Vol. 30(11–12), pp. 590-600.
- Lasch, F., Robert, F., & Le Roy, F. (2013). 'Regional determinants of ICT new firm formation'. *Small Business Economics*, Vol. 40(3), pp. 671-686.
- Liu, X., Wright, M., Filatotchev, I., Dai, O., & Lu, J. (2010). 'Human mobility and international knowledge spillovers: evidence from high-tech small and medium enterprises in an emerging market'. *Strategic Entrepreneurship Journal*, Vol. 4(4), pp. 340-355.
- Lorentzen, A. (2008). 'Knowledge networks in local and global space'. *Entrepreneurship & Regional Development*, Vol. 20(6), pp. 533-545.
- Marcketti, S. B., Niehm, L. S., & Fuloria, R. (2006). 'An Exploratory Study of Lifestyle Entrepreneurship and Its Relationship to Life Quality'. *Family and Consumer Sciences Research Journal*, Vol. 34(3), pp. 241-259.
- Martynovich, M., & Henning, M. (2016). LABOUR FORCE BUILDING IN A RAPIDLY EXPANDING SECTOR. CRA Working Paper 2016:3.
 Gothenburg, Sweden: Centre for Regional Analysis, School of Business, Economics and Law, Gothenburg University
- Maskell, P. (2014). 'Accessing remote knowledge—the roles of trade fairs, pipelines, crowdsourcing and listening posts'. *Journal of Economic Geography*, Vol. 14(5), pp. 883-902.
- Michelacci, C., & Silva, O. (2007). 'Why So Many Local Entrepreneurs?'. Review of Economics and Statistics, Vol. 89(4), pp. 615-633.
- Mulder, C. H., & van Ham, M. (2005). 'Migration histories and occupational achievement'. *Population, Space and Place,* Vol. 11(3), pp. 173-186.

- Nanda, R., & Sørensen, J. B. (2010). 'Workplace Peers and Entrepreneurship'. Management Science, Vol. 56(7), pp. 1116-1126.
- Neffke, F., Hartog, M., Boschma, R., & Henning, M. (2014). Agents of structural change. The role of firms and entrepreneurs in regional diversification.
 Papers in Evolutionary Economic Geography (PEEG) 1410. Utrecht, the Netherlands: Utrecht University.
- Nijkamp, P. (2003). 'Entrepreneurship in a Modern Network Economy'. *Regional Studies*, Vol. 37(4), pp. 395-405.
- Østergaard, C. R., & Dalum, B. (2011). Evolution of the geographical concentration pattern of the Danish IT sector. In H. Bathelt, M. P. Feldman, & D. F. Kogler (Eds.), *Beyond Territory: Dynamic Geographies of Knowledge Creation, Diffusion, and Innovation*. New York: Routledge.
- Pellenbarg, P. H., van Wissen, L. J. G., & van Dijk, J. (2002). Firm Migration. In P. McCann (Ed.), *Industrial Location Economics*. Cheltenham and Northampton, MA: Edward Elgar, pp. 110-150.
- Qian, H., & Acs, Z. J. (2013). 'An absorptive capacity theory of knowledge spillover entrepreneurship'. Small Business Economics, Vol. 40(2), pp. 185-197.
- Reuschke, D. (2014). 'Self-employment, Internal Migration and Place Embeddedness'. *Population, Space and Place*, Vol. 20(3), pp. 235-249.
- Reuschke, D. (2015). 'Self-Employment as a Route In and Out of Britain's South East'. *Regional Studies*, Vol. 49(4), pp. 665-680.
- Rigby, R. (2008). 'Mobility puts paid to long commutes and the office slog'. *Financial Times* Vol. February 5, pp. 14.
- Sacchetti, S. (2009). 'Why, Where and With Whom Do You Link? The Nature and Motivations of Linkages Within and Outside an Italian Local System'. *Regional Studies*, Vol. 43(2), pp. 197-209.
- Saxenian, A. (2006). *The New Argonauts: Regional Advantage in a Global Economy*. Cambridge, MA: Harvard University Press.
- Shane, S. (2000). 'Prior Knowledge and the Discovery of Entrepreneurial Opportunities'. *Organization Science*, Vol. 11(4), pp. 448-469.
- Shane, S., & Stuart, T. (2002). 'Organizational Endowments and the Performance of University Start-ups'. *Management Science*, Vol. 48(1), pp. 154-170.

- Simmie, J., & Strambach, S. (2006). 'The contribution of KIBS to innovation in cities: an evolutionary and institutional perspective'. *Journal of Knowledge Management*, Vol. 10(5), pp. 26-40.
- Sørensen, J. B. (2007). 'Bureaucracy and Entrepreneurship: Workplace Effects on Entrepreneurial Entry'. *Administrative Science Quarterly*, Vol. 52(3), pp. 387-412.
- Sørensen, J. B., & Sharkey, A. J. (2014). 'Entrepreneurship as a Mobility Process'. *American Sociological Review*, Vol. 79(2), pp. 328-349.
- Sorenson, O., & Audia, P. G. (2000). 'The Social Structure of Entrepreneurial Activity: Geographic Concentration of Footwear Production in the United States, 1940–1989'. *American Journal of Sociology*, Vol. 106(2), pp. 424-462.
- Stam, E. (2007). 'Why Butterflies Don't Leave: Locational Behavior of Entrepreneurial Firms'. *Economic Geography*, Vol. 83(1), pp. 27-50.
- Stam, E. (2010). Entrepreneurship, evolution and geography. In R. Boschma & R. Martin (Eds.), *The Handbook of Evolutionary Economic Geography*. Cheltenham and Northampton, MA: Edward Elgar, pp. 139-161.
- Varga, A., & Schalk, H. (2004). 'Knowledge Spillovers, Agglomeration and Macroeconomic Growth: An Empirical Approach'. *Regional Studies*, Vol. 38(8), pp. 977-989.
- Vona, F., & Consoli, D. (2015). 'Innovation and skill dynamics: a life-cycle approach'. *Industrial and Corporate Change*, Vol. 24(6), pp. 1393-1415.
- Warhurst, C., Lockyer, C., & Dutton, E. (2006). 'IT jobs: opportunities for all?'. New Technology, Work and Employment, Vol. 21(1), pp. 75-88.
- Westlund, H., & Bolton, R. (2003). 'Local Social Capital and Entrepreneurship'. Small Business Economics, Vol. 21(2), pp. 77-113.
- Wright, M. (2011). Entrepreneurial Mobility. In D. D. Bergh & D. J. Ketchen (Eds.), Building Methodological Bridges (Research Methodology in Strategy and Management, Volume 6) pp. 137-159.

Appendices

Person	<i>t-</i> 6	t-5	t-4	t-3	t-2	t-1	t (entry)
А	Växjö	Växjö	Växjö	Växjö	Växjö	Växjö	Växjö
В	Linköping	Linköping	Linköping	Linköping	Linköping	Linköping	Växjö
С	Stockholm	Stockholm	Stockholm	Stockholm	Stockholm	Stockholm	Växjö
D	Linköping	Linköping	Linköping	Linköping	Linköping	Växjö	Växjö
Е	Stockholm	Stockholm	Stockholm	Stockholm	Stockholm	Växjö	Växjö
F	Linköping	Linköping	Stockholm	Stockholm	Stockholm	Växjö	Växjö
G	Linköping	Linköping	Stockholm	Stockholm	Stockholm	Stockholm	Växjö
н	Växiö	Stockholm	Stockholm	Stockholm	Stockholm	Växiö	Växiö

A1. Spatial biographies of individuals. Hypothetical examples

The table presents (some) possible combinations of spatial relocations of individuals. For demonstration purposes, it is assumed that individuals A-H perform an entrepreneurial entry at year t in a local labor market of Växjö. Then, provided their recorded employment histories, the following categorical values with respect to spatial biography variables are assigned to these individuals.

Person	Entry location	Prior relocation	Metropolitan emp. tenure
А	Local	Stayer	0
В	Mover from non-metro	One-time mover	0
с	Mover from metro	One-time mover	6
D	Local	One-time mover	0
Е	Local	One-time mover	5
F	Local	Onward mover	3
G	Mover from metro	Onward mover	4
н	Local	Returner	4

Note, for example, that the difference between individuals B and D is that the former moved to Växjö at the year of entry, while the latter moved to Växjö a year before entry. That is why individual B is classified as a mover from a non-metropolitan region and individual D – as a local founder, while both of them are one-time movers according to their extended spatial biography. Besides, none of them has any employment tenure in a metropolitan region. In that respect, even being classified within the same category according to one of the spatial biography variables, doesn't mean that they will be classified together with respect to another spatial biography variable.

A2. Transitions to entrej	preneur	ship. San	nple sum	mary										
	z	Mean	Std. dev.	-	2	3	4	5	U		7	8	6	10
1. Entrepreneur	101789	0.0825	0.2751	٢										
2. Sex	101789	0.7363	0.4407	0.0894	-									
3. Age	101789	34.6147	9.9740	0.0713	0.0040	-								
4. Age squared	101789	1297.6570	765.3007	0.0794	0.0054	0.9884	-							
5. Higher education	101789	0.6028	0.4893	-0.0141	0.1025	0.0397	0.009	4						
6. Sector experience	101789	1.3570	2.3340	-0.0291	0.0626	0.3190	0.297	5 0.09	58	_				
7. Unemployment spell	101789	1.0127	2.0006	0.0254	-0.0022	-0.3396	-0.294	8 -0.06	82 -0.2	660	-			
8. Entrepreneurial experience	101789	0.0800	0.2713	0.2029	0.0638	0.1341	0.131	4 -0.00	60 0.06	648 -0.	0879	-		
9. Entry location	101789	0.4796	0.7613	-0.0515	0.0584	-0.0133	3 -0.015	6 0.06	27 0.03	262 -0.	- 0982	0.0296	. -	
10. Prior relocation	101789	2.0543	1.0664	0.0099	0.0557	-0.0980	-0.098	7 0.06	83 -0.0	320 0.	0050 (0.0054	0.4887	-
A3. Post-entry survival.	Sample	e summar	Ż											
	z	Mean	Std. dev.	-	2	3	4	5	9	7	80	6	10	1
1. Age	6580	37.2737	11.6481	-										
2. Age squared	6580	1524.9870	938.5607	0.9882	-									
3. Sex	6580	0.8748	0.3310	-0.078	-0.0641	-								
4. Higher education	6580	0.5757	0.4943	0.0478	0.0334	0.0514	-							
5. Sector experience	6580	1.0164	1.8953	0.2344	0.2146	0.0483	0.0874	-						
6. Unemployment spell	6580	1.2202	2.0545	-0.2482	-0.2102	-0.0453	-0.0362	-0.2345	-					
7. Entrepreneurial experience	6580	0.0853	0.2793	0.1313	0.1297	-0.0177	0.0000	0.0944	-0.0854	-				
8. Entry location	6580	0.7030	1.4991	0.0679	0.0484	0.0188	0.0344	0.1437	-0.2039	-0.0277	-			
9. Metropolitan employment tenure	6580	2.1279	1.1078	-0.1214	-0.1194	0.0636	0.0062	-0.0289	0.0043	-0.0947	0.2539	-		
10. Firm employment at entry	6580	1.1103	0.7127	0.0168	0.0095	-0.0001	-0.0125	0.0699	-0.0542	0.1184	-0.0133	-0.0371	-	
11. Regional IT employment at entry (log)	6580	6.0821	1.4094	-0.0676	-0.0634	0.0251	0.1070	0.0367	-0.0028	-0.0088	-0.0429	-0.048	0.0218	-

37

MEDDELANDEN FRÅN LUNDS UNIVERSITETS GEOGRAFISKA INSTITUTION. AVHANDLINGAR

I.	<i>Herman Richter:</i> Skånes karta från mitten av 1500-talet till omkring 1700 : bidrag till en historisk-kartografisk undersökning. (1929)
II.	Josef Westin: Kulturgeografiska studier inom Nätra-, Näske- och Utbyåarnas flodområden samt angränsande kusttrakter. (1930)
III.	<i>Herman Richter</i> och <i>Wilhelm Norlind:</i> Orbis Arctoi Nova et Accurata Delineatio Auctore Andrea Bureo Sueco 1626. (1936)
IV.	Sven Björnsson: Sommen-Åsundenområdet : en geomorfologisk studie. (1937)
V.	<i>Arne Sandell:</i> Tektonik och morfologi inom dalformationen med omgivande urbergsterräng. (1941)
VI.	Sven Dahl: Torna och Bara : studier i Skånes bebyggelse- och näringsgeografi före 1860. (1942)
VII.	<i>Karl Erik Bergsten:</i> Isälvsfält kring norra Vättern : fysisk- geografiska studier. (1943)
VIII.	<i>Carl Erik Nordenskjöld:</i> Morfologiska studier inom övergångsområdet mellan Kalmarslätten och Tjust. (1944)
IX.	Sven Björnsson: Blekinge : en studie av det blekingska kulturlandskapet. (1946)
Х.	<i>Karl Erik Bergsten:</i> Östergötlands bergslag : en geografisk studie. (1946)
XI.	<i>Tor Holmquist:</i> Den halländska vinterfiskehamnsfrågan. (1947)
XII.	<i>Olof Ängeby:</i> Landformerna i nordvästra Jämtland och angränsande delar av Nord-Tröndelag. (1947)

XIII.	Axel Wennberg: Lantbebyggelsen i nordöstra Östergötland 1600-1875. (1947)
XIV.	<i>Lars Bjerning:</i> Skånes jord- och stenindustri : dess utveckling, lokalisering och betydelse ur näringsgeografisk synvinkel. (1947)
XV.	<i>Allan Weinhagen:</i> Norbergs bergslag samt Gunnilbo och Ramnäs till omkring 1820 : studier i områdets närings- och bebyggelsegeografi. (1947)
XVI.	<i>Helge Stålberg:</i> Smålands skogs- och träförädlingsindustrier : en närings-geografisk studie. (1947)
XVII.	<i>Folke Lägnert:</i> Veteodlingen i södra och mellersta Sverige. (1949)
XVIII.	<i>Yngve Nilsson:</i> Bygd och näringsliv i norra Värmland : en kulturgeografisk studie. (1950)
XIX.	Olof Ängeby: Evorsionen i recenta vattenfall. (1951)
XX.	Karl Erik Bergsten: Sydsvenska födelseortsfält. (1951)
XXI.	<i>Folke Lägnert:</i> Valmanskåren på skånes landsbygd 1911- 1948. (1952)
XXII.	<i>Olof Nordström:</i> Relationer mellan bruk och omland i östra Småland 1750-1900. (1952)
XXIII.	Arvid Bergdahl: Israndsbildningar i östra Syd- och Mellansverige med särskild hänsyn till åsarna. (1953)
XXIV.	Sven E Behrens: Morfometriska, morfogenetiska och tektoniska studier av de nordvästskånska urbergsåsarna, särskilt Kullaberg. (1953)
XXV.	<i>Torsten Hägerstrand:</i> Innovationsförloppet ur korologisk synpunkt. (1953)
XXVI.	<i>Gunhild Weimarck:</i> Studier över landskapets förändring inom Lönsboda, Örkeneds socken, nordöstra Skåne. (1953)
XXVII.	<i>Ingemar Larsson:</i> Structure and landscape in western Blekinge, southeast Sweden. (1954)

XXVIII.	Sven Godlund: Busstrafikens framväxt och funktion i de urbana influensfälten. (1954)
XXIX.	<i>Folke Lägnert:</i> Syd- och mellansvenska växtföljder. Del I : de äldre bruknings-systemens upplösning under 1800-talet. (1955)
XXX.	<i>Olof Ängeby:</i> Toppkonstans, erosionsytor och passdalar i Jämtland och Tröndelag. (1955)
XXXI.	Gunnar Johnsson: Glacialmorfologiska studier i södra Sverige. (1956)
XXXII.	<i>Folke Lägnert:</i> Syd- och mellansvenska växtföljder. Del II : 1900-talet. (1956)
XXXIII.	<i>Olof Nordström:</i> Befolkningsutveckling och arbetskraftsproblem i östra Småland 1800-1955. (1957)
XXXIV.	<i>Sven Godlund:</i> Befolkning - regionsjukhus - resmöjligheter - regioner. (1958)
XXXV.	<i>Elis Pålsson:</i> Gymnasiers rekrytering och lokalisering. (1958)
XXXVI.	<i>Harald Svensson:</i> Glaciation och morfologi : en glacialgeografisk studie i ett tvärsnitt genom Skanderna mellan södra Helgelandskusten och Kultsjödalen. (1959)
XXXVII.	<i>Erik Ljungner:</i> Nahuel Huapi : ein geographischer Querschnitt durch die Anden in Patagonien. (1959)
XXXVIII.	<i>Nils Lewan:</i> Om pendling mellan bostad och arbetsplats : en undersökning med material från sydvästra Skåne. (1960)
XXXIX.	<i>Åke Mattsson:</i> Morphologische Studien in Südschweden und auf Bornholm über die nichtglaziale Formenwelt der Felsenskulptur. (1962)
XL.	<i>Stig Nordbeck:</i> Framställning av kartor med hjälp av siffermaskiner. (1964)
XLI.	Olof Nordström: Svensk glasindustri 1550-1960. (1962)
XLII.	Jan Davidsson: Littoral processes and morphology on scandian flatcoasts. (1963)

XLIII.	<i>Martin Markgren:</i> Detaljmorfologiska studier i fast berg och blockmaterial : geomorfologisk studie inom Fennoskandia med Skåne. (1962-1963)
XLIV.	<i>Martin Markgren:</i> Geomorphological studies in Fennoscandia. II : chute slopes in northern Fennoscandia. A : regional studies. (1964)
XLV.	<i>Martin Markgren:</i> Geomorphological studies in Fennoscandia. II : chute slopes in northern Fennoscandia. B : systematic studies. (1964)
XLVI.	Lennart Améen: Stadsbebyggelse och domänstruktur. (1964)
XLVII.	Arvid Bergdahl: Det glaciala landskapet. (1961)
XLVIII.	<i>Olof Nordström - Solveig Mårtensson:</i> Turism på Öland (1966)
XLIX.	<i>Jan O Mattsson:</i> The temperature climate of potato crops. (1966)
L.	Nils Lewan: Landsbebyggelse i förvandling. (1967)
LI.	Gösta Nordholm: Skånes äldre ekonomiska geografi. (1967)
LII.	Sven Godlund - Torsten Hägerstrand - Bengt Svanström: Samhälls-utvecklingen och samhällsplaneringen. (1967)
LIII.	Tor Fr Rasmussen: Storbyutvikling og arbeidsreiser. (1966)
LIV.	<i>Erik Fagerlund - Harald Svensson - Sven Lindqvist m fl:</i> Infrarödtermografi : principer och naturgeografiska tillämpningar. (1967)
LV.	<i>Lars Eldblom:</i> Structure foncière, organisation et structure sociale. (1968)
LVI.	Knut Norborg: Jordbruksbefolkningen i Sverige. (1968)
LVII.	Gunhild Weimarck: Ulfshult (1968)
LVIII.	Rune Järvenstedt - Sven Lindqvist - Jan O Mattsson m fl: Televisionssystem i naturgeografisk forskning. (1968)

LIX.	Arne Jakobsson: Omflyttningen i Sverige 1950-1960. (1969)
LX.	<i>Åke Hillefors:</i> Västsveriges glaciala historia och morfologi. (1969)
LXI.	Sven Lindqvist: Bebyggelseklimatiska studier. (1970)
LXII.	Torsten Hägerstrand, Gunnar Törnqvist m fl: Urbaniseringen i Sverige. (SOU 1970:14)
LXIII.	<i>Bengt Sahlberg:</i> Interregionala kontaktmönster : personkontakter inom svenskt näringsliv – en flygpassagerarstudie. (1970)
LXIV.	<i>Björn Hedberg:</i> Kontaktsystem inom svenskt näringsliv : en studie av organisationers externa personkontakter. (1970)
LXV.	<i>Mats G Engström:</i> Regional arbetsfördelning : nya drag i förvärvsarbetets geografiska organisation i Sverige. (1970)
LXVI.	<i>Torsten Persson:</i> Geomorphological studies in the south-Swedish highlands. (1972)
LXVII.	<i>Dewitt Davis Jr:</i> A factoral ecology of Malmö 1965 : the social geography of a city. (1972)
LXVIII.	<i>Zoltan Petery:</i> Studier i svensk regionplanering : regionplanering enligt byggnadslagen i mindre regioner. (1972)
LXIX.	Tommy Book: Stadsplan och järnväg i Norden. (1974)
LXX.	Hans Abrahamson: Studier i jordbrukets omstrukturering. (1974)
LXXI.	Christer Persson: Kontaktarbete och framtida lokaliseringsförändringar. (1974)
LXXII.	<i>Ulf Helldén:</i> Karst : en studie av Artfjällets karstområde samt jämförande korrosionsanalyser från Västspetsbergen och Tjeckoslovakien. (1974)
LXXIII.	<i>Jànos Szegö:</i> Befolkningstäthet, markanvändning, planering (vol 1 o 2). (1974)

LXXIV.	<i>Raul Nino Guerrero:</i> Rural to urban drift of the unemployed in Colombia. (1975)
LXXV.	<i>Ulf Erlandsson:</i> Företagsutveckling och utrymmesbehov. (1975)
LXXVI.	<i>Sture Öberg:</i> Methods of describing physical access to supply points. (1976)
LXXVII.	<i>Bo Lenntorp:</i> Paths in space-time environments : a time- geographic study of movement possibilities of individuals. (1976)
LXXVIII.	<i>Richard Åhman:</i> Palsar i Nordnorge : en studie av palsars morfologi, utbredning och klimatiska förutsättningar i Finnmarks och Troms fylke. (1977)
LXXIX.	<i>Björn Gyllström:</i> The organization of production as a space- modelling mechanism in underdeveloped countries. (1977)
LXXX.	<i>Anders Järnegren - Fosco Ventura:</i> Tre samhällens förändringshistoria : exploateringen av den fysiska miljön i historisk belysning. (1977)
LXXXI.	<i>Tommy Book:</i> Stadsplan och järnväg i Storbritannien och Tyskland. (1978)
LXXXII.	<i>Jan O Mattsson - Leif Börjesson:</i> Lokalklimatiska temperaturstudier inom ett skånskt fruktodlingsdistrikt med särskilt beaktande av frostläntheten. (1978)
LXXXIII.	Bjørn Terje Asheim: Regionale ulikheter i levekår. (1979)
LXXXIV.	<i>Solveig Mårtensson:</i> On the formation of biographies in space-time environments. (1979)
LXXXV.	<i>Erik Wallin:</i> Vardagslivets generativa grammatik - vid gränsen mellan natur och kultur. (1980)
LXXXVI.	<i>Reinhold Castensson:</i> Välja för framtid - om markanvändningsval och förtroendemannainflytande i kommunal planering. (1980)
LXXXVII.	<i>Kerstin Montal:</i> Industri och vatten : den vattenförorenande industrins lokaliseringsproblem i Malmöhus län. (1980)

LXXXVIII.	<i>Tommy Carlstein:</i> Time resources, society and ecology : on the capacity for human interaction in space and time in preindustrial societies. (1980)
LXXXIX.	Jonas Åkerman: Studies on periglacial geomorphology in west Spitsbergen. (1980)
XC.	Leif Engh: Karstområdet vid Lummelunds bruk, Gotland, med speciell hänsyn till Lummelundagrottan. (1980)
XCI.	<i>Karna Lidmar-Bergström:</i> Pre-quaternary geomorphological evolution in southern Fennoscandia. (1982)
XCII.	<i>Lars-Olof Olander:</i> Staten, kommunerna och servicen : tiden kring kommun-reformen i ett ekonomiskt – geografiskt perspektiv. (1984)
XCIII.	<i>Bo Malmström</i> och <i>Owe Palmér:</i> Glacial och periglacial geomorfologi på Varangerhalvön, Nordnorge : geomorfologisk kartering med analys av glaciala former och blockhav. (1984)
XCIV.	<i>Franz-Michael Rundquist:</i> Hybrid maize diffusion in Kenya : policies, diffusion patterns and consequences. (1984)
XCV.	<i>Girma Yadeta:</i> Dynamic processes of development in marginal areas : a case study from the pokot of north west Kenya. (1985)
XCVI.	<i>Anders Sporrek:</i> Food marketing and urban growth in Dar Es Salaam. (1985)
XCVII.	<i>Rolf Nyberg:</i> Debris flows and slush avalanches in northern Swedish Lappland : distribution and geomorphological significance. (1985)
XCVIII.	<i>Lennart Olsson:</i> An integrated study of desertification - applications of remote sensing, GIS and spatial models in semi-arid Sudan. (1985)
XCIX.	<i>Mikael Stern:</i> Census from heaven? : population estimates with remote sensing techniques. (1985)

C.	<i>Katarina Olsson:</i> Remote sensing for fuelwood resources and land degradation studies in Kordofan, the Sudan. (1985)
CI.	<i>Göran Loman:</i> The climate of a sugar beet stand - dynamics, impact on the crop and possibilities of improvement. (1986)
CI.	<i>Eric Clark:</i> The rent gap and urban change : case studies in Malmö 1860-1985. (1987)
CII.	<i>Karin Hall-Könyves:</i> Remote sensing of cultivated lands in the south of Sweden. (1988)
CIII.	<i>Eva Ahlcrona:</i> The impact of climate and man on land transformation in central Sudan : applications of remote sensing. (1988)
CIV.	Kerstin Cederlund: Offentlig verksamhet : sysselsättning territoriellt och funktionellt. (1988)
CV.	<i>Per Olof Hallin:</i> Tid för omställning : om hushålls anpassningsstrategier vid en förändrad energisituation. (1989)
CVI.	<i>Jens Möller:</i> Godsen och den agrara revolutionen : arbetsorganisation, domän-struktur och kulturlandskap på skånska gods under 1800-talet. (1989)
CVII.	<i>Juha Uitto:</i> The Kenyan conundrum : a regional analysis of population growth and primary education in Kenya. (1989)
CVIII.	<i>Ola Jonsson:</i> Informationsteknologi och arbete : fallstudier inom svensk sjukvård. (1989)
CIX.	<i>Tora Friberg:</i> Kvinnors vardag. Kvinnors arbete och liv : anpassnings-strategier i tid och rum. (1990)
CX.	<i>Tomas Nihlén:</i> Eolian processes in southern Scandinavia and the Mediterra-nean area. (1990)
CXI.	<i>Anders Löfgren:</i> Att flytta hemifrån : boendets roll i ungdomars vuxenblivande ur ett situationsanalytiskt perspektiv. (1990)

CXII.	<i>Irma Guillén:</i> Cuidad Guayana – en stad, två världar : en studie av ett regionalt utvecklingsprojekts lokala effekter. (1991)
CXIII.	<i>Hans Holmén:</i> Building organizations for rural development : state and cooperatives in Egypt. (1991)
CXIV.	<i>Petter Pilesjö:</i> GIS and remote sensing for soil erosion studies in semi-arid environments : estimation of soil erosion parameters at different scales. (1992)
CXV.	Ann-Cathrine Åquist: Tidsgeografi i samspel med samhällsteori. (1992)
CXVI.	<i>José da Cruz:</i> Disaster and society : the 1985 Mexican earthquakes. (1993)
CXVII.	<i>Tomas Germundsson:</i> Landsbygdens egnahem : egnahemsrörelsen, småbruket och landskapet i sydsvenskt perspektiv. (1993)
CXVIII.	<i>Ann-Katrin Bäcklund:</i> JUST-IN-TIME : hur industriella rationaliseringsstrate-gier formar arbetsdelning och kompetens. (1994)
CXIX.	<i>Jon Knudsen:</i> Kulturspredning i et strukturelt perspektiv : eksemplifisert ved politisk og religiøs endring under moderniseringen av det norske samfunn. (1994)
CXX.	<i>Tone Haraldsen:</i> Teknologi, økonomi og rom : en teoretisk analyse av relasjoner mellom industrielle og territorielle endringsprosesser. (1994)
CXXI.	<i>Peter Jönsson:</i> Wind climate during the instumental period and recent wind erosion in southern Scandinavia. (1994)
CXXII.	<i>Peter Schlyter</i> . Palaeo-wind abrasion in southern Scandinavia : field and laboratory studies. (1995)
CXXIII.	<i>Jun Yamashita:</i> Spatial interaction and spatial structure : a study of public facility location. (1995)
CXXIV.	<i>Mats Riddersporre:</i> Bymarker i backspegel : odlingslandskapet före kartornas tid. (1995)

CXXV.	Anders Schærström: Pathogenic paths? : a time geographical approach in medical geography. (1996)
CXXVI.	<i>Lars Eklundh:</i> AVHRR NDVI for monitoring and mapping of vegetation and drought in east African environments. (1996)
CXXVII.	<i>Magnus Jirström:</i> In the wake of the green revolution : environmental and socio-economic consequences of intensive rice agriculture – the problem of weeds in Muda, Malaysia. (1996)
CXXVIII.	<i>Stefan Anderberg:</i> Flödesanalys i den hållbara utvecklingens tjänst : reflektioner kring en "metabolism" – studie av Rhenområdets utveckling. (1996)
CXXIX.	<i>Karl-Johan Lundquist:</i> Företag, regioner och internationell konkurrens : om regionala resursers betydelse. (1996)
CXXX.	<i>Badr-Eldin Taha Osman:</i> GIS-hydrological modelling in aridlands : a geographical synthesis of surface waters for the African Red Sea region in the Sudan. (1996)
CXXXI.	<i>Marie Stenseke:</i> Bonden och landskapet : ägares och brukares relationer till markerna och förutsättningarna för en uthållig markanvänding. (1997)
CXXXII.	<i>Kristina Blennow:</i> Spatial variation in near-ground radiation and low temperature – interactions with forest vegetation. (1997)
CXXXIII.	<i>Lennart Runesson:</i> Tomträtt : ett markpolitiskt instrument i upplösning. (1997)
CXXXIV.	<i>Johan Hultman:</i> The eco-ghost in the machine : reflexions on space, place and time in environmental geography. (1998)
CXXXV.	Jonas Ardö: Remote sensing of forest decline in the Czech Republic. (1998)
CXXXVI.	<i>Per Hillbur:</i> The knowledge arena : approaching agroforestry and competing knowledge systems – a challenge for agricultural extensionl (1998)

CXXXVII.	<i>Tom Mels:</i> Wild landscapes : the cultural nature of Swedish national parks. (1999)
CXXXVIII.	<i>Carolyn Hannan-Andersson:</i> Promoting equality between women and men in bilateral development cooperation : concepts, goals, rationales and institu-tional arrangements. (2000)
CXXXIX.	<i>Nikolaus Solakius:</i> The parnassus zone, central Greece. (2000)
CXL.	Jonathan Seaquist: Mapping primary production for the west African Sahel using satellite data. (2001)
CXLI.	<i>Karin Book och Lena Eskilsson:</i> Stadens struktur : varför och hur? (2001)
CXLII.	<i>Maria Wikhall</i> : Universiteten och kompetenslandskapet : effekter av den högre utbildningens tillväxt och regionala spridning i Sverige. (2001)
CXLIII.	<i>Rannveig Olafsdottir</i> : Land degradation and climate in Iceland : a spatial and temporal assessment. (2002)
CXLIV.	<i>Marie Ekström</i> : Relationships between atmospheric circulation and wind erosion in southern Sweden and Australia. (2002)
CXLV.	<i>Maj-Lena Finnander Linderson</i> : The spatial distribution of precipitation in Scania, southern Sweden : observations, model simulations and statistical downscaling. (2002)
CXLVI.	<i>Richard Ek:</i> Öresundsregion - bli till! : de geografiska visionernas diskursiva rytm. (2003)
CXLVII.	<i>Olivia Louw:</i> Exploring the culture of non-payment in the post-apartheid South Africa. (2003)
CXLVIII.	<i>Cecilia Kjellman:</i> Ta plats eller få plats? : studier av marginaliserade människors förändrade vardagsliv (2003)
CXLIX.	<i>Christina Scholten:</i> Kvinnors försörjningsrum : hegemonins förvaltare och murbräckor (2003)

CL.	<i>Micael Runnström:</i> Land degradation and mitigation in northern China : evaluated from the biological production (2003)
CLI.	<i>Sara Brogaard:</i> Recent changes in land use and productivity in agro-pastoral Inner Mongolia, China (2003)
CLII.	Jan-Henrik Nilsson: Östersjöområdet : studier av interaktion och barriärer (2003)
CLIII.	<i>Thomas Hickler:</i> Towards an integrated ecology through mechanistic modelling of ecosystem structure and functioning (2004)
CLIV.	<i>Andreas Persson:</i> Hydrological modelling, topographical influence and yield mapping in precision agriculture (2004)
CLV.	<i>Maria Olsrud:</i> Mechanisms of below-ground carbon cycling in subarctic ecosystems (2004)
CLVI.	<i>Sandra C. Fernández:</i> Farewell to the peasantry? : (Post)modernising rural Mexico : the case of the ejido peasants in the Isthmus of Tehuantepec (2004)
CLVII.	<i>Andrés P. Leskó</i> : To own the phone : spatial diffusion, ownership and regulation of telephone service in Argentina, 1878-1990 (2004)
CLVIII.	<i>Henrik Svensson</i> : Öppna och slutna rum : enskiftet och de utsattas geografi : husmän, bönder och gods på den skånska landsbygden under 1800-talets första hälft (2005)
CLIX.	<i>Pablo Morales</i> : Modeling carbon and water fluxes in European terrestrial ecosystems (2006)
CLX.	<i>Emmanuel S. Gritti</i> : Global changes and European terrestrial ecosystems (2006)
CLXI.	<i>Ola Thufvesson</i> : Kreativitetens yttre villkor : miljöer, rörlighet och nobelpristagare (2006)
CLXII.	<i>Deniz Koca</i> : Impacts of regional climate change on Swedish forests : an evaluation using process-based regional ecosystem modelling approach (2006)

CLXIII.	<i>Bodil Elmqvist</i> : Livelihood diversification and land use change in the Sahel : an interdisciplinary analysis of gum arabic in Sudan (2006)
CLXIV.	Jan Vang-Lauridsen: Industrial dynamics and the spatial organization of industries (2006)
CLXV.	Heidi Wiig Aslesen: Innovation in an urban context (2006)
CLXVI.	<i>Torbjörn Johansson</i> : Temporal and spatial variability of carbon cycling in a subarctic landscape (2006)
CLXVII.	<i>Anders Lund Hansen:</i> Space wars and the new urban imperialism (2006)
CLXVIII.	<i>Lars Coenen</i> : Faraway, so close! : the changing geographies of regional innovation (2006)
CLXIX.	<i>Pontus Olofsson</i> : Remote sensing of carbon balance across Scandinavian forests (2007)
CLXX.	<i>Margareta Rämgård</i> : The power of place : existential crises and place security in the context of pregnancy (2006)
CLXXI.	<i>Helena Eriksson</i> : Leaf area index of Scandinavian forests : methods using in situ and remotely sensed data (2007)
CLXXII.	<i>Ransom Lekunze</i> : Corporate social responsibility and development : the case of the Chad Cameroon Oil Pipeline Project (2007)
CLXXIII.	<i>Alla Yurova</i> : Hydrological aspects of the carbon balance in a boreal catchment : a model study (2007)
CLXXIV.	<i>Jerker Moodysson</i> : Sites and modes of knowledge creation : on the spatial organization of biotechnology innovation (2007)
CLXXV.	<i>Yahia Mohamed-Mahmood</i> : Chinese development assistance and West African agriculture : a shifting approach to foreign aid? (2007)
CLXXVI.	<i>Høgni Kalsø Hansen:</i> The urban turn and the location of economic activities (2008)

CLXXVII.	<i>Helene Bogren:</i> Flytta eller stanna? : betydelsen av plats och platsförankring för den kvalificerade arbetskraftens internationella migration (2008)
CLXXVIII.	<i>Lotten Jönsson Johansson:</i> Semi-natural grasslands : landscape, history and plant species diversity : a spatio- temporal study of an agricultural landscape on Öland, Sweden (2008)
CLXXIX.	<i>Carin Nilsson:</i> Windstorms in Sweden : variations and impacts (2008)
CLXXX.	<i>Margareta Johansson:</i> Changing lowland permafrost in northern Sweden : multiple drivers of past and future trends (2008)
CLXXXI.	<i>Martin Svensson Henning:</i> Industrial dynamics and regional structural change : geographical perspectives on economic evolution (2009)
CLXXXII.	<i>Brynhildur Bjarnadóttir:</i> Carbon stocks and fluxes in a young Siberian larch (Larix sibirica) plantation in Iceland (2009)
CLXXXIII.	<i>Magnus Lund:</i> Peatlands at a threshold : greenhouse gas dynamics in a changing climate (2009)
CLXXXIV.	<i>Marcin Jackowicz-Korczyński:</i> Land-atmosphere interactions of a subarctic palsa mire (2009)
CLXXXV.	<i>Nicklas Guldåker:</i> Krishantering, hushåll och stormen Gudrun : att analysera hushålls krishanteringsförmåga och sårbarheter (2009)
CLXXXVI.	<i>Nicodemus Mandere:</i> Alternative agriculture and rural development: a case study of sugar beet cultivation in Kenya (2009)
CLXXXVII.	<i>Anna Wramneby:</i> The role of vegetation-climate feedbacks in regional earth system dynamics (2010)
CLXXXVIII.	<i>Mikhail Mastepanov:</i> Towards a changed view on greenhouse gas exchange in the Arctic : new findings and improved techniques (2010)

CLXXXIX.	<i>Evelin Urbel-Piirsalu:</i> The Estonian forest sector in transition to sustainability? : capturing sustainability with the help of integrated assessment (2010)
CXC.	<i>Marie Vandewalle:</i> Effects of past and current land use on biodiversity : from a semi-natural grassland on Öland (Sweden) to a European perspective (2010)
CXCI.	<i>Maria Andrea Nardi:</i> Rural development and territorial dynamics in the provinceof Misiones, Argentina (2011)
CXCII.	<i>Torbern Tagesson:</i> Land-atmosphere exchange of carbon in a high-Arctic wet tundra ecosystem (2011)
CXCIII.	<i>Per Schubert:</i> Model development for estimating carbon dioxide exchange in Nordic forests and peatlands with MODIS time series data (2011)
CXCIV.	<i>Mabel Ndakaripa Munyuki-Hungwe:</i> In search of 'community' in Zimbabwe's fast track resettlement area of Mazowe district (2011)
CXCV.	<i>Oliver Purschke:</i> Plant community assembly and biodiversity : a spatio-temporal perspective (2011)

MEDDELANDEN FRÅN INSTITUTIONEN FÖR KULTURGEOGRAFI OCH EKONOMISK GEOGRAFI. AVHANDLINGAR

I.	<i>Johanna Bergman Lodin:</i> Engendered promises, gendered challenges. Changing patterns of labor, control and benefits among smallholder households growing NERICA in Uganda (2012)
II.	<i>Dobloug Tore Anstein</i> : Høyvekstbedrifter og regionale finansieringssystemer (2012)
III.	<i>Patrik Olsson:</i> Ömse sidor om vägen; Allén och landskapet i Skåne 1700-1900 (2012)
IV.	<i>Ingrid Helene Garmann Johnsen:</i> Social Capital and Regional Innovation Systems – Bridging approaches and broadening understanding of knowledge sharing and innovation (2012)
V.	<i>Roman Martin:</i> Knowledge Bases and the Geography of Innovation (2012)
VI.	<i>Monica Plechero:</i> The changing geography of innovation – Chinese and Indian regions and the global flows of innovation (2012)
VII.	<i>Erik Jönsson:</i> Fields of Green and Gold: Territorial hunger, rural planning, and the political ecologies of high-end golf (2013)
VIII.	<i>Elena Zukauskaite:</i> Institutions and the Geography of Innovation: A Regional Perspective (2013)
IX.	<i>Carl-Johan Sanglert:</i> Att skapa plats och göra rum – Landskapsperspektiv på det historiska värdets betydelse och funktion i svensk planering och miljövård (2013)
Χ.	<i>Tiina Lovisa Solbär:</i> Anthropogenic Open Land in Boreal Landscapes – Investigations into the Creation and Maintenance of Arable Fields on Swedish Farms (2014)

XI.	<i>Ståle Holgersen:</i> The rise (and fall?) of post-industrial Malmö – Investigations of city-crisis dialectics (2014)
XII.	<i>Wim Carton</i> : Fictitious Carbon, Fictitious Change? Environmental Implications of the Commodification of Carbon (2016)
XIII.	<i>Hanna Martin:</i> Innovation for tackling grand challenges: Cleantech industry dynamics and regional context (2016)
XIV.	<i>Srilata Sircar:</i> Between the highway and the red dirt track: Subaltern Urbanization and Census Towns in India (2016)
XV.	<i>Mikhail Martynovich:</i> General purpose technology diffusion and labour market dynamics: A spatio-temporal perspective (2016)



The diffusion of the information technology has led to the profound transformation of economies all over the world. Among other things, it has left a deep impact on the functioning of the labour market and people's relationship to the world of work. This dissertation aims at qualifying the role of the labour market in the process of economic transformation induced by the introduction and diffusion of the information technology.

The dissertation integrates the theoretical perspectives on the long-term economic development with the insights from the labour market studies into the evolutionary model of economic change that links the behaviour of economic agents at the micro-level of the economy with the macro-level outcomes and determinants of this behaviour. Empirically, the dissertation investigates the co-evolutionary dynamics of industrial restructuring and worker reallocation across and within regional labour markets in Sweden in the period between 1985 and 2010.

The findings of this dissertation indicate that the dynamics of technological change and its implications for functioning of the regional labour markets should be approached through the prism of technology diffusion process which unfolds *in time and space*.



LUND UNIVERSITY Faculty of Social Sciences Department of Human Geography ISBN 978-91-7753-069-5

